SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT
PURSUANT TO SECTION 13 OR 15(d) OF THE
SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): MAY 14, 2001

VECTOR GROUP LTD. (Exact name of registrant as specified in its charter)

DELAWARE

(State or other jurisdiction of incorporation)

1-5759 (Commission File Number) 65-0949535

(I.R.S. Employer Identification No.)

100 S.E. SECOND STREET, MIAMI, FLORIDA (Address of principal executive offices)

33131 (Zip Code)

(305) 579-8000 (Registrant's telephone number, including area code)

ITEM 5. OTHER EVENTS.

On May 14, 2000, Vector Group Ltd.'s wholly-owned subsidiary BGLS Inc. completed the issuance at a discount of \$60 million principal amount of 10% senior secured notes due March 31, 2006 in a private placement to certain institutional investors. BGLS received proceeds from the offering of \$49.7 million before fees and expenses of approximately \$3.2 million. The notes were priced to provide the purchasers with a 15.75% yield to maturity.

On May 16, 2001, Vector entered into a Stock Purchase Agreement with High River Limited Partnership, an investment entity owned by Carl C. Icahn, in which High River agreed to purchase for \$50 million 1,639,344 shares of Vector's common stock at a price of \$30.50 per share. The closing of the purchase of the shares is conditioned upon the expiration or termination of the waiting period under the Hart-Scott-Rodino Act and the listing of the shares on the New York Stock Exchange. High River has agreed not to sell or transfer the shares in the public markets for a one-year period following the closing.

A press release announcing the closing of the BGLS note placement and the execution of the agreement with High River was issued by Vector on May 16, 2001.

The summary of the foregoing transactions is qualified in its entirety by reference to the text of the related agreements, which are included as exhibits hereto and are incorporated herein by reference.

ITEM 7. FINANCIAL STATEMENTS, PRO FORMA FINANCIAL INFORMATION AND EXHIBITS.

(c) The following Exhibits are provided in accordance with the provisions of Item 601 of Regulation S-K and are filed herewith unless otherwise noted.

EXHIBIT INDEX

- Note Purchase Agreement, dated as of May 14, 2001, between BGLS Inc. and TCW Leveraged Income Trust, L.P., TCW Leveraged Income Trust II, L.P., TCW LINC II CBO Ltd., POWRS 1997-2, Captiva II Finance Ltd. and AIMCO CDO, Series 2000-A (the "Purchasers"), relating to the 10% Senior Secured Notes due March 31, 2006 (the "Notes"), including the form of Note (the "Note Purchase Agreement").
- Collateral Agency Agreement, dated as of May 14, 2001, by and among BGLS Inc., Brooke Group Holding Inc., Vector Group Ltd., New Valley Holdings, Inc., United States Trust Company of New York, as collateral agent for the benefit of the holders of the Notes pursuant to the Note Purchase Agreement (the "Collateral Agent"), and the Purchasers.
- 10.3 Pledge and Security Agreement, dated as of May 14, 2001 between BGLS Inc. and the Collateral Agent.
- 10.4 Pledge and Security Agreement, dated as of May 14, 2001, between New Valley Holdings, Inc. and the Collateral Agent.
- 10.5 Pledge and Security Agreement, dated as of May 14, 2001, between Brooke Group Holding Inc. and the Collateral Agent.
- Acknowledgment and Pledge Agreement, dated as of May 14, 2001, between Vector Group Ltd. and the Collateral Agent.
- 10.7 Account Control Agreement, dated as May 14, 2001, between BGLS Inc., Bank of America, N.A. and the Collateral Agent.
- 10.8 Stock Purchase Agreement, dated May 16, 2001, between High River Limited Partnership and Vector Group Ltd.
- 10.9 Press release of Vector Group Ltd. issued on May 16, 2001.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

VECTOR GROUP LTD.

By: /s/ JOSELYNN D. VAN SICLEN

Joselynn D. Van Siclen

Vice President and Chief Financial Officer

Date: May 17, 2001

BGLS INC.

\$60,000,000

10% SENIOR SECURED NOTES DUE MARCH 31, 2006

NOTE PURCHASE AGREEMENT

Dated as of May 14, 2001

TABLE OF CONTENTS

	PAI		
1.	AUTHORIZ	ZATION OF NOTES1	
2.	SALE AND	PURCHASE OF NOTES	
3.	CLOSING.	2	
4.	CONDITIO	ONS TO CLOSING	
	4.1	Representations and Warranties	
	4.2	Performance; No Default	
	4.3	Compliance Certificates	
	4.4	Opinion of Counsel	
	4.5	Purchase Permitted By Applicable Law, Consents of Third Parties, etc	
	4.6	Payment of Certain Fees and Expenses	
	4.7	Private Placement Number3	
	4.8	Changes in Corporate Structure	
	4.9	Proceedings and Documents	
	4.10	Rating Letters4	
	4.11	Pledged Stock and Pledged Notes4	
	4.12	Filings, Registrations and Recordings	
	4.13	Lien Searches	
	4.14	Financing Documents	
	4.15	Group Accounts5	
5.	REPRESEN	ITATIONS AND WARRANTIES OF THE COMPANY5	
	5.1	Organization; Power and Authority5	
	5.2	Authorization, etc	
	5.3	Disclosure5	
	5.4	Organization and Ownership of Shares of Subsidiaries; Affiliates6	
	5.5	Financial Statements6	
	5.6	Compliance with Laws, Other Instruments, etc7	
	5.7	Governmental Authorizations, etc	
	5.8	Litigation; Observance of Agreements, Statutes and Orders	
	5.9	Taxes	
	5.10	Title to Property; Leases	
	5.11	Licenses, Permits, etc8	
	5.12	Compliance with ERISA8	
	5.13	Private Offering by the Company9	
	5.14	Use of Proceeds; Margin Regulations9	
	5.15	Existing Indebtedness; Future Liens	
	5.16	Foreign Assets Control Regulations, etc	
	5.17	Status under Certain Statutes	
	5.18	Environmental Matters 11	

SECT		F	PAGE
	5.19	Security Agreements	
	5.20	Fees	
	5.21	Solvency	
	5.22	Assets and Operations of Certain Subsidiaries	. 12
	5.23	Vector Group Ltd.; Affiliated Transactions	. 12
	5.24	Company Assets	. 12
	5.25	Group Accounts	. 13
6.	REPRESEN	ITATIONS OF THE PURCHASERS	.13
7.	PAYMENT	AND REPAYMENT	. 14
	7.1	Maturity	
	7.2	Optional Prepayments	
	7.3	Allocation of Partial Prepayments	
	7.4	Maturity; Surrender, etc	. 15
	7.5	Purchase of Notes	
	7.6	Accreted Value	
8.	COVENANT	S	.16
	8.1	Payment of Notes	.16
	8.2	Maintenance of Office or Agency	.16
	8.3	Limitations on Restricted Payments	. 17
	8.4	Limitation on Indebtedness	
	8.5	Limitation on Transactions with Affiliates and Investments	
	8.6	Investment Company Act	
	8.7	Limitation on Liens	
	8.8	No Violation of the Margin Rules	
	8.9	Financial Reports	
	8.10	Waiver of Stay, Extension or Usury Laws	
	8.11	[RESERVED.]	
	8.12	Compliance with Laws	
	8.13	Existence	
	8.13	Payment of Taxes and Other Claims	
	8.15	Maintenance of Properties	
	8.16	Compliance Certificate; Inspection Rights	
	8.17	Ownership of NV Holdings; Additional Covenants Relating to NV Holdings	. 26
	8.18	Restrictions on Issuance of Additional Equity Interests	. 26
	8.19	Opinions, Appraisals, Etc	.27
	8.20	Ownership of Restricted Subsidiaries	
	8.21	Group Accounts	
	8.22	[RESERVED.]	
	8.23	Fiscal Year	
	8.24	Offer to Purchase Upon Triggering Asset Sale	
	8.25	Offer to Purchase Upon Change of Control	
	8.26	Dividend and Other Payment Restrictions Affecting Subsidiaries	
	8.27	Rule 144A Information	. 32

SECTI	ON 		PAGE
	8.28 8.29 8.30	Limitation on Restricted Subsidiary Business	32
9.	ADDITIONAL 9.1 9.2 9.3 9.4 9.5 9.6	L COVENANTS. Vector Tobacco Credit Facility. [RESERVED.]. Tax Sharing Payments. Tax Matters. Voting of Notes by Group Executives and New Valley. Transfer of Company Equity Interests.	33 34 34 35
10.	SUCCESSOR 10.1 10.2	PERSON Mergers, Consolidations Successor Person Substituted	36
11.	EVENTS OF	DEFAULT	38
12.	REMEDIES (12.1 12.2 12.3	ON DEFAULT, ETC	40 41
13.	SECURITY 1 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11	INTEREST. Pledge and Security Interest. Release of Collateral. Reliance on Opinion of Counsel. Purchaser May Rely. Payment of Expenses. Suits to Protect the Collateral. Collateral Agent's Duties. Sale of Collateral. Dealings with Collateral. Annual Reports. Securities Account.	41 42 42 42 42 43 43 44
14.	REGISTER (14.1 14.2 14.3 14.4	OF NOTES; EXCHANGE AND SUBSTITUTION OF NOTES; LEGENDS. Register of Notes Transfer and Exchange of Notes Replacement of Notes Legends on Notes.	46 46 47
15.	PAYMENTS (15.1 15.2	ON NOTES Place of Payment Home Office Payment	48

SECTION I			
16.	EXPENSES, 16.1 16.2	ETC. Transaction Expenses. Survival.	49
17.	SURVIVAL	OF REPRESENTATIONS AND WARRANTIES; ENTIRE AGREEMENT	49
18.	AMENDMENT 18.1 18.2 18.3 18.4	AND WAIVER. With Consent of Holders. Revocation and Effect of Consents. Notation on Notes. Rights of Holders to Receive Payment.	50 51 51
19.	NOTICES		51
20.	REPRODUCT	ION OF DOCUMENTS	52
21.	CONFIDENT	IAL INFORMATION	52
22.	MISCELLAN 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10	Certificate and Opinion as to Conditions Precedent Statements Required in Certificate or Opinion. Third Party Beneficiary. Successors and Assigns. Payments Due on Non-Business Days. Severability. Construction. Counterparts. Captions. Governing Law; Jurisdiction; Waiver of Jury Trial.	53 54 54 54 55 55 55
Schedu Schedu Exhibi Exhibi Exhibi Exhibi Exhibi Exhibi Exhibi Exhibi	ule B it A it B-1 it B-2 it C it D it E it 4.3(a). it 4.4	List of Purchas Definit: Applicable Accretion Percenta Form of Nover and Secretary Agreem Form of Collateral Agency Agreem Form of Secretary's Certifice Matters to be Covered in Legal Opin Liggett and VTUSA Product Line Informati	ions ages Note ment ment ment ment cate cate nion

BGLS INC.

100 S.E. Second Street Miami, Florida 33131

10% SENIOR SECURED NOTES DUE MARCH 31, 2006

As of May 14, 2001

TO THE PURCHASERS WHOSE NAMES APPEAR IN THE ACCEPTANCE FORM AT THE END HEREOF:

Ladies and Gentlemen:

BGLS Inc., a Delaware corporation (the "COMPANY"), agrees with you (the "PURCHASERS") as follows:

1. AUTHORIZATION OF NOTES.

The Company will authorize the issue and sale of \$60,000,000 aggregate principal amount of its 10% Senior Secured Notes Due March 31, 2006 (the "NOTES", such term to include any such Notes issued in substitution therefor pursuant to SECTION 14 of this Agreement). The Notes shall be substantially in the form of EXHIBIT A, with such changes therefrom, if any, as may be approved by each Purchaser and the Company. Certain capitalized terms used in this Agreement are defined in SCHEDULE B; references to a "SCHEDULE" or an "EXHIBIT" are, unless otherwise specified, to a Schedule or an Exhibit attached to this Agreement.

2. SALE AND PURCHASE OF NOTES.

Subject to the terms and conditions of this Agreement, the Company will issue and sell to you, and you will purchase from the Company, at the Closing provided for in SECTION 3, Notes in the principal amount specified opposite your

name in SCHEDULE A at the purchase price of 82.88445% of the principal amount thereof. The obligations of each Purchaser hereunder are several and not joint obligations.

3. CLOSING.

The sale and purchase of the Notes shall occur at the offices of Milbank, Tweed, Hadley & McCloy LLP, 601 South Figueroa Street, 30th Floor, Los Angeles, California 90017, at 9:00 a.m., Los Angeles time, at a closing (the "CLOSING") on May 14, 2001 or on such other Business Day thereafter on or prior to May 31, 2001 as may be agreed upon by the Company and the Majority Holders. At the Closing, the Company will deliver to you the Notes to be purchased by you in the form of a single Note (or such greater number of Notes in denominations of at least \$100,000 as you may request) dated the date of the Closing and registered in your name (or in the name of your nominee), against delivery by you of immediately available funds in the amount of the purchase price therefor by wire transfer. If at the Closing the Company shall fail to tender such Notes to you as provided above in this SECTION 3, or any of the conditions specified in SECTION 4 shall not have been fulfilled to your satisfaction, you shall, at your election, be relieved of all further obligations under this Agreement.

4. CONDITIONS TO CLOSING.

Your obligation to purchase and pay for the Notes to be sold to you at the Closing is subject to the fulfillment to your satisfaction, prior to or at the Closing, of the following conditions:

4.1 REPRESENTATIONS AND WARRANTIES.

4.2 PERFORMANCE; NO DEFAULT.

Each Document Party shall have performed and complied with all agreements and conditions contained in the Note Documents required to be performed or complied with by it prior to or at the Closing, and after giving effect to the issue and sale of the Notes, no Default or Event of Default shall have occurred and be continuing. Neither the Company nor any of the Subsidiaries shall have entered into any transaction since December 31, 2000 in excess of \$1,000,000, that would have been prohibited by SECTION 8.3, 8.4, 8.5 or 8.7 had such Section applied since such date except as listed on SCHEDULE 4.2.

4.3 COMPLIANCE CERTIFICATES.

(a) OFFICERS' CERTIFICATE. The Company shall have delivered to you and to the Collateral Agent an Officers' Certificate, dated the date of the Closing and substantially in the form of EXHIBIT 4.3(A) attached hereto, certifying that the conditions specified in this SECTION 4 have been fulfilled.

(b) SECRETARY'S CERTIFICATE. Each Document Party shall have delivered to you and to the Collateral Agent a certificate, dated the date of the Closing and substantially in the form of EXHIBIT 4.3(B) attached hereto, certifying as to the resolutions attached thereto and other corporate proceedings relating to the authorization, execution and delivery of each Note Document to which it is a party.

4.4 OPINION OF COUNSEL.

You shall have received opinions in form and substance satisfactory to you and to the Collateral Agent, each dated the date of the Closing, from McDermott, Will & Emery, counsel for the Company, and Richard J. Lampen, Executive Vice President and Special Counsel of the Company, covering the matters set forth in EXHIBIT 4.4 (and the Company hereby instructs its counsel to deliver such opinions to you).

4.5 PURCHASE PERMITTED BY APPLICABLE LAW, CONSENTS OF THIRD PARTIES, FTC.

On the date of the Closing, your purchase of Notes shall (i) be permitted by the laws and regulations of each jurisdiction to which you are subject, (ii) not violate any applicable law or regulation (including, without limitation, Regulation T, U or X of the Board of Governors of the Federal Reserve System) and (iii) not subject you to any tax, penalty or liability under or pursuant to any applicable law or regulation, which law or regulation was not in effect on the date hereof. If requested by you, you shall have received an Officers' Certificate from any Document Party certifying as to such matters of fact as you may reasonably specify to enable you to determine whether such purchase is so permitted. All consents of third parties necessary to consummate the transactions contemplated in the Note Documents shall have been obtained.

4.6 PAYMENT OF CERTAIN FEES AND EXPENSES.

Without limiting the provisions of SECTION 16.1, the Company shall have paid on or before the Closing (i) a funding fee of \$1,200,000 to the Purchasers or their respective designees, pro rata, in proportion to the amount of Notes to be purchased by each Purchaser and (ii) the reasonable fees, charges and disbursements of Milbank, Tweed, Hadley & McCloy LLP, to the extent reflected in a statement of such counsel rendered to the Company prior to the Closing.

4.7 PRIVATE PLACEMENT NUMBER.

A Private Placement number issued by Standard & Poor's CUSIP Service Bureau shall have been obtained for the Notes.

4.8 CHANGES IN CORPORATE STRUCTURE.

Except as specified in SCHEDULE 4.8, no Document Party shall have changed its jurisdiction of incorporation or been a party to any merger or consolidation and shall have succeeded to all or any substantial part of the liabilities of any other entity, at any time following the date of the most recent financial statements referred to in SCHEDULE 5.5.

4.9 PROCEEDINGS AND DOCUMENTS.

All corporate and other proceedings in connection with the transactions contemplated by the Note Documents and all documents and instruments incident to such transactions shall be reasonably satisfactory to you and your counsel, and you shall have received all such counterpart originals or certified or other copies of such documents as you may reasonably request.

4.10 RATING LETTERS.

The delivery to the Purchasers of an Officers' Certificate of the Company to the effect that attached thereto is a true and correct copy of (i) a letter signed by Moody's Investors Service, Inc. confirming that the Notes have been rated at least B3 by Moody's Investors Service, Inc. and that such rating is in full force and effect on the date of Closing; and (ii) a letter signed by Standard & Poor's Ratings Services confirming that the Notes have been rated at least B by Standard & Poor's Ratings Services and that such ratings are in full force and effect on the date of Closing.

4.11 PLEDGED STOCK AND PLEDGED NOTES.

The Collateral Agent shall have received (i) the certificates representing the Pledged Stock pursuant to the Security Agreements, together with an undated stock power for each such certificate executed in blank by a duly authorized officer of the pledgor thereof in form and substance satisfactory to the Collateral Agent, and (ii) each promissory note pledged to the Collateral Agent pursuant to the Security Agreements endorsed (without recourse) in blank (or accompanied by an executed transfer form in blank in form and substance satisfactory to the Collateral Agent) by the Company.

4.12 FILINGS, REGISTRATIONS AND RECORDINGS.

Each document (including, without limitation, any Uniform Commercial Code financing statement) required by the Security Agreements, under current law, under law scheduled to go into effect or reasonably requested by the Collateral Agent or any Purchaser to be filed, registered or recorded in order to create in favor of the Collateral Agent, for the benefit of the Purchasers, a perfected Lien on the Collateral described therein, prior and superior in right to any other Person (other than with respect to Liens expressly permitted by SECTION 8.7) shall be in proper form for filing, registration or recordation.

4.13 LIEN SEARCHES.

The Purchasers shall have received the results of a recent Uniform Commercial Code, tax and judgment lien search in each of the jurisdictions where assets of the Company and the Subsidiary Pledgors are located or where reasonably requested by any Purchaser, and such search shall have revealed no Liens on any assets of the Company or any Subsidiary Pledgor other than Permitted Liens on Collateral other than Pledged Stock and Cash and cash equivalents.

that:

4.14 FINANCING DOCUMENTS.

A true and complete copy of each Material financing document of the Company and the Restricted Subsidiaries shall have been delivered to the Purchasers, accompanied by an Officer's Certificate to the effect that all such documents are true and complete copies of all Material financing documents, as amended, modified and supplemented through the date hereof, of the Company and the Restricted Subsidiaries.

4.15 GROUP ACCOUNTS.

The Group Accounts shall have been established. All Cash and cash equivalents held by the Company and the Subsidiary Pledgors shall have been placed in a Group Account, and the aggregate amount on the deposit in all Checking Accounts shall not exceed \$500,000 on the Closing Date.

5. REPRESENTATIONS AND WARRANTIES OF THE COMPANY.

The Company represents and warrants to you and the Collateral Agent

5.1 ORGANIZATION; POWER AND AUTHORITY.

Each Document Party is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation, and is duly qualified as a foreign corporation and is in good standing in each jurisdiction in which such qualification is required by law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Each Document Party has the corporate power and authority to own or hold under lease the properties it purports to own or hold under lease, to transact the business it transacts and proposes to transact, to execute and deliver each Note Document to which it is a party and to perform the provisions of such Note Document.

5.2 AUTHORIZATION, ETC.

Each Note Document has been duly authorized by all necessary corporate action on the part of each Document Party party thereto, and each of this Agreement and the Security Agreements constitutes, and upon execution and delivery thereof each Note will constitute, a legal, valid and binding obligation of each Document Party party thereto enforceable against such Document Party in accordance with its terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

5.3 DISCLOSURE.

Except as disclosed in SCHEDULE 5.3, this Agreement, the documents, certificates or other writings delivered to you by or on behalf of the Company in connection with the transactions contemplated hereby and the financial statements listed in SCHEDULE 5.5, taken as a whole, do not contain any untrue statement of a material fact or omit to state any material fact necessary to

make the statements therein not misleading in light of the circumstances under which they were made. Except as expressly described in SCHEDULE 5.3, or in one of the documents, certificates or other writings identified therein, or in the financial statements listed in SCHEDULE 5.5, since December 31, 2000, there has been no change in the financial condition, operations, business, affairs, assets, properties or prospects of the Company or any Subsidiary except changes that individually or in the aggregate could not reasonably be expected to have a Material Adverse Effect. There is no fact known to the Company that could reasonably be expected to have a Material Adverse Effect that has not been set forth herein or in the other documents, certificates and other writings delivered to you by or on behalf of the Company specifically for use in connection with the transactions contemplated hereby.

5.4 ORGANIZATION AND OWNERSHIP OF SHARES OF SUBSIDIARIES: AFFILIATES.

- (a) SCHEDULE 5.4 contains complete and correct lists (i) of the Company's Subsidiaries, showing, as to each Subsidiary, the correct name thereof, the jurisdiction of its organization, and the percentage and number of shares of each class of its capital stock or other similar Equity Interests outstanding owned by the Company and each Subsidiary, (ii) of the Company's Affiliates, other than Subsidiaries, and (iii) of the Company's directors and senior officers.
- (b) All of the outstanding shares of capital stock or other similar Equity Interests of each Subsidiary shown in SCHEDULE 5.4 as being owned by the Company and its Subsidiaries have been validly issued, are fully paid and nonassessable and are owned by the Company or a Subsidiary free and clear of any Lien (except as otherwise disclosed in SCHEDULE 5.4).
- (c) Each Subsidiary identified in SCHEDULE 5.4 is a corporation or other legal entity duly organized, validly existing and in good standing under the laws of its jurisdiction of organization, and is duly qualified as a foreign corporation or other legal entity and is in good standing in each jurisdiction in which such qualification is required by law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Each such Subsidiary has the corporate or other power and authority to own or hold under lease the properties it purports to own or hold under lease and to transact the business it transacts and proposes to transact.
- (d) No Subsidiary is a party to, or otherwise subject to any legal restriction or any agreement (other than this Agreement, the agreements listed on SCHEDULE 5.4 and customary limitations imposed by corporate law statutes) restricting the ability of such Subsidiary to pay dividends out of profits or make any other similar distributions of profits to the Company or any of its Subsidiaries that owns outstanding shares of capital stock or similar equity interests of such Subsidiary.

5.5 FINANCIAL STATEMENTS.

The Company has delivered to each Purchaser copies of the financial statements of the Company and its Subsidiaries listed on SCHEDULE 5.5. All of said financial statements (including in each case the related schedules and notes) fairly present in all material respects the consolidated financial position of the Company and its Subsidiaries as of the respective dates

specified in such Schedule and the consolidated results of their operations and cash flows for the respective periods so specified and have been prepared in accordance with GAAP consistently applied throughout the periods involved except as set forth in the notes thereto (subject, in the case of any interim financial statements, to normal year-end adjustments).

5.6 COMPLIANCE WITH LAWS, OTHER INSTRUMENTS, ETC.

The execution, delivery and performance by each Document Party of each Note Document to which it is a party will not (i) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of the Company or any Subsidiary under, any indenture, mortgage, deed of trust, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other agreement or instrument to which the Company or any Subsidiary is bound or by which the Company or any Subsidiary or any of their respective properties is bound or affected, (ii) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental or Regulatory Authority applicable to the Company or any Subsidiary or (iii) violate any provision of any statute or other rule or regulation of any Governmental or Regulatory Authority applicable to the Company or any Subsidiary.

5.7 GOVERNMENTAL AUTHORIZATIONS, ETC.

No consent, approval or authorization of, or registration, filing (other than the filing of any financing statements contemplated in any Note Document) or declaration with, any Governmental or Regulatory Authority is required in connection with the execution, delivery or performance by the Document Parties of any Note Document.

5.8 LITIGATION; OBSERVANCE OF AGREEMENTS, STATUTES AND ORDERS.

- (a) Except as disclosed in SCHEDULE 5.8, there are no actions, suits or proceedings pending or, to the knowledge of the Company, threatened against or affecting the Company or any Subsidiary or any property of the Company or any Subsidiary in any court or before any arbitrator of any kind or before or by any Governmental or Regulatory Authority that, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.
- (b) Neither the Company nor any Subsidiary is in default under any term of any agreement or instrument to which it is a party or by which it is bound, or any order, judgment, decree or ruling of any court, arbitrator or Governmental or Regulatory Authority or is in violation of any applicable law, ordinance, rule or regulation (including without limitation Environmental Laws) of any Governmental or Regulatory Authority, which default or violation, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

5.9 TAXES.

The Company and its Subsidiaries have filed all tax returns that are required to have been filed in any jurisdiction, and have paid all taxes shown to be due and payable on such returns and all other taxes and assessments levied upon them or their properties, assets, income or franchises, to the extent such taxes and assessments have become due and payable and before they have become delinquent, except for any taxes and assessments (i) the amount of which is not individually or in the aggregate Material or (ii) the amount, applicability or

validity of which is currently being contested in good faith by appropriate proceedings and with respect to which the Company or a Subsidiary, as the case may be, has established adequate reserves in accordance with GAAP. The charges, accruals and reserves on the books of the Company and its Subsidiaries in respect of Federal, state or other taxes for all fiscal periods are adequate. The Company knows of no basis for any tax or assessment that is not being contested or for which adequate reserves have not been established in accordance with GAAP that could reasonably be expected to have a Material Adverse Effect. The Federal income tax liabilities of the Company and its Subsidiaries have been paid for all fiscal years up to and including the fiscal year ended December 31, 2000.

5.10 TITLE TO PROPERTY; LEASES.

The Company and its Subsidiaries have good and marketable title to their respective properties that individually or in the aggregate are Material, including all such properties reflected in the most recent audited balance sheet referred to in SECTION 5.5 or purported to have been acquired by the Company or any Subsidiary after said date (except as sold or otherwise disposed of in the ordinary course of business), in each case free and clear of Liens prohibited by this Agreement. All leases that the Company or any Subsidiary is party to as lessee and that individually or in the aggregate are Material are valid and subsisting and are in full force and effect in all material respects.

5.11 LICENSES, PERMITS, ETC.

Except as disclosed in SCHEDULE 5.11,

- (a) the Company and its Subsidiaries own or possess all licenses, permits, franchises, authorizations, patents, copyrights, service marks, trademarks and trade names, or rights thereto, that individually or in the aggregate are Material, without known conflict with the rights of others;
- (b) to the best knowledge of the Company, no product of the Company or any of its Subsidiaries infringes in any material respect on any license, permit, franchise, authorization, patent, copyright, service mark, trademark, trade name or other right owned by any other Person; and
- (c) to the best knowledge of the Company, there is no Material violation by any Person of any right of the Company or any of its Subsidiaries with respect to any patent, copyright, service mark, trademark, trade name or other right owned or used by the Company or any of its Subsidiaries.

5.12 COMPLIANCE WITH ERISA.

(a) The Company and each ERISA Affiliate have operated and administered each Plan in compliance with all applicable laws except for such instances of noncompliance as have not resulted in and could not reasonably be expected to result in a Material Adverse Effect. Neither the Company nor any ERISA Affiliate has incurred any liability pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans (as defined in section 3 of ERISA), and no event, transaction or condition has occurred or

exists that could reasonably be expected to result in the incurrence of any such liability by the Company or any ERISA Affiliate, or in the imposition of any Lien on any of the rights, properties or assets of the Company or any ERISA Affiliate, in either case pursuant to Title I or IV of ERISA or to such penalty or excise tax provisions or to section 401(a)(29) or 412 of the Code, other than such liabilities or Liens as would not be individually or in the aggregate Material.

- (b) The present value of the aggregate benefit liabilities under each of the Plans (other than Multiemployer Plans), determined as of the end of such Plan's most recently ended plan year on the basis of the actuarial assumptions specified for funding purposes in such Plan's most recent actuarial valuation report, did not exceed the aggregate current value of the assets of such Plan allocable to such benefit liabilities. The term "BENEFIT LIABILITIES" has the meaning specified in section 4001 of ERISA, and the terms "CURRENT VALUE" and "PRESENT value" have the meaning specified in section 3 of ERISA.
- (c) The Company and its ERISA Affiliates have not incurred withdrawal liabilities (and are not subject to contingent withdrawal liabilities) under section 4201 or 4204 of ERISA in respect of Multiemployer Plans that individually or in the aggregate are Material.
- (d) The Expected Postretirement Benefit Obligation of the Company and its Subsidiaries is not Material.
- (e) The execution and delivery of this Agreement and the issuance and sale of the Notes hereunder will not involve any transaction that is subject to the prohibitions of section 406 of ERISA or in connection with which a tax could be imposed pursuant to section 4975(c)(1)(A)-(D) of the Code.

5.13 PRIVATE OFFERING BY THE COMPANY.

Neither the Company nor any of its Affiliates has directly or through any agent, (i) sold, offered for sale, solicited offers to buy or otherwise negotiated in respect of, any security (as defined in the Securities Act) which is or will be integrated with any sale of the Notes in a manner that would require registration under the Securities Act or any state securities laws of any such securities or (ii) engaged in any form of general solicitation or general advertising in connection with any offer and sale of the Notes or in any manner involving a public offering within the meaning of Section 4(2) of the Securities Act.

5.14 USE OF PROCEEDS; MARGIN REGULATIONS.

No part of the proceeds from the sale of the Notes hereunder will be used, directly or indirectly, for the purpose of buying or carrying any Margin Stock within the meaning of Regulation U of the Board of Governors of the Federal Reserve System (12 CFR 221), or for the purpose of buying or carrying or trading in any securities under such circumstances as to involve the Company in a violation of Regulation X of said Board (12 CFR 224) or to involve any broker or dealer in a violation of Regulation T of said Board (12 CFR 220). Margin Stock does not constitute more than 25% of the value of the consolidated assets

of the Company and its Subsidiaries and the Company does not have any present intention that Margin Stock will constitute more than 25% of the value of such assets. None of the Company, Brooke Holding, Liggett, New Valley, NV Holdings, Brooke Overseas, Research or VTUSA is engaged principally in the business of purchasing or carrying Margin Stock. As used in this SECTION 5.14, the term "purpose of buying or carrying" shall have the meanings assigned to it in said Regulation U.

5.15 EXISTING INDEBTEDNESS; FUTURE LIENS.

- (a) Except as described therein, SCHEDULE 5.15 sets forth a complete and correct list of all outstanding Indebtedness of the Company and its Subsidiaries as of May 14, 2001 since which date there has been no Material change in the amounts, interest rates, sinking funds, installment payments or maturities of the Indebtedness of the Company or its Subsidiaries. Neither the Company nor any Subsidiary is in default and no waiver of default is currently in effect, in the payment of any principal or interest on any Indebtedness of the Company or such Subsidiary and no event or condition exists with respect to any Indebtedness of the Company or any Subsidiary that would permit (or that with notice or the lapse of time, or both, would permit) one or more Persons to cause such Indebtedness to become due and payable before its stated maturity or before its regularly scheduled dates of payment.
- (b) Except as disclosed in SCHEDULE 5.15, neither the Company nor any Subsidiary Pledgor has agreed or consented to cause or permit in the future (upon the happening of a contingency or otherwise) any of its property, whether now owned or hereafter acquired, to be subject to a Lien not permitted by SECTION 8.7
- (c) All outstanding Indebtedness of Unrestricted Subsidiaries is Non-Recourse Indebtedness.
- (d) The only Liggett Senior Credit Facility in effect as of the Closing Date is the Loan and Security Agreement, dated as of March 8, 1994, by and between Congress Financial Corporation and Liggett Group Inc., as amended, modified and supplemented through the date hereof.

5.16 FOREIGN ASSETS CONTROL REGULATIONS, ETC.

Neither the sale of the Notes by the Company hereunder nor its use of the proceeds thereof will violate the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto.

5.17 STATUS UNDER CERTAIN STATUTES.

Neither the Company nor any Subsidiary is subject to regulation under the Investment Company Act, the Public Utility Holding Company Act of 1935, as amended, the Interstate Commerce Act, as amended, or the Federal Power Act, as amended.

5.18 ENVIRONMENTAL MATTERS.

Neither the Company nor any Subsidiary has knowledge of any claim or has received any notice of any claim, and no proceeding has been instituted raising any claim against the Company or any of its Subsidiaries or any of their respective real properties now or formerly owned, leased or operated by any of them or other assets, alleging any damage to the environment or violation of any Environmental Laws, except, in each case, such as could not reasonably be expected to result in a Material Adverse Effect. Except as otherwise disclosed to you in writing,

- (a) neither the Company nor any Subsidiary has knowledge of any facts which would give rise to any claim, public or private, of violation of Environmental Laws or damage to the environment emanating from, occurring on or in any way related to real properties now or formerly owned, leased or operated by any of them or to other assets or their use, except, in each case, such as could not reasonably be expected to result in a Material Adverse Effect;
- (b) neither the Company nor any of its Subsidiaries has stored any Hazardous Materials on real properties now or formerly owned, leased or operated by any of them and has not disposed of any Hazardous Materials in a manner contrary to any Environmental Laws in each case in any manner that could reasonably be expected to result in a Material Adverse Effect; and
- (c) all buildings on all real properties now owned, leased or operated by the Company or any of its Subsidiaries are in compliance with applicable Environmental Laws, except where failure to comply could not reasonably be expected to result in a Material Adverse Effect.

5.19 SECURITY AGREEMENTS.

The Security Agreements are effective to create in favor of the Collateral Agent, for the benefit of the Holders, a legal, valid and enforceable security interest in the Collateral, including, without limitation any Cash or cash equivalents held by the Company, described therein and the proceeds thereof. In the case of the Pledged Stock described in the Pledge Agreements, when any stock certificates representing such Pledged Stock are delivered to the Collateral Agent in the State of New York, and in the case of the other Collateral described in the Security Agreements, when financing statements in appropriate form are filed in the offices specified on Annex 2 to each Pledge Agreement (which financing statements have been duly completed and executed and delivered to the Collateral Agent) and such other filings and actions specified on Annex 2 to each Pledge Agreement have been completed (all of which filings and actions have been duly completed) the Security Agreements shall constitute a fully perfected Lien on, and security interest in, all right, title and interest of the Company in such Collateral and the proceeds thereof, as security for the Secured Obligations (as defined in each Pledge Agreement), in each case prior and superior in right to any other Person (except, in the case of Collateral other than Pledged Stock and Cash and cash equivalents, Permitted Liens).

5.20 FEES.

Other than Jefferies & Company, Inc., whose fee shall be the sole obligation of the Company, no Person shall have received a fee for investment banking services in connection with the transactions contemplated by any Note Document.

5.21 SOLVENCY.

Each of Vector, the Company, NV Holdings, New Valley and any Restricted Subsidiary is and after giving effect to the transactions contemplated in the Note Documents will be Solvent.

- 5.22 ASSETS AND OPERATIONS OF CERTAIN SUBSIDIARIES.
- (a) Brooke Overseas has no Material assets and has no Material liabilities.
- (b) NV Holdings has no Material assets other than as described in SECTION 5.4 and has no Material liabilities other than as pursuant to the Note Documents to which it is a party.
- (c) New Valley owns directly or indirectly all of Vector's Material direct and indirect current interests in business operations and properties in the former Soviet Union.
 - 5.23 VECTOR GROUP LTD.; AFFILIATED TRANSACTIONS
- (a) Vector owns 100% of the issued and outstanding Equity Interests of the Company and is primarily engaged in the business of owning the Equity Interests of the Company.
- (b) Vector has no Material assets other than Equity Interests in the Company and Cash and cash equivalents and has no Material liabilities.
- (c) Except as set forth on SCHEDULE 5.23(C), neither the Company nor any Restricted Subsidiary has any agreement, contract, understanding or arrangement with Vector that provides for payment, transfer of assets or provision of services in excess of \$100,000.
- (d) Except as set forth on SCHEDULE 5.23(D), neither the Company nor any Restricted Subsidiary has any agreement, contract, understanding or arrangement with an Unrestricted Subsidiary that provides for payment, transfer of assets or provision of services in excess of \$100,000.
- (e) Except as set forth on SCHEDULE 5.23(E), neither the Company nor any Restricted Subsidiary has any agreement, contract, understanding or arrangement with an Affiliate of Vector (other than the Company of one of its Subsidiaries) that provides for payment, transfer of assets or provision of services in excess of \$100,000.

5.24 COMPANY ASSETS

The Company has no Material Assets other than its ownership of 100% of the Equity Interests of Brooke Holding, VTUSA, NV Holdings, Research and Brooke Overseas and Cash and cash equivalents.

5.25 GROUP ACCOUNTS

Each Group Account has been duly established. All Cash and cash equivalents of the Company and each Subsidiary Pledgor are being held in Group Accounts, and the aggregate balance of the Checking Accounts does not exceed \$500,000 as of the Closing Date.

6. REPRESENTATIONS OF THE PURCHASERS.

Each Purchaser hereby, severally and not jointly, represents and warrants to, and agrees with, the Company that:

- (a) such Purchaser understands and acknowledges that the Notes have not been registered under the Securities Act based in part, on reliance that the issuance of the Notes is exempt from registration under Section 4(2) of the Securities Act and, therefore, the Notes may not be offered or sold except pursuant to an effective registration statement under, or an exemption from the registration requirements of, Securities Act;
- (b) such Purchaser (i) has not and, absent an effective registration statement permitting resale of such Notes by such Purchaser, will not solicit offers for, or offer to sell, the Notes by any form of general solicitation or general advertising (as those terms are used in Regulation D under the Securities Act) or in any manner involving a public offering within the meaning of Section 4(2) of the Securities Act, (ii) is acquiring one or more Notes hereunder for its own account, for investment purposes only and not with a view to any distribution thereof that would violate the Securities Act or the securities laws of any state of the United States or any applicable jurisdiction and (iii) absent an effective registration statement permitting resale of such Notes by such Purchaser, will not offer, sell, assign, transfer, pledge, encumber or otherwise dispose of the Notes except pursuant to an available exemption from the registration requirements of the Securities Act and in compliance with applicable state securities laws; furthermore, upon the request of the Company, such Purchaser shall deliver, or cause to be delivered, an opinion of counsel, certifications and/or other information requested by the Company and a certificate of transfer in the form appearing on the Note having been completed and delivered by the transferor to the Company prior to any such disposition; and
- (c) such Purchaser is an "accredited investor" as such term is defined in Rule 501(a) promulgated under Regulation D of the Securities Act and such Purchaser has the knowledge and experience in financial and business matters as are necessary in order to evaluate the merits and risks of an investment in the Notes; such Purchaser is relying on its own diligence in connection with its investment hereunder; and such Purchaser is not relying on any other Purchaser to provide such Purchaser with any information with respect to the offer and sale of the Notes or the Company generally and is not relying on the completeness or accuracy of any information provided by any other Purchaser; and such Purchaser has been given access to all books, records and other information of the Company which such Purchaser has desired to review and given the opportunity to ask questions of and receive answers from the Company in connection with the sale and purchase of the Notes hereunder.

Each Purchaser understands that the Company, and, with respect to its opinion delivered pursuant to this Agreement, counsel to the Company will rely

upon the accuracy and truth of the foregoing representations, warranties and agreements with respect to making a determination that an exemption from the registration requirement of the Securities Act is available pursuant to Section 4(2) thereof in connection with the issuance of the Notes hereunder, and the Purchasers hereby consent to such reliance.

7. PAYMENT AND REPAYMENT.

7.1 MATURITY.

As provided therein, the entire unpaid principal amount of the Notes, together with any accrued and unpaid interest, shall be due and payable on March 31, 2006 (the "MATURITY DATE").

7.2 OPTIONAL PREPAYMENTS.

- (a) Before May 14, 2003, the Company may on any one or more occasions redeem up to 35% of the aggregate principal amount of Notes originally issued under this Agreement in amounts equal to \$1,000,000 or integral multiples thereof at a redemption price of 105% of the Accreted Value thereof, plus accrued and unpaid interest to the redemption date, with the net cash proceeds of one or more (x) Equity Offerings or (y) Vector Equity Offerings of an amount not less than \$5,000,000; PROVIDED that:
 - (i) at least 65% of the aggregate principal amount of Notes issued under this Agreement remains outstanding immediately after the occurrence of such redemption (excluding Notes held by the Company and its Subsidiaries); and
 - (ii) the redemption must occur within forty-five (45) days of the date of the closing of the Equity Offering.
- (b) From and after May 14, 2003, the Company may, at its option, upon notice as provided below, prepay at any time all, or part of, the Notes, in an amount not less than, in the case of a partial prepayment, the lesser of (x) \$5,000,000 and (y) the aggregate principal amount of the Notes then outstanding, at the prepayment amounts (expressed as percentages of Accreted Value) PLUS accrued and unpaid interest thereon, if any, to the applicable prepayment date, if redeemed during the twelve-month period beginning on May 14 of the years indicated below.

YEAR	PERCENTAGE
2003	103.0%
2004	102.0%
2005 and thereafter	100.0%

The Company will give each Holder written notice of each prepayment under this SECTION 7.2 not less than thirty (30) days and not more than sixty (60) days prior to the date fixed for such prepayment. Each such notice shall specify such date, the aggregate principal amount and aggregate Accreted Value of the Notes to be prepaid on such date, the principal amount and Accreted Value of each Note held by such Holder to be prepaid, the applicable Accreted Value

Premium and the interest to be paid on the prepayment date with respect to such principal amount being prepaid, and shall be accompanied by an Officers' Certificate as to the estimated prepayment amount due in connection with such prepayment (calculated as if the date of such notice were the date of the prepayment), setting forth the details of such computation. Two Business Days prior to such prepayment, the Company shall deliver to each Holder whose Notes are to be redeemed an Officers' Certificate specifying the calculation of such Accreted Value Premium and the interest to be paid to such Holder as of the specified prepayment date.

7.3 ALLOCATION OF PARTIAL PREPAYMENTS.

In the case of each partial prepayment of the Notes, the principal amount of the Notes to be prepaid shall be allocated among all of the Notes to be prepaid in proportion, as nearly as practicable, to the respective unpaid principal amounts thereof not theretofore called for prepayment.

7.4 MATURITY; SURRENDER, ETC.

In the case of each prepayment of Notes pursuant to this SECTION 7, SECTION 8.24 or Section 8.25, the Accreted Value and the applicable Accreted Value Premium of each Note to be prepaid shall mature and become due and payable on the date fixed for such prepayment, together with interest on the related principal amount accrued to such date. From and after such date, unless the Company shall fail to pay such Accreted Value or Accreted Value Premium when so due and payable, together with the interest, if any, as aforesaid, interest on the related principal amount shall cease to accrue. Any Note paid or prepaid in full shall be surrendered to the Company and cancelled and shall not be reissued, and no Note shall be issued in lieu of any prepaid principal amount of any Note.

7.5 PURCHASE OF NOTES.

Neither the Company nor Vector shall, or shall permit, any of their respective Affiliates (other than a Group Executive or New Valley pursuant to SECTION 9.5) to purchase, redeem, prepay or otherwise acquire, directly or indirectly, any of the outstanding Notes except upon the payment or prepayment of the Notes in accordance with the terms of this Agreement and the Notes unless an offer to purchase Notes on the same terms is made to all Primary Holders. The Company shall promptly cancel all Notes acquired by it or any Affiliate of the Company (other than a Group Executive or New Valley) pursuant to any payment, prepayment, redemption or purchase of Notes, and the Company and any of its Affiliates shall be prohibited from voting any Notes that it may hold prior to such cancellation. Vector shall cause the Company promptly to cancel all Notes acquired by it or any Affiliate of Vector pursuant to any payment, prepayment, redemption or purchase of Notes, and Vector and any of its Affiliates shall be prohibited from voting any Notes that it may hold prior to such cancellation. No Notes may be issued in substitution or exchange for any such Notes purchased by the Company, Vector or any of their respective Affiliates.

7.6 ACCRETED VALUE.

The term "ACCRETED VALUE" for any specified date means, with respect to the principal amount at maturity of any Note, the amount (rounded to the nearest cent) determined by multiplying the Applicable Accretion Percentage by such principal amount.

- (i) The term "APPLICABLE ACCRETION PERCENTAGE" means: in the event that the Accreted Value is to be determined as of a date set forth on SCHEDULE C hereto (each a "MONTHLY ACCRUAL DATE"), the Applicable Accretion Percentage shall equal the percentage set forth on SCHEDULE C hereto for such Monthly Accrual Date;
- (ii) in the event that the Accreted Value is to be determined as of a specified date other than a Monthly Accrual Date, the Applicable Accretion Percentage shall equal the sum of (a) the Applicable Accretion Percentage for the Monthly Accrual Date immediately preceding such specified date and (b) an amount equal to the product of (x) the Applicable Accretion Percentage for the immediately following Monthly Accrual Date LESS the Applicable Accretion Percentage for the immediately preceding Monthly Accrual Date multiplied by (y) a fraction, the numerator of which is the number of days from the immediately preceding Monthly Accrual Date to the specified date, using a 360-day year of twelve 30-day months, and the denominator of which is 30; or
- (iii) if the specified date occurs after the last Monthly Accrual Date, the Applicable Accretion Percentage will equal 100%.

8. COVENANTS.

The Company covenants that so long as any of the Notes are outstanding:

8.1 PAYMENT OF NOTES.

The Company shall pay the principal of and interest on the Notes on the dates and in the manner provided in the Notes.

The Company shall pay interest on overdue principal at the rate borne by the Notes plus 2% and shall pay interest on overdue installments of interest at the same rate, to the extent lawful.

8.2 MAINTENANCE OF OFFICE OR AGENCY.

The Company will maintain in the Borough of Manhattan, the City of New York, an office or agency where Notes may be presented or surrendered for payment, where Notes may be surrendered for registration of transfer or exchange and where notices and demands to or upon the Company in respect of the Notes and this Agreement may be served. The Company hereby initially appoints United States Trust Company of New York as its office or agent for each of said

purposes. The Company will give prompt written notice to the Holders of the location, and any change in the location, of such office or agency. If at any time the Company shall fail to maintain any such required office or agency or shall fail to furnish the Holders with the address thereof, such presentations, surrenders, notices and demands may be made or served at the address of the Company set forth in SECTION 19.

The Company may also from time to time designate one or more other offices or agencies where the Notes may be presented or surrendered for any or all such purposes and may from time to time rescind such designations; PROVIDED, HOWEVER, that no such designation or rescission shall in any manner relieve the Company of its obligation to maintain an office or agency in the Borough of Manhattan, the City of New York, for such purposes.

8.3 LIMITATIONS ON RESTRICTED PAYMENTS.

At any time that the Company does not hold the Required Cash Holdings on the BGLS Balance Sheet, the Company shall not, and shall not permit any of its Restricted Subsidiaries to, directly or indirectly: (i) declare or pay any dividend or make any distribution on account of the Company's or any of its Subsidiaries' Equity Interests (including, without limitation, any payment in connection with any merger or consolidation involving the Company) (other than dividends or distributions payable in Equity Interests (other than Disqualified Equity Interests) of the Company or dividends or distributions payable to the Company or any Restricted Subsidiary of the Company); (ii) purchase, redeem or otherwise acquire or retire for value any Equity Interests of the Company or any direct or indirect parent of the Company or other Affiliate or Subsidiary of the Company (other than any such Equity Interests owned by the Company or any Restricted Subsidiary of the Company); (iii) make any payment (other than regularly scheduled interest payments) on or purchase, redeem, defease or otherwise acquire or retire for value any Indebtedness (including, without limitation, pay any amount owed under any guarantee of the obligations of another Person) (other than the Notes) that is subordinated to or pari passu with the Notes (unless, in the case of pari passu Indebtedness only, such purchase, redemption, defeasance, acquisition, or retirement is made, or offered (if applicable), pro rata with the Notes), except for any scheduled repayment or at the final maturity thereof; (iv) make any Restricted Investment, (v) pay any Expected Postretirement Benefit Obligations in excess of \$1,000,000 in any Purchase Agreement Year or (vi) make any payment (including, without limitation, the payment of any Shadow Dividends), transfer any assets or provide any services in an Affiliated Transaction with Vector or any Affiliate of Vector (including, without limitation, any Unrestricted Subsidiary, but excluding the Company or a Restricted Subsidiary) or any Affiliated Senior Manager other than Permitted Vector Expenses in any Purchase Agreement Year in excess of the sum of (x) \$9,500,000 and (y) the amount of Excess Interest Income received during such Purchase Agreement Year (all such payments and other actions set forth in clauses (i) through (vi) above being collectively referred to as "RESTRICTED PAYMENTS"), unless, at the time of and after giving effect to such Restricted Payment:

- (a) no Default or Event of Default shall have occurred and be continuing or would occur as a consequence thereof;
- (b) the Leverage Ratio, after giving pro forma effect to such Restricted Payment, is less than (i) 2.25 to 1.00, if such Restricted Payment is to be made on or prior to June 30, 2002 and (ii) 2.00 to 1.00, if such Restricted Payment is to be made after June 30, 2002; and

(c) either, at the election of the Company, (i) such Restricted Payment, together with the aggregate of all other Restricted Payments made by the Company and its Restricted Subsidiaries beginning on the first day of the most recent fiscal quarter commencing after the most recent Drop-Below Date (excluding Restricted Payments permitted below), is less than the sum of (x) 50% of the Consolidated Net Income (adjusted to exclude any amounts that are otherwise included in this CLAUSE(C)(I) to the extent there would be, and to avoid, any duplication in the crediting of any such amounts) of the Company for the period (taken as one accounting period) from the beginning of the first fiscal quarter commencing after the most recent Drop-Below Date to the end of the Company's most recently ended fiscal quarter thereafter for which internal financial statements are available at the time of such Restricted Payment (or, if such Consolidated Net Income for such period is a deficit, less 100% of such deficit), PLUS (y) to the extent that any Restricted Investment that was made after the most recent Drop-Below Date is sold for cash or otherwise liquidated or repaid for cash, the amount of net proceeds received by the Company or a Restricted Subsidiary with respect to such Restricted Investment; or

(ii) such Restricted Payment, together with the aggregate of all other Restricted Payments made by the Company and its Restricted Subsidiaries, beginning on the first day of the Drop-Below Quarter (excluding Restricted Payments permitted below), is less than the sum of (x) 50% of the Consolidated Net Income (adjusted to exclude any amounts that are otherwise included in this CLAUSE (C)(II) to the extent there would be, and to avoid, any duplication in the crediting of any such amounts) of the Company for the period (taken as one accounting period) from the beginning of the Drop-Below Quarter to the end of the Company's most recently ended fiscal quarter thereafter for which internal financial statements are available at the time of such Restricted Payment (or, if such Consolidated Net Income for such period is a deficit, less 100% of such deficit), PLUS (y) to the extent that any Restricted Investment that was made after the most recent Drop-Below Date is sold for cash or otherwise liquidated or repaid for cash, the amount of net proceeds received by the Company or a Restricted Subsidiary with respect to such Restricted Investment;

PROVIDED, HOWEVER, that when calculating the amount of the Restricted Payments that may be made subsequent to any Drop-Below Date, the Company shall elect whether to use CLAUSE (C)(I) or CLAUSE (C)(II) prior to or on the date that financial statements for the Drop-Below Quarter are delivered to the Holders pursuant to SECTION 8.9 and so indicate such election in an Officer's Certificate delivered to the Holders with such financial statements, and the Company shall use CLAUSE (C)(I) or CLAUSE (C)(II) to calculate the amount of permissible Restricted Payments as so elected until the next succeeding date on which the Company maintains the Required Cash Holdings on the BGLS Balance Sheet.

The Company and its Restricted Subsidiaries shall be prohibited from making any Restricted Payments during any Black-Out Period. Within fifteen (15) days of the end of any fiscal quarter of the Company during which a Drop-Below Date occurs and Company does not have Required Cash Holdings on the BGLS Balance Sheet on the last day of such fiscal quarter, the Company shall deliver to each Holder an Officer's Certificate (i) acknowledging that the Company and its Restricted Subsidiaries may not make Restricted Payments except in compliance with the preceding paragraph until it once again maintains the Required Cash

Holdings on the BGLS Balance Sheet and (ii) warranting that no Restricted Payments were made during any Black-Out Period during such most-recently ended fiscal quarter.

The foregoing provisions will not prohibit:

- (i) Permitted Payments; and
- (ii) payments permitted pursuant to SECTION 9.3(A).

The amount of all Restricted Payments (other than cash) shall be the fair market value (evidenced by a resolution of the Board of Directors set forth in an Officers' Certificate delivered to the Holders) on the date of the Restricted Payment of the asset(s) proposed to be transferred or services proposed to be provided by the Company or such Restricted Subsidiary, as the case may be, pursuant to the Restricted Payment. If the Company does not have the Required Cash Holdings on the BGLS Balance Sheet at the time of making any Restricted Payment, not later than the date of making any Restricted Payment, the Company shall deliver to the Holders an Officers' Certificate stating that such Restricted Payment is permitted and setting forth the basis upon which the calculations required by this SECTION 8.3 were computed, which calculations may be based upon the Company's latest available financial statements.

8.4 LIMITATION ON INDEBTEDNESS.

- (a) The Company shall not, and shall not permit any Restricted Subsidiary to, incur Indebtedness other than Indebtedness owed to the Company or any Restricted Subsidiary, UNLESS on the date of the incurrence of such Indebtedness:
 - (1) the Leverage Ratio is less than 2.50 to 1.00 after giving pro forma effect to the incurrence of such Indebtedness; and
 - (2) no Default or Event of Default shall have occurred or be continuing or would occur as a consequence thereof.
- (b) The Company shall not guarantee the Indebtedness of any Unrestricted Subsidiary.
- (c) The Company shall not permit any Unrestricted Subsidiary to incur any Indebtedness other than Non-Recourse Indebtedness.
- (d) The Company shall not permit Brooke Holding to incur any Indebtedness.
- (e) For purposes of determining compliance with, and the outstanding principal amount of any particular Indebtedness incurred pursuant to and in compliance with, this Agreement, the amount of Indebtedness issued at a price that is less than the principal amount thereof shall be equal to the amount of liability in respect thereof determined in accordance with GAAP.

8.5 LIMITATION ON TRANSACTIONS WITH AFFILIATES AND INVESTMENTS.

- (a) The Company shall not, and shall not permit any of its Restricted Subsidiaries to engage, directly or indirectly, in any Affiliated Transaction with an Affiliate of the Company EXCEPT (i) any direct payment to or reimbursement of Vector or any Affiliated Senior Manager by the Company or any of its Restricted Subsidiaries of Permitted Vector Expenses not exceeding in any Purchase Agreement Year the sum of (x) \$9,500,000 and (y) the amount of Excess Interest Income received during such Purchase Agreement Year, (ii) as contemplated in SECTION 8.5(B) or (C), (iii) any transaction between any Restricted Subsidiaries and (iv) any transaction between the Company and any Restricted Subsidiary; PROVIDED, HOWEVER, that the Company and its Restricted Subsidiaries may enter into Affiliated Transactions with Unrestricted Subsidiaries so long as (I) the Company or such Restricted Subsidiary would be permitted to do so pursuant to SECTION 8.3, (II) such Affiliated Transaction is fair to the Company or such Restricted Subsidiary as the case may be from a financial point of view as evidenced by a resolution of the Board of Directors of the Company or such Restricted Subsidiary and (III) in the event the amount of such Affiliated Transaction is in excess of \$3,000,000 the Company or such Restricted Subsidiary shall have received an opinion of a Designated Investment Bank that such Affiliated Transaction is fair to the Company or such Restricted Subsidiary from a financial point of view. The Company shall not permit any of its Unrestricted Subsidiaries to engage, directly or indirectly, in any Affiliated Transaction with a person controlled by the Company (other than (i) any Unrestricted Subsidiary or (ii) the Company or any Restricted Subsidiary to the extent permitted by the preceding sentence) unless the terms of such Affiliated Transaction are no less favorable to such Unrestricted Subsidiary, as the case may be, than would have been obtainable in an arms-length transaction with unrelated persons. The Company shall not, and shall not permit any Restricted Subsidiary to, engage, directly or indirectly, in any Affiliated Transaction with a person controlled by the Company (other than (i) another Restricted Subsidiary or the Company or (ii) an Unrestricted Subsidiary to extent permitted by the first sentence of this SECTION 8.5(A)). The Company shall not, and shall not permit any Restricted Subsidiary to make, directly or indirectly, an Investment in an Affiliate of the Company (other than the Company or another Restricted Subsidiary); PROVIDED, HOWEVER, that the Company and its Restricted Subsidiaries may make Investments in Unrestricted Subsidiaries to the extent permitted by SECTION 8.3.
- (b) So long as any Notes remain outstanding, the Company shall not, and shall not permit any of its Restricted Subsidiaries to, directly or indirectly, enter into any Affiliated Transaction with an Affiliated Senior Manager for consideration in excess of \$100,000 in any Purchase Agreement Year (except as permitted by PARAGRAPH (A) above if such Affiliated Senior Manager had been an Affiliate of the Company or PARAGRAPH (c) below).
- (c) The foregoing PARAGRAPHS (A) and (b) shall not prevent (i) the Company from making Restricted Payments in accordance with SECTION 8.3; (ii) the Company and any of its Subsidiaries or Affiliates from entering into securities brokerage and securities underwriting transactions with subsidiaries of Ladenburg Thalmann Financial Services, Inc. ("LTFS") at such subsidiary's usual and customary rates and on usual and customary terms, so long as such rates and terms are in accordance with securities industry practice for comparable brokerage firms; (iii) any Disposition of Assets effected in compliance with SECTION 10.1; (iv) payments of the type permitted pursuant to SECTION 9.3(A); (v) guarantees of Indebtedness of Subsidiaries of the Company by Vector; (vi) Vector from making capital contributions to the Company and (vii) the incurrence of Indebtedness by the Company and any Restricted Subsidiary owing to any Unrestricted Subsidiary so long as such Indebtedness is incurred in accordance

with SECTION 8.4 and the total cost of capital to the Company or such Restricted Subsidiary of such Indebtedness is less than 12% per annum; PROVIDED, HOWEVER, that the limitation contained in the foregoing CLAUSE (VII) on cost of capital to the Company or any Restricted Subsidiary shall be deemed not to include any Equity Interests in Vector or warrants for Equity Interests in Vector issued to an Unrestricted Subsidiary in connection with and as additional consideration for such Unrestricted Subsidiary extending such Indebtedness.

- (d) Nothing in this SECTION 8.5 or anywhere else in this Agreement shall be deemed to prohibit the payment of dividends, tax sharing payments or management fees from Liggett to Brooke Holding or from Brooke Holding to the Company.
- (e) The Company shall not, and shall not permit any Restricted Subsidiary to, incur any Indebtedness owing to any Group Executive.
- (f) In the event that any aircraft owned by the Company or any Restricted Subsidiary is used by Vector or any Affiliate of Vector other than the Company or a Restricted Subsidiary, Vector or such Affiliate of Vector shall compensate the Company or such Restricted Subsidiary, as the case may be, for such use in an amount equal to the cost of such use.
- (g) Nothing in this Agreement shall be deemed to prohibit a merger between New Valley or a wholly owned Subsidiary of New Valley and Vector or a wholly owned Subsidiary of Vector; PROVIDED, HOWEVER, that such wholly owned subsidiary shall not be the Company or any successor to the Company pursuant to SECTION 10.1.

8.6 INVESTMENT COMPANY ACT.

The Company shall not register as, or be required to register as, an "investment company" as defined under the Investment Company Act. The Company shall deliver to the Holders, within thirty (30) days after the date thereof, written notice of any event which, but for the provisions of Rule 3a-2 (or any successor rule) under the Investment Company Act, would have caused it to be deemed an "investment company" as defined in the Investment Company Act.

8.7 LIMITATION ON LIENS.

Except for (i) Liens arising under the Security Documents and (ii) Permitted Liens on Collateral other than Pledged Stock and Cash and cash equivalents, the Company shall not, and shall not permit any Subsidiary Pledgor to, directly or indirectly, create, incur, assume or permit to exist any Lien on or with respect to any Collateral or any income or profits therefrom.

8.8 NO VIOLATION OF THE MARGIN RULES.

 $\qquad \qquad \text{The Company shall not, directly or indirectly, use the proceeds of the Notes to purchase or carry, or to extend credit to others for \\$

the purpose of purchasing or carrying, any Margin Stock in violation of Regulations T, U or X of the Board of Governors of the Federal Reserve System.

8.9 FINANCIAL REPORTS.

The Company shall deliver to the Holders, within fifteen (15) days after it files with the SEC, copies of the quarterly and annual reports and the information, documents, and other reports (or copies of such portions of any of the foregoing as the SEC may by rules and regulations prescribe) which the Company is required to file with the SEC pursuant to Section 13 or 15(d) of the Exchange Act. If the Company is not subject to the requirements of Section 13 or 15(d) of the Exchange Act, then it shall deliver to the Holders, within fifteen (15) days after it would have been required to file with the SEC, financial statements, including any notes thereto (and, in the case of a fiscal year end, an auditors' report by a firm of established national reputation reasonably satisfactory to the Majority Holders), comparable to that which it would have been required to include in such quarterly or annual reports, information, documents or other reports, as the case may be, if it were subject to the requirements of Section 13 or 15(d) of the Exchange Act.

So long as the Notes remain outstanding, the Company shall cause any annual reports to stockholders, containing audited consolidated financial statements, and any other financial reports furnished by the Company to stockholders and quarterly or other financial reports filed with the SEC and furnished by the Company to stockholders or proxy or information statements furnished to stockholders, to be delivered to the Holders (no later than 10 Business Days after the date such materials are mailed or made available to the stockholders) and will cause to be disclosed in such annual reports as of the date of the most recent financial statements in each such report the amount available for dividends and other payments pursuant to SECTION 8.3.

So long as the Notes remain outstanding, the Company shall cause to be delivered to the Holders copies of (i) (A) consolidated audited annual financial statements prepared in accordance with GAAP of (w) Liggett, (x) VTUSA, (y) New Valley and (z) LTFS and (B) a calculation of the Leverage Ratio as of the last day of the Reference Period most recently ended which calculation shall set forth in detail how the numeric amount of the Leverage Ratio was derived, within one hundred five (105) days of close of the fiscal year of the Company, and (ii) (A) consolidated quarterly financial statements prepared in accordance with GAAP of (w) Liggett, (x) VTUSA, (y) New Valley and (z) LTFS and (B) a calculation of the Leverage Ratio as of the last day of the Reference Period most recently ended which calculation shall set forth in reasonable detail how the numeric amount of the Leverage Ratio was derived, within sixty (60) days of the end of each of the first three fiscal quarters of each fiscal year of the Company. Concurrently with the delivery of the financial statements of Liggett and VTUSA required by the preceding sentence, the Company shall deliver to the Holders income statements for each brand of Liggett and VTUSA which income statements shall include the items for each brand set forth on EXHIBIT 8.9.

All annual consolidated financial statements of the Company and its Subsidiaries shall be accompanied by a report thereon from PricewaterhouseCoopers or another firm of high established national reputation reasonably satisfactory to the Majority Holders.

The Company shall provide to the Holders within 10 Business Days of delivery to the Holders the reports or financial statements referred to in the first paragraph of this SECTION 8.9, an Officers' Certificate setting forth for the fiscal quarter or fiscal year, as the case may be, covered by such reports or financial statements (i) the aggregate amount of all dividends paid by the Company, (ii) the estimated amount (reasonably determined) as of the last day of such fiscal year or quarter of net operating loss, capital loss and tax credit carryover ("TAX SHARING ATTRIBUTES") utilizable by each Subsidiary of the Company which is included in the VGR Group for purposes of determining the amount due under any tax sharing agreement or arrangement whether or not in writing, with respect to which any such Subsidiary may be required to make a payment, (iii) all Restricted Payments, to the extent not disclosed pursuant to clause (i) above, and, when the Company is not maintaining the Required Cash Holdings on the BGLS Balance Sheet, the sum of clause (c)(i) or (c)(ii), as applicable, in the first paragraph of SECTION 8.3, (iv) in the event that Vector is no longer required to file periodic reports with the SEC pursuant to the Exchange Act, all information relating to compensation and management and consulting fees paid to Group Executives required to be included in periodic reports filed by Vector with the SEC on the date hereof, (v) the computation of all payments in respect of tax sharing arrangements, (vi) the calculation of the Net Available Proceeds of all Company Asset Sales consummated in such fiscal quarter or fiscal year, (vii) a detailed schedule of the Indebtedness of the Company and the Restricted Subsidiaries (including, without limitation, any guarantees of the obligations of other Persons) which schedule shall include, without limitation, the principal amounts, interest rates, maturity dates, amortization schedule and all other material information about each type of Indebtedness listed and (viii) the balance of all Group Accounts as of the most recent date for which such information is available.

The Company shall deliver to the Holders a schedule of the actual Tax Sharing Attributes of the Subsidiaries of the Company not later than fifteen (15) days after the filing of the consolidated federal income tax return of the VGR Group.

8.10 WAIVER OF STAY, EXTENSION OR USURY LAWS.

The Company covenants (to the extent that it may lawfully do so) that it will not at any time insist upon, plead, or in any manner whatsoever claim or take the benefit or advantage of any stay or extension law or any usury law or other law which would prohibit or forgive the Company from paying all or any portion of the principal or Accreted Value of or interest and Accreted Value Premium (if any) on the Notes as contemplated herein, wherever enacted, now or at any time hereafter in force, or which may affect the covenants or the performance of this Agreement; and (to the extent that it may lawfully do so) the Company hereby expressly waives all benefit or advantage of any such law, and covenants that it will not hinder, delay or impede the execution of any power herein granted to the Holders, but will suffer and permit the execution of every such power as though no such law had been enacted.

8.11 [RESERVED.]

8.12 COMPLIANCE WITH LAWS.

The Company shall and shall cause each of its Subsidiaries to comply with all applicable statutes, rules, regulations, orders and restrictions of the United States of America, states and municipalities, and of any Governmental or

Regulatory Authority, in respect of the conduct of its businesses and the ownership of its properties, except such as are being contested in good faith and by appropriate proceedings in such manner as not to cause any material adverse effect upon the business, properties, operations, condition (financial or other) of the Company and the Subsidiaries taken as a whole and except for such noncompliances as will not in the aggregate have a Material Adverse Effect.

8.13 EXISTENCE.

Subject to SECTION 10, the Company shall, and shall cause each of its Subsidiaries to, do or cause to be done all things necessary to preserve and keep in full force and effect its corporate or other existence and its rights (charter and statutory) and franchises; PROVIDED, HOWEVER, that the Company shall not be required to preserve any right or franchise if the Board of Directors of the Company shall determine that the preservation thereof is no longer desirable in the conduct of the business of the Company and that the loss thereof is not disadvantageous in any material respect to the Holders.

8.14 PAYMENT OF TAXES AND OTHER CLAIMS.

The Company shall pay or discharge or cause to be paid or discharged, before the same shall become delinquent (1) all taxes, assessments and governmental charges (including withholding taxes and any penalties, interest and additions to taxes) levied or imposed upon the Company or any Subsidiary or upon the income, profits or property of the Company or any Subsidiary (including, without limitation, any taxes levied on, or in respect of, the Collateral) and (2) all lawful claims for labor, materials and supplies which, if unpaid, might by law become a Lien upon the property of the Company or any Subsidiary; PROVIDED, HOWEVER, that neither the Company nor any Subsidiary shall be required to pay or discharge or cause to be paid or discharged any such tax, assessment, charge or claim whose amount, applicability or validity is being contested in good faith by appropriate proceedings and for which disputed amounts adequate reserves have been made.

8.15 MAINTENANCE OF PROPERTIES.

- (a) The Company shall cause all properties used or useful in the conduct of its business or the business of any Subsidiary to be maintained and kept in good condition, repair and working order and supplied with all necessary equipment and shall cause to be made all necessary repairs, renewals, replacements, betterments and improvements thereof, all as in the judgment of the Company may be necessary, so that the business carried on in connection therewith may be properly and advantageously conducted at all times; PROVIDED, HOWEVER, that nothing in this SECTION 8.15 shall prevent the Company from discontinuing the operation or maintenance of any of such properties, if such discontinuance is, in the judgment of the Board of Directors of the Company, desirable in the conduct of the business of the Company and not disadvantageous in any material respect to the Holders.
- (b) The Company shall not substitute or exchange any Equity Interests of New Valley included in the Collateral for other Property other than in connection with any sale permitted by SECTION 8.5(G), SECTION 8.17(E) or SECTION 13.8.

8.16 COMPLIANCE CERTIFICATE; INSPECTION RIGHTS.

(a) The Company shall deliver to the Holders within ninety (90) days after the end of its fiscal year (which on the date hereof is the calendar year) an Officers' Certificate of the Company stating that a review of its activities during the preceding fiscal year has been made under the supervision of the signing Officers with a view to determining whether it has kept, observed, performed and fulfilled its obligations under this Agreement and further stating, as to each such Officer signing such certificate, that to the best of his or her knowledge, during such preceding fiscal year it has kept, observed, performed and fulfilled each and every such covenant and no Default or Event of Default occurred during such year. If such signers do know of such a Default or Event of Default and its status with particularity and the action the Company or the Subsidiary concerned is taking with respect thereto.

The Company shall deliver to the Holders, forthwith upon becoming aware of (i) any Default or Event of Default in the performance of any covenant, agreement or condition contained in this Agreement, (ii) any default under any other mortgage, indenture or instrument to which the Company or any Restricted Subsidiary is a party, which together with all other mortgages, indentures or instruments under which the Company or any Restricted Subsidiary is then in default, evidences Indebtedness in excess of \$3,000,000 in the aggregate or (iii) any Account Notice Event, an Officers' Certificate specifying with particularity such Default, Event of Default or default.

- (b) The Company shall permit the representatives of any Holder of Notes:
 - (i) NO DEFAULT if no Default or Event of Default then exists, at the expense of such Holder and upon reasonable prior notice to the Company, to visit the principal executive office of the Company, to discuss the affairs, finances and accounts of the Company and the Subsidiaries with the Company's officers, and (with the consent of the Company, which consent will not be unreasonably withheld) its independent public accountants, and (with the consent of the Company, which consent will not be unreasonably withheld) to visit the other offices and properties of the Company and each Subsidiary, all at such reasonable times and as often as may be reasonably requested in writing; and
 - (ii) DEFAULT if a Default or Event of Default then exists, at the expense of the Company, to visit and inspect any of the offices or properties of the Company or any Subsidiary, to examine all their respective books of account, records, reports and other papers, to make copies and extracts therefrom, and to discuss their respective affairs, finances and accounts with their respective officers and independent public accountants (and by this provision the Company authorizes said accountants to discuss the affairs, finances and accounts of the Company and the Subsidiaries), all at such times and as often as may be requested.

- 8.17 OWNERSHIP OF NV HOLDINGS; ADDITIONAL COVENANTS RELATING TO NV HOLDINGS.
- (a) So long as the Notes are outstanding, the Company shall continue to own all the Equity Interests of NV Holdings and any options, warrants or other rights to acquire such Equity Interests.
- (b) Subject to SECTION 8.5(G), the Company shall cause NV Holdings' sole function to be to hold Equity Interests and Investments in New Valley. The Company shall not permit NV Holdings to conduct any business activities other than to hold such Equity Interests, to make such payments and to conduct incidental activities required in order to oversee and manage its interests in New Valley permitted pursuant to this SECTION 8.17.
 - (c) The Company shall not permit NV Holdings to incur any Indebtedness.
- (d) Except as contemplated by the NV Holdings Pledge Agreement, the Company shall not permit NV Holdings to, directly or indirectly, create, incur, assume or permit to exist any Lien on or with respect to any of its assets, including, without limitation, the New Valley Assets, or any income or profits therefrom including any dividends and distributions received thereon or distributable with respect thereto, or interest therein.
- (e) Subject to SECTION 8.5(G), the Company shall not permit NV Holdings to make, directly or indirectly, any Investments in cash or by delivery of Property (x) in any Person, whether by the acquisition of Indebtedness or other obligation, security or Equity Interest, or by loan, advance, extension of credit or capital contribution, or otherwise, or (y) in any Property, except (i) purchases of Equity Interests in New Valley so long as such Equity Interests shall be pledged hereunder and under the Security Agreements or (ii) Investments in Publicly-held Companies to the extent received as a result of any dividend in kind, distribution, spin off, split up, corporate reorganization or similar transaction with respect to New Valley in which such Investment is received without the payment of further consideration by NV Holdings so long as such Investments shall be pledged hereunder and under the Security Agreements.
- (f) The Company shall not permit NV Holdings to (i) consolidate with or merge with or into any other Person, (ii) transfer (by lease, assignment, sale or otherwise) in a single transaction or through a series of transactions, all or substantially all of its properties and assets, as an entirety or substantially as an entirety, to another Person or group of Persons or (iii) adopt a Plan of Liquidation.
 - 8.18 RESTRICTIONS ON ISSUANCE OF ADDITIONAL EQUITY INTERESTS.
- (a) So long as any Notes remain outstanding, the Company shall not permit any Restricted Subsidiary to issue, directly or indirectly, any additional Equity Interests or options, warrants, convertible or exchangeable securities or other rights to acquire such Equity Interests to any Person other than the Company or a Restricted Subsidiary, except (other than in the case of Brooke Holding and VTUSA) for shares of non-voting, non-convertible, non-exchangeable preferred stock the issuance of which (i) would not at the time of issuance or thereafter cause such Subsidiary to cease to be a member of a

hypothetical affiliated group for federal income tax purposes of which the Company is the parent and (ii) complies with PARAGRAPH (B) below.

(b) So long as any Notes remain outstanding, the Company shall not permit any Liggett Subsidiary or Subsidiary of VTUSA to issue, directly or indirectly, any preferred stock unless (i) the cumulative liquidation preferences of all such preferred stock issuances is less than \$10,000,000; (ii) such preferred stock has no right to participate with other Equity Interests in the sharing of the assets of the issuer after satisfaction of its stated liquidation preference; (iii) such preferred stock is not issued to any Group Executive or any other Vector Expanded Affiliate (other than the Company, any of its Subsidiaries or any officer or director of such Subsidiary who would not be a Vector Expanded Affiliate but for such status as an officer or director), (iv) the Board of Directors of such Subsidiary shall have determined that such transaction is fair to such Subsidiary from a financial point of view and (v) such Subsidiary shall have received an opinion from a Designated Investment Bank

8.19 OPINIONS, APPRAISALS, ETC.

Whenever an opinion, appraisal or similar document (other than an Officers' Certificate) is required to be provided to the Holders or the Collateral Agent under this Agreement, any Security Agreement or otherwise in connection with a transaction contemplated hereby, the person providing such opinion, appraisal or similar document shall not be an Affiliate or Associate of the Company or any Group Executive or any other Vector Expanded Affiliate nor a person subject to an agreement to cause such person to become such an Affiliate or Associate or any other Vector Expanded Affiliate.

8.20 OWNERSHIP OF RESTRICTED SUBSIDIARIES.

Except as permitted under SECTION 8.18(A) and SECTION 13.8, the Company shall continue to own, directly or indirectly, all of the Equity Interests and any options, warrants or other rights to acquire such Equity Interests of the Restricted Subsidiaries, and such ownership shall be directly held by the Company (including, for the avoidance of doubt, any successor thereto under SECTION 10) in the case of Brooke Holding and VTUSA.

8.21 GROUP ACCOUNTS.

- (a) All Cash and cash equivalents held or received by the Company and the Subsidiary Pledgors at any time shall be held in the Group Accounts. The Company shall cause any Cash and cash equivalents held at the end of any Business Day by the Subsidiary Pledgors in excess of the difference between (i) \$500,000 and (ii) the amount of Cash and cash equivalents held in the BGLS Checking Account at the end of such Business Day to be transferred to the Company and placed in the Securities Account.
- (b) The aggregate balance of the Checking Accounts shall not exceed 500,000 at the end of any Business Day.
- (c) The Company shall not, and shall not permit any Subsidiary Pledgor to, establish or maintain any (i) "securities account" as defined in the Uniform

Commercial Code or (ii) any "deposit account" as defined in the Uniform Commercial Code other than the Group Accounts.

- (d) The Company shall keep only Cash and cash equivalents in the Securities ${\sf Account.}$
 - 8.22 [RESERVED.]
 - 8.23 FISCAL YEAR.

The Company will not, for purposes of determining compliance with this Agreement, change the last day of its fiscal year from December 31 of each year.

8.24 OFFER TO PURCHASE UPON TRIGGERING ASSET SALE.

Within thirty (30) days following the consummation of a Triggering Asset Sale, a Restricted Subsidiary Asset Sale if required by SECTION 8.29 or a Vector Equity Offering pursuant to SECTION 8.25, (the "AVAILABLE PROCEEDS OFFER DATE"), the Company shall make an offer to all Holders (an "AVAILABLE PROCEEDS OFFER") to apply the applicable Available Proceeds Offer Purchase Amount to the acquisition of Notes at a purchase price (the "AVAILABLE PROCEEDS OFFER PURCHASE PRICE") equal to (i) in the case of an Available Proceeds Purchase Date prior to May 14, 2003, 105% of the accreted value of the Notes to be purchased plus accrued interest to the Available Proceeds Purchase Date and (ii) in the case of an Available Proceeds Purchase Date and price payable if the Company were redeeming such Notes pursuant to SECTION 7.2(B), plus accrued interest to such Available Proceeds Purchase Date.

The Company shall give the Holders notice that it is making an Available Proceeds Offer pursuant to this SECTION 8.24. Notice of an Available Proceeds Offer shall be mailed by the Company to all Holders not less than thirty (30) days nor more than sixty (60) days before the Available Proceeds Purchase Date. The Available Proceeds Offer shall remain open from the time of mailing until five (5) days before the Available Proceeds Purchase Date. The notice shall contain all instructions and materials necessary to enable such Holders to accept the Available Proceeds Offer. The notice, which shall govern the terms of the Available Proceeds Offer, shall state:

- (1) that the Available Proceeds Offer is being made pursuant to this SECTION 8.24;
- (2) the aggregate principal amount of Notes for which the Available Proceeds Offer is being made, the amount of the interest that shall have accrued thereon to the Available Proceeds Purchase Date, the purchase price (including the amount of accrued interest) for the Notes, the Available Proceeds Offer Purchase Amount and the Available Proceeds Purchase Date;
- (3) that any Note not purchased pursuant to the Available Proceeds Offer shall continue to accrue interest;

- (4) that any Note accepted for payment and for which payment is made pursuant to the Available Proceeds Offer shall cease to accrue interest after the Available Proceeds Purchase Date;
- (5) that Holders electing to have a Note purchased pursuant to an Available Proceeds Offer shall be required to deliver the form entitled "Option of Holder to Elect Purchase" attached to the Note completed and duly executed, to the Company at the address specified in the notice at least five (5) days before the Available Proceeds Purchase Date;
- (6) that Holders shall be entitled to withdraw their election if the Company receives, not later than five (5) days prior to the Available Proceeds Purchase Date, a telegram, telex, facsimile transmission or letter setting forth the name of the Holder, the principal amount of the Note the Holder delivered for purchase and a statement that such Holder is withdrawing his election to have the Note purchased:
- (7) (x) that if Holders have accepted the Available Proceeds Offer with respect to Notes in a principal amount in excess of the principal amount of Notes for which the Available Proceeds Offer is being made, the Company shall purchase such Notes on a pro rata basis (with such adjustments as may be deemed appropriate by the Company so that only Notes in denominations of \$1,000 or integral multiples of \$1,000 shall be acquired) and (y) the procedures set forth in the second full succeeding paragraph of this SECTION 8.24; and
- (8) that Holders whose Notes were purchased only in part will be issued new Notes equal in principal amount to the unpurchased portion of the Notes surrendered.

Each such notice shall be accompanied by a copy of the information regarding the Company required to be contained in a Quarterly Report on Form 10-Q for the fiscal quarter of the Company ended immediately prior to the Available Proceeds Offer Date or, if such fiscal quarter ended less than forty-five (45) days prior to the Available Proceeds Offer Date in the case of the first three fiscal quarters or ninety (90) days prior to the Available Proceeds Offer Date in the case of the fourth fiscal quarter, for the preceding fiscal quarter. Notwithstanding the foregoing, if the Available Proceeds Offer Date is during the second fiscal quarter of the Company, such notice may include only a copy of the information required to be contained in an Annual Report to Shareholders pursuant to Rule 14a-3 under the Exchange Act for the fiscal year of the Company ending immediately prior to such Available Proceeds Offer Date.

Each Holder accepting an Available Proceeds Offer (an "ACCEPTING HOLDER") may include in such notice of acceptance an agreement to have prepaid, in addition to the Available Proceeds Offer Purchase Amount allocable to each Note of such Holder, all or any part of the balance of the principal amount of each such Note, specifying the maximum principal amount of each such Note, which such Accepting Holder is willing to have prepaid. Upon receipt of all timely acceptances from Accepting Holders, the Company shall allocate that portion of the Available Proceeds Offer Purchase Amount that would have been allocated to the Holders who did not accept the Available Proceeds Offer among the Notes of

Accepting Holders in proportion to the respective Available Proceeds Offer Purchase Amount allocated to the Notes of Accepting Holders. If the portion of the Available Proceeds Offer Purchase Amount thus allocated to the Notes of an Accepting Holder would exceed the maximum principal amount of such Notes which such Accepting Holder has agreed to have prepaid, such excess shall be allocated among the Notes of Accepting Holders who have agreed to accept prepayments in an amount which still exceeds the amount of prepayments previously allocated to them pursuant to this paragraph in proportion to the respective Available Proceeds Offer Purchase Amount allocable to the Notes of such Accepting Holders; and such allocation shall be repeated as many times as shall be necessary until the Available Proceeds Offer Purchase Amount has been fully allocated or until it is no longer possible to allocate the Available Proceeds Offer Purchase Amount without exceeding the maximum principal amount of the Notes which all Accepting Holders respectively have agreed to have prepaid.

Before 11:00 a.m., New York City time, on an Available Proceeds Purchase Date, the Company shall (i) deliver payment to each Holder, in U.S. Legal Tender, amounts sufficient to pay the Available Proceeds Purchase Price of all Notes or portions thereof held by such so accepted and (ii) deliver to each relevant Holder a new Note equal in principal amount to any unpurchased portion of the Note surrendered.

An Available Proceeds Offer shall be made by the Company in compliance with all applicable laws, including, without limitation, the requirements of Regulation 14E under the Exchange Act, any other tender offer rules under the Exchange Act and all other applicable Federal and state securities laws.

8.25 OFFER TO PURCHASE UPON CHANGE OF CONTROL.

In the event of a Change of Control, each Holder shall have the option to cause the Company to purchase the Notes held by such Holder in whole but not in part, at a price (the "CHANGE OF CONTROL PURCHASE PRICE") equal to 101% of the Accreted Value thereof, plus accrued interest to the date of purchase (which date shall be no less than twenty-five (25) Business Days and no more than fifty (50) Business Days following the delivery of notice to the Holders of such Change of Control) (the "CHANGE OF CONTROL PURCHASE DATE"). Within five (5) Business Days after any Change of Control, the Company shall notify the Holders in writing (i) of such Change of Control and (ii) that each Holder has the option to cause the Company to repurchase the Notes owned by such Holder pursuant to this SECTION 8.25. Each Holder electing to have its Notes purchased pursuant to this SECTION 8.25 is required to deliver to the Company, no later than 5:00 p.m., New York City time, on the date that is five (5) Business Days prior to the Change of Control Purchase Date, the form attached to the Notes entitled "Option of Holder to Elect Purchase" completed and duly executed. Each Holder who elects to have its Notes purchased shall be entitled to revoke its election by delivering a written notice of such revocation to the Company prior to 5:00 p.m., New York City time, on the date that is five (5) Business Days prior to the Change of Control Purchase Date. To the extent permitted by law, such revocation shall be irrevocable. All questions as to the validity, eligibility (including time of receipt) and acceptance of any Note for repurchase or any notice of revocation will be determined by the Company, which determination shall be final and binding.

Before 11:00 a.m., New York City time, on a Change of Control Purchase Date, the Company shall deliver payment to each Holder, in U.S. Legal Tender in an amount sufficient to pay the Change of Control Purchase Price of all Notes owned by such Holder.

Any offer by the Company to purchase Notes following a Change of Control shall be made in compliance with all applicable laws, including, without limitation, the requirements of Regulation 14E under the Exchange Act, any other tender offer rules under the Exchange Act and all other applicable Federal and state securities laws.

As used herein, "CHANGE OF CONTROL" means Bennett S. LeBow (other than by his death or Incapacity) either (i) ceasing to control the Company (including, for the avoidance of doubt, any successor thereto under Section 10) or (ii) ceasing to be, together with his Immediate Family, the beneficial owner (as described in Rule 13d-3 under the Exchange Act) of at least 25% of Vector Voting Power; PROVIDED, HOWEVER, that Bennett S. LeBow, together with his Immediate Family, may be the beneficial owner of less than 25% of Vector Voting Power without causing a "Change of Control" if (x) his and his Immediate Family's collective beneficial ownership of Vector Voting Power falls below 25% as a result of a Vector Equity Offering and the dilution associated therewith after the date of this Agreement, and (y) the Company causes the Net Vector Offering Proceeds of such Vector Equity Offering to be used FIRST, to the extent that Vector or the Company elects (or is required by the terms of any Indebtedness of any Restricted Subsidiary), for the Retirement of Restricted Subsidiary Indebtedness, and, SECOND, in the event that Cumulative Net Remaining Proceeds then exceed \$5,000,000, to redeem the Notes pursuant to an Available Proceeds Offer in accordance with SECTION 8.24. For the purposes of clause (i) of the preceding sentence, "control," when used with respect to any Person, means the power to direct the management and policies of such person, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise.

8.26 DIVIDEND AND OTHER PAYMENT RESTRICTIONS AFFECTING SUBSIDIARIES.

The Company shall not, and shall not permit any of its Subsidiaries to, directly or indirectly, create or otherwise cause or suffer to exist or become effective any encumbrance or restriction on the ability of any Restricted Subsidiary to (a)(i) pay dividends or make any other distributions to the Company or any of its Restricted Subsidiaries (A) on its Equity Interests or (B) with respect to any other interest or participation in, or measured by, its profits, or (ii) pay any indebtedness owed to the Company or any of its Restricted Subsidiaries, (b) make loans or advances to the Company or any of its Restricted Subsidiaries, except for such encumbrances or restrictions existing under or by reason of:

- (i) this Agreement and the Notes;
- (ii) applicable law;
- (iii) any instrument governing Indebtedness or Equity Interests of a Person acquired by the Company or any of its Restricted Subsidiaries as in effect at the time of such acquisition (except to the extent such Indebtedness was incurred in connection with or in

contemplation of such acquisition), which encumbrance or restriction is not applicable to any Person, or the properties or assets of any Person, other than the Person, or the property or assets of the Person, so acquired, PROVIDED that, in the case of Indebtedness, such Indebtedness was permitted by the terms of this Agreement to be incurred:

- (iv) by reason of customary non-assignment provisions in leases and licenses entered into in the ordinary course of business and consistent with past practices;
- (v) purchase money obligations or Capitalized Lease Obligations for property acquired in the ordinary course of business that impose restrictions of the nature described in CLAUSE (C) above only on the property so acquired;
- (vi) agreements relating to the financing of the acquisition of real or tangible personal property acquired after the date of this Agreement, PROVIDED, that such encumbrance or restriction relates only to the property which is acquired and in the case of any encumbrance or restriction that constitutes a Lien, such Lien constitutes a Permitted Lien: or
- (vii) any encumbrance or restriction pursuant to any Liggett Senior Credit Facility in effect on the date hereof and any encumbrance or restriction pursuant to any Liggett Senior Credit Facility entered into after the date hereof; PROVIDED, HOWEVER, that any encumbrance or restriction contained in any Liggett Senior Credit Facility entered into after the date hereof is no less favorable as a whole to the Holders of the Notes than the encumbrances and restrictions contained in any Liggett Senior Credit Facility in effect on the date hereof.

8.27 RULE 144A INFORMATION.

The Company shall (a) upon the request of any Holder make available the information necessary to permit sales pursuant to Rule 144A and (b) take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell Notes without registration under the Securities Act pursuant to the exemptions provided by Rule 144A. Upon the request of any Holder, the Company shall deliver to such Holder a written statement as to whether they have complied with such information requirements.

8.28 LIMITATION ON RESTRICTED SUBSIDIARY BUSINESS.

The Company shall not permit any Restricted Subsidiary to engage in any business other than a Related Business.

8.29 LIMITATION OF RESTRICTED SUBSIDIARY ASSET SALES.

No Restricted Subsidiary may enter into any Restricted Subsidiary Asset Sale unless:

- (a) the Leverage Ratio would have been less than 2.00 to 1.00 for the Reference Period most recently ended had the contemplated Restricted Subsidiary Asset Sale occurred on the first day of such Reference Period;
- (b) no Default or Event of Default shall have occurred and be continuing or would occur as a consequence of such Restricted Subsidiary Asset Sale; and
- (c) the Board of Directors of such Restricted Subsidiary shall have determined that such Restricted Subsidiary Asset Sale is fair to such Restricted Subsidiary from a financial point of view;

PROVIDED, HOWEVER, that in the event that after giving pro forma effect to any Restricted Subsidiary Asset Sale, the Company would be in compliance with CLAUSES (B) and (c) above but not clause (a), the Company may still permit such Restricted Subsidiary Asset Sale if both (1) the Company causes the proceeds of such Restricted Subsidiary Asset Sale to be used, FIRST, to the extent that the Company elects (or is required by the terms of any Indebtedness of any Restricted Subsidiary), for the Retirement of Restricted Subsidiary Indebtedness, and, SECOND, in the event that Cumulative Net Remaining Proceeds then exceed \$5,000,000, to redeem the Notes pursuant to an Available Proceeds Offer in accordance with SECTION 8.24 and (2) after giving pro forma effect to such Restricted Subsidiary Asset Sale, the Leverage Ratio is lower than it was prior to giving such pro forma effect.

8.30 TOBACCO LITIGATION BONDS.

In the event that the Company or a Restricted Subsidiary agrees to or becomes required to post a Tobacco Litigation Bond, the Company shall forthwith, and no later than one (1) Business Day thereafter, deliver an Officer's Certificate to each Holder setting forth the circumstances surrounding and the amount of such Tobacco Litigation Bond.

9. ADDITIONAL COVENANTS.

The Company covenants that so long as any of the Notes are outstanding:

- 9.1 VECTOR TOBACCO CREDIT FACILITY.
- (a) Vector hereby agrees that it shall not accept any payments from VTUSA or Vector Tobacco Ltd. pursuant to the Vector Tobacco Credit Facility. In the event that Vector receives any payments from any Person under the Vector Tobacco Credit Facility, it shall promptly transfer the full amount of such payments to the Company.
- (b) The Company shall cause VTUSA and Vector Tobacco Ltd. to make payments owed under the Vector Tobacco Credit Facility and any related documentation solely to the Company.

9.2 [RESERVED.]

- 9.3 TAX SHARING PAYMENTS.
- (a) (i)(A) Within one hundred twenty (120) days after the last day of each calendar year, the Company shall pay to Vector with respect to such calendar year, an amount equal to the sum of (x) The Tax Payment for such calendar year and (y) the Tax Audit Payment for such calendar year.
 - (1) The Tax Payment for a calendar year shall equal (x) the Tax Percentage for such calendar year multiplied by (y) Company Net Income for such calendar year, less Company Credits for such calendar year.
 - (2) The Tax Audit Payment for a calendar year shall equal (i) any amounts payable by Vector as common parent of the VGR Group or as equivalent agent with respect to state and local income tax returns in connection with any audit or equivalent proceeding with respect to federal, state or local income taxes of BGLS or any Subsidiary for any year beginning on or after January 1, 1996 (including any expenses of contesting such audits).
 - (B) Within fifteen (15) days prior to the last day of each of the calendar quarters of the Company, the Company may make interest free advances to Vector in an amount not exceeding the Tax Payment for such quarter. The Tax Payment for such quarter shall equal an amount not exceeding the amount determined by multiplying the Tax Percentage for the calendar year of the Company in which such quarter ends by Company Group Net Income for such quarter as reasonably, and in good faith, determined by the Company (which determination for the last calendar quarter in each Purchase Agreement Year shall be made by a resolution of the Board of Directors of the Company) and deducting therefrom Company Credits, as reasonably, and in good faith, determined by the Company (which determination for the last calendar quarter in each Purchase Agreement Year shall be made by a resolution of the Board of Directors of the Company).
 - (C) Any advances with respect to a calendar year under SECTION 9.3(A)(I)(B) above less than or equal to the amount determined for such calendar year under SECTION 9.3(A)(I)(A) above shall be considered forgiven and treated as a distribution to Vector. If the advances under SECTION 9.3(A)(I)(B) above exceed the amount determined under SECTION 9.3(A)(I)(A) above, the excess shall be repaid to the Company within fifteen (15) days of notice from the Company to Vector of such excess. Such notice shall be mailed by the Company no less than five (5) days after the filing of the federal income tax return of the Company for the relevant calendar year with the Internal Revenue Service.
 - (D) Upon ten (10) days' notice by Vector to BGLS certifying that a Tax Audit Payment is payable by Vector, BGLS shall pay Vector the amount of such Tax Audit Payment.

- (ii) Within ten (10) days prior to the due date of any actual tax payment by Vector with respect to any combined, consolidated or unitary state and/or local intangibles or net worth tax payable with respect to a combined, consolidated or unitary group which includes the Company, the Company may remit to Vector the amount of such tax so payable except to the extent such tax is attributable to Vector on a separate company basis.
- (b) Notwithstanding anything provided herein to the contrary, the Company shall not make any payment pursuant to SECTION 9.3(A)(I) hereof but excluding any payments referred to in SECTION 9.3(A)(II) hereof, unless immediately upon the receipt of any such payment Vector shall make or cause to be made the cash capital contributions or payments required under SECTION 9.4(B).

9.4 TAX MATTERS.

- (a) Vector shall terminate any tax sharing agreement currently in place between it and NV Holdings. Vector shall not receive any payments with respect to federal, state and local income taxes of the Company and its Subsidiaries other than the payments provided for in SECTION 9.3 and any such payments payable to Vector by the Company or its Subsidiaries shall instead be paid to the Company.
- (b) Immediately upon the receipt of any, payments from BGLS pursuant to SECTION 9.3(A)(I), Vector shall make a cash capital contribution to the Company in an amount equal to the excess of (i) the cumulative amount of any payments to Vector from the Company pursuant to SECTION 9.3(A) made on or after January 1, 2001 as reduced by any prior capital contributions from Vector pursuant to this SECTION 9.4, over (ii) the sum of (x) aggregate Tax Audit Payments payable by Vector and (y) the cumulative amount of any federal income taxes (and, where the relevant payment includes an amount with respect to state or local income taxes determined on a consolidated or combined basis ("STATE COMBINED TAXES"), the amount of such State Combined Taxes) actually payable by Vector (and not refundable to Vector as the result of a loss carryback or otherwise) to the relevant taxing authority with respect to the periods beginning on or after January 1, 1996 through and including the period with respect to which such Tax Sharing Payment was made ("RELEVANT PERIODS") which are attributable to the Company Group under the principles set forth in Treasury Regulation Sections 1.1552-1(a)(1) and 1.1502-33(d)(2). The amount of the capital contributions required under this SECTION 9.4 with respect to any payment made by BGLS with respect to estimated tax liability ("ESTIMATED TAX PAYMENTS") shall be made on an estimated basis at the time such Estimated Tax Payments are made. To the extent such estimated capital contributions with respect to a taxable year exceed the amount of capital contributions determined to be required based on a complete taxable year, the Company shall pay to Vector the amount of such excess.
- (c) The Company shall not receive any payments with respect to federal, state and local taxes of the Company and the members of the Company Group expect as provided in this Section 9.4(c). The members of the Company Group (including the Company) shall pay to the Company in sufficient time to for the Company to make its payments to Vector pursuant to Sections 9.3 and 9.4 the amounts payable by the Company to Vector pursuant to such Sections. The amounts payable by the member of the Company Group (including the Company) to the Company shall be

apportioned among such members (including the Company) under the principles set forth in Treasury Regulations Sections 1.1552-1(a)(1) and 1.1502-33(d)(2). If Vector makes any capital contributions to the Company pursuant to Section 9.4(b), the Company shall recompute the amounts apportionable to the members of the Company Group (including the Company) taking into account the net amount paid by the Company to Vector (i.e. the amount paid by the Company to Vector less the amount contributed by Vector to the Company) and applying the principles set forth in Treasury Regulations Sections 1.1552-1(a)(1) and 1.1502-33(d)(2). The Company shall then directly or indirectly contribute such amounts to the members of the Company Group so that the net amount paid by members of the Company (i.e. the amount paid by the members of the Company, as reduced by amounts contributed to such members or retained by the Company) are equal to the recomputed amounts apportionable to them.

- (d) In the event of a change of the obligor or a change in the position of the obligor within the VGR Group or of Vector ceasing to be the common parent of the VGR Group, Vector agrees to cause tax sharing agreements or arrangements to be entered into among members of the VGR Group (to the extent any such member is not prohibited from entering into such agreements or arrangements under the terms of any indenture or other debt agreement to which is it a party) so that (i) the position of the obligor as to its payment with respect to its tax liability and that of its subsidiaries and the position of the common parent of the VGR Group as to its receipt of amounts with respect to such tax liability and (ii) the position of all direct and indirect subsidiaries of the obligor with respect to payments to the obligor is preserved in all material respects.
- (e) Whenever any payment is made by any corporation to the common parent on an estimated basis, the payment shall, to the extent not remitted to the appropriate taxing authority on the first business day following the day of receipt by the common parent, be immediately contributed by the common parent to the obligor hereunder.

9.5 VOTING OF NOTES BY GROUP EXECUTIVES AND NEW VALLEY.

Neither any Group Executive nor New Valley shall be permitted to vote any Notes that any of them may hold from time to time under any circumstances, including, without limitation, in the event of a bankruptcy of the Company. In the event of a bankruptcy of the Company, all Group Executives and New Valley shall be deemed to be an entity designated in 11 U.S.C. ss.1126(e) for purposes of determining acceptance of allowed claims pursuant to 11 U.S.C. ss.1126(c).

9.6 TRANSFER OF COMPANY EQUITY INTERESTS

Vector shall not sell, transfer, hypothecate, pledge, transfer or in any way dispose of any Equity Interest in the Company.

10. SUCCESSOR PERSON.

10.1 MERGERS, CONSOLIDATIONS.

The Company shall not (i) consolidate with or merge with or into any other Person, (ii) transfer (by lease, assignment, sale or otherwise) in a single transaction or through a series of transactions, all or substantially all

of its properties and assets, as an entirety or substantially as an entirety, to another Person or group of affiliated persons (a "DISPOSITION OF ASSETS") or (iii) adopt a Plan of Liquidation, unless:

- (1) the Company shall be the continuing Person, or the Person (if other than the Company) (or, in the case of CLAUSE (II), one Person to which assets are transferred) formed by such consolidation or into which the Company is merged or to which the properties and assets of the Company are transferred or leased as an entirety or substantially as an entirety or pursuant to a Plan of Liquidation, shall be a corporation organized and existing under the laws of the United States or any State thereof or the District of Columbia and shall expressly assume, by an agreement supplemental hereto, executed and delivered to the Holders, in form satisfactory to the Holders, all the obligations of the Company under the Notes, this Agreement and the Security Agreements;
- (2) the person (or, in the case of CLAUSE (II), one person to which assets are transferred, which person shall be the same person fulfilling the conditions set forth in CLAUSE (1)) formed by such consolidation or surviving such merger or to which the properties and assets of the Company, are transferred or leased as an entirety or substantially as an entirety or pursuant to a Plan of Liquidation shall have a Tangible Net Worth immediately after such transaction, equal to or greater than that of the Company, immediately preceding, and without giving effect to, such transaction;
- (3) immediately before and immediately after giving effect to such transaction, no Default or Event of Default shall have occurred and be continuing;
- (4) the Leverage Ratio shall not be greater than 2.00 to 1.00 for the Reference Period most recently ended had the contemplated transaction occurred on the first day of such Reference Period; and
- (5) the Company shall have delivered to the Holders an Officers' Certificate of the Company and an Opinion of Counsel, each stating that such consolidation, merger, transfer or lease and such supplemental agreement comply with this SECTION 10 and that all conditions precedent herein provided relating to such transaction have been complied with.

10.2 SUCCESSOR PERSON SUBSTITUTED.

Upon any consolidation or merger, or any transfer or lease of assets of the Company in accordance with SECTION 10.1, the successor person formed by such consolidation or into which the Company is merged or to which such transfer is made shall succeed to, and be substituted for, and may exercise every right and

power of, the Company under this Agreement with the same effect as if such successor person had been named as the Company herein. When a successor person assumes all of the obligations of the Company hereunder and under the Notes, except in the case of succession by lease, the Company shall be released from such obligations.

11. EVENTS OF DEFAULT.

An "EVENT OF DEFAULT" shall exist if any of the following conditions or events shall occur and be continuing:

- (1) the Company defaults in the payment of interest on any Notes when the same becomes due and payable and the default continues for a period of thirty (30) days;
- (2) the Company defaults in the payment of the principal or Accreted Value of, or Accreted Value Premium on, any Notes when the same becomes due and payable at maturity, upon acceleration, upon redemption, pursuant to SECTION 7.2, 8.24 or 8.25 hereof or otherwise;
- (3) any Document Party fails to comply with any of its other agreements contained in the Note Documents and the default continues for the period and after the notice specified below or any representation or warranty made by any Document Party in any Note Document shall have been untrue in any material respect when made;
- (4) Vector, Research, New Valley, VTUSA, Brooke Overseas, Liggett, Brooke Holding, any Group Executive or any other Person, as applicable, fails to comply with (A) SECTION 7.5, 8.5, 8.25, 9.1, 9.3, 9.4, 9.5 or 9.6 or (B) its acknowledgment and undertaking under any Note Document, and the default continues for the period and after the notice specified below;
- (5) an Other Obligation Payment Default or Other Obligation Acceleration Default shall have occurred; PROVIDED, HOWEVER, that an Other Obligation Payment Default and Other Obligation Acceleration Default shall not constitute an Event of Default unless, at such time, one or more Other Obligation Payment Defaults or Other Obligation Acceleration Defaults shall have occurred and be continuing with respect to Other BGLS Group Debt the outstanding principal amount of which exceeds in the aggregate \$3,000,000;
- (6) the Company, any Restricted Subsidiary, NV Holdings, New Valley or any New Valley Transferee pursuant to or within the meaning of any Bankruptcy Law (A) becomes insolvent, (B) fails generally to pay its debts as they become due, (C) admits in writing its inability to pay its debts generally as they become due, (D) commences a voluntary case or proceeding, (E) consents to the entry of a judgment, decree or order for relief against it in an involuntary case or proceeding, (F) consents to or acquiesces to the appointment of a Custodian of it or for any part of its property, (G) consents to or acquiesces in the institution of a bankruptcy or an insolvency proceeding against it, (H) applies for, consents to or acquiesces in the appointment of or taking possession by a Custodian of it or for any part of its property, (I) makes a general assignment for the benefit of its creditors or (J) takes any corporate action in furtherance of or to facilitate, conditionally or otherwise, any of the foregoing; PROVIDED that no such

event with respect to any New Valley Transferee referred to in this CLAUSE (6) shall constitute an "Event of Default" unless such event and all other such events with respect to any New Valley Transferee have a material adverse effect on the financial condition of New Valley and its subsidiaries taken as a whole;

- (7) a court of competent jurisdiction enters a judgment, decree or order for relief in respect of the Company, any Restricted Subsidiary, NV Holdings, New Valley or any New Valley Transferee in an involuntary case or proceeding under any Bankruptcy Law which shall (A) approve as properly filed a petition seeking reorganization, arrangement, adjustment or composition of it, (B) appoint a Custodian of it or for any part of its property or (C) order the winding-up or liquidation of its affairs; and such judgment, decree or order shall remain unstayed and in effect for a period of thirty (30) consecutive days; or any bankruptcy or insolvency petition or application is filed, or any bankruptcy or insolvency proceeding is commenced, against the Company, any Restricted Subsidiary, NV Holdings, New Valley or any New Valley Transferee and such petition, application or proceeding is not dismissed within sixty (60) days; or any warrant of attachment is issued against any portion of the property of the Company, any Restricted Subsidiary, NV Holdings, New Valley or any New Valley Transferee which is not released within sixty (60) days of service; PROVIDED that no such event with respect to any New Valley Transferee referred to in this CLAUSE (7) shall constitute an "Event of Default" unless such event and all other such events with respect to any New Valley Transferee have a material adverse effect on the financial condition of New Valley and its subsidiaries taken as a whole;
- (8) final judgments not covered by insurance for the payment of money which, in the case of any one of the Company, any Restricted Subsidiary, NV Holdings, New Valley or any New Valley Transferee considered individually or in the aggregate at any one time exceed \$250,000 shall be rendered against the Company, any Restricted Subsidiary, NV Holdings, New Valley or any New Valley Transferee by a court of competent jurisdiction and shall remain undischarged or unbonded for a period (during which execution shall not be effectively stayed) of forty-five (45) days after such judgment becomes final and nonappealable and the default continues for the period and after the notice specified below; PROVIDED that no such judgment with respect to any New Valley Transferee referred to in this clause (8) shall constitute an "Event of Default" unless such default and all other such judgments with respect to any New Valley Transferee have a material adverse effect on the financial condition of New Valley and its subsidiaries taken as a whole; or
- (9) (a) any of the Security Agreements or any provision thereof (i) shall cease to be in full force and effect in any material respect, (ii) shall cease to give the Collateral Agent on behalf of the Holders a valid and perfected Lien on the Collateral intended to be covered thereby or (iii) shall cease to be enforceable against the Company for any reason, (b) the repudiation by the Company of any of its obligations under the Security Agreements or (c) the Company shall default in the due performance or observance of any term, covenant or agreement on their part to be performed or observed pursuant to the Security Agreements for a period of thirty (30) days after Company first becomes aware of the existence of such failure or ten (10) days after written notice thereof is delivered to Company by the Collateral Agent or any Holder of such failure (or such shorter period following such notice as may be required in any Security Agreement).

The term "BANKRUPTCY LAW" means Title 11, U.S. Code or any similar Federal or state law for the relief of debtors. The term "CUSTODIAN" means any receiver, trustee, assignee, liquidator, sequestrator or similar official under any Bankruptcy Law.

A Default under CLAUSE (3), (4) or (8) above (other than any Defaults under SECTIONS 8.3, 8.4, 8.5, 8.6, 8.7, 8.8, the second paragraph of 8.16(A) and 10.1 of this Agreement, which Defaults shall be Events of Default without the notice or passage of time specified in this paragraph) is not an Event of Default until the Majority Holders notify the Company, or, in the case of a Default under said CLAUSE (4), any Group Executive or Vector (as applicable) and the Company, of the Default, and the Company does not cure the Default within thirty (30) days after receipt of the notice. The notice must specify the Default, demand that it be remedied and state that the notice is a "Notice of Default."

12. REMEDIES ON DEFAULT, ETC.

12.1 ACCELERATION.

If an Event of Default (other than an Event of Default specified in SECTION 11(6) or (7) as a result of a case or proceeding in which the Company is the subject debtor) occurs and is continuing, the Holders of at least 25% in aggregate principal amount of the Notes then outstanding may, by notice to the Company declare the Accreted Value and accrued interest to the date of acceleration on the Notes then outstanding (if not then due and payable) to be and become due and payable and, upon any such declaration, the same shall be and become due and payable. If an Event of Default specified in SECTION 11(6) or (7) as a result of a case or proceeding in which the Company is the subject debtor occurs, the Accreted Value and accrued interest on the Notes then outstanding shall IPSO FACTO become and be immediately due and payable without any declaration or other act on the part of any Holder. Except as otherwise provided in this Agreement, upon payment of the Accreted Value of and interest, together with any default interest, on the Notes all of the Company's obligations under the Notes and this Agreement shall terminate. The Majority Holders may rescind an acceleration and its consequences if (i) all existing Events of Default, other than the non-payment of the principal or Accreted Value of, or Accreted Value Premium on, the Notes which has become due solely by such declaration of acceleration, have been cured or waived, (ii) to the extent the payment of such interest is lawful, interest on overdue installments of interest and overdue principal, Accreted Value or Accreted Value Premium, which has become due otherwise than by such declaration of acceleration, has been paid and (iii) the rescission would not conflict with any judgment or decree of a court of competent jurisdiction. No rescission of an acceleration under the preceding sentence shall extend to or affect any subsequent Event of Default or Default or impair any right consequent thereon.

If an Event of Default occurs by reason of any willful action (or inaction) taken (or not taken) by or on behalf of any Group Executive, Vector, the Company or any Subsidiary with the intention of avoiding payment of the Accreted Value Premium that the Company would have had to pay if the Company then had elected to redeem the Notes pursuant to SECTION 7.2 hereof, then, upon acceleration of the Notes, an equivalent premium shall also become and be immediately due and payable, to the extent permitted by law, anything in this

Agreement or in the Notes to the contrary notwithstanding. If an Event of Default occurs by reason of any willful action (or inaction) taken (or not taken) by or on behalf of the Company with the intention of avoiding the prohibition on redemption of the Notes prior to such date, then, upon acceleration of the Notes, an additional premium shall also become and be immediately due and payable to the extent permitted by law in an amount, for each twelve-month period beginning on May 14 of the years set forth below, (expressed as a percentage of the Accreted Value of the Notes on the date of payment that would otherwise be due but for the provisions of this sentence):

12/11	PERCENTAGE
2001	105% 103% 102%
2005 and thereafter	100%

12.2 OTHER REMEDIES.

If any Default or Event of Default has occurred and is continuing, and irrespective of whether any Notes have become or have been declared immediately due and payable under SECTION 12.1, the Holder of any Note at the time outstanding may proceed to protect and enforce the rights of such Holder by an action at law, suit in equity or other appropriate proceeding, whether for the specific performance of any agreement contained herein or in any Note, or for an injunction against a violation of any of the terms hereof or thereof, or in aid of the exercise of any power granted hereby or thereby or by law or otherwise.

12.3 NO WAIVERS OR ELECTION OF REMEDIES, EXPENSES, ETC.

No course of dealing and no delay on the part of any Holder of any Note in exercising any right, power or remedy shall operate as a waiver thereof or otherwise prejudice such Holder's rights, powers or remedies. No right, power or remedy conferred by this Agreement or by any Note upon any Holder thereof shall be exclusive of any other right, power or remedy referred to herein or therein or now or hereafter available at law, in equity, by statute or otherwise. Without limiting the obligations of the Company under SECTION 16, the Company will pay to the Holder of each Note on demand such further amount as shall be sufficient to cover all costs and expenses of such Holder incurred in any enforcement or collection under this SECTION 12, including, without limitation, reasonable attorneys' fees, expenses and disbursements.

13. SECURITY INTEREST.

13.1 PLEDGE AND SECURITY INTEREST.

The Security Interest as now or hereafter in effect shall be held for the equal and ratable benefit and security of the Holders, without preference, priority or distinction of any thereof over any other by reason of difference in time of issuance, sale or otherwise, and for the enforcement of the payment of principal of and interest on the Notes, in accordance with their terms.

The Collateral shall consist of the Property described in and subject to the Lien of the Security Agreements. Prior to the issuance of the Notes, the Company will execute and deliver, file and record, all instruments and documents necessary to subject the Collateral to the Security Interest.

The Company shall not, and shall not permit any of its Subsidiaries to, dispose of any of the Collateral except in accordance with this Agreement and the Security Agreements.

So long as any Notes remain outstanding, the Company shall apply, and cause each Document Party to apply, the Net Available Proceeds of any sale of the Collateral in accordance with SECTION 8.24 of this Agreement.

13.2 RELEASE OF COLLATERAL.

The Security Interest in any Collateral shall automatically and without any action by the Collateral Agent or the Holders be released upon any sale of such Collateral by any Document Party pursuant to the terms of this Agreement and the Security Agreements.

13.3 RELIANCE ON OPINION OF COUNSEL.

The Collateral Agent shall, before taking any action under this SECTION 13 or the Security Agreements, be entitled to receive an Opinion of Counsel in form and substance satisfactory to the Collateral Agent, stating (a) the legal effect of such action, (b) the steps necessary to consummate the same and perfect the Collateral Agent's security interest with respect to the Collateral, (c) that such action will not be in contravention of the provisions of this Agreement and (d) that such opinion shall be full protection to the Collateral Agent for any action taken or not taken in reliance thereon.

13.4 PURCHASER MAY RELY.

A purchaser in good faith of the Collateral or any part thereof or interest therein which is purported to be transferred or granted by the Collateral Agent as provided in this SECTION 13 or the Security Agreements may rely on the authority of the Collateral Agent to execute a transfer, grant or release, and shall not be bound to ascertain or inquire as to the satisfaction of any conditions precedent to the exercise of such authority, or to see the application of the purchase price therefor.

13.5 PAYMENT OF EXPENSES.

On demand of the Collateral Agent, the Company forthwith shall pay, or satisfactorily provide for, all reasonable expenditures incurred by the Collateral Agent under this SECTION 13 and the Security Agreements, and all such sums shall be a Lien prior to the Notes upon the Collateral and shall be secured thereby.

13.6 SUITS TO PROTECT THE COLLATERAL.

To the extent permitted under the Security Agreements, the Collateral Agent shall have power to institute and to maintain such suits and proceedings as it may deem expedient to prevent any impairment of the Collateral by any acts

which may be unlawful or in violation of the Security Agreements or this Agreement, including the power to institute and maintain suits or proceedings to restrain the enforcement of or compliance with any legislative or other governmental enactment, rule or order that may reasonably be believed to be unconstitutional or otherwise invalid if the enforcement of, or compliance with, such enactment, rule or order would impair the security hereunder or be prejudicial to the interests of the Holders or of the Collateral Agent.

13.7 COLLATERAL AGENT'S DUTIES.

The powers conferred upon the Collateral Agent by this SECTION 13 and the Security Agreements are solely to protect its interest and the interest of the Holders in the Collateral and shall not impose any duty upon the Collateral Agent to exercise any such powers except as expressly provided in this Agreement and the Security Agreements. The Collateral Agent shall be under no duty whatsoever to any Document Party to make or give any presentment, demand for performance, notice of nonperformance, protest, notice of protest, notice of dishonor, or other notice or demand in connection with any Collateral, or to take any steps necessary to preserve any rights against third parties except as expressly provided in this Agreement. The Collateral Agent shall not be liable to any Document Party for failure to collect or realize upon any or all of the Collateral, or for any delay in so doing, nor shall the Collateral Agent be under any duty to any Document Party to take any action whatsoever with regard thereto. The Collateral Agent shall have no duty to any Document Party to comply with any recording, filing, or other legal requirements necessary to establish or maintain the validity, priority or enforceability of, or the Collateral Agent's rights in or to, any of the Collateral, except with regard to the safekeeping of any Collateral perfected by the Collateral Agent's possession thereof, and, with regard to such safekeeping, the Collateral Agent shall be liable only for any damages that result from its willful misconduct or gross negligence.

13.8 SALE OF COLLATERAL.

So long as (i) no Default or Event of Default shall have occurred and be continuing or would occur as a result of any contemplated sale of Collateral and (ii) the Leverage Ratio would be less than 2.00 to 1.00 for the Reference Period most-recently ended after giving pro forma effect to any contemplated sale of Collateral, the Company may, or may permit any other Document Party to, sell, transfer or otherwise dispose of any of the Collateral (other than a Disposition of Assets effected in compliance with SECTION 10.1, PROVIDED that no such Collateral may be sold, transferred or otherwise disposed of, directly or indirectly to any Vector Expanded Affiliate (other than the Company or, in the case of any Equity Interests in New Valley, NV Holdings) and PROVIDED further that in the case of the Equity Interests of Liggett, Brooke Holding, Research, VTUSA or NV Holdings, no such Equity Interests shall be sold, transferred or otherwise disposed of unless (i) in the case of Liggett or Brooke Holding, all of such Equity Interests so held by the Company are sold as part of the same transaction or series of substantially contemporaneous related transactions, (ii) any of the Designated Investment Banks has delivered an opinion to the Company that such transaction is fair to the Company (assuming for such purpose that the Company consists only of the Company and its Subsidiaries determined as

of the date of the rendering of such opinion), (iii) such Equity Interests are sold only for Cash or cash equivalents and/or securities (such securities, "SECURITY PROCEEDS") and customary contract rights that are incidental to such securities or to the effectuation of the transaction, (iv) any Cash or cash equivalents received are deposited in a Company Account and (v) all Security Proceeds are pledged to the Collateral Agent pursuant to the Security Agreements; PROVIDED that no such Security Proceeds may be sold or otherwise transferred by the Company (A) to any Affiliate or (B) for any consideration except Cash or cash equivalents which shall be placed into a Company Account.

13.9 DEALINGS WITH COLLATERAL.

The Company shall, and shall cause each other Document Party to, only take action in respect of the Collateral in good faith and with due regard to the preservation of the value of the Collateral (including, without limitation, the sale, purchase and voting of the Collateral).

13.10 ANNUAL REPORTS.

On or within thirty (30) days before each anniversary of the date and execution and delivery of this Agreement, the Company shall furnish to the Collateral Agent and each Holder one or more Opinions of Counsel stating (1) what action has been taken with respect to the recording, registering, filing, re-recording, reregistering and re-filing of this Agreement and the Security Agreements and all necessary financing statements, notifications of secured transactions and other instruments as may be necessary to make effective and maintain the Security Interest contemplated hereby and thereby and reciting the details of such action (including the jurisdictions in which such actions were taken), or stating that no such action was required, and (2) what, if any, action of the foregoing character may reasonably be expected to become necessary during the next year to so maintain the Security Interest contemplated by this Agreement and the Security Agreements.

13.11 SECURITIES ACCOUNT.

(a) The Holders shall not instruct the Collateral Agent to deliver a Notice of Sole Control to the Securities Intermediary under the Account Control Agreement unless an Account Notice Event shall have occurred and be continuing; PROVIDED, HOWEVER, that if an Account Notice Event ceases to be continuing after the delivery of a Notice of Sole Control, each Holder shall promptly instruct the Collateral Agent to deliver a Notice of End of Sole Control upon such Holder's receiving an Officer's Certificate from the Company that such Account Notice Event has been cured as well as any evidence reasonably requested by such Holder that such Account Notice Event has been cured. The Company hereby acknowledges that the Majority Holders may instruct the Collateral Agent to deliver a Notice of Sole Control to the Securities Intermediary when an Account Notice Event has occurred and is continuing. Each Holder hereby acknowledges that the Majority Holders may instruct the Collateral Agent to deliver a Notice of End of Sole Control upon the Majority Holders receiving an Officer's Certificate from the Company that such Account Notice Event has been cured as well as evidence reasonably requested by the Majority Holders without regard to information requested by any other Holder that such Account Notice Event has been cured.

- (b) In the event that the Majority Holders instruct the Collateral Agent to deliver a Notice of Sole Control due to an Other Obligation Payment Default, the Holders, subject to SECTION 13.11(D) and SECTION 13.11(E), shall not instruct the Collateral Agent to take any further action with respect to the Securities Account (other than to protect the rights of the Company, the Collateral Agent and the Holders with respect thereto against outside parties) until (i) such Account Notice Event has been cured at which point the Holders shall instruct the Collateral Agent to deliver a Notice of End of Sole Control in accordance with SECTION 13.11(A) or (ii) five (5) Business Days after such Other Obligation Payment Default has become an Other Obligation Acceleration Default.
- (c) In the event that the Majority Holders instruct the Collateral Agent to deliver a Notice of Sole Control due to an Other Obligation Acceleration Default, the Holders, subject to SECTION 13.11(D) and SECTION 13.11(E), shall not instruct the Collateral Agent to take any further action with respect to the Securities Account (other than to protect the rights of the Company, the Collateral Agent and the Holders with respect thereto against outside parties) until five (5) Business Days after the Holders have instructed the Collateral Agent to deliver such Notice of Sole Control.
- (d) In the event that an Account Notice Event caused by an Other Obligation Payment Default or an Other Obligation Acceleration Default could be cured by the payment by the Company or a Restricted Subsidiary of an amount not to exceed \$5,000,000, the Holders, upon the written request of the Company, shall instruct the Collateral Agent (i) to withdraw funds from the Securities Account equal to the amount necessary to cure such Account Notice Event and (ii) to deliver such funds to the Company in exchange for an Officer's Certificate from the Company certifying that the Company will cause such funds to be used to cure such Account Notice Event, and the Company shall use such funds to cure such Account Notice Event; PROVIDED, HOWEVER, that in the case of an Account Notice Event caused by an Other Obligation Acceleration Default, the Holders shall be under no obligation to instruct the Collateral Agent to deliver funds from the Securities Account to the Company to cure such Account Notice Event if the Company has not made a written request to the Holders to give such instruction to the Collateral Agent prior to the fifth (5th) Business Day after the Holders have instructed the Collateral Agent to deliver a Notice of Sole Control.
- (e) In the event that an Account Notice Event caused by an Other Obligation Payment Default or an Other Obligation Acceleration Default could be cured by the payment by the Company or a Restricted Subsidiary of an amount in excess of \$5,000,000, the Holders, upon the written request of the Company, shall instruct the Collateral Agent (i) to withdraw funds from the Securities Account equal to the amount necessary to cure such Account Notice Event and (ii) to deliver such funds to the Company in exchange for an Officer's Certificate from the Company certifying that (x) the Company will cause such funds to be used to cure such Account Notice Event, and (y) after giving pro forma effect to the cure of such Account Notice Event, the Leverage Ratio would be less than 2.50 to 1.00, and the Company shall use such funds to cure such Account Notice Event; PROVIDED, HOWEVER, that the Holders shall be under no obligation to deliver a written instruction to the Collateral Agent pursuant to this SECTION 13.11(E) unless after giving pro forma effect to the cure of such Account Notice Event, the Leverage Ratio would be less than 2.50 to 1.00; and PROVIDED,

FURTHER, that in the case of an Account Notice Event caused by an Other Obligation Acceleration Default, the Holders shall be under no obligation to instruct the Collateral Agent to deliver funds from the Securities Account to the Company to cure such Account Notice Event if the Company has not made a written request to the Holders to give such instruction to the Collateral Agent prior to the fifth (5th) Business Day after the Holders have instructed the Collateral Agent to deliver a Notice of Sole Control.

- (f) In the event that the Majority Holders instruct the Collateral Agent to deliver a Notice of Sole Control due to a Non-Grace Period Covenant Acceleration Default, the Holders shall not instruct the Collateral Agent to take any further action with respect to the Securities Account (other than to protect the rights of the Company, the Collateral Agent and the Holders with respect thereto against outside parties) until five (5) Business Days after the Holders have instructed the Collateral Agent to deliver such Notice of Sole Control.
- 14. REGISTER OF NOTES; EXCHANGE AND SUBSTITUTION OF NOTES; LEGENDS.

14.1 REGISTER OF NOTES.

The Company shall keep at its principal executive office a register for the registration and registration of transfers of Notes. The name and address of each Holder of one or more Notes, each transfer thereof and the name and address of each transferee of one or more Notes shall be registered in such register. Prior to due presentment for registration of transfer, the Person in whose name any Note shall be registered shall be deemed and treated as the owner and Holder thereof for all purposes hereof, and the Company shall not be affected by any notice or knowledge to the contrary. The Company shall give to any Holder of a Note promptly upon request therefor, a complete and correct copy of the names and addresses of all registered Holders of Notes.

14.2 TRANSFER AND EXCHANGE OF NOTES.

Upon surrender of any Note at the principal executive office of the Company for registration of transfer or exchange (and in the case of a surrender for registration of transfer, duly endorsed or accompanied by a written instrument of transfer duly executed by the registered Holder of such Note or his attorney duly authorized in writing and accompanied by the address for notices of each transferee of such Note or part thereof), the Company shall execute and deliver, at the Company's expense (except as provided below), one or more new Notes (as requested by the Holder thereof) in exchange therefor, in an aggregate principal amount equal to the unpaid principal amount of the surrendered Note; PROVIDED, that in connection with any such transfer, the Holder requesting the transfer shall provide to the Company evidence reasonably satisfactory to the Company that the transfer is to a "qualified institutional buyer" or an "accredited investor," as such terms are defined in Rules 144A and 501, respectively, of the Securities Act, and the transfer is exempt from the registration requirements of the Securities Act and, if the transfer is to an entity or person other than a "qualified institutional buyer," the Company shall be provided with an opinion of counsel reasonably satisfactory to the Company that the transfer is so exempt from the registration requirements of the Securities Act. Each such new Note shall be payable to such Person as such Holder may request and shall be substantially in the form of EXHIBIT A. Each

such new Note shall be dated and bear interest from the date to which interest shall have been paid on the surrendered Note or dated the date of the surrendered Note if no interest shall have been paid thereon. The Company may require payment of a sum sufficient to cover any stamp tax or governmental charge imposed in respect of any such transfer of Notes. Notes shall not be transferred in denominations of less than \$100,000, PROVIDED that if necessary to enable the registration of transfer by a Holder of its entire holding of Notes, one Note may be in a denomination of less than \$100,000. Any transferee, by its acceptance of a Note registered in its name (or the name of its nominee), shall be deemed to have made the representation set forth in SECTION 6.2.

14.3 REPLACEMENT OF NOTES.

Upon receipt by the Company of notice from any Holder of Notes of the loss, theft, destruction or mutilation of any Note, and

- (a) in the case of loss, theft or destruction, of indemnity reasonably satisfactory to it, or
 - (b) in the case of mutilation, upon surrender and cancellation thereof,

the Company at its own expense shall execute and deliver, in lieu thereof, a new Note, dated and bearing interest from the date to which interest shall have been paid on such lost, stolen, destroyed or mutilated Note or dated the date of such lost, stolen, destroyed or mutilated Note if no interest shall have been paid thereon.

14.4 LEGENDS ON NOTES.

Each Note issued pursuant to this Agreement shall bear one or more legends in substantially the following form:

"THIS NOTE HAS NOT BEEN registered under the Securities Act of 1933, as amended (the "Securities Act"), OR ANY STATE SECURITIES LAWS. NEITHER THIS SECURITY NOR ANY INTEREST OR PARTICIPATION HEREIN may be offered, sold, ASSIGNED, TRANSFERRED, pledged, ENCUMBERED or otherwise DISPOSED OF IN THE ABSENCE OF SUCH registration UNLESS SUCH TRANSACTION IS EXEMPT FROM, OR NOT SUBJECT TO, SUCH REGISTRATION AND ANY APPLICABLE STATE SECURITIES LAWS HAVE BEEN COMPLIED WITH.

THE HOLDER OF THIS NOTE BY ITS ACCEPTANCE HEREOF AGREES NOT TO OFFER, SELL, ASSIGN, TRANSFER OR OTHERWISE DISPOSE OF THIS NOTE, PRIOR TO THE DATE THAT IS TWO YEARS (OR SUCH SHORTER PERIOD THAT MAY HEREAFTER BE PROVIDED UNDER RULE 144(k) UNDER THE SECURITIES ACT AS PERMITTING RESALES BY NON-AFFILIATES OF RESTRICTED SECURITIES WITHOUT RESTRICTION) AFTER THE LATER OF THE ORIGINAL ISSUE DATE HEREOF AND THE LAST DATE ON WHICH THE COMPANY OR ANY AFFILIATE OF THE COMPANY WAS THE OWNER OF THIS NOTE (OR ANY PREDECESSOR OF SUCH SECURITY) EXCEPT (A) TO THE COMPANY, (B) PURSUANT TO A REGISTRATION STATEMENT WHICH HAS BEEN DECLARED EFFECTIVE UNDER THE SECURITIES ACT, (C) FOR SO LONG AS THIS NOTE IS ELIGIBLE FOR RESALE PURSUANT TO RULE 144A UNDER THE SECURITIES ACT, TO A "QUALIFIED INSTITUTIONAL BUYER" (AS DEFINED IN RULE 144A) THAT PURCHASES FOR ITS OWN ACCOUNT OR FOR THE ACCOUNT OF A QUALIFIED

INSTITUTIONAL BUYER TO WHOM NOTICE IS GIVEN THAT THE TRANSFER IS BEING MADE IN RELIANCE ON RULE 144A, (D) IN AN OFFSHORE TRANSACTION MEETING THE REQUIREMENTS OF RULE 903 OR 904 UNDER THE SECURITIES ACT, (E) TO AN INSTITUTIONAL "ACCREDITED INVESTOR" WITHIN THE MEANING OF RULE 501(A)(1), (2), (3) OR (7) UNDER THE SECURITIES ACT THAT IS PURCHASING THIS NOTE FOR ITS OWN ACCOUNT FOR INVESTMENT PURPOSES AND NOT WITH A VIEW TO, OR FOR OFFER OR SALE IN CONNECTION WITH, ANY DISTRIBUTION IN VIOLATION OF THE SECURITIES ACT OR (F) PURSUANT TO ANOTHER AVAILABLE EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT, SUBJECT TO THE COMPANY'S AND THE TRANSFER AGENT'S RIGHT PRIOR TO ANY SUCH OFFER, SALE OR TRANSFER PURSUANT TO CLAUSE (D), (E) OR (F) TO REQUIRE THE DELIVERY OF AN OPINION OF COUNSEL, CERTIFICATIONS AND/OR OTHER INFORMATION SATISFACTORY TO THE COMPANY AND THE REGISTRAR, AND IN EACH OF THE FOREGOING CASES, A CERTIFICATE OF TRANSFER IN THE FORM APPEARING ON THIS NOTE IS COMPLETED AND DELIVERED BY THE TRANSFEROR TO THE TRANSFER AGENT."

15. PAYMENTS ON NOTES.

15.1 PLACE OF PAYMENT.

Subject to SECTION 15.2, payments of principal and Accreted Value of and interest and Accreted Value Premium (if any) on the Notes becoming due and payable on the Notes shall be made in New York, New York at the principal office of United States Trust Company of New York in such jurisdiction. The Company may at any time, by notice to each Holder of a Note, change the place of payment of the Notes so long as such place of payment shall be either the principal office of the Company in such jurisdiction or the principal office of a bank or trust company in such jurisdiction.

15.2 HOME OFFICE PAYMENT.

So long as you or your nominee shall be the Holder of any Note, and notwithstanding anything contained in SECTION 15.1 or in such Note to the contrary, the Company shall pay all sums becoming due on such Note for principal and Accreted Value of and interest and Accreted Value Premium (if any) on the Notes by the method and at the address specified for such purpose below your name in SCHEDULE A, or by such other method or at such other address as you shall have from time to time specified to the Company in writing for such purpose, without the presentation or surrender of such Note or the making of any notation thereon, except that upon written request of the Company made concurrently with or reasonably promptly after payment or prepayment in full of any Note, you shall surrender such Note for cancellation, reasonably promptly after any such request, to the Company at its principal executive office or at

the place of payment most recently designated by the Company pursuant to SECTION 15.1. Prior to any sale or other disposition of any Note held by you or your nominee you will, at your election, either endorse thereon the amount of principal paid thereon and the last date to which interest has been paid thereon or surrender such Note to the Company in exchange for a new Note or Notes pursuant to SECTION 15.2. The Company will afford the benefits of this SECTION 15.2 to any Person that is the direct or indirect transferee of any Note purchased by you under this Agreement and that has made the same agreement relating to such Note as you have made in this SECTION 15.2.

16. EXPENSES, ETC.

16.1 TRANSACTION EXPENSES.

Whether or not the transactions contemplated hereby are consummated, the Company will pay all costs and expenses incurred by you and each other Holder of a Note in connection with such transactions and in connection with any amendments, waivers or consents under or in respect of this Agreement, the Security Agreements or the Notes (whether or not such amendment, waiver or consent becomes effective), including, without limitation: (a) the reasonable costs, expenses and fees of Milbank, Tweed, Hadley & McCloy LLP and, if reasonably required, local or other counsel; (b) the costs and expenses incurred in enforcing or defending (or determining whether or how to enforce or defend) any rights under any Note Document or in responding to any subpoena or other legal process or informal investigative demand issued in connection with any Note Document, or by reason of being a Holder of any Note; and (c) the costs and expenses, including financial advisors' fees, incurred in connection with the insolvency or bankruptcy of the Company or any Subsidiary or in connection with any work-out or restructuring of the transactions contemplated hereby and by the Notes. The Company will pay, and will save you and each other Holder of a Note harmless from, all claims in respect of any fees, costs or expenses if any, of brokers and finders (other than those retained by you).

16.2 SURVIVAL.

The obligations of the Company under this SECTION 16 will survive the payment or transfer of any Note, the enforcement, amendment or waiver of any provision of this Agreement or the Notes, and the termination of this Agreement.

17. SURVIVAL OF REPRESENTATIONS AND WARRANTIES; ENTIRE AGREEMENT.

All representations and warranties contained herein shall survive the execution and delivery of each Note Document, the purchase or transfer by you of any Note or portion thereof or interest therein and the payment of any Note, and may be relied upon by any subsequent Holder of a Note, regardless of any investigation made at any time by or on behalf of you or any other Holder of a Note. All statements contained in any certificate or other instrument delivered by or on behalf of the Company pursuant to this Agreement shall be deemed representations and warranties of the Company under this Agreement. Subject to

the preceding sentence, the Note Documents embody the entire agreement and understanding between you and the Company and supersede all prior agreements and understandings relating to the subject matter hereof.

18. AMENDMENT AND WAIVER.

18.1 WITH CONSENT OF HOLDERS.

Subject to SECTION 18.4 and the consent of the Collateral Agent or any depositary bank if required under any Security Agreement, with the consent of the Majority Holders by written act of said holders delivered to the Company, the Company and the Majority Holders may amend or supplement this Agreement, the Security Agreements or the Notes for the purpose of adding any provisions to or changing in any manner or eliminating any of the provisions of this Agreement, the Security Agreements or the Notes or of modifying in any manner the rights of the holders under this Agreement, the Security Agreements or the Notes. Subject to SECTION 18.4 and the consent of the Collateral Agent or any depositary bank if required under any Security Agreement, the Majority Holders may waive compliance by the Company with any provision of this Agreement, the Security Agreements or the Notes. Notwithstanding any of the above, however, no such amendment, supplement or waiver shall, without the consent of the holder of each outstanding Note affected thereby:

- (i) reduce the percentage of outstanding principal amount of Notes whose holders must consent to an amendment, supplement or waiver of any provision of this Agreement, the Security Agreements or the Notes:
- (ii) reduce the rate or extend the time for payment of interest on any Note; $\,$
- $\mbox{(iii)}$ reduce the principal amount or Accreted Value of any Note;
- (iv) change the Maturity Date or alter the repayment provisions hereunder;
- (v) waive a Default in the payment of the principal or Accreted Value of or interest or Accreted Value Premium, if any, with respect to any Note;
- (vi) make any changes to SECTION 18.4 or this third sentence of this SECTION 18.1;
- (vii) make any amendment, modification or supplement to any Note that is not offered to be made equally to all holders of Notes; or
- (viii) make the principal or Accreted Value of or interest or Accreted Value Premium on any Note payable with anything other than U.S. Legal Tender.

It shall not be necessary for the consent of the holders of Notes to approve the particular form of any proposed amendment, supplement or waiver, but it shall be sufficient if such consent approves the substance thereof.

After an amendment, supplement or waiver under this SECTION 18.1 becomes effective, the Company shall mail to the holders of Notes affected thereby a notice briefly describing the amendment, supplement or waiver. Any failure of the Company to mail such notice, or any defect therein, shall not, however, in any way impair or affect the validity of any such amendment, supplement or waiver.

After an amendment, supplement or waiver under this SECTION 18.1 becomes effective, it shall bind each holder of Notes.

18.2 REVOCATION AND EFFECT OF CONSENTS.

Until an amendment, waiver or supplement becomes effective, a consent to it by a Holder of Notes is a continuing consent by such Holder and every subsequent Holder of a Note or portion of a Note that evidences the same debt as such consenting Holder's Note, even if notation of the consent is not made on any Note. However, any such Holder or subsequent holder may revoke the consent as to its Note or portion of its Note by written notice to the Company or the Person designated by the Company as the Person to whom consents should be sent if such revocation is received by the Company or such Person before the date on which the Holders of the requisite principal amount of Notes have consented (and not theretofore revoked such consent) to the amendment, supplement or waiver.

18.3 NOTATION ON NOTES.

If an amendment, supplement or waiver changes the terms of a Note, the Company may require the Holder of the Note to deliver it to the Company or require the Holder to put an appropriate notation on the Note. The Company may place an appropriate notation on the Note about the changed terms and return it to the Holder. Alternatively, if the Company so determines, the Company in exchange for the Note shall issue a new Note that reflects the changed terms. Any failure to make the appropriate notation or to issue a new Note shall not affect the validity of such amendment, supplement or waiver.

18.4 RIGHTS OF HOLDERS TO RECEIVE PAYMENT.

Notwithstanding any other provision of this Agreement, the right of any Holder to receive payment of principal and Accreted Value or interest and Accreted Value Premium, if any, on the Notes or after the respective due dates expressed in this Agreement and in the Notes or to bring suit for the enforcement of any such payment on or after such respective dates, shall not be impaired or affected without the consent of such Holder.

19. NOTICES.

All notices and communications provided for hereunder shall be in writing and sent (a) by telecopy if the sender on the same day sends a confirming copy of such notice by a recognized overnight delivery service (charges prepaid), or (b) by registered or certified mail with return receipt requested (postage prepaid), or (c) by a recognized overnight delivery service (with charges prepaid). Any such notice must be sent:

- (i) if to you or your nominee, to you or it at the address specified for such communications in SCHEDULE A, or at such other address as you or it shall have specified to the Company in writing,
- (ii) if to any other Holder of any Note, to such Holder at such address as such other Holder shall have specified to the Company in writing, or
- (iii) if to the Company, to the Company at their address set forth at the beginning hereof to the attention of the General Counsel, or at such other address as the Company shall have specified to the Holder of each Note in writing.

Notices under this SECTION 19 will be deemed given when actually received if sent by telecopy, upon the succeeding Business Day if sent by overnight courier and three (3) days after deposit in the US mail if sent by registered or certified mail.

20. REPRODUCTION OF DOCUMENTS.

This Agreement and all documents relating thereto, including, without limitation, (a) consents, waivers and modifications that may hereafter be executed, (b) documents received by you at the Closing (except the Notes themselves), and (c) financial statements, certificates and other information previously or hereafter furnished to you, may be reproduced by you by any photographic, photostatic, microfilm, microcard, miniature photographic or other similar process, and you may destroy any original document so reproduced. The Company agrees and stipulates that, to the extent permitted by applicable law, any such reproduction shall be admissible in evidence as the original itself in any judicial or administrative proceeding (whether or not the original is in existence and whether or not such reproduction was made by you in the regular course of business) and any enlargement, facsimile or further reproduction of such reproduction shall likewise be admissible in evidence. This SECTION 20 shall not prohibit the Company or any other Holder of Notes from contesting any such reproduction to the same extent that it could contest the original, or from introducing evidence to demonstrate the inaccuracy of any such reproduction.

21. CONFIDENTIAL INFORMATION.

For the purposes of this SECTION 21, "CONFIDENTIAL INFORMATION" means information delivered to you by or on behalf of the Company or any Subsidiary in connection with the transactions contemplated by or otherwise pursuant to this Agreement that is proprietary in nature PROVIDED that such term does not include information that (a) was publicly known or otherwise known to you prior to the time of such disclosure, (b) subsequently becomes publicly known through no act or omission by you or any person acting on your behalf, (c) otherwise becomes known to you other than through disclosure by the Company or any Subsidiary or (d) constitutes financial statements delivered to you under SECTION 8.9 that are otherwise publicly available. You will maintain the confidentiality of such Confidential Information in accordance with procedures adopted by you in good faith to protect confidential information of third parties delivered to you, PROVIDED that you may deliver or disclose Confidential Information to (i) your

directors, officers, employees, agents, attorneys and affiliates (to the extent such disclosure reasonably relates to the administration of the investment represented by your Notes), (ii) your financial advisors and other professional advisors who agree to hold confidential the Confidential Information substantially in accordance with the terms of this SECTION 21, (iii) any other Holder of any Note, (iv) any Person to which you sell or offer to sell such Note or any part thereof or any participation therein (if such Person has agreed in writing prior to its receipt of such Confidential Information to be bound by the provisions of this SECTION 21), (v) any Person from which you offer to purchase any security of the Company (if such Person has agreed in writing prior to its receipt of such Confidential Information to be bound by the provisions of this SECTION 21), (vi) any federal or state regulatory authority having jurisdiction over you, (vii) the National Association of Insurance Commissioners or any similar organization, or any nationally recognized rating agency that requires access to information about your investment portfolio or (viii) any other Person to which such delivery or disclosure may be necessary or appropriate (w) to effect compliance with any law, rule, regulation or order applicable to you, (x) in response to any subpoena or other legal process, (y) in connection with any litigation to which you are a party or (z) if an Event of Default has occurred and is continuing, to the extent you may reasonably determine such delivery and disclosure to be necessary or appropriate in the enforcement or for the protection of the rights and remedies under your Notes or this Agreement; provided that in the case of clauses (w), (x) and (y), you shall promptly notify the Company of such proposed delivery of disclosure and cooperate with the Company so that they may seek an appropriate protection order or otherwise seek appropriate protection for the confidentiality of the Confidential Information. Each Holder of a Note, by its acceptance of a Note, shall be deemed to have agreed to be bound by and to be entitled to the benefits of this SECTION 21 as though it were a party to this Agreement. On reasonable request by the Company in connection with the delivery to any Holder of a Note of information required to be delivered to such Holder under this Agreement or requested by such Holder (other than a Holder that is a party to this Agreement or its nominee), such Holder shall enter into an agreement with the Company embodying the provisions of this SECTION 21.

In connection with any confidential information provided to the Holders pursuant to this Agreement, the Company shall use its best efforts to advise the Holders whether any information to be provided is likely to include material, non-public information with regard to any Publicly-held Company and afford the Holders an opportunity to elect not to receive any such material, non-public information in connection with the exercise of their rights under this Agreement.

22. MISCELLANEOUS.

22.1 CERTIFICATE AND OPINION AS TO CONDITIONS PRECEDENT.

Upon any request or application by the Company to the Collateral Agent to take any action under any Note Document, the Company shall (i) for any actions taken pursuant to SECTION 13 furnish to the Collateral Agent and (ii) for any other actions taken under this Agreement, furnish at the request of the Collateral Agent:

(i) an Officers' Certificate (which shall include the statements set forth in SECTION 22.2) stating that, in the opinion of the signers, all conditions precedent, if any, provided for in this Agreement relating to the proposed action have been complied with; and

(ii) an Opinion of Counsel (which shall include the statements set forth in SECTION 22.2) stating that, in the opinion of such counsel, all such conditions precedent have been complied with.

22.2 STATEMENTS REQUIRED IN CERTIFICATE OR OPINION.

Each Officers' Certificate of the Company and each Opinion of Counsel with respect to compliance with a condition or covenant provided for in this Agreement shall include:

- (i) a statement that the person making such certificate or opinion has read such covenant or condition;
- (ii) a brief statement as to the nature and scope of the examination or investigation upon which the statements or opinions contained in such certificate or opinion are based;
- (iii) a statement that, in the opinion of such person, he has made such examination or investigation as is necessary, to enable him to express an informed opinion as to whether or not such covenant or condition has been complied with; and
- (iv) a statement as to whether or not, in the opinion of each such person, such condition or covenant has been complied with; provided, however, that with respect to matters of fact an Opinion of Counsel may rely on Officers' Certificates of the Company or certificates of public officials.

22.3 THIRD PARTY BENEFICIARY.

To the extent necessary to carry out its duties under the Security Agreements and to effect the purposes thereof, the Collateral Agent is hereby expressly made a third-party beneficiary of the provisions of this Agreement.

22.4 SUCCESSORS AND ASSIGNS.

All covenants and other agreements contained in this Agreement by or on behalf of any of the parties hereto bind and inure to the benefit of their respective successors and assigns (including, without limitation, any subsequent Holder of a Note) whether so expressed or not.

22.5 PAYMENTS DUE ON NON-BUSINESS DAYS.

Anything in this Agreement or the Notes to the contrary notwithstanding, any payment of principal and Accreted Value of and interest and Accreted Value Premium (if any) on the Notes that is due on a date other than a Business Day shall be made on the next succeeding Business Day without including the additional days elapsed in the computation of the interest payable on such next succeeding Business Day.

22.6 SEVERABILITY.

Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall (to the full extent permitted by law) not invalidate or render unenforceable such provision in any other jurisdiction.

22.7 CONSTRUCTION.

Each covenant contained herein shall be construed (absent express provision to the contrary) as being independent of each other covenant contained herein, so that compliance with any one covenant shall not (absent such an express contrary provision) be deemed to excuse compliance with any other covenant. Where any provision herein refers to action to be taken by any Person, or which such Person is prohibited from taking, such provision shall be applicable whether such action is taken directly or indirectly by such Person.

22.8 COUNTERPARTS.

This Agreement may be executed in any number of counterparts, each of which shall be an original but all of which together shall constitute one instrument. Each counterpart may consist of a number of copies hereof, each signed by less than all, but together signed by all, of the parties hereto. Delivery of an executed counterpart of a signature page of this Agreement by facsimile transmission shall be effective as delivery of a manually executed counterpart of this Agreement.

22.9 CAPTIONS.

The headings of the sections of this document have been inserted for convenience of reference only and shall not be deemed a part of this Agreement.

22.10 GOVERNING LAW; JURISDICTION; WAIVER OF JURY TRIAL.

THIS AGREEMENT, THE NOTES AND ALL ISSUES HEREUNDER AND THEREUNDER, INCLUDING (WITHOUT LIMITATION) THE DETERMINATION OF THE MAXIMUM LAWFUL RATE OF INTEREST THAT MAY BE CONTRACTED FOR, CHARGED OR RECEIVED WITH RESPECT TO THE NOTES SHALL BE CONSTRUED IN ACCORDANCE WITH AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF NEW YORK INCLUDING, WITHOUT LIMITATION, ss.ss. 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW AND NYCPLR 327(b). TO THE FULLEST EXTENT IT MAY EFFECTIVELY DO SO UNDER APPLICABLE LAW, EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF ANY NEW YORK STATE COURT SITTING IN THE BOROUGH OF MANHATTAN IN NEW YORK CITY OR ANY FEDERAL COURT SITTING IN THE BOROUGH OF MANHATTAN IN NEW YORK CITY IN RESPECT OF ANY SUIT,

ACTION OR PROCEEDING, WHETHER IN TORT, CONTRACT OR OTHERWISE, ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OF THE OTHER NOTE DOCUMENTS OTHER THAN ANY SUIT, ACTION OR PROCEEDING BROUGHT BY ANY HOLDER TO ENFORCE ANY RIGHT OR EXERCISE ANY REMEDY UNDER THE SECURITY AGREEMENTS (COLLECTIVELY, "COLLATERAL ACTIONS"), AND IRREVOCABLY AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH SUIT, ACTION OR PROCEEDING (OTHER THAN COLLATERAL ACTIONS) SHALL BE HEARD AND DETERMINED ONLY IN ANY SUCH COURT. EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES, TO THE FULLEST EXTENT IT MAY EFFECTIVELY DO SO UNDER APPLICABLE LAW, TRIAL BY JURY AND ANY OBJECTION WHICH IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT AND ANY CLAIM THAT ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM. NOTHING HEREIN SHALL AFFECT THE RIGHT OF ANY HOLDER OR ITS AGENTS TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR TO COMMENCE LEGAL PROCEEDINGS OR OTHERWISE PROCEED AGAINST THE COMPANY OR ANY OF ITS AFFILIATES IN ANY COLLATERAL ACTION IN ANY OTHER JURISDICTION.

* * * *

If you are in agreement with the foregoing, please sign the form of agreement on the accompanying counterpart of this Agreement and return it to the Company, whereupon the foregoing shall become a binding agreement between you and the Company.

Very truly yours,

BGLS INC.

By: /s/ RICHARD J. LAMPEN

Name:Richard J. Lampen

Title: Executive Vice President

TCW LEVERAGED INCOME TRUST, L.P. TCW Advisers (Bermuda), Ltd. as its General Partner

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall

Title: Managing Director

By: TCW Investment Management Company

as Investment Adviser

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title:

TCW LEVERAGED INCOME TRUST, II L.P. By: TCW (LINC II), L.P. às its General Partner

By: TCW Advisers (Bermuda), Ltd. its General Partner

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall

Title: Managing Director

By: TCW Investment Management Company as Investment Adviser

By: /s/ MARK ATTANASIO

Name: Mark Attanasio Title: Group Director

TCW LINC III CBO LTD.

By: TCW Investment Management Company as Collateral Manager

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio Title: Group Managing Director

POWRs 1997-2 (Participating Obligations with Residuals 1997-2)

By: Citibank Global Asset Management Its Investment Advisor

By: TCW Asset Management Company Its Portfolio Manager

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall

Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title: Group Managing Director

Captiva II Finance Ltd. By: TCW Advisors, Inc. Its Financial Manager

By: /s/ DARRYL L. SCHALL

Namo: Darryl I Schall

Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title: Group Managing Director

AIMCO CDO, SERIES 2000-A By: Allstate Investment Management Company
Its Collateral Manager

By: TCW Asset Management Company Its Investment Advisor

/s/ DARRYL L. SCHALL By:

> Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio Title: Group Managing Director

SCHEDULE A

\$

INFORMATION RELATING TO PURCHASER

NAME AND ADDRESS OF PURCHASER NOTES TO BE PURCHASED

[NAME OF PURCHASER]

(1) All payments by wire transfer of immediately available funds to:

with sufficient information to identify the source and application of such funds.

- (2) All notices of payments and written confirmations of such wire transfers:
- (3) All other communications:
- (4) Address for Delivery of Notes:

SCHEDULE B

DEFINED TERMS

As used herein, the following terms have the respective meanings set forth below or set forth in the Section hereof following such term:

"ACCEPTING HOLDER" is defined in SECTION 8.24.

"ACCOUNT CONTROL AGREEMENT" means the Account Control Agreement, substantially in the form of EXHIBIT B-2 among the Company, the Collateral Agent and Bank of America, as amended, modified and supplemented from time to time.

"ACCOUNT NOTICE EVENT" means (i) the Company defaults in payment of interest on any Notes when the same becomes due and payable and the default continues for a period of thirty (30) days, (ii) the Company defaults in the payment of the principal or Accreted Value of, or Accreted Value Premium on, any Notes when the same becomes due and payable at maturity or upon redemption, pursuant to SECTION 7.2, 8.24 or 8.25, (iii) a Non-Grace Period Covenant Acceleration Default shall have occurred and be continuing, (iv) an Other Obligation Payment Default or an Other Obligation Acceleration Default shall have occurred and be continuing, PROVIDED, however, that an Other Obligation Payment Default or Other Obligation Acceleration Default shall not constitute an Account Notice Event unless one or more Other Obligation Payment Defaults and Other Obligation Acceleration Defaults shall have occurred and be continuing with respect to Other BGLS Group Debt the outstanding principal amount of which exceeds in the aggregate \$5,000,000, or (v) any Person (other than the Collateral Agent) shall have commenced the enforcement or foreclosure of any Lien of such Person on the Securities Account.

"ACCRETED VALUE" is defined in SECTION 7.6.

"ACCRETED VALUE PREMIUM" means the difference between (i) the percentage of Accreted Value at which a Note is to be purchased by the Company pursuant to SECTIONS 7.2, 8.24, and 8.25 and (ii) 100%.

"AFFILIATE" of any specified Person means any other Person directly or indirectly controlling or controlled by or under direct or indirect common control with such specified Person. For the purposes of this definition, "CONTROL" when used with respect to any Person means the power to direct the management and policies of such Person, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise; and the terms "AFFILIATED," "CONTROLLING" and "CONTROLLED" have meanings correlative to the foregoing.

"AFFILIATED SENIOR MANAGER" means (i) any Vector Senior Manager, (ii) any spouse, descendant, parent, siblings, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law or sister-in-law of any Vector Senior Manager, (iii) any trust created by any Vector Senior Manager

or with regard to which any Vector Senior Manager serves as a trustee or co-trustee, (iv) any Affiliate of any of the foregoing and (v) any Person in which any one or more of the foregoing shall own more than a 10% economic interest or control more than 10% of the voting power; PROVIDED, HOWEVER, that the term "Affiliated Senior Manager" shall not include (a) the Company, (b) New Valley or (c) any Person controlled by the Company or New Valley if any one or more of the foregoing listed in clauses (i) through (v) above do not have more than a 10% economic interest or control more than 10% of the voting power in such Person, in each case other than through economic interests in or voting power in the Company or Vector.

"AFFILIATED TRANSACTION" means, with respect to any Person, (x) any direct or indirect sale, issuance, lease, transfer, exchange, purchase or other disposition of securities, assets, property or services to or from, or any contract, agreement, understanding or other transaction or series of related transactions with or for the benefit of, or any payment, loan or advance to or for the benefit of, such Person and (y) any transaction or series of related transactions, whether monetary or non-monetary or otherwise in-kind, pursuant to, on account of or in connection with which such Person receives, obtains, or otherwise becomes entitled to, either immediately or in the future, any form of economic benefit, regardless of whether or not such Person provides any consideration therefor (including, without limitation, the payment of money, the transfer of property or rights, the waiver of claims or rights, or the forbearance from exercising rights, asserting claims or the like) and, in the case of a transaction or series of related transactions described in either of the foregoing CLAUSE (X) or (y), without regard to the fairness of the terms of such transaction or series of related transactions; PROVIDED, HOWEVER, that a transaction which benefits the Company or any of its Subsidiaries or New Valley generally shall not constitute an Affiliated Transaction with respect to any Affiliated Senior Manager or any other holder of Equity Interests in the Company for the foregoing purpose solely because it indirectly benefits such Affiliated Senior Manager or such other holder solely as a result of such Affiliated Senior Manager's or such other holder's direct or indirect ownership interests in the Company. With regard to CLAUSE (Y), which, as is the case with CLAUSE (X), is specifically intended to be construed as broadly as possible and without limiting the generality thereof, and by way of example only, a transaction would be an Affiliated Transaction between New Valley and an Affiliated Senior Manager if, in connection therewith, such Affiliated Senior Manager received (a) payment for a covenant not to compete, (b) payment for the cancellation of a claim against, or a right to receive assets or property from, New Valley, (c) payment for the cancellation of or amendment to a contract with New Valley, (d) payment with respect to employment with or consulting services provided to New Valley, (e) a brokerage, investment banking or finders fee or (f) the right to purchase or acquire securities of, or assets or other property from, New Valley whether or not full value was to be paid therefor.

"AGREEMENT" means this Note Purchase Agreement.

"APPLICABLE ACCRETION PERCENTAGE" is defined in SECTION 7.6.

"ASSOCIATE" has the meaning provided in Rule 12b-2 promulgated by the SEC under the Exchange Act.

"AVAILABLE PROCEEDS OFFER" is defined in SECTION 8.24.

"AVAILABLE PROCEEDS OFFER DATE" is defined in SECTION 8.24.

"AVAILABLE PROCEEDS OFFER PURCHASE AMOUNT" means, with respect to any Available Proceeds Offer, an amount equal to (i) in the case of an Available Proceeds Offer pursuant to a Triggering Asset Sale, the Cumulative Net Available Proceeds calculating on the date of commencement of such Available Proceeds Offer and (ii) in the case of an Available Proceeds Offer pursuant to SECTION 8.25 or 8.29, the difference between the proceeds of such Restricted Subsidiary Asset Sale and, the amount of Cumulative Net Remaining Proceeds.

"AVAILABLE PROCEEDS OFFER PURCHASE PRICE" is defined in SECTION 8.24.

"AVAILABLE PROCEEDS PURCHASE DATE" means any date on which Notes are purchased pursuant to an Available Proceeds Offer.

"AVERAGE WEEKLY CASH BALANCE" means, for any period, the average weekly balance of Cash and cash equivalents held by the Company.

"BANK OF AMERICA" means Bank of America, N.A.

"BANKRUPTCY LAW" is defined in SECTION 11.

"BGLS ALLOCABLE AMOUNT" is defined in SECTION 9.3(A)(I)(A).

"BGLS BALANCE SHEET" means the balance sheet of the Company without consolidation with any other Person. $\,$

"BGLS CHECKING ACCOUNT" means Account #01596108029 of the Company at Bank of America.

"BGLS PLEDGE AGREEMENT" means a Pledge and Security Agreement substantially in the form of EXHIBIT B-1, between the Company and the Collateral Agent for the benefit of the Holders, as amended, modified or supplemented from time to time.

"BLACK-OUT PERIOD" means any period commencing on any Drop-Below Date and ending on the earlier of (i) next date on which the Company holds the Required Cash Holdings on the BGLS Balance Sheet and (ii) the first day of the next succeeding fiscal quarter of the Company.

 $$\tt "BROOKE\ HOLDING"\ means\ Brooke\ Group\ Holding\ Inc.,\ a\ Delaware\ corporation,\ or\ any\ successor\ thereto.$

"BROOKE HOLDING CHECKING ACCOUNT" means Account #03751459059 of Brooke Holding at Bank of America.

"BROOKE HOLDING PLEDGE AGREEMENT" means a Pledge and Security Agreement substantially in the form of EXHIBIT D, between Brooke Holding and the Collateral Agent for the benefit of the Holders, as amended, modified or supplemented from time to time. $$\tt "BROOKE"$ OVERSEAS" means Brooke (Overseas) Ltd., a Delaware corporation, and any successor thereto.

"BUSINESS DAY" means any day other than a Saturday, a Sunday or a day on which commercial banks in (i) New York City, New York or (ii) solely with respect to the Checking Accounts, the jurisdiction in which the Checking Accounts are located are required or authorized to be closed.

"CAPITALIZED LEASE OBLIGATION" means indebtedness represented by obligations under a lease that is required to be capitalized for financial reporting purposes in accordance with GAAP, and the amount of such indebtedness shall be the capitalized amount of such obligations determined in accordance with GAAP.

"CASH OR CASH EQUIVALENTS" means (i) U.S. Legal Tender or direct noncallable obligations of, or noncallable obligations guaranteed by, the United States of America or agencies thereof for the timely payment of which obligation or guarantee the full faith and credit of the United States of America or such agency is pledged, (ii) investments in commercial paper or master notes maturing within two hundred seventy (270) days from the date of acquisition thereof and having (or the issuer thereof having), at such date of acquisition, the highest credit rating obtainable from Standard & Poor's Rating Group or from Moody's Investors Service, Inc., (iii) investments in certificates of deposit, banker's acceptances, time deposits or "late cash" deposits/funding agreements maturing or puttable to the issuer thereof within one hundred eighty (180) days from the date of acquisition thereof issued or guaranteed by or placed with, and money market deposit accounts issued or offered by, any domestic office of any commercial bank organized under the laws of, or licensed to conduct a banking or trust business in, the United States of America or any State thereof which has a combined capital and surplus and undivided profits of not less than \$500,000,000 whose long-term debt is rated at least "A" or the equivalent thereof by both Standard & Poor's Rating Group and Moody's Investors Service, Inc; (iv) fully collateralized repurchase agreements with a term of not more than thirty (30) days for securities described in clause (i) above and entered into with a financial institution satisfying the criteria described in clause (iii) above, and (v) investments in Nations Funds Cash Reserves Money Market Mutual Fund.

"CHANGE OF CONTROL" is defined in SECTION 8.25.

"CHANGE OF CONTROL PURCHASE DATE" is defined in SECTION 8.25.

"CHANGE OF CONTROL PURCHASE PRICE" is defined in SECTION 8.25.

"CHECKING ACCOUNTS" means the BGLS Checking Account, the Brooke Holding Checking Account and the NV Holdings Checking Account.

"CLOSING" is defined in SECTION 3.

"CODE" means the Internal Revenue Code of 1986, as amended from time to time, and the rules and regulations promulgated thereunder from time to time.

"COLLATERAL" means the Property described in, or from time to time subject to the Lien of, the Security Agreements.

"COLLATERAL AGENCY AGREEMENT" means the Collateral Agency Agreement, substantially in the form of EXHIBIT E, by and among the Purchasers, the Collateral Agent and the Document Parties as amended, modified and supplemented from time to time.

"COLLATERAL AGENT" means United States Trust Company of New York, solely in its capacity as Collateral Agent under the Collateral Agency Agreement and the Security Agreements.

"COMPANY" means BGLS, Inc., or any successor(s) thereto that shall have become such in the manner prescribed in SECTION 10.2.

"COMPANY ACCOUNT" means, collectively, the BGLS Checking Account and the Securities Account.

"COMPANY ASSET SALE" means (i) any conveyance, sale, transfer, assignment or other disposition of any Collateral or (ii) any conveyance, sale, transfer, assignment or other disposition of all or substantially all of the assets of Liggett, Research or VTUSA; PROVIDED that "Company Asset Sale" shall not include any Disposition of Assets effected in compliance with SECTION 10.1.

"COMPANY GROUP" means the Company and its Subsidiaries which would be eligible to join in a consolidated federal (and where applicable, consolidated or combined state or local) income tax return with the Company.

"COMPANY GROUP CREDITS" means credits against tax liability available to the Company in the relevant period under the Code.

"COMPANY NET INCOME" means the taxable income of the Company (including capital gains and capital losses, but as to capital losses, only to the extent they do not exceed capital gains for the relevant period) determined as if the Company had filed a separate consolidated or combined income tax return with respect to the relevant calendar quarter with, for federal income taxes purposes, the Company as common parent or year, as the case may be, based on the amounts that would be, or would be estimated to be, reported on the federal income tax return of the Company for the relevant calendar year and adjusted under the principles set forth in Treasury Regulation Section 1.1552-1(a)(2)(ii) (and computed without regard to any the Company net operating loss carryforwards).

"CONFIDENTIAL INFORMATION" is defined in SECTION 21.

"CONSOLIDATED EBITDA" means, with respect to the Company for any period, without duplication, Consolidated Net Income of the Company for such period PLUS, without duplication and to the extent reflected as a charge in the statement of such Consolidated Net Income for such period, the sum of (a) Consolidated Income Taxes, (b) Consolidated Interest Expense, (c) consolidated depreciation expense of the Company and its Restricted Subsidiaries, (d)

consolidated amortization of intangibles (including, but not limited to, goodwill) of the Company and its Restricted Subsidiaries and (e) any other non-cash charges reducing Consolidated Net Income (excluding any such non-cash charge to the extent it represents an accrual of or reserve for cash charges in any future period or amortization of a prepaid cash expense that was paid in a prior period not included in the calculation); and MINUS, to the extent included in the statement of such Consolidated Net Income for such period, the sum of (x)interest income; PROVIDED, HOWEVER, that in the event the Company holds on the BGLS Balance Sheet an Average Weekly Cash Balance in excess of \$50,000,000 for such period, the Company shall not be required to exclude Excess Interest Income, (y) any extraordinary, unusual or non-recurring income or gains (including, whether or not otherwise includable as a separate item in the statement of such Consolidated Net Income for such periods, gains on the sales of assets outside of the ordinary course of business) and (z) any other non-cash income except the amount of any non-cash charge to Consolidated Net Income for a Tobacco Litigation Expense which non-cash charge is later reversed, all as determined on a consolidated basis for the Company and its Restricted Subsidiaries. For the purposes of calculating Consolidated EBITDA for any Reference Period (i) if at any time during such Reference Period the Company or one of its Restricted Subsidiaries shall have made any Material Disposition, the Consolidated EBITDA for such Reference Period shall be reduced by an amount equal to the Consolidated EBITDA (if positive) attributable to the property that is the subject of such Material Disposition for such Reference Period or increased by an amount equal to the Consolidated EBITDA (if negative) attributable thereto for such Reference Period and (ii) if during such Reference Period, the Company or one of its Restricted Subsidiaries shall have made a Material Acquisition, Consolidated EBITDA for such Reference Period shall be calculated after giving pro forma effect thereto as if such Material Acquisition occurred on the first day of such Reference Period. As used in this definition, "Material Acquisition" means any acquisition of property or series of related acquisitions of property that (a) constitutes assets comprising all or substantially all of an operating unit of a business or constitutes all or substantially all of the voting Equity Interests of a Person and (b) involves the payment of consideration by the Company and its Restricted Subsidiaries in excess of \$1,000,000; and "Material Disposition" means any disposition of property or series of related dispositions of property (whether by lease, assignment, sale or otherwise) that yields gross proceeds to the Company or any of its Restricted Subsidiaries in excess of \$1,000,000.

"CONSOLIDATED INCOME TAXES" means, with respect to the Company for any period, taxes imposed upon the Company or other payments required to be made by the Company by any Governmental or Regulatory Authority which taxes or other payments are calculated by reference to the income or profits of the Company or the Company and its Restricted Subsidiaries (to the extent such income or profits were included in computing Consolidated Net Income for such period), regardless of whether such taxes or payments are required to be remitted to any Governmental or Regulatory Authority.

"CONSOLIDATED INTEREST EXPENSE" means, with respect to the Company for any period, the total interest expense of the Company and the Restricted Subsidiaries, whether paid or accrued, plus, to the extent not included in such interest expense:

- (1) interest expense attributable to Capitalized Lease Obligations;
- (2) amortization of debt discount and debt issuance cost;

- (3) accrued interest;
- (4) non-cash interest expense;
- (5) commissions, discounts and other fees and charges owed with respect to letters of credit and bankers' acceptance financing;
- (6) interest actually paid by the Company or any such Restricted Subsidiary under any guarantee of Indebtedness or other obligation of any other Person;
- (7) net costs associated with Interest Swaps and Hedging Obligations (including amortization of fees);
- (8) the consolidated interest expense of the Company and its Restricted Subsidiaries that was capitalized during such period;
- (9) the product of (a) all dividends paid in Cash or cash equivalents or Indebtedness or accrued during such period on any Disqualified Equity Interests of the Company or on preferred stock of its Restricted Subsidiaries payable to a party other than the Company or a Restricted Subsidiary, times (b) a fraction, the numerator of which is one and the denominator of which is one minus the then current combined federal, sate, provincial and local statutory tax rate of the Company, expressed as a decimal, in each case, on a consolidated basis and in accordance with GAAP.

For purposes of the foregoing, total interest expense will be determined after giving effect to any net payments made or received by the Company and its Restricted Subsidiaries with respect to Interest Swaps and Hedging Obligations.

"CONSOLIDATED NET INCOME" of the Company means, without duplication, for any period, the after-tax net income (loss) of the Company and its Restricted Subsidiaries for such period on a consolidated basis as determined in accordance with GAAP, adjusted by excluding therefrom (a) net extraordinary gains or net extraordinary losses, as the case may be, of the Company and its Restricted Subsidiaries during such period, (b) net gains or losses (less all fees and expenses relating thereto) in respect of dispositions of assets (other than in the ordinary course of business) by the Company and its Restricted Subsidiaries during such period, (c) the income (or loss) of any other Person (other than a Restricted Subsidiary) in which the Company or any Restricted Subsidiary has an ownership interest, except to the extent of the amount of dividends or other distributions actually paid to such Person or its subsidiaries by such other Person during such period, (d) net income of any other Person combined with the Company or any Restricted Subsidiary on a "pooling of interests" basis attributable to any period prior the date of combination, (e) the net income of any Restricted Subsidiary during such period to the extent that the declaration or payment of dividends or similar distributions by that Restricted Subsidiary of that income is not at the time permitted, directly or indirectly, by operation of the terms of its charter or any agreement (including any shareholder agreement), instrument, judgment, decree, order, statute, rule or governmental regulations applicable to that Restricted Subsidiary (except to the extent of the amount of cash dividends or

distributions to the Company or other Restricted Subsidiaries (subject, in the case of a dividend to another Restricted Subsidiary, to the limitation contained in this clause (e)) from such Restricted Subsidiary during such period) and (f) the amount of any Shadow Dividends (but only to the extent such Shadow Dividends are counted as an expense); PROVIDED, HOWEVER, that (1) the amount of any Tobacco Litigation Bond of the Company or any Restricted Subsidiary deemed non-refundable or forfeited shall be expensed against Consolidated Net Income on the date that such amount is deemed non-refundable or forfeited; (2) to the extent that the amount of an appealable judgment for which a Tobacco Litigation Bond has been posted is counted as a charge against net income of the Company or a Restricted Subsidiary in accordance with GAAP, the amount of such charge shall be added back when calculating Consolidated Net Income; (3) to the extent that the present value of any Tobacco Claim Obligation not immediately due and payable is counted as a charge against net income in accordance with GAAP, the amount of such charge shall be added back when calculating Consolidated Net Income; and (4) any payment in respect of a Tobacco Claim Obligation shall be expensed against Consolidated Net Income on the date that such payment is made, except that the forfeiture of a Tobacco Litigation Bond in satisfaction of all or a portion a Tobacco Claim Obligation shall be expensed in the amount forfeited in accordance with clause (1) of this proviso rather than this clause (4).

"CUMULATIVE NET AVAILABLE PROCEEDS" means, at any time of determination, the sum of (i) the Net Available Proceeds received by the Company and its Subsidiaries with respect to all Company Asset Sales consummated since (x) the Available Proceeds Offer caused by a Company Asset Sale immediately preceding such time of determination or (y) if no Available Proceeds Offer Date caused by a Company Asset Sale shall have occurred, the date hereof PLUS (ii) all amounts deemed to be Net Available Proceeds since the Available Proceeds Offer Date caused by a Company Asset Sale immediately preceding such time of determination pursuant to CLAUSE (A) or (B) of the second proviso in the definition of "Net Available Proceeds".

"CUMULATIVE NET REMAINING PROCEEDS" means, at any time of determination, the sum of (i) the Net Available Proceeds and Net Vector Offering Proceeds received by the Vector, Company and its Subsidiaries with respect to all Restricted Subsidiary Asset Sales and Vector Equity Offerings consummated since (x) the Available Proceeds Offer caused by SECTION 8.25 or 8.29 immediately preceding such time of determination or (y) if no Available Proceeds Offer Date caused by SECTION 8.25 and 8.29 shall have occurred, the date hereof PLUS (ii) all amounts deemed to be Net Available Proceeds since the Available Proceeds Offer Date caused by SECTION 8.25 or 8.29 immediately preceding such time of determination pursuant to CLAUSE (A) or (b) of the second proviso in the definition of "Net Available Proceeds" MINUS, all amounts of such Net Available Proceeds and Net Vector Offering Proceeds used for Retirement of Restricted Subsidiary Indebtedness since the last Available Proceeds Offer Date caused by SECTION 8.25 or 8.29.

"CUSTODIAN" is defined in SECTION 11.

"DEFAULT" means an event or condition the occurrence or existence of which would, with the lapse of time or the giving of notice or both, become an Event of Default.

"DESIGNATED INVESTMENT BANK" means any of the following:
Deutsche Banc Alex. Brown Inc., Bear Stearns Securities Corp., Morgan Chase
Securities, Inc., CIBC World Markets, Credit Suisse First Boston, Warburg Dillon
Read, Goldman Sachs Group, Inc., Chase H&Q, J.P. Morgan Chase, Lazard LLC,

Lehman Brothers Holdings Inc., Merrill Lynch & Co., Banc of America Securities LLC, Morgan Stanley Dean Witter & Co., Prudential Securities Incorporated, Salomon Smith Barney Holdings, UBS Warburg, Duff & Phelps, Jefferies & Company, Inc. and Houlihan Lokey Howard & Zukin.

"DISPOSITION OF ASSETS" is defined in SECTION 10.1.

"DISQUALIFIED EQUITY INTEREST" means any Equity Interest of a Person that, by its terms or by the terms of any security into which it is convertible or exchangeable, is, or upon the happening of an event would be, required to be redeemed or retired, or at the option of the holder, repurchased, in whole or in part by such person or any subsidiary of such person, or has, or upon the happening of an event would have, a redemption or similar payment due, on or prior to the date that is ninety-one (91) days after the Maturity Date.

"DOCUMENT PARTY" means the Company, NV Holdings, Brooke Holding and Vector.

"DROP-BELOW DATE" means any date on which the Cash and cash equivalents held by the Company drops below the Required Cash Holdings on the BGLS Balance Sheet.

"DROP-BELOW QUARTER" means any fiscal quarter of the Company during which a Drop-Below Date occurs and subsequent to such Drop-Below Date the Company fails to maintain the Required Cash Holdings on the BGLS Balance Sheet at any time from such Drop-Below date until the end of such fiscal quarter.

"ENVIRONMENTAL LAWS" means any and all Federal, state, local, and foreign statutes, laws, regulations, ordinances, rules, judgments, orders, decrees, permits, concessions, grants, franchises, licenses, agreements or governmental restrictions relating to pollution and the protection of the environment or the release of any materials into the environment, including but not limited to those related to hazardous substances or wastes, air emissions and discharges to waste or public systems.

"EQUITYHOLDERS' EQUITY" means at any date with respect to any Person the amount of consolidated stockholders' equity (deficit) or partners' capital (excluding stockholders' equity or partners' capital attributable to Disqualified Equity Interests of such Person) that would appear on the balance sheet of such person at such date, in accordance with GAAP.

"EQUITY INTERESTS" means, with respect to any Person, any capital stock of such Person and shares, interests, participations or other ownership interests (however designated) of any Person and any rights (other than debt securities convertible into corporate stock), warrants and options to purchase any of the foregoing, including (without limitation) each class of common stock and preferred stock of such Person if such Person is a corporation and each general and limited partnership interest of such Person if such Person is a partnership.

"EQUITY OFFERING" means an offering of common stock of the Company of an amount of not less than \$5,000,000.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and the rules and regulations promulgated thereunder from time to time in effect.

"ERISA AFFILIATE" means any trade or business (whether or not incorporated) that is treated as a single employer together with the Company under section 414 of the Code.

"ESTIMATED TAX PAYMENTS" is defined in SECTION 9.4(B).

"EVENT OF DEFAULT" is defined in SECTION 11.

"EXCESS INTEREST INCOME" means the product of (i) a fraction the (x) numerator of which is the difference (if positive) between (A) the Average Weekly Cash Balance for a Reference Period and (B) \$50,000,000 and (y) the denominator of which is the Average Weekly Cash Balance for such Reference Period and (ii) the amount of interest income earned by the Company on Cash and cash equivalents that it holds during such Reference Period.

"EXCHANGE ACT" means the Securities Exchange Act of 1934, as

amended.

Accounts.

"EXPECTED POSTRETIREMENT BENEFIT OBLIGATION" means any expected postretirement benefit obligation (determined as of the last day of the Company's most recently ended fiscal year in accordance with Financial Accounting Standard Board Statement No. 106, without regard to liabilities attributable to continuation coverage mandated by section 4980B of the Code) of the Company and its Subsidiaries.

"FISCAL YEAR" means the fiscal year period beginning January 1 of each calendar year and ending on the following December 31.

"GAAP" means generally accepted accounting principles as in effect from time to time in the United States of America.

"GOVERNMENTAL OR REGULATORY AUTHORITY" means any court, tribunal, arbitrator, authority, agency, commission, official or other instrumentality of the United States or any state, county, city or other political subdivision or of any foreign country or political subdivision thereof.

"GROUP ACCOUNTS" means the Securities Account and the Checking

"GROUP EXECUTIVE" means (i) any Affiliated Senior Manager or (ii) as of the date hereof, any chairman of the Board, president, chief executive officer, chief operating officer, executive vice president, vice president, treasurer, assistant treasurer, secretary, assistant secretary and any other senior officer of or person performing similar functions on behalf of the Company or any Restricted Subsidiary and any successor of any of the foregoing.

"HAZARDOUS MATERIAL" means any and all pollutants, toxic or hazardous wastes or any other substances that might pose a hazard to health or safety, the removal of which may be required or the generation, manufacture, refining, production, processing, treatment, storage, handling, transportation, transfer, use, disposal, release, discharge, spillage, seepage, or filtration of which is or shall be restricted, prohibited or penalized by any applicable law (including, without limitation, asbestos, urea formaldehyde foam insulation and polychlorinated biphenyls).

"HOLDER" means, with respect to any Note, the Person in whose name such Note is registered in the register maintained by the Company pursuant to SECTION 14.1.

"IMMEDIATE FAMILY" means an individual Person's current spouse, parents, grandparents, siblings, children, children's spouses, grandchildren or grandchildren's spouses or any trusts, estates or partnerships for the benefit of or controlled by any of the foregoing.

"INCAPACITY" means the inability to perform fully the duties of an individual's office, which inability has continued for ninety (90) consecutive days.

"INDEBTEDNESS" of any Person means, without duplication,

- (a) all liabilities and obligations, contingent or otherwise, of such any Person, to the extent such liabilities and obligations would appear as a liability upon the consolidated balance sheet of such Person in accordance with GAAP, (1) in respect of borrowed money (whether or not the recourse of the lender is to the whole of the assets of such Person or only to a portion thereof), (2) evidenced by bonds, notes, debentures or similar instruments, or (3) representing the balance deferred and unpaid of the purchase price of any property or services, except those incurred in the ordinary course of its business that would constitute ordinarily a trade payable to trade creditors;
- (b) all liabilities and obligations, contingent or otherwise, of such Person (1) evidenced by bankers' acceptances or similar instruments issued or accepted by banks, (2) relating to any Capitalized Lease Obligation, or (3) evidenced by a letter of credit or a reimbursement obligation of such Person with respect to any letter of credit;
- (c) all net obligations of such Person under Interest Swap and Hedging Obligations;
 - (d) all outstanding and unsatisfied Tobacco Claim Obligations;
- (e) the amount of any Tobacco Litigation Bond of such Person to the extent such Person has not incurred "Indebtedness" pursuant to clauses (a), (b), (c), (d), (f), (g) or (h) of this definition of "Indebtedness" in order to obtain or procure the funds necessary to post such Tobacco Litigation Bond; PROVIDED, HOWEVER, that if at any time all or a portion of a Tobacco Litigation Bond has been (i) deemed non-refundable or is otherwise forfeited by such Person or (ii) is released or returned to such Person, the amount of such Tobacco Litigation Bond described the foregoing clauses (i) and (ii) shall cease to be "Indebtedness";
- (f) all liabilities and obligations of others of the kind described in the preceding clause (a), (b), (c) or (d) that such Person has guaranteed or provided credit support or that is otherwise its legal liability or which are secured by any assets or property of such Person;

- (g) any and all deferrals, renewals, extensions, refinancing and refundings (whether direct or indirect) of, or amendments, modifications or supplements to, any liability of the kind described in any of the preceding clauses (a), (b), (c), (d) or (e), or this clause (f), whether or not between or among the same parties; and
- (h) all Disqualified Equity Interests of such Person (measured at the greater of its voluntary or involuntary maximum fixed repurchase price plus accrued and unpaid dividends).

For purposes hereof, the "maximum fixed repurchase price" of any Disqualified Equity Interests which does not have a fixed repurchase price shall be calculated in accordance with the terms of such Disqualified Equity Interests as if such Disqualified Equity Interests were purchased on any date on which Indebtedness shall be required to be determined pursuant to the Indenture, and if such price is based upon, or measured by, the fair market value of such Disqualified Equity Interests, such fair market value to be determined in good faith by the board of directors of the issuer (or managing general partner of the issuer) of such Disqualified Equity Interests.

The amount of any Indebtedness outstanding as of any date shall be (1) the accreted value thereof, in the case of any Indebtedness issued with original issue discount, but the accretion of original issue discount in accordance with the original terms of Indebtedness issued with an original issue discount will not be deemed to be an incurrence and (2) the principal amount thereof, together with any interest thereon that is more than thirty (30) days past due, in the case of any other Indebtedness.

"INTEREST SWAP AND HEDGING OBLIGATION" means any obligation of any Person pursuant to any interest rate swap agreement, interest rate cap agreement, interest rate collar agreement, interest rate exchange agreement, currency exchange agreement or any other agreement or arrangement designed to protect against fluctuations in interest rates or currency values, including, without limitation, any agreement whereby, directly or indirectly, such Person is entitled to receive from time to time periodic payments calculated by applying either a fixed or floating rate of interest on a stated notional amount in exchange for periodic payments made by such Person calculated by applying a fixed or floating rate of interest on the same notional amount.

"INVESTMENT" means, for any Person, any loan, advance, stock purchase, capital contribution, or other investment by such Person in any other Person and includes any other direct or indirect investment of any kind (including, without limitation, guarantees of indebtedness, provisions of letters of credit and other similar items).

"INVESTMENT COMPANY ACT" means the Investment Company Act of 1940, as amended.

"LEVERAGE RATIO" means, as at any date, the ratio of (a) the sum of (i) the aggregate outstanding Indebtedness of the Company and its Restricted Subsidiaries as of the date of calculation on a consolidated basis in accordance with GAAP and (ii) to the extent not included in the foregoing clause Liggett.

(i), the amount of any Indebtedness of another Person (other than the Company or a Restricted Subsidiary) that is (A) guaranteed by the Company or a Restricted Subsidiary or (B) secured by a Lien on the assets of the Company or a Restricted Subsidiary to (b) the Company's Consolidated EBITDA for the Reference Period most recently ended.

"LIEN" means any mortgage, pledge, lien, encumbrance, charge or adverse claim affecting title or resulting in an encumbrance against real or personal property, or a security interest of any kind (including any conditional sale or other title retention agreement, any lease in the nature thereof, any option or other agreement to sell and any filing of or agreement to give any financing statement under the Uniform Commercial Code (or equivalent statutes) of any jurisdiction).

 $\mbox{\tt "LIGGETT"}$ means Liggett Group Inc., a Delaware corporation, and any successor thereto.

"LIGGETT SENIOR CREDIT FACILITY" means, with respect to any Liggett Subsidiary, one or more debt facilities or commercial paper facilities in an aggregate amount not to exceed \$50,000,000 with banks or other institutional lenders providing for revolving credit loans, term loans, receivables financing (including through the sale of receivables to such lenders or to special purpose entities formed to borrow from such lenders against such receivables) or letters of credit, in each case, as amended, restated, modified, renewed, refunded, replaced or refinanced in whole or in part from time to time (and whether or not with the original administrative agent and lenders or another administrative agent or agents or other lenders and whether provided under the original credit facility or any other credit or other agreement or indenture); PROVIDED, HOWEVER, no more than \$10,000,000 of Indebtedness under any Liggett Senior Credit Facility may be in the form of term loans.

"LIGGETT SUBSIDIARY" means any of Liggett or any Subsidiary of

"LTFS" is defined in SECTION 8.5(C).

"MAJORITY HOLDERS" means (i) prior to the Closing, the TCW Funds, (ii) at any time that the TCW Funds hold 25% or more of the aggregate principal amount of the Notes outstanding at such time, the TCW Funds and (iii) at any time that the TCW Funds hold less than 25% of the aggregate principal amount of the Notes outstanding at such time, the holders of Notes holding not less than a majority in the aggregate principal amount of all Notes outstanding at the time held by all holders (other than the Company and its Affiliates) of Notes.

"MARGIN STOCK" means "margin stock" as such term is defined in Regulation U of the Board of Governors of the Federal Reserve System.

"MASTER SETTLEMENT AGREEMENT" means (i) that Master Settlement Agreement between Settling States (as defined therein including the states and territories of Alabama, Alaska, American Samoa, Arizona, Arkansas, California, Colorado, Connecticut, Delaware, District of Columbia, Georgia, Guam, Hawaii, Idaho, Illinois, Indiana, Iowa, Kansas, Kentucky, Louisiana, Maine, Maryland, Massachusetts, Michigan, Missouri, Montana, Nebraska, Nevada, New Hampshire, New Jersey, New Mexico, New York, North Carolina, North Dakota, Northern Mariana Islands, Ohio, Oklahoma, Oregon, Pennsylvania, Puerto Rico, Rhode Island, South Carolina, South Dakota, Tennessee, Utah, Vermont, Virginia, Virgin Islands, Washington, West Virginia, Wisconsin, Wyoming) and Participating Manufacturers

(as defined therein including Philip Morris Inc., RJ Reynolds Inc., Brown & Williamson Tobacco Corporation, Lorillard Tobacco Company and Liggett Group Inc.) and signatories thereto, dated November of 1998, as modified, amended or supplemented; PROVIDED, that any such modification, amendment or supplement does not materially increase the obligations of any Liggett Subsidiary and (ii) similar agreements between any Liggett Subsidiary and any of the states of Florida, Mississippi, Minnesota and Texas that do not provide for payments by the Liggett Subsidiaries in excess of the product of (i) \$500,000 and (ii) the sum of (A) the number of whole Purchase Agreement Years elapsed between the date hereof and the date of any such calculation and (B) one (1).

"MATERIAL" means material in relation to the business, operations, affairs, financial condition, assets, net worth or properties of the Company and the Subsidiaries taken as a whole.

"MATERIAL ADVERSE EFFECT" means a material adverse effect on (a) the business, operations, affairs, financial condition, prospects, net worth, assets or properties of the Company and the Subsidiaries taken as a whole, or (b) the ability of any Document Party to perform its respective obligations under this Agreement, the Security Agreements and the Notes or (c) the validity or enforceability of this Agreement, the Security Agreements or the Notes.

"MATURITY DATE" is defined in SECTION 7.1.

"MONTHLY ACCRUAL DATE" is defined in SECTION 7.6.

"MULTIEMPLOYER PLAN" means any Plan that is a "multiemployer plan" (as such term is defined in section 4001(a)(3) of ERISA).

"NET AVAILABLE PROCEEDS" means, with respect to any Company Asset Sale or Restricted Subsidiary Asset Sale, the aggregate amount of all cash and the fair market value (which, for this purpose, shall mean the sale value of the Property conveyed, sold, transferred, assigned or otherwise disposed of in such Company Asset Sale or Restricted Subsidiary Asset Sale which could be realized in an arm's length sale at such time between an informed and willing buyer and an informed and willing seller, under no compulsion to buy or sell, respectively) of all consideration received by the Company or any Subsidiary directly or indirectly (including through any Subsidiary) in connection with such Company Asset Sale or Restricted Subsidiary Asset Sale; PROVIDED that Net Available Proceeds shall be net of (i) the amount of any legal, title and recording tax expenses, commissions and other fees and expenses paid or payable in cash by the Company or any of its Subsidiaries as a result of such Company Asset Sale or Restricted Subsidiary Asset Sale, (ii) the amount of any Federal, state, local or foreign income or other taxes estimated to be actually payable by the Company or any of its Subsidiaries as a result of such Company Asset Sale or Restricted Subsidiary Asset Sale which are attributable to the Company Group under the principles of Treasury Regulation Sections 1.1552-1(a)(1) and 1.1502-33(d)(2), after taking into account all relevant tax attributes, (iii) reasonable brokerage or other customary selling commissions paid or payable in

cash by the Company or any of its Subsidiaries, (iv) reasonable third-party expenses incurred by the Company or any of its Subsidiaries in preparing for sale the Property which is the subject of such Company Asset Sale or Restricted Subsidiary Asset Sale, (v) in the case of any sale of all or substantially all of the assets of Liggett , any repayment of Indebtedness of Liggett, provided that such Indebtedness is not owed to the Company, such Indebtedness was not incurred in contemplation of such Company Asset Sale or Restricted Subsidiary Asset Sale (and the Company shall have provided to the Holders an Officer's Certificate of the Company to such effect) and such Indebtedness when incurred represented a borrowing by Liggett of cash in an amount equal to not less than the face amount of the Indebtedness and no more than \$20,000,000 aggregate principal amount of such Indebtedness owed to any Subsidiaries of the Company may be deducted pursuant to this clause (v) in one or more successive applications of this clause (v) in the aggregate and (vi) in the case of any sale of all or substantially all of the assets of Liggett, any consideration deemed received by the Company or any of its Subsidiaries as a result of the assumption of any Indebtedness by any Person in connection with such Company Asset Sale or Restricted Subsidiary Asset Sale, provided that such Indebtedness was not incurred in contemplation of such Company Asset Sale or Restricted Subsidiary Asset Sale (and the Company shall have provided to the Holders an Officer's Certificate of the Company to such effect); and PROVIDED, FURTHER, that (a) if any expenses, commissions or fees deducted above from Net Available Proceeds are not actually paid by the Company or any of its Subsidiaries within one year of the applicable Company Asset Sale or Restricted Subsidiary Asset Sale, such expenses, commissions or fees, to the extent not paid, shall be deemed to constitute Net Available Proceeds for purposes of the next Available Proceeds Offer made by the Company, (b) if at any date the sum of (I) any Indebtedness owed to any Subsidiary and assumed as contemplated by clause (vi) PLUS (II) the aggregate Indebtedness referred to in CLAUSE (V) owed to any Subsidiary exceeds \$20,000,000 cumulatively from the date of the original issuance of the Notes (such excess, "EXCESS INDEBTEDNESS"), any direct or indirect payment on or retirement of any Excess Indebtedness shall be deemed to be Net Available Proceeds received on the date of such payment or retirement; and (c) if any Indebtedness owed to the Company is assumed as contemplated by clause (vi), any direct or indirect payment on or retirement of any such Indebtedness shall be deemed to be Net Available Proceeds received on the date of such payment or retirement; and PROVIDED, FURTHER, that Net Available Proceeds shall include any earnings thereon received by the Company or any of its Subsidiaries until such Net Available Proceeds are applied in accordance with SECTION 8.24.

"NET VECTOR OFFERING PROCEEDS" means the cash proceeds received from any Vector Equity Offering, net of attorneys' fees, investment banking fees, accountants' fees, underwriting discounts and commissions and other customary fees and expenses actually incurred in connection therewith.

"NEW VALLEY" means $\mbox{New Valley Corporation, a Delaware corporation, and any successor thereto.$

"NEW VALLEY ASSETS" means any Equity Interest or other Investment in New Valley.

"NEW VALLEY EQUITY" means any Equity Interests in New Valley.

"NEW VALLEY TRANSFEREE" means (i) any Material Subsidiary of New Valley or (ii) any transferee of assets of New Valley if the value of all securities in such transferee maintained by New Valley represents a material portion of the assets of New Valley.

"NON-GRACE PERIOD COVENANT ACCELERATION DEFAULT" means the Company has defaulted on the payment of the principal or Accreted Value of, or Accreted Value Premium on, any Notes when the same becomes due and payable as a result of an acceleration due to non-compliance with SECTIONS 8.3, 8.4, 8.5, 8.6, 8.7, 8.8, the second paragraph of 8.16(A) and 10.1 of this Agreement.

"NON-RECOURSE INDEBTEDNESS" means Indebtedness:

- (1) as to which neither the Company nor any Restricted Subsidiary (a) provides any guarantee or credit support of any kind (including any undertaking, guarantee, indemnity, agreement or instrument that would constitute Indebtedness) or (b) is directly or indirectly liable (as a guarantor or otherwise);
- (2) no default with respect to which (including any rights that the holders thereof may have to take enforcement action against an Unrestricted Subsidiary) would permit (upon notice, lapse of time or both) any holder of any other Indebtedness of the Company (other than the Notes) or any Restricted Subsidiary to declare a default under such other Indebtedness or cause the payment thereof to be accelerated or payable prior to its stated maturity; and
- (3) the terms of which result in there being no recourse against any of the assets of the Company or its Restricted Subsidiaries.

"NON-TAX SHARING SUBSIDIARY" is defined in SECTION 9.4(C).

"NOTES" is defined in SECTION 1.

"NOTE DOCUMENTS" means this Agreement, the Security Agreements and the Notes.

"NOTICE OF END OF SOLE CONTROL" means a notice of end of sole control over the Securities Account delivered by the Collateral Agent pursuant to the Account Control Agreement.

"NOTICE OF SOLE CONTROL" means a notice of sole control over the Securities Account delivered by the Collateral Agent pursuant to the Account Control Agreement.

"NV HOLDINGS" means New Valley Holdings, Inc., a Delaware corporation, and any successor thereto. $\,$

"NV HOLDINGS CHECKING ACCOUNT" means Account #01596321075 of NV Holdings at Bank of America.

"NV HOLDINGS PLEDGE AGREEMENT" means a Pledge and Security Agreement substantially in the form of EXHIBIT C, dated as of the date of the Closing, between NV Holdings and the Collateral Agent for the benefit of the Holders, as amended, modified or supplemented from time to time.

"NVH ALLOCABLE AMOUNTS" is defined in SECTION 9.4(C).

"NVH TAX AGREEMENT" is defined in SECTION 9.4(C).

"OBLIGATIONS" means any principal, interest, penalties, fees, indemnifications, reimbursements, damages and other liabilities payable under the documentation governing any Indebtedness.

"OFFICER" means the Chairman of the Board, the President, any Vice President, the Treasurer, the Controller, the Secretary or the Assistant Secretary of the Company.

"OFFICERS' CERTIFICATE" means a certificate signed by two Officers or by an Officer and an Assistant Treasurer of the Company, and otherwise complying with the requirements of SECTIONS 22.1 and 22.2, except an Officers' Certificate given pursuant to SECTIONS 7.2 and 8.16 shall be signed by a principal operating officer, principal financial officer or a principal accounting officer.

"OPINION OF COUNSEL" means a written opinion from legal counsel who is reasonably acceptable to the Holders or the Collateral Agent (as the case may be) (which counsel may be an employee of or counsel to the Company), and otherwise complying with the requirements of this Agreement.

"OTHER BGLS GROUP DEBT" means any bond, debenture, note or other evidence of Indebtedness or any mortgage, indenture or other instrument under which there may be issued or by which there may be secured or evidenced any Indebtedness of the Company, or any Restricted Subsidiary other than the Note Documents.

"OTHER OBLIGATION ACCELERATION DEFAULT" means the Company, or any Restricted Subsidiary, defaults on any Other BGLS Group Debt, and as a result of such default, such Indebtedness of the Company or such Restricted Subsidiary, becomes due prior to its stated maturity, whether or not subordinated.

"OTHER OBLIGATION PAYMENT DEFAULT" means the Company, or any Restricted Subsidiary, defaults in the payment of any principal of or interest on any Other BGLS Group Debt, and such default extends beyond any period of grace provided with respect thereto.

"PBGC" means the Pension Benefit Guaranty Corporation referred to and defined in ERISA or any successor thereto.

"PERMITTED INVESTMENT" means an Investment by the Company or a Restricted Subsidiary in:

- (a) a Restricted Subsidiary or a Person which will, upon making such Investment becomes a Restricted Subsidiary; PROVIDED, HOWEVER, that such Person is engaged solely in Related Businesses prior to such Investment;
- (b) another Person if as a result of such Investment such other Person is merged or consolidated with or into, or transfers or

conveys all or substantially all its assets to the Company or a Restricted Subsidiary; PROVIDED, HOWEVER, that such Person is engaged solely in Related Businesses prior to such Investment; or

(c) Cash or cash equivalents.

"PERMITTED LIEN" means:

- (a) Liens created by the Security Agreements;
- (b) Liens imposed by any Governmental or Regulatory Authority for taxes, assessments or charges not yet due or that are being contested in good faith and by appropriate proceedings if adequate reserves with respect thereto are maintained on the books of the Company in accordance with GAAP;
- (c) carriers', warehousemen's, mechanics', materialmen's, repairmen's or other like Liens arising in the ordinary course of business that are not overdue for a period of more than thirty (30) days or that are being contested in good faith and by appropriate proceedings and Liens securing judgments but only to the extent for an amount and for a period not resulting in an Event of Default under CLAUSE (8) of SECTION 11;
- (d) pledges or deposits under worker's compensation, unemployment insurance and other social security legislation;
- (e) deposits to secure the performance of bids, trade contracts (other than for Indebtedness), leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business; and
- (f) easements, rights-of-way, restrictions and other similar encumbrances incurred in the ordinary course of business and encumbrances consisting of zoning restrictions, easements, licenses, restrictions on the use of property or minor imperfections in title thereto that, in the aggregate, are not material in amount, and that do not in any case materially detract from the value of the property subject thereto or interfere with the ordinary conduct of the business of the Company.

"PERMITTED PAYMENT" means the making of any Restricted Payment so long as the Company maintains the Required Cash Holdings on the BGLS Balance Sheet after giving pro forma effect to such Restricted Payment as measured on the date such Restricted Payment was made.

"PERMITTED VECTOR EXPENSES" means (i) costs and operating expenses of Vector, (ii) compensation of Vector Senior Managers allocable to Vector, and (iii) any management or consulting fees paid by the Company and the Restricted Subsidiaries to Vector.

Agreement.

"PERSON" means an individual, partnership, corporation, limited liability company, association, trust, unincorporated organization, or a government or agency or political subdivision thereof.

"PLAN" means an "employee benefit plan" (as defined in section 3(3) of ERISA) that is or, within the preceding five (5) years, has been established or maintained, or to which contributions are or, within the preceding five (5) years, have been made or required to be made, by the Company or any ERISA Affiliate or with respect to which the Company or any ERISA Affiliate may have any liability.

"PLAN OF LIQUIDATION" with respect to a Person, means a plan (including by operation of law) that provides for, contemplates or the effectuation of which is preceded or accompanied by (whether or not substantially contemporaneously) (i) the sale, lease, conveyance or other disposition of all or substantially all of the assets of such Person otherwise than as an entirety or substantially as an entirety and (ii) the distribution of all or substantially all of the proceeds of such sale, lease, conveyance or other disposition and all or substantially all of the remaining assets of such Person to holders of Equity Interests of such Person.

"PLEDGE AGREEMENTS" means, collectively, the BGLS Pledge Agreement, the NV Holdings Pledge Agreement, the Brooke Holding Pledge Agreement and any other pledge agreement, mortgage, security agreement or similar agreement which evidences the Liens on the Collateral granted by the Company or any Subsidiary Pledgor to the Collateral Agent.

"PRIMARY HOLDER" means any Holder, which together with all Affiliates of such Holder, holds in excess of \$10,000,000 in aggregate principal amount of Notes.

"PROPERTY" or "PROPERTY" means any property of any kind or nature whatsoever, real, personal or mixed (including fixtures), whether tangible or intangible.

"PUBLICLY-HELD COMPANY" means any corporation which has a class of equity interests registered pursuant to Section 12(b) or 12(g) of the Exchange Act.

"PURCHASE AGREEMENT YEAR" means the period from and including the date hereof to but excluding the date of the first anniversary hereof, the period from and including the date of the first anniversary hereof to but excluding the date of the second anniversary date hereof, the period from and including the date of the second anniversary hereof to but excluding the date of the third anniversary hereof, the period from and including the date of the third anniversary hereof to but excluding the date of the fourth anniversary hereof and the period from and including the date of the fourth anniversary hereof to and including the Maturity Date.

"PURCHASER" is defined in the first paragraph of this

"REFERENCE PERIOD" means any period of four consecutive fiscal quarters for which financial statements are available.

"RELATED BUSINESS" means any business which is the same as or ancillary to the tobacco businesses of any Restricted Subsidiary on the date hereof.

"RELEVANT PERIODS" is defined in SECTION 9.4(B).

"REQUIRED CASH HOLDINGS" means at least \$50,000,000 in Cash or cash equivalents.

"RESEARCH" means Vector Research Ltd., a Delaware corporation, or any successor thereto. $\,$

"RESTRICTED INVESTMENT" means an Investment other than a Permitted Investment.

"RESTRICTED PAYMENT" is defined in SECTION 8.3.

"RESTRICTED SUBSIDIARY" means (i) Brooke Holding, (ii) any Liggett Subsidiary, (iii) VTUSA and any Subsidiary of VTUSA, (iv) Research and any Subsidiary of Research and (v) any Subsidiary of the Company that is acquired or formed after the date of this Agreement other than any Subsidiary of New Valley or Brooke Overseas.

"RESTRICTED SUBSIDIARY ASSET SALE" means any sale, lease, transfer or other disposition (or series of related sales, leases, transfers or dispositions) of Equity Interests of a Restricted Subsidiary, property or other assets, including by way of a sale/leaseback transaction (each referred to for the purposes of this definition as a "disposition"), by any Restricted Subsidiary (including any disposition by means of merger, consolidation or similar transaction) other than (i) a disposition to any Restricted Subsidiary or the Company, (ii) a disposition of Property or assets in the ordinary course of business, (iii) dispositions of inventory in the ordinary course of business, (iv) a conveyance, sale, transfer, assignment or other disposition covered by the definition of "Company Asset Sale" and (v) sales of obsolete or worn-out equipment.

"RETIREMENT OF RESTRICTED SUBSIDIARY INDEBTEDNESS" means the prepayment, repayment or purchase of Indebtedness of the Restricted Subsidiaries and in connection with any such prepayment, repayment or purchase of Indebtedness of Restricted Subsidiaries, the retirement of such Indebtedness and the permanent reduction of the related loan commitment (if any) in the principal amount of the Indebtedness so prepaid, repaid or repurchased.

"RULE 144A" means Rule 144A promulgated under the Securities Act.

"SEC" means the Securities and Exchange Commission.

"SECURITIES ACCOUNT" means the "Account" as defined in the Account Control Agreement.

"SECURITIES ACT" means the Securities Act of 1933, as amended from time to time.

"SECURITIES INTERMEDIARY" has the meaning ascribed thereto in the Account Control Agreement.

"SECURITY AGREEMENTS" means the Pledge Agreements, the Account Control Agreement, the Vector Pledge Agreement and the Collateral Agency Agreement.

"SECURITY INTEREST" means the Lien on the Collateral which may be created by the Security Agreements in favor of the Holders.

"SECURITY PROCEEDS" is defined in SECTION 13.8.

"SHADOW DIVIDEND" means any payment or distribution made to any Group Executive based on the amount of any dividends or other distributions that would have been paid to such Group Executive had such Group Executive been at the record date for such dividends or other distributions a holder of shares of Vector Common Stock issuable upon exercise of any then unexercised option or options to acquire Vector Common Stock.

"SOLVENT" means when used with respect to any Person at the time of determination, that: (a) (i) the fair value of such Person's assets (both at fair valuation and at present fair saleable value) is in excess of the total amount of its known liabilities, (ii) such Person is then able and expects to be able to pay its debts as they mature (taking into account the timing and amounts of cash to be received by such Person and the amounts to be payable on or in respect of the debts of such Person) and (iii) such Person has capital sufficient to carry on its business as conducted and as proposed to be conducted, with contingent liabilities (such as, without limitation, guaranties and pension plan liabilities), being computed for the purposes of the foregoing at the amounts that, in light of all the facts and circumstances existing at the time, represent the amounts that can reasonably be expected to become actual or matured liabilities; and (b) in any case, such Person is not "insolvent" within the meaning of (i) Section 2 of the Uniform Fraudulent Transfer Act or Uniform Fraudulent Consequence Act of any applicable jurisdiction (and, with respect to such Act, is not properly to be presumed to be insolvent under Section 2(b) thereof), (ii) Section 101 of Title 11 of the United States Code or (iii) any comparable law or regulation applicable in the circumstances.

"STATE COMBINED TAXES" is defined in SECTION 9.4(B).

"SUBSIDIARY" means, as to any Person, any corporation, association or other business entity in which such Person or one or more of its Subsidiaries or such Person and one or more of its Subsidiaries owns sufficient equity or voting interests to enable it or them (as a group) ordinarily, in the absence of contingencies, to elect a majority of the directors (or Persons performing similar functions) of such entity, and any partnership or joint venture if more than a 50% interest in the profits or capital thereof is owned by such Person or one or more of its Subsidiaries or such Person and one or more of its Subsidiaries (unless such partnership or joint venture can and does ordinarily take major business actions without the prior approval of such Person or one or more of its Subsidiaries). Unless the context otherwise clearly requires, any reference to a "Subsidiary" is a reference to a Subsidiary of the Company.

"SUBSIDIARY PLEDGOR" means any of NV Holdings or Brooke

Holding.

"TANGIBLE NET WORTH" of any Person at any date means the Equityholders' Equity of such Person at such date, less the sum of the following: any writedown to reflect an impairment in value that is other than temporary and the book value of all assets which should be classified as intangibles (without duplication of deductions in respect of items already deducted in arriving at surplus and retained earnings) including goodwill, minority interests, research and development costs, trademarks, trade names,

copyrights, patents and franchises, unamortized debt discount and expense, such calculation to be reviewed by such Person's independent public accountants based on such procedures as are necessary to enable such accountants, under professional accounting standards, to report to the Holders in writing that in connection with such procedures no matter came to their attention that caused them to believe that such calculations should be adjusted.

"TAX AUDIT PAYMENT" is defined in SECTION 9.3(A)(I)(A).

"TAX PERCENTAGE" means the maximum combined marginal rate of Federal, state and local income tax applicable in the relevant year to a Delaware corporation the principal place of business of which is in New York City (the rate applicable to corporations doing business in New York City being selected solely because the laws of the State of New York govern the terms of this Agreement).

"TAX SHARING ATTRIBUTES" is defined in SECTION 8.9.

"TCW FUNDS" means (a) prior to the Closing, any Purchaser who is an Affiliate of the Trust Company of the West and (b) after the Closing, any Holder who is an Affiliate of Trust Company of the West.

"TOBACCO CLAIM OBLIGATION" means all obligations and liabilities, regardless of whether bonded or unbonded, to any plaintiff or other party in satisfaction or discharge of any final, nonappealable judgment, binding mediation or arbitration or settlement agreement (excluding the Master Settlement Agreement) relating to tobacco claims.

"TOBACCO LITIGATION BOND" means (i) any supersedeas or other bond that any Person commits to post in anticipation of or as a result of a judgment in a tobacco-related litigation or (ii) any amount paid or posted by any Person in connection with the Stipulation and Agreed Order Regarding Stay of Execution Pending Review and Related Matters by and among the plaintiffs in HOWARD A. ENGLE ET AL. V. R.J. REYNOLDS ET AL. and Philip Morris Incorporated, Lorillard Tobacco Co., Lorillard, Inc., Brooke Holding and Liggett.

"TOBACCO LITIGATION EXPENSE" means any amount expended in regard to defending, litigating, mediating, arbitrating, negotiating, researching, settling or discharging any tobacco-related claim including, without limitation, Tobacco Claim Obligations and any amounts paid under the Master Settlement Agreement.

"TRIGGERING ASSET SALE" means a Company Asset Sale (i) the Net Available Proceeds of which equal or exceed \$5,000,000 or (ii) the Net Available Proceeds of which will cause the Cumulative Net Available Proceeds, as of the date such Company Asset Sale is consummated, to equal or exceed \$5,000,000.

"U.S. LEGAL TENDER" means such coin or currency of the United States of America as at the time of payment is legal tender for the payment of public and private debts.

"UNIFORM COMMERCIAL CODE" means the Uniform Commercial Code as in effect from time to time in the State of New York.

"UNRESTRICTED SUBSIDIARY" means any Subsidiary of the Company other than a Restricted Subsidiary.

 $\mbox{"VECTOR"}$ means Vector Group Ltd., a Delaware corporation, and any successor thereto.

"VECTOR COMMON STOCK" means common stock, par value \$.10 per share, of Vector.

"VECTOR EQUITY OFFERING" means an offering of Equity Interests in Vector by Vector.

"VECTOR EXPANDED AFFILIATE" means any director or employee of Vector or any of its Affiliates or Associates.

"VECTOR PLEDGE AGREEMENT" means the Acknowledgment and Pledge Agreement, dated as of May 14, 2001, by and among Vector, the Collateral Agent and the Purchasers, as amended, modified and supplemented from time to time.

"VECTOR SENIOR MANAGER" means each of the following, as of the date hereof, any chairman of the Board, president, chief executive officer, chief operating officer, executive vice president, vice president, treasurer, assistant treasurer, secretary, assistant secretary and any other senior officer of or person performing similar functions on behalf of Vector, New Valley or any Unrestricted Subsidiary and any successor of any of the foregoing.

"VECTOR TOBACCO CREDIT FACILITY" means the Loan Facility Agreement, dated as of March 6, 2001, from the Company and Vector to VTUSA and Vector Tobacco Ltd., as amended, modified and supplemented from time to time.

"VECTOR VOTING POWER" the total voting power of Voting Interests of Vector. $\ensuremath{^{\circ}}$

 $\,$ "VGR GROUP" means the affiliated group filing a consolidated income tax return of which Vector is the common parent.

"VOTING INTERESTS" of a corporation, limited liability company, partnership or other entity means all classes of Equity Interests of such corporation, limited liability company, partnership or other entity then outstanding and normally entitled to vote in the election of directors, management committee members or persons fulfilling similar functions.

 $\mbox{"VTUSA"}$ means Vector Tobacco (USA) Ltd., a Delaware corporation, and any successor thereto.

"WHOLLY OWNED SUBSIDIARY" means, at any time, any Subsidiary one hundred percent (100%) of all of the equity interests (except directors' qualifying shares) and voting interests of which are owned by any one or more of the Company and the Company's other Wholly Owned Subsidiaries at such time.

SCHEDULE C

APPLICABLE ACCRETION PERCENTAGES

C-1

FXHTBTT A

FORM OF NOTE

THIS NOTE HAS NOT BEEN registered under the Securities Act of 1933, as amended (the "Securities Act"), OR ANY STATE SECURITIES LAWS. NEITHER THIS SECURITY NOR ANY INTEREST OR PARTICIPATION HEREIN may be offered, sold, ASSIGNED, TRANSFERRED, pledged, ENCUMBERED or otherwise DISPOSED OF IN THE ABSENCE OF SUCH registration UNLESS SUCH TRANSACTION IS EXEMPT FROM, OR NOT SUBJECT TO, SUCH REGISTRATION AND ANY APPLICABLE STATE SECURITIES LAWS HAVE BEEN COMPLIED WITH.

THE HOLDER OF THIS NOTE BY ITS ACCEPTANCE HEREOF AGREES NOT TO OFFER, SELL, ASSIGN, TRANSFER OR OTHERWISE DISPOSE OF THIS NOTE, PRIOR TO THE DATE THAT IS TWO (2) YEARS (OR SUCH SHORTER PERIOD THAT MAY HEREAFTER BE PROVIDED UNDER RULE 144(k) UNDER THE SECURITIES ACT AS PERMITTING RESALES BY NON-AFFILIATES OF RESTRICTED SECURITIES WITHOUT RESTRICTION) AFTER THE LATER OF THE ORIGINAL ISSUE DATE HEREOF AND THE LAST DATE ON WHICH THE COMPANY OR ANY AFFILIATE OF THE COMPANY WAS THE OWNER OF THIS NOTE (OR ANY PREDECESSOR OF SUCH SECURITY) EXCEPT (A) TO THE COMPANY, (B) PURSUANT TO A REGISTRATION STATEMENT WHICH HAS BEEN DECLARED EFFECTIVE UNDER THE SECURITIES ACT, (C) FOR SO LONG AS THIS NOTE IS ELIGIBLE FOR RESALE PURSUANT TO RULE 144A UNDER THE SECURITIES ACT, TO A "QUALIFIED INSTITUTIONAL BUYER" (AS DEFINED IN RULE 144A) THAT PURCHASES FOR ITS OWN ACCOUNT OR FOR THE ACCOUNT OF A QUALIFIED INSTITUTIONAL BUYER TO WHOM NOTICE IS GIVEN THAT THE TRANSFER IS BEING MADE IN RELIANCE ON RULE 144A, (D) IN AN OFFSHORE TRANSACTION MEETING THE REQUIREMENTS OF RULE 903 OR 904 UNDER THE SECURITIES ACT, (E) TO AN INSTITUTIONAL "ACCREDITED INVESTOR" WITHIN THE MEANING OF RULE 501(A)(1), (2), (3) OR (7) UNDER THE SECURITIES ACT THAT IS PURCHASING THIS NOTE FOR ITS OWN ACCOUNT FOR INVESTMENT PURPOSES AND NOT WITH A VIEW TO, OR FOR OFFER OR SALE IN CONNECTION WITH, ANY DISTRIBUTION IN VIOLATION OF THE SECURITIES ACT OR (F) PURSUANT TO ANOTHER AVAILABLE EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT, SUBJECT TO THE COMPANY'S AND THE TRANSFER AGENT'S RIGHT PRIOR TO ANY SUCH OFFER, SALE OR TRANSFER PURSUANT TO CLAUSE (D), (E) OR (F) TO REQUIRE THE DELIVERY OF AN OPINION OF COUNSEL, CERTIFICATIONS AND/OR OTHER INFORMATION SATISFACTORY TO THE COMPANY AND THE REGISTRAR, AND IN EACH OF THE FOREGOING CASES, A CERTIFICATE OF TRANSFER IN THE FORM APPEARING ON THIS NOTE IS COMPLETED AND DELIVERED BY THE TRANSFEROR TO THE TRANSFER AGENT.

BGLS INC.

10% SENIOR SECURED NOTE DUE MARCH 31, 2006

No.	[]		May	14,	2001
\$[1	PPN	#055	432-	A*-9

FOR PURPOSES OF SECTION 1273 OF THE INTERNAL REVENUE CODE OF 1986, AS AMENDED (THE "CODE"), THIS NOTE HAS ORIGINAL ISSUE DISCOUNT. FOR PURPOSES OF SECTION 1273 OF THE CODE, THE ISSUE PRICE IS \$_______ AND THE AMOUNT OF ORIGINAL ISSUE DISCOUNT IS \$______. FOR PURPOSES OF SECTION 1275 OF THE CODE, THE ISSUE DATE OF THIS NOTE IS MAY 14, 2001. FOR PURPOSES OF SECTION 1272 OF THE CODE, THE YIELD TO MATURITY (COMPOUNDED SEMI-ANNUALLY) IS 15.07%.

Payments of principal and Accreted Value of and interest and Accreted Value Premium (if any) on the Notes are to be made in lawful money of the United States of America by the method and at the address specified with respect to such Holder in SCHEDULE A to the Note Purchase Agreement.

This Note is one of a series of 10% Senior Secured Notes Due March 31, 2006 (herein called the "NOTES") issued pursuant to a Note Purchase Agreement, dated as of May 14, 2001, as it may be amended, modified or supplemented from time to time (the "NOTE PURCHASE AGREEMENT"), between the Company and the Purchasers named therein and is entitled to the benefits thereof. Each Holder of this Note will be deemed, by its acceptance hereof, to have agreed to the confidentiality provisions set forth in SECTION 21 of the Note Purchase Agreement.

This Note is secured as provided in the Note Documents. Reference is hereby made to the Note Documents for a description of the properties and assets in which a security interest has been granted, the nature

Exhibit A-2

and extent of the security, the terms and conditions upon which the security interests were granted and the rights of the Holder of this Note in respect thereof

This Note is registered as to both principal and stated interest with the Company pursuant to United States Treasury Regulation Section 5f.103-1 and may be transferred only by the surrender of a Note by the transferor and the issuance by the Company of a new Note to the transferee. As provided in the Note Purchase Agreement, upon surrender of this Note for registration of transfer, duly endorsed, or accompanied by a written instrument of transfer duly executed, by the registered holder hereof or such holder's attorney duly authorized in writing, a new Note of this series for a like principal amount will be issued to, and registered in the name of, the transferee. Prior to due presentment for registration of transfer, the Company may treat the person in whose name this Note is registered as the owner hereof for the purpose of receiving payment and for all other purposes, and the Company will not be affected by any notice to the contrary.

The Company will make required prepayments of principal on the dates and in the amounts specified in the Note Purchase Agreement. This Note is also subject to optional prepayment, in whole or from time to time in part, at the times and on the terms specified in the Note Purchase Agreement, but not otherwise.

If an Event of Default, as defined in the Note Purchase Agreement, occurs and is continuing, the principal or Accreted Value of or interest and Accreted Value Premium (if any) on this Note may be declared or otherwise become due and payable in the manner, at the price (including any applicable Accreted Value Premium) and with the effect provided in the Note Purchase Agreement.

NEITHER ANY GROUP EXECUTIVE (AS DEFINED IN THE NOTE PURCHASE AGREEMENT) OR NEW VALLEY CORPORATION SHALL BE PERMITTED TO VOTE ANY NOTES THAT ANY OF THEM MAY HOLD FROM TIME TO TIME UNDER ANY CIRCUMSTANCES, INCLUDING, WITHOUT LIMITATION, IN THE EVENT OF A BANKRUPTCY OF THE COMPANY. IN THE EVENT OF A BANKRUPTCY OF THE COMPANY, ALL GROUP EXECUTIVES AND NEW VALLEY CORPORATION SHALL BE DEEMED TO BE ENTITIES DESIGNATED IN 11 U.S.C. SS.1126(E) FOR PURPOSES OF DETERMINING ACCEPTANCE OF ALLOWED CLAIMS PURSUANT TO 11 U.S.C. SS.1126(C).

 $\hbox{ This Note shall be construed and enforced in accordance with the laws of the State of New York. } \\$

BGLS INC.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen

Title: Executive Vice President

Exhibit A-3

OPTION OF HOLDER TO ELECT PURCHASE

$ \hbox{ If you want to elect to have this Note purchased by t Company pursuant to Section 8.24 or 8.25 of the Note Purchase Agreement the box below: } \\$	
[] Section 8.24 [] Section 8.25	
If you want to elect to have only part of the Note pu by the Company pursuant to Section 8.24 or Section 8.25 of this Agreeme the amount you elect to have purchased: \$	
Name:	
(Print Name)	
Date: Your Signature:	
(Sign exactly as your name appears on the Note)	
Tax Identification No.:	

Exhibit A-4

COLLATERAL AGENCY AGREEMENT

This Collateral Agency Agreement (this "AGREEMENT") is made and entered into as of May 14, 2001 by and among (i) BGLS Inc., a Delaware corporation (the "COMPANY"), (ii) Brooke Group Holding Inc., a Delaware corporation ("BROOKE HOLDING"), (iii) Vector Group Ltd., a Delaware corporation ("VECTOR"), (iv) New Valley Holdings, Inc., a Delaware corporation ("NV HOLDINGS"), (parties (i) through (iv) the "COMPANY PARTIES"), (v) United States Trust Company of New York, a New York banking corporation ("U.S. TRUST") as collateral agent (the "COLLATERAL AGENT") appointed pursuant to this Agreement with reference to that certain Note Purchase Agreement (the "NOTE PURCHASE AGREEMENT") dated as of even date herewith under which the Company shall issue to the purchasers (the "PURCHASERS") \$60,000,000 in aggregate principal amount of the Company's 10% Senior Secured Notes due March 31, 2006 (the "NOTES"), (vi) the Purchasers under the Note Purchase Agreement and (vii) such other parties who may become a party pursuant to SECTION 7.13 hereof. Defined terms appear in SECTION 1.

WHEREAS, the obligations of the Company under the Notes and the Note Purchase Agreement are secured by the Security Agreements;

WHEREAS, pursuant to the Security Agreements, the Company Parties have granted to the Collateral Agent, for the benefit of the holders of the Notes (the "HOLDERS"), a security interest in and liens on the Collateral (as defined in the Note Purchase Agreement); and

WHEREAS, it is a condition precedent to the consummation of the transactions contemplated by the Note Purchase Agreement that this Agreement shall have been duly executed and delivered.

NOW THEREFORE, in consideration of the foregoing and the mutual agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

Section 1. TERMS AND DEFINITIONS.

1.01 DEFINITIONS. Capitalized terms used but not defined herein have the respective meanings given to such terms in the Note Purchase Agreement. As used in this Agreement, the following terms shall have the following respective meanings:

"DEFAULT" and "EVENT OF DEFAULT" shall have the respective meanings given to such terms in the Note Purchase Agreement.

"PERMITTED SECURITIES" has the meaning ascribed thereto in the Account Control Agreement.

"RESPONSIBLE OFFICER" when used with respect to the Collateral Agent, means any officer within the Corporate Trust Administration of the Collateral Agent (or any successor group of the Collateral Agent) with direct responsibility for the administration of the Note Documents and also means, with

respect to a particular corporate trust matter, any other officer to whom such matter is referred, but only for so long as it is referred, because of his knowledge of and familiarity with the particular subject.

1.02 INTERPRETATION.

- (a) Unless the context otherwise indicates, words expressed in the singular shall include the plural and vice versa.
- (b) Headings of sections herein are solely for convenience of reference, do not constitute a part hereof and shall not affect the meaning, construction or effect hereof.
- (c) The words "herein," "hereof," "hereby," "hereunder," and other words of similar import refer to this Agreement as a whole and not to any particular Section or subdivision hereof.
- (d) Unless the context otherwise indicates, any reference to a "Section" or "Schedule" refers to a Section of or a Schedule to this Agreement, respectively.
- 1.03 HOLDERS' ACTION OR INSTRUCTIONS. Any action permitted to be taken by the Holders under this Agreement, or any direction that the Holders may give to the Collateral Agent, may be taken or given, as the case may be, only by the Majority Holders unless specifically stated otherwise herein.
 - Section 2. APPOINTMENT, COMPENSATION AND AUTHORIZATION OF THE COLLATERAL AGENT.
- 2.01 APPOINTMENT OF THE COLLATERAL AGENT. U.S. Trust is hereby appointed as Collateral Agent for the Holders and is irrevocably authorized and empowered to (i) hold the Collateral for the benefit of the Holders, (ii) exercise such authority, rights, powers, and duties hereunder as specifically are delegated to and accepted by the Collateral Agent hereunder and (iii) take such other action in connection with the foregoing as the Holders may from time to time direct in accordance with the terms and conditions of this Agreement, the Note Purchase Agreement and the Security Agreements. U.S. Trust hereby accepts its appointment as Collateral Agent and agrees to perform the duties of the Collateral Agent specified herein, in the Note Purchase Agreement and in the Security Agreements and to exercise the powers granted hereby and thereby, in either case in accordance with the terms hereof or thereof, as the case may be.
- 2.02 COMPENSATION OF THE COLLATERAL AGENT. As compensation for its services as Collateral Agent, the Company agrees to pay or cause to be paid the Collateral Agent on the Closing Date and, so long as this Agreement remains in effect, annually on the first day of April, the sum agreed between the Company and the Collateral Agent in writing and all the reasonable fees, costs and expenses incurred in good faith by the Collateral Agent (including, without limitation, the reasonable fees and disbursements of its counsel and other advisers as the Collateral Agent reasonably elects to retain) (i) arising in connection with the preparation, execution, delivery, performance, modification and termination of this Agreement and the Security Agreements in good faith in connection with the administration of the Collateral, the sale or other disposition thereof pursuant to Security Agreements and the preservation, protection, defense or enforcement of the Collateral Agent's rights under the

Security Agreements and in and to the Collateral or (ii) incurred in good faith by the Collateral Agent in connection with the resignation or removal of the Collateral Agent pursuant to SECTION 5.01. Additionally, the Company agrees to (A) indemnify and hold harmless the Collateral Agent from any present or future claim or liability for any stamp or other similar tax and any penalties or interest with respect thereto, that may be assessed, levied or collected by any jurisdiction in connection with this Agreement, the Note Purchase Agreement, the Security Agreements or any Collateral and (B) pay or to reimburse the Collateral Agent for any and all amounts in respect of all search, filing, recording and registration fees, taxes, excise taxes and other similar imposts that may be payable or determined to be payable in respect of the execution, delivery, filing, performance and enforcement of this Agreement, the Note Purchase Agreement, the Security Agreements and all documents (including, without limitation, financing statements) provided for herein or therein.

2.03 AUTHORIZATION OF SECURITY AGREEMENTS.

The Purchasers hereby authorize the Collateral Agent's entering into the Security Agreements on their behalf.

2.04 SECURITIES ACCOUNT NOTICES.

- (a) The Collateral Agent shall deliver a Notice of Sole Control promptly after receiving written instructions from the Majority Holders to deliver such Notice of Sole Control. The Collateral Agent shall not deliver a Notice of Sole Control absent delivery by the Majority Holders of such written instructions.
- (b) The Collateral Agent shall deliver a Notice of End of Sole Control promptly after receiving the written instructions of the Majority Holders to deliver such Notice of End of Sole Control. The Collateral Agent shall not deliver a Notice of End of Sole Control absent delivery by the Majority Holders of such written instructions.
- (c) During the continuance of an Account Notice Event, the Majority Holders shall provide the Collateral Agent with written instructions as to Permitted Securities in which the balance of the funds on deposit in the Securities Account shall be invested.
- (d) During the continuance of an Account Notice Event, if the Majority Holders deliver written instructions to the Collateral Agent for the Collateral Agent to withdraw a specified amount of funds from the Securities Account and deliver such funds to the Company in exchange for an Officer's Certificate containing information to be specified by the Majority Holders, the Collateral Agent shall withdraw such specified amount of funds from the Securities Account and deliver such specified amount of funds to the Company in exchange for an Officer's Certificate containing the specified information.

2.05 WITHDRAWALS FROM SECURITIES ACCOUNT DURING ACCOUNT NOTICE

EVENT.

In the event of an Account Notice Event, the Majority Holders may deliver written instructions to the Collateral Agent to make the withdrawals from the Securities Account specified in Section 13.11 of the Note Purchase Agreement.

Section 3. DUTIES, POWERS AND RIGHTS OF THE COLLATERAL AGENT.

3.01 SPECIFIC DUTIES OF THE COLLATERAL AGENT. The Collateral Agent shall have the following duties:

- (a) upon the receipt by it of written instructions of the Majority Holders, execute and deliver on behalf of the Holders such documents as the Majority Holders shall deem necessary or appropriate and provide to the Collateral Agent from time to time to maintain the perfection of any lien in, to or upon the Collateral or any portion thereof, that has been, are or will be granted in favor of the Collateral Agent pursuant to the Security Agreements;
- (b) accept, on behalf of the Holders, any part of the Collateral delivered to it, including, without limitation, any certificated securities, instruments and documents, and accept, on behalf of the Holders, any new Collateral given as security for the Secured Obligations, and execute and deliver, on behalf of the Holders, such documents or instruments as the Majority Holders deem necessary or appropriate and provide to the Collateral Agent with written instructions of the Majority Holders to evidence the creation of any lien with respect thereto and to perfect such lien;
- (c) upon the receipt by it of written instructions of the Majority Holders, release the Collateral or any portion thereof from any liens thereon that were created pursuant to the Security Agreements;
- (d) furnish to the Holders, promptly upon receipt thereof, duplicates of all reports, notices, requests, demands, certificates and other documents received by it under this Agreement, the Security Agreements, the Note Purchase Agreement or other documents provided for herein or therein;
- (e) provide to the Holders a copy of all written notices received from the Company or any Subsidiary Pledgor with respect to any capital stock or securities that constitute Collateral and, upon receipt by it of written instructions of the Majority Holders, exercise all rights and powers determined by the Majority Holders that are appurtenant to any such capital stock or securities that become a part of the Collateral, including, without limitation, the right to vote stock, to receive dividends or other distributions, and to grant or refrain from granting any consent or waiver, all in accordance with such written instructions;
- (f) inform the Holders in writing of the existence of any Default or Event of Default promptly upon learning of the same; PROVIDED, HOWEVER, that the Collateral Agent shall not be deemed to have any knowledge whatsoever of any Default or Event of Default unless a Responsible Officer of the Collateral Agent has actually received written notice stating that a Default or an Event of Default has occurred from any of the Holders or the Company Parties;

- (g) upon receipt by it of written instructions of the Majority Holders, take those actions determined by the Majority Holders as necessary to protect and preserve the Collateral and realize on and foreclose upon the Collateral, including, without limitation, initiating (at the expense of the Company) and defending any and all actions or proceedings in any court of law or equity that may be brought affecting any of the Collateral or any portion thereof or otherwise pursue any remedies available to any Holder or to it in respect of the Collateral or any portion thereof, which actions may include, without limitation, initiating and conducting any public or private sale or pursuing any other actions or remedies relating to the Collateral or any portion thereof; provided, HOWEVER, that the Collateral Agent shall be under no obligation to exercise any of its rights and powers under this Agreement unless it shall have received security and indemnity satisfactory to it against any loss, liability or expense;
- (h) provide, at the written instructions of the Majority Holders, notice required by the Note Purchase Agreement or the Security Agreements, or by law, to the Company, or any other party entitled thereto, in order to take any actions required or authorized to be taken under this Agreement or specified in written instructions of the Majority Holders;
- (i) receive any and all amounts of any kind paid pursuant to the Security Agreements and receive proceeds of the Collateral subsequent to an Event of Default and apply such amounts or proceeds as specified in SECTION 4.01; and
- (j) take, or refrain from taking, such other actions (but only such actions that are set forth in this Agreement) as the Majority Holders shall from time to time direct by written instruction; PROVIDED, HOWEVER, that the Collateral Agent may, in its sole discretion, refrain from taking such action (other than an action required or necessary to discharge any duty under SECTION 5.01 below) if the taking of such action would expose it to liability, financial or otherwise for which it does not receive adequate protection.

3.02 DUTIES LIMITED.

(a) The Collateral Agent shall be obligated to perform such duties and only such duties as specifically set forth in this Agreement and no implied covenants or obligations shall be read into this Agreement against the Collateral Agent. The Collateral Agent shall be obligated to take any actions or exercise any rights, powers or remedies which are discretionary with the Collateral Agent under this Agreement only as may be specified in a written notice from the Majority Holders; PROVIDED, HOWEVER, that the Collateral Agent shall not be required to take any actions specified in a written notice if the provisions of this Agreement expressly prohibit such action. The Collateral Agent shall be deemed to have exercised reasonable care in the custody and preservation of the Collateral in its possession if the Collateral is accorded treatment substantially equal to that which the Collateral Agent accords to similar property held by it as collateral agent or in a comparable capacity, it being understood that the Collateral Agent shall not have any responsibility for (a) ascertaining or taking any action with respect to calls, conversions,

exchanges, maturities or similar matters relative to any property held as Collateral, whether or not the Collateral Agent has or is deemed to have knowledge or notice of such matters or (b) taking any steps to maintain the value of any property held as Collateral or to preserve rights against any parties with respect thereto. Except as expressly provided herein or in the Note Purchase Agreement or the Security Agreements, the Collateral Agent shall not have any duty or obligation, express or implied, to:

- (i) manage, control, use, maintain, sell, dispose of, purchase, bid for or otherwise deal with the Collateral or any portion thereof, or to otherwise take or refrain from taking any action under, or in connection with this Agreement, the Note Purchase Agreement or any Security Agreement, except to the extent required by law;
- (ii) take any action that relates to, materially affects, or impairs the amounts that the Holders may recover from disposition of the Collateral, including, without limitation, any election or waiver of remedies available under the Security Agreements, or with respect to the Collateral or the manner of foreclosure upon the same; any determination of the order and timing of foreclosure upon any portion of the Collateral or of the amount of any credit bid to be entered at any public or private, judicial, or nonjudicial sale of the Collateral; the pursuit of any remedies against the Company or any of its Subsidiaries following the completion of foreclosure upon the Collateral; the compromise or settlement of any claims against the Company or any of its Subsidiaries, including without limitation the conduct of any negotiations relating to the same or with a view toward the termination of any pending foreclosure proceedings;
- $\mbox{(iii)}$ obtain or maintain insurance on the Collateral or any other insurance;
- (iv) pay or discharge any tax, assessment or other governmental charge or any lien or encumbrance of any kind owing with respect to, or assessed or levied against, any part of the Collateral;
- (v) take any action or omit to take any action provided for in the Security Agreements;
 - (vi) advance any monies for any purpose;
- (vii) except at the specific written instructions of the Majority Holders, record or file the Security Agreements, any other document or any other instrument provided to it referred to herein or therein with respect to any lien; or
- (viii) except as provided in Section 2.04, deliver a Notice of Sole Control or a Notice of End of Sole Control.
- (b) In addition to and not in limitation of the provisions of SECTION 3.02(A), under no circumstances shall the Collateral Agent have any duty or obligation to take any actions hereunder other than those under Section 5.01, even if instructed to do so by the Majority Holders or if expressly set forth herein, if the Collateral Agent determines, in its sole and absolute discretion, that such actions would subject it to liability or expense for which satisfactory indemnity to the Collateral Agent has not been provided hereunder or otherwise.

- (c) Except as otherwise provided herein at the written instruction of the Majority Holders, the Collateral Agent shall have no obligation or liability in respect of the recording, rerecording, filing or refiling of any instruments, documents, financing statements or continuation statements or to take any other action hereunder with respect to the security interests created hereby or pursuant to the Security Agreements, and the Collateral Agent shall have no obligation to monitor the status of the security interests as a perfected security interest created hereunder or under the Security Agreements.
- 3.03 SPECIFIC POWERS OF THE COLLATERAL AGENT. In addition to all powers necessary, appropriate, desirable or incidental to the Collateral Agent's performance of the specific duties set forth in SECTION 3.01, the Collateral Agent is hereby empowered and authorized to do, in its sole and absolute discretion, any and all of the following in connection with its performance of such duties; PROVIDED, HOWEVER, that in no event shall it have any obligation to do so:
- (a) establish bank accounts in its name with the right to be the only party authorized to draw from such account or accounts;
- (b) employ such persons, firms or professionals as it shall reasonably deem appropriate or desirable in connection with the performance of its duties hereunder, including, without limitation, appraisers, auctioneers, stockbrokers, custodians of securities, fiduciaries, commercial banks, investment banks, accountants and attorneys; and
- (c) execute and deliver, as Collateral Agent and on behalf of the Majority Holders, any agreements, escrow instructions, bills of sale, releases, applications or any other documents related to or in any way connected with any disposition of the Collateral, or any portion thereof, permitted under this Agreement or directed by the Majority Holders in accordance with the terms hereof; PROVIDED, HOWEVER, that in the event it is unwilling or unable for any reason to execute and deliver such documents, then it promptly shall notify the Holders of such unwillingness or inability and shall request execution and delivery of such documents by the Holders.
- 3.04 WRITTEN INSTRUCTIONS. Any written request or written instructions required or permitted to be given hereunder to the Collateral Agent shall be given exclusively by the Majority Holders. In the event that the Collateral Agent receives written instructions from the Majority Holders, that the Collateral Agent determines, in its sole and absolute discretion, to be ambiguous, inconsistent, in conflict with other instructions previously received or otherwise insufficient to direct the actions of the Collateral Agent, then the Collateral Agent shall have no obligation whatsoever to take or refrain from taking any action pursuant to such written instructions, but shall instead do the following:
- (a) FIRST, seek additional written instructions from the Majority Holders reasonably satisfactory to it; or
- (b) SECOND, if the Collateral Agent is reasonably dissatisfied with the further instructions or does not receive further instructions pursuant to SECTION 3.04(A), resign as Collateral Agent in accordance with this Agreement.

The Collateral Agent shall not be liable to any party hereto (or any Person claiming by, through or under such party) by reason of its actions under this SECTION 3.04.

- 3.05 RELIANCE. In acting with respect to this Agreement, the Note Purchase Agreement or the Security Agreements, the Collateral Agent shall be entitled to rely conclusively:
- (a) on any communication reasonably believed by it to be genuine and to have been made, sent or signed by the Person by whom it purports to have been made, sent or signed;
- (b) as to any matters of fact that might reasonably be expected to be within the knowledge of the Holders, the Company Parties or the Securities Intermediary, on a certificate signed by or on behalf of any of the Holders, the Company Parties or the Securities Intermediary;
- (c) on the advice or services of any persons, firms or professionals employed by it pursuant to SECTION 3.03(B) and rely upon the opinions and statements of any professional advisor so employed; and
- (d) on any resolution, statement, certificate, instrument, opinion, report, notice, request, consent, order, bond or other paper or document that it reasonably believes to be genuine and to have been signed or presented by the proper person or, in the case of cables, facsimile transmissions, telecopies and telexes, to have been sent by the proper person. The Collateral Agent shall not be responsible, may conclusively rely upon and shall be protected, indemnified and held harmless by the Company Parties for the sufficiency or accuracy of the form of, or the execution, validity, value or genuineness of any document or property received, held or delivered by it hereunder, or the signature or endorsement thereon, or for any description therein, nor shall the Collateral Agent be responsible or liable in any respect on account of the identity, authority or rights of the Persons executing or delivering or purporting to execute or deliver any document, property or this Agreement absent the gross negligence or willful misconduct of the Collateral Agent.

The Collateral Agent shall not be liable to any party hereto for any consequence of any such relying, acting, or refraining to act. Nothing in this SECTION 3.05 shall impair the right of the Collateral Agent in its discretion to take or omit to take any action that the Collateral Agent deems proper to take or omit to take if such action or omission is not inconsistent with any notice or direction from the Majority Holders; and that the Collateral Agent shall not be under any obligation to take any action that is discretionary to the Collateral Agent under this Agreement, the Note Purchase Agreement or the Security Agreements except as may be specified in a written instruction of the Majority Holders.

- 3.06 NO RESPONSIBILITY. The Collateral Agent does not assume any responsibility for:
- (a) any failure or delay in performance or breach by Vector, the Company or any Subsidiaries of the Company of any of their respective obligations under the Note Purchase Agreement or the Security Agreements or the Securities Intermediary under the Account Control Agreement;

- (c) the legality, validity, effectiveness, adequacy or enforceability of the Note Purchase Agreement or the Security Agreements or the priority or perfection of the Security Interest;
- (d) the validity, enforceability or sufficiency of an agreement or instrument, the sufficiency of any amount on deposit in any Group Account to satisfy any obligation of any Company Party, any depreciation of diminution in the value of any Collateral or any income thereon, or the effect of any withdrawals or other transactions in the Group Accounts by or at the direction of any Company Party; or
- (e) the terms and conditions of any other document, instrument or agreement in connection herewith, except those agreements to which it is a party.

As to any event or occurrence in which neither the Collateral Agent nor any Person acting on its behalf is a participant, the Collateral Agent shall be conclusively presumed to have no knowledge of such event or occurrence, absent gross negligence or willful misconduct, except to the extent that a Responsible Officer of the Collateral Agent shall have received a written notice from any of the Holders or any Company Party with respect thereto.

3.07 COLLATERAL AGENT PROTECTED. The Collateral Agent shall be protected fully in acting or refraining to act upon any certificate, statement, instrument, opinion, report, notice, request, consent, order, bond or paper or document reasonably believed by it to be genuine and to have been signed or presented by the proper party or parties. The Collateral Agent shall not be liable for any error in judgment made in good faith by the Collateral Agent unless the Collateral Agent was grossly negligent in ascertaining the pertinent facts or acted intentionally in bad faith. The Collateral Agent may consult with legal counsel with significant experience in transactions of the type contemplated by the Note Purchase Agreement, and the advice of such counsel shall constitute full and complete protection in respect of any action taken, suffered or omitted by it under this Agreement in good faith and in accordance with such advice of counsel. The Collateral Agent may execute any of its powers hereunder or perform any duties hereunder either directly or through agents, attorneys or custodians, and the Collateral Agent shall not be responsible for any misconduct or negligence on the part of, or for the supervision of, any agent, attorney or custodian appointed with due care by it hereunder; PROVIDED, HOWEVER, that as between the other parties hereto and the Collateral Agent, all such powers and duties are those of the Collateral Agent as provided hereunder.

3.08 LIMITATION ON LIABILITY.

- (a) The Collateral Agent may not be relieved from liability for its own grossly negligent action, its own grossly negligent failure to act, or its own willful misconduct, except that:
 - (i) the Collateral Agent shall not be liable for any error of judgment made in good faith by a Responsible Officer, unless it is proved that the Collateral Agent was grossly negligent in ascertaining the pertinent facts; and

(ii) the Collateral Agent shall not be liable with respect to any action it takes or omits to take in good faith in accordance with a direction received by it pursuant to this Agreement or the other Note Documents:

3.09 LIABILITY FOR MONEY AND INTEREST. The Collateral Agent shall not be liable for any interest or any money received by it except as the Collateral Agent may agree in writing. Money held in trust by the Collateral Agent need not be segregated from other funds except as required by law.

Section 4. APPLICATION OF PROCEEDS OF COLLATERAL.

4.01 APPLICATION OF PROCEEDS OF COLLATERAL. The receipt of any amounts on behalf of the Holders under the Note Purchase Agreement, the Security Agreements or otherwise with respect to the Collateral and the proceeds of any sale, enforcement or other disposition of any of the Collateral or any other distribution in respect of the Collateral shall be applied by the Holders and the Collateral Agent first, to the payment of all proper costs incurred by the Collateral Agent in the collection thereof (including stamp or other taxes in respect of the transfer or sale of any Collateral and the reasonable compensation, expenses and the disbursements of the Collateral Agent and its counselors) and then in accordance with Section 5.13 of the applicable Pledge Agreement.

Section 5. RESIGNATION OR REMOVAL OF COLLATERAL AGENT.

5.01 RESIGNATION OR REMOVAL OF THE COLLATERAL AGENT. The Collateral Agent may, by written notice to the Holders, at any time resign its agency under this Agreement. The Majority Holders may remove the Collateral Agent by written notice to the Collateral Agent. No such resignation or removal shall become effective, unless and until a successor Collateral Agent under this Agreement is appointed and has accepted the appointment, with such successor Collateral Agent to be appointed by the Majority Holders; PROVIDED, HOWEVER, that if no successor Collateral Agent shall have been so appointed and shall have accepted such appointment within thirty (30) days after the retiring Collateral Agent's giving notice of resignation or after notice to the retiring Collateral Agent of the retiring Collateral Agent's removal, as the case may be, then the retiring Collateral Agent may apply to any court of competent jurisdiction, at the expense of the Company, to appoint a successor Collateral Agent to act until such time as a successor shall have been appointed by the Holders. Upon the acceptance of any appointment as Collateral Agent hereunder by a successor Collateral Agent, such successor Collateral Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Collateral Agent, and the retiring Collateral Agent shall be discharged from any further duties and obligations under this Agreement except the duty to execute and deliver any documents necessary to vest or confirm the vesting of such rights, powers, privileges, and duties in such successor Collateral Agent and to deliver possession of any Collateral in the possession of such retiring Collateral Agent to such successor Collateral Agent so long as all sums owing to the Collateral Agent have been paid pursuant to Section 2.02. After the retiring Collateral Agent's resignation or removal hereunder as Collateral Agent, each reference herein to a place for giving of notice or deliveries to the Collateral Agent shall be deemed to refer to the principal office of the successor Collateral Agent or such other office of the successor Collateral Agent as it may specify to each party hereto.

Section 6. INDEMNIFICATION.

6.01 INDEMNIFICATION.

- (a) The Company agrees to pay, indemnify and hold the Collateral Agent and each director, officer, employee, agent, bailee or other person acting on behalf of the Collateral Agent, and each stockholder of any thereof, harmless from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses (including, without limitation, the reasonable fees and disbursements of counsel and other advisers) or disbursements of any kind or nature whatsoever with respect to the execution, delivery, enforcement, performance and administration of this Agreement, including, without limitation, any amendment hereto, or to the Security Agreements, or in connection with the transactions contemplated by this Agreement, the Security Agreements and the Note Purchase Agreement (including arising from the ordinary negligence of the person seeking indemnification), unless arising from the gross negligence or willful misconduct of the person seeking indemnification.
- (b) In any suit, proceeding or action brought by the Collateral Agent under or with respect to the Collateral for any sum owing thereunder, or to enforce any provisions hereof or of the Security Agreements, the Company shall save, indemnify and keep the Collateral Agent (including its successors, assigns, directors, officers, employees, agents, bailees or any other person acting on behalf of the Collateral Agent) and each stockholder of any thereof and the Holders harmless from and against all expense, loss or damage suffered by reason of any defense, set-off, counterclaim, recoupment or reduction of liability whatsoever of the obligee thereunder, arising out of a breach by the Company of any obligation hereunder or thereunder or arising out of any other agreement, indebtedness or liability at any time owing to or in favor of such obligee or its successors from the Company, and all such obligations of the Company shall be and remain enforceable against and only against such Person and shall not be enforceable against the Collateral Agent or the Holders.
- (c) The obligations of the Company under this SECTION 6.01 shall survive the termination or modification of the other provisions of this Agreement and shall survive the commencement of a case under any applicable bankruptcy law on behalf of or against the Company or any other proceeding for the reorganization, management, adjustment of debt, dissolution or liquidation on behalf of or against any such Person and shall survive any dissolution of any such Person and shall survive the resignation or removal of the Collateral Agent.

Section 7. MISCELLANEOUS.

7.01 NOTICES. All notices and other communications provided for in this Agreement shall be given or made by facsimile transmission, first class mail, overnight delivery or personal delivery to the intended recipient to the address specified below each party's name on the signature pages hereto, or at such other address, or to the attention of such other officer, as any such party shall have furnished in writing to each other party. Except as otherwise provided in this Agreement, all such communications shall be deemed to have been duly given when transmitted by facsimile transmission, subject to telephone confirmation of receipt and the provision immediately thereafter of a copy by first class mail, overnight delivery or personal delivery, or, in the case of a

mailed notice, when duly deposited in the U.S. mails, first class postage prepaid, in each case given or addressed as aforesaid or upon actual receipt by the Collateral Agent.

- 7.02 NO PARTNERSHIP OR JOINT VENTURE. Nothing contained in this Agreement, and no action taken by the Collateral Agent or the Holders (or any of them) pursuant hereto, is intended to constitute or shall be deemed to constitute a partnership, association or joint venture.
- 7.03 COUNTERPARTS. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Delivery of an executed counterpart of a signature page of this Agreement by facsimile transmission shall be effective as delivery of a manually executed counterpart of this Agreement.
- $7.04\ \text{HEADINGS}.$ The section and other headings contained in this Agreement are for reference purposes only and shall not affect the meaning or interpretation of this Agreement.
- 7.05 PAYMENTS. All payments hereunder shall be made in U.S. dollars in immediately available funds.
- 7.06 REMEDIES CUMULATIVE, ETC. Each right, power and remedy provided in this Agreement for the benefit of the Collateral Agent now or hereafter existing at law or in equity or by statute or otherwise shall be cumulative and concurrent and shall be in addition to every other right, power or remedy provided for in this Agreement or now or hereafter existing at law or in equity or by statute or otherwise. The exercise or partial exercise by the Collateral Agent of any one or more of such rights, powers or remedies shall not preclude the simultaneous or later exercise by the Collateral Agent of all such other rights, powers or remedies, and no failure or delay on the part of the Collateral Agent to exercise any such right, power or remedy shall operate as a waiver thereof.
- 7.07 THE COMPANY'S SECURED OBLIGATIONS ABSOLUTE, ETC. The obligations of the Company under this Agreement shall be absolute and unconditional and shall remain in full force and effect without regard to, and shall not be released, discharged or in any way impaired by any circumstance whatsoever, including without limitation: (a) any amendment or modification of the Notes, the Note Purchase Agreement, the Security Agreements or any instrument provided for herein or therein, or any assignment, transfer or other disposition of any thereof; (b) any waiver, consent, extension, indulgence or other action or inaction under or in respect of any such instrument or any exercise or non-exercise of any right, remedy, power or privilege under or in respect of any such instrument or this Agreement; (c) any bankruptcy, insolvency, reorganization, arrangement, readjustment, composition, liquidation or similar proceeding with respect to the Company or any of its properties or creditors; or (d) any limitation on the liability or obligations of the Company under any such instrument or any invalidity or unenforceability, in whole or in part, of any such instrument or any term thereof regardless of whether the Company shall have notice or knowledge of any of the foregoing.
- 7.08 TERMINATION. This Agreement shall terminate upon the receipt by the Collateral Agent of written notice from the Majority Holders of the indefeasible payment (or prepayment) in full of the principal or accreted

value of, and the premium, if any, and interest on, all the Notes, in accordance with the terms of the Note Purchase Agreement and the Notes, and the indefeasible payment of all other amounts then owing to the Collateral Agent hereunder, under the Note Purchase Agreement, the Notes and the Security Agreements. At the time of such termination, the Collateral Agent, at the request and expense of the Company, will execute and deliver to the Company a proper instrument or instruments provided to it acknowledging the satisfaction and termination of this Agreement, and will duly assign, transfer and deliver to the Company such of the Collateral as has not yet theretofore been sold or otherwise applied or released pursuant to this Agreement, together with any moneys at the time held by the Collateral Agent or the Holders hereunder in connection with the Collateral.

7.09 FURTHER ASSURANCES, ETC. The Company, at its expense, shall duly execute, acknowledge and deliver all such instruments and take all such action as the Collateral Agent may request in order further to effectuate the purposes of this Agreement and to carry out the terms hereof.

7.10 CHOICE OF LAW. This Agreement shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the law of the State of New York, excluding choice-of-law principles of the law of such State that would require the application of the laws of a jurisdiction other than such State.

7.11 AMENDMENTS AND WAIVERS. Any terms of this Agreement may be amended and the observance of any term hereof may be waived (either generally or in a particular instance and either retroactively or prospectively) only with the written consent of each Company Party (at any time prior to and during the continuance of an Event of Default), the Collateral Agent and the Majority Holders. Any amendment or waiver effected in accordance with this SECTION 7.11 shall be binding upon the Company Parties and the Collateral Agent.

7.12 ENTIRE AGREEMENT. This Agreement, together with the other Security Agreements, embodies the entire agreement and understanding among the parties hereto as to the subject matter hereof and supersedes all prior agreements and understandings relating to the subject matter hereof.

7.13 SUCCESSORS AND ASSIGNS. This Agreement shall be binding upon the successors and/or assigns of each party hereto.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement by their duly authorized officers as of the day and year first above

BGLS INC.

By: /s/ RICHARD J. LAMPEN

Name:Richard J. Lampen

Title: Executive Vice President

ADDRESS FOR NOTICES:

100 S.E. Second Street Miami, Florida 33131 Telephone: (305) 579-8000 Facsimile: (305) 579-8009

BROOKE GROUP HOLDING INC.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen

Title: Executive Vice President

ADDRESS FOR NOTICES:

100 S.E. Second Street Miami, Florida 33131

Telephone: (305) 579-8000 Facsimile: (305) 579-8009

VECTOR GROUP LTD.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen

Title: Executive Vice President

ADDRESS FOR NOTICES:

100 S.E. Second Street Miami, Florida 33131 Telephone: (305) 579-8000 Facsimile: (305) 579-8009

NEW VALLEY HOLDINGS, INC.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen

Title: Executive Vice President

ADDRESS FOR NOTICES:

100 S.E. Second Street Miami, Florida 33131 Telephone: (305) 579-8000 Facsimile: (305) 579-8009

UNITED STATES TRUST COMPANY OF NEW YORK, as Collateral Agent

By: /s/ PATRICIA GALLAGHER

Name: Patricia Gallagher

Title: Assistant Vice President

ADDRESS FOR NOTICES:

114 West 47th Street, 25th Floor New York, NY 10036-1532 Telephone: (212) 852-1664 Facsimile: (212) 852-1626

Attention: Patricia Gallagher

TCW LEVERAGED INCOME TRUST, L.P.,
By: TCW Advisers (Bermuda), Ltd.,
as its General Partner

By: /s/ DARRYL L. SCHALL

/s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: TCW_Investment Management Company,

as Investment Adviser

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title: Group Managing Director

ADDRESS FOR NOTICES:

Trust Company of the West 11100 Santa Monica Boulevard, Suite 2000

Los Angeles, CA 90025 Telephone: (310) 235-5956 Facsimile: (310)235-5966

Attention: Alena Tabora

WITH A COPY TO:

Milbank, Tweed, Hadley & McCloy LLP 601 South Figueroa Street, 30th Floor

Los Angeles, California 90017 Telephone: (213) 892-4333 Facsimile: (213) 629-5063

TCW LEVERAGED INCOME TRUST II, L.P.
By: TCW (LINC II), L.P.
as its General Partner

By: TCW Advisers (Bermuda), Ltd., its General Partner

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: TCW Investment Management Company,

as Investment Adviser

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title: Group Managing Director

ADDRESS FOR NOTICES:

Trust Company of the West
11100 Santa Monica Boulevard, Suite 2000

Los Angeles, CA 90025 Telephone: (310) 235-5956 Facsimile: (310)235-5966

Attention: Alena Tabora

WITH A COPY TO:

Milbank, Tweed, Hadley & McCloy LLP 601 South Figueroa Street, 30th Floor

Los Angeles, California 90017 Telephone: (213) 892-4333 Facsimile: (213) 629-5063

TCW LINC III CBO LTD.

By: TCW Investment Management Company, as Collateral Manager

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title: Group Managing Director

ADDRESS FOR NOTICES:

Trust Company of the West 11100 Santa Monica Boulevard, Suite 2000 Los Angeles, CA 90025

Telephone: (310) 235-5956 Facsimile: (310)235-5966

Attention: Alena Tabora

WITH A COPY TO:

Milbank, Tweed, Hadley & McCloy LLP 601 South Figueroa Street, 30th Floor

Los Angeles, California 90017 Telephone: (213) 892-4333 Facsimile: (213) 629-5063

POWRs 1997-2 (Participating Obligations with Residuals 1997-2)

By: Citibank Global Asset Management, Its Investment Advisor

100 1110000110110 71001001

By: TCW Asset Management Company, Its Portfolio Manager

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title: Group Managing Director

ADDRESS FOR NOTICES:

Trust Company of the West 11100 Santa Monica Boulevard, Suite 2000

Los Angeles, CA 90025 Telephone: (310) 235-5956 Facsimile: (310) 235-5966

Attention: Alena Tabora

WITH A COPY TO:

Milbank, Tweed, Hadley & McCloy LLP 601 South Figueroa Street, 30th Floor

Los Angeles, California 90017 Telephone: (213) 892-4333 Facsimile: (213) 629-5063

CAPTIVA II FINANCE LTD.,

By: TCW Advisors, Inc.

Its Financial Manager

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title: Group Managing Director

ADDRESS FOR NOTICES:

Trust Company of the West 11100 Santa Monica Boulevard, Suite 2000

Los Angeles, CA 90025 Telephone: (310) 235-5956 Facsimile: (310) 235-5966

Attention: Alena Tabora

WITH A COPY TO:

Milbank, Tweed, Hadley & McCloy LLP 601 South Figueroa Street, 30th Floor Los Angeles, California 90017

Los Angeles, California 90017 Telephone: (213) 892-4333 Facsimile: (213) 629-5063

AIMCO CDO, SERIES 2000-A

By: Allstate Investment Management Company,

Its Collateral Manager

By: TCW Asset Management Company,

Its Investment Advisor

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title: Group Managing Director

ADDRESS FOR NOTICES:

Trust Company of the West 11100 Santa Monica Boulevard, Suite 2000

Los Angeles, CA 90025 Telephone: (310) 235-5956 Facsimile: (310)235-5966

Attention: Alena Tabora

WITH A COPY TO:

Milbank, Tweed, Hadley & McCloy LLP 601 South Figueroa Street, 30th Floor

Los Angeles, California 90017 Telephone: (213) 892-4333 Facsimile: (213) 629-5063

PLEDGE AND SECURITY AGREEMENT

PLEDGE AND SECURITY AGREEMENT, dated as of May 14, 2001 (as amended, modified or supplemented from time to time, this "AGREEMENT") between BGLS INC., a corporation duly organized and validly existing under the laws of the State of Delaware (the "COMPANY") and United States Trust Company of New York, a New York banking corporation, as collateral agent (together with its successors and assigns, the "COLLATERAL AGENT") on behalf of the holders (the "HOLDERS") of the 10% Senior Secured Notes due March 31, 2006 (the "NOTES") issued pursuant to that certain Note Purchase Agreement, dated as of May 14, 2001 (as amended, supplemented or modified from time to time, the "NOTE PURCHASE AGREEMENT") between the Company and the Holders.

Section 1. DEFINITIONS. Terms defined in the Note Purchase Agreement are used herein as defined therein. In addition, as used herein:

"ACCOUNT CONTROL AGREEMENT" means the Account Control Agreement, dated as of the date hereof, substantially in the form of Exhibit B-2 to the Note Purchase Agreement among the Company, the Collateral Agent and Bank of America, N.A., as amended, modified and supplemented from time to time.

"ACCOUNTS" has the meaning ascribed thereto in Section 3(d) hereof.

 ${\tt "BROOKE\ HOLDING"}$ means Brooke Group Holding, Inc. a Delaware corporation and any successor thereto.

"BROOKE HOLDING CONVERTIBLE SECURITIES" means any securities or rights that are convertible into or exchangeable for Equity Interests of Brooke Holding.

"BROOKE OVERSEAS" means Brooke (Overseas) Ltd., a Delaware corporation, and any successor thereto.

"BROOKE OVERSEAS CONVERTIBLE SECURITIES" means any securities or rights that are convertible into or exchangeable for Equity Interests of Brooke Overseas

"COLLATERAL" has the meaning ascribed thereto in Section 3 hereof.

"COMPANY ACCOUNTS" means, collectively, the BGLS Checking Account and the Securities Account.

"COMMERCIAL TORT CLAIM" means (i) prior to July 1, 2001, a claim of the Company arising in tort and (ii) on or after July 1, 2001, has the meaning ascribed thereto in the Uniform Commercial Code.

"COPYRIGHT COLLATERAL" means all Copyrights, whether now owned or hereafter acquired by the Company.

"COPYRIGHTS" means all copyrights, copyright registrations and applications for copyright registrations, including, without limitation, all renewals, extensions, income, royalties, damages and payments now or hereafter due and/or payable under and with respect thereto, including, without limitation, damages and payments for past or future infringements thereof, the right to sue for past, present and future infringements thereof, and all rights corresponding thereto throughout the world.

"DEPOSIT ACCOUNT" means any deposit account (as defined in the Uniform Commercial Code).

"DOCUMENTS" has the meaning ascribed thereto in Section 3(k) hereof.

"ENVIRONMENTAL LAWS" means any and all present and future Federal, state, local and foreign laws, rules or regulations, and any orders or decrees, in each case as now or hereafter in effect, relating to the regulation or protection of human health, safety or the environment or to emissions, discharges, releases or threatened releases of pollutants, contaminants, chemicals or toxic or hazardous substances or wastes into the indoor or outdoor environment, including, without limitation, ambient air, soil, surface water, ground water, wetlands, land or subsurface strata, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of pollutants, contaminants, chemicals or toxic or hazardous substances or wastes.

"EQUIPMENT" has the meaning ascribed thereto in Section 3(i) hereof.

"EQUITY INTERESTS" means, with respect to any Person, any capital stock of such Person and shares, interests, participations or other ownership interests (however designated) of any Person and any rights (other than debt securities convertible into corporate stock), warrants and options to purchase any of the foregoing, including (without limitation) each class of common stock and preferred stock of such Person if such Person is a corporation and each general and limited partnership interest of such Person if such Person is a partnership.

"FARM PRODUCTS" has the meaning ascribed thereto in the Uniform Commercial Code. $\begin{tabular}{ll} \hline \end{tabular}$

"INITIAL PLEDGED BROOKE HOLDING SHARES" means all Equity Interests of Brooke Holding represented by the certificates of Brooke Holding identified in Annex 1 hereto.

"INITIAL PLEDGED BROOKE OVERSEAS SHARES" means all Equity Interests of Brooke Overseas represented by the certificates identified in Annex 1 hereto.

"INITIAL PLEDGED NEW VALLEY SHARES" means all Equity Interests of New Valley represented by the certificates identified in Annex 1 hereto.

"INITIAL PLEDGED NV HOLDINGS SHARES" means all Equity Interests of NV Holdings represented by the certificates identified in Annex 1 hereto.

"INITIAL PLEDGED RESEARCH SHARES" means all Equity Interests of Research represented by the certificates identified in Annex 1 hereto.

"INITIAL PLEDGED VTUSA SHARES" means all Equity Interests of VTUSA represented by the certificates identified in Annex 1 hereto.

"INSTRUMENTS" has the meaning ascribed thereto in Section 3(e) hereof.

"INTELLECTUAL PROPERTY" means, collectively, all Copyright Collateral, all Patent Collateral and all Trademark Collateral.

"INVENTORY" has the meaning ascribed thereto in Section 3(g) hereof.

"ISSUERS" means, collectively, the respective corporations identified in Annex 1 hereto under the caption "Issuer".

"MOTOR VEHICLES" means motor vehicles, tractors, trailers and other like property whether or not the title thereto is governed by a certificate of title or ownership.

"NEW VALLEY" means New Valley Corporation, a Delaware corporation, and any successor thereto. $\,$

"NEW VALLEY COMMON STOCK" means common shares of New Valley, par value \$.01 per share.

"NEW VALLEY CONVERTIBLE SECURITIES" means any securities that are convertible into or exchangeable for Equity Interests of New Valley.

"NOTES" has the meaning ascribed thereto in the first paragraph hereof.

"NV HOLDINGS" means New Valley Holdings, Inc., a Delaware corporation, and any successor thereto. $\,$

"NV HOLDINGS CONVERTIBLE SECURITIES" means any securities or rights that are convertible into or exchangeable for Equity Interests of NV Holdings.

"PATENT COLLATERAL" means all Patents, whether now owned or hereafter acquired by the Company.

"PATENTS" means all patents and patent applications, including, without limitation, the inventions and improvements described and claimed therein together with the reissues, divisions, continuations, renewals, extensions and continuations-in-part thereof, all income, royalties, damages and payments now or hereafter due and/or payable under and with respect thereto, including, without limitation, damages and payments for past or future infringements thereof, the right to sue for past, present and future infringements thereof, and all rights corresponding thereto throughout the world.

"PERMITTED SECURITIES" has the meaning ascribed thereto in the Account Control Agreement.

"PLEDGED STOCK" has the meaning ascribed thereto in Section 3(a) hereof.

"RESEARCH" means Vector Research Ltd., a Delaware corporation, and any successor thereto.

"RESEARCH CONVERTIBLE SECURITIES" means any securities or rights that are convertible into or exchangeable for Equity Interests of Research.

"SECURED OBLIGATIONS" means, collectively, all obligations and liabilities of any kind or nature, present or future, absolute or contingent, of (i) the Company arising under the Note Documents, (ii) NV Holdings arising under the NV Holdings Pledge Agreement or any other undertaking or agreement delivered by NV Holdings in connection with any other Note Document, (iii) VTUSA arising under any undertaking or agreement delivered by VTUSA in connection with any Note Document, (iv) Brooke Overseas arising under any undertaking or agreement delivered by Brooke Overseas in connection with any Note Document, (v) New Valley arising under any undertaking or agreement delivered by New Valley in connection with any Note Document, (vi) Brooke Holding arising under the Brooke Holding Pledge Agreement or any other undertaking or agreement delivered by Brooke Holding, (vii) Research arising under any undertaking, or agreement delivered by Research in connection with any Note Document, (viii) Vector arising under the Vector Pledge Agreement or any other undertaking or agreement delivered by Vector in connection with any other Note Document or (ix) Liggett arising under any undertaking or agreement delivered by Liggett in connection with any Note Document.

"STOCK COLLATERAL" means, collectively, the Collateral described in clauses (a) through (c) of Section 3 hereof and the proceeds of and to any such property and, to the extent related to any such property or such proceeds, all books, correspondence, credit files, records, invoices and other papers.

"TRADEMARK COLLATERAL" means all Trademarks, whether now owned or hereafter acquired by the Company. Notwithstanding the foregoing, the Trademark Collateral does not and shall not include any Trademark that would be rendered invalid, abandoned, void or unenforceable by reason of its being included as part of the Trademark Collateral.

"TRADEMARKS" means all trade names, trademarks and service marks, logos, trademark and service mark registrations, and applications for trademark and service mark registrations, including, without limitation, all renewals of trademark and service mark registrations, all rights corresponding thereto throughout the world, the right to recover for all past, present and future infringements thereof, all other rights of any kind whatsoever accruing thereunder or pertaining thereto, together, in each case, with the product lines and goodwill of the business connected with the use of, and symbolized by, each such trade name, trademark and service mark.

"UNIFORM COMMERCIAL CODE" means the Uniform Commercial Code as in effect from time to time in the State of New York.

"VECTOR" means Vector Group Ltd., a Delaware corporation, and any successor thereto.

"VTUSA" means Vector Tobacco (USA) Ltd., a Delaware corporation, and any successor thereto.

"VTUSA CONVERTIBLE SECURITIES" means any securities that are convertible into or exchangeable for Equity Interests of VTUSA.

Section 2. REPRESENTATIONS AND WARRANTIES. The Company represents and warrants and covenants to the Collateral Agent for the benefit of the Holders and the Collateral Agent that:

- (a) except for the Security Interest, the Company is the sole legal and equitable owner of the Collateral, holds the same free and clear of all Liens, charges, encumbrances and security interests of any kind and nature and will make no other assignment, pledge, mortgage, hypothecation, transfer or other disposition of the Collateral, other than the sale or application of the Collateral in compliance with the provisions of Sections 8.24 and 13.8 of the Note Purchase Agreement;
- (b) the Company has good, right and legal title to the Collateral and will defend its title thereto against the claims of all Persons whomsoever and will maintain and preserve the Security Interest as long as this Agreement shall remain in full force and effect;
- (c) no financing statement or other public notice with respect to all or any part of the Collateral is on file or of record in any public office, except as have been filed by the Collateral Agent pursuant to this Agreement or as are permitted by the Note Purchase Agreement;
- (d) except for the filing of UCC-1 financing statements, no consent or approval of, or other action by, and no notice to or filing with, any Governmental or Regulatory Authority or securities exchange, was or is necessary as a condition (i) to the validity of the pledge provided for herein or for the execution, delivery or performance of this Agreement by the Company or (ii) other than filings with (A) the United States Patent and Trademark Office in connection with the assignment of the Patents, Trademarks and Copyrights or (B) the Securities Act, and regulations thereunder and any applicable blue sky laws, for the exercise by the Collateral Agent of the voting or other rights provided for in this Agreement or the remedies in respect of the Collateral pursuant to this Agreement;
- (e) as of the date hereof: (i) the Initial Pledged NV Holdings Shares are the only issued and outstanding Equity Interests of NV Holdings and (ii) there are no issued and outstanding NV Holdings Convertible Securities, and NV Holdings is not subject to any

obligation, contingent or otherwise, to issue in the future any Equity Interests or any such NV Holdings Convertible Securities;

- (f) the Initial Pledged NV Holdings Shares are duly authorized, validly issued, fully paid and nonassessable;
- (g) as of the date hereof: (i) the Initial Pledged VTUSA Shares are the only issued and outstanding Equity Interests of VTUSA and (ii) there are no issued and outstanding VTUSA Convertible Securities, and VTUSA is not subject to any obligation, contingent or otherwise, to issue in the future any additional Equity Interests or any such VTUSA Convertible Securities;
- (h) the Initial Pledged VTUSA Shares are duly authorized, validly issued, fully paid and nonassessable;
- (i) as of the date hereof: (i) the Initial Pledged Brooke Overseas Shares are the only issued and outstanding Equity Interests of Brooke Overseas and (ii) there are no issued and outstanding Brooke Overseas Convertible Securities, and Brooke Overseas is not subject to any obligation, contingent or otherwise, to issue in the future any additional Equity Interests or any such Brooke Overseas Convertible Securities;
- (j) the Initial Pledged Brooke Overseas Shares are duly authorized, validly issued, fully paid and nonassessable;
- (k) as of the date hereof: (i) the Initial Pledged Brooke Holding Shares are the only issued and outstanding shares of capital stock of Brooke Holding and (ii) there are no issued and outstanding Brooke Holding Convertible Securities, and Brooke Holding is not subject to any obligation, contingent or otherwise, to issue in the future any additional Equity Interests or any such Brooke Holding Convertible Securities;
- (1) the Initial Pledged Brooke Holding Shares are duly authorized, validly issued, fully paid and nonassessable;
- (m) as of the date hereof: (i) the Initial Pledged Research Shares are the only issued and outstanding shares of Research and (ii) there are no issued and outstanding Research Convertible Securities, and Research is not subject to any obligation, contingent or otherwise, to issue in the future any Equity Interests or any such Research Convertible Securities;
- (n) the Initial Pledged Research Shares are duly authorized, validly issued, fully paid and nonassessable;
- (o) as of the date hereof: (i) there are 22,813,063 shares of New Valley Common Stock issued and outstanding and such shares are the only issued and outstanding Equity Interests of New Valley and (ii) there are no issued and outstanding New Valley Convertible Securities, and New Valley is not subject to any obligation, contingent or

otherwise, to issue in the future any additional Equity Interests except:

- (1) warrants exercisable for 17,867,566 shares of New Valley Common Stock; and
- (2) options to purchase 1,271,088 shares of New Valley Common Stock and options to purchase warrants exercisable for 584,000 shares of New Valley Common Stock.
- (p) the Initial Pledged New Valley Shares are duly authorized, validly issued, fully paid and nonassessable;
- (q) the Company has full corporate power and authority to execute, deliver and perform this Agreement;
- (r) except for the Note Documents, the Company is not now and will not become a party to any voting trust or other agreement or undertaking with respect to the exercise of the voting or consent rights associated with any of the Stock Collateral (other than an agreement or undertaking in respect of a sale of the Stock Collateral, which sale is or will be in compliance with the provisions of Section 13.8 of the Note Purchase Agreement), and there are no other restrictions on the exercise of such rights;
- (s) except for the Note Documents, the Company is not now and will not become a party to any agreement or undertaking pursuant to which any Issuer or any other Person could have the right to purchase any of the Stock Collateral (other than an agreement or undertaking in respect of a sale of the Stock Collateral, which sale is or will be in compliance with the provisions of Section 13.8 of the Note Purchase Agreement), and at the time of delivery of any Stock Collateral to the Collateral Agent there will not be (and the Company will not thereafter permit to exist) any other restrictions on the transfer of the Stock Collateral;
- (t) except for the Note Documents, the Company has the right to vote, pledge and grant a security interest in, or otherwise transfer, the Collateral free of any liens (except as set forth in (a) above);
- (u) the security interests granted pursuant to this Agreement (i) upon completion of the filings and other actions specified on Annex 2 (which, in the case of all filings and other documents referred to on said Annex, have been delivered to the Collateral Agent in duly executed form) will constitute valid perfected security interests in all of the Collateral in favor of the Collateral Agent enforceable in accordance with the terms hereof against all creditors of the Company

and all Persons purporting to purchase any Collateral from the Company and (ii) are prior to all other Liens on the Collateral in existence on the date hereof except for Liens permitted by the Note Purchase Agreement which have priority over the Liens on the Collateral held by the Collateral Agent by operation of law;

- (v) the grant and perfection of the Security Interest in the Stock Collateral for the benefit of the Collateral Agent and the Holders, in accordance with the terms hereof, will not violate the registration requirements of the Securities Act, any provisions of any other applicable federal, state or foreign securities laws, or any applicable general corporation law or other applicable law;
- (w) each Issuer is a corporation duly organized, validly existing and in good standing under the laws of the state of its incorporation and has full power and authority to own or lease its properties and assets and to carry on its business;
- (x) the stock powers delivered to the Collateral Agent with respect to the Pledged Stock have been, and at the time of the delivery of any other stock powers to the Collateral Agent in connection with the pledge of any additional Stock Collateral such stock powers will be, duly executed;
- (y) the pledge of the Initial Pledged Shares pursuant to this Agreement, together with the transfer of such Initial Pledged Shares to the Collateral Agent, creates a valid and perfected first priority security interest in the Stock Collateral, in favor of the Collateral Agent for the benefit of the Collateral Agent and the Holders, securing the due and punctual payment and performance in full of the Secured Obligations in accordance with the respective terms thereof;
- (z) the Company is incorporated in the State of Delaware and the location of its chief executive office or sole place of business on the date hereof is specified on Annex 3;
- (aa) none of the Collateral consists of, or is the proceeds (as defined in the Uniform Commercial Code) of, Farm Products;
- (bb) on the date hereof, the Inventory and the Equipment (other than Motor Vehicles) are kept at the locations listed on Annex 3;
- (cc) Annex 4 lists all Intellectual Property owned by the Company in its own name on the date hereof;
 - (i) On the date hereof, all material Intellectual Property is valid, subsisting, unexpired and enforceable, has not been abandoned and does not infringe the intellectual property rights of any other Person;

- (ii) Except as set forth in Annex 4, on the date hereof, none of the Intellectual Property is the subject of any licensing or franchise agreement pursuant to which the Company is the licensor or franchisor;
- (iii) No holding, decision or judgment has been rendered by any Governmental or Regulatory Authority which would limit, cancel or question the validity of, or the Company's rights in, any Intellectual Property in any respect; and
- (iv) No action or proceeding is pending or, to the knowledge of the Company threatened, or the date hereof seeking to limit, cancel or question the validity of any Intellectual Property or the Company's ownership interest therein;
- (dd) subject to Section 5.01(b), all Instruments owned by the Company are listed on Annex 5 and have been delivered to the Collateral Agent;
- (ee) each Instrument which is a promissory note issued to and held by the Company (including any promissory note evidencing loans made by the Company to any of its Affiliates or Associates, any Vector Expanded Affiliate or any Group Executive) constitutes the legal, valid and binding obligation of the obligor with respect thereto, enforceable in accordance with its terms, subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing;
- (ff) any instruments of assignment and transfer delivered pursuant to Section 5.01(b) have been, or will be, duly executed;
- $\mbox{(gg)}$ all Deposit Accounts of the Company are listed on Annex 6;
 - (hh) the Company has no Commercial Tort Claims;
- (ii) the Company shall not permit NV Holdings to sell, assign, transfer or otherwise dispose of, or pledge, hypothecate or otherwise encumber, any of its Equity Interests in New Valley which are pledged pursuant to the NV Holdings Pledge Agreement, unless any such sale, assignment, transfer or disposition is in compliance with the provisions of Section 13.8 of the Note Purchase Agreement; and
- (jj) the Company shall not permit Brooke Holding to sell, assign, transfer or otherwise dispose of, or pledge, hypothecate or

otherwise encumber, any of its Equity Interests in Liggett which are pledged pursuant to the Brooke Holding Pledge Agreement, unless any such sale, assignment, transfer or disposition is in compliance with provisions of Section 13.8 of the Note Purchase Agreement.

Section 3. COLLATERAL. As collateral security for the prompt payment in full when due (whether at stated maturity, by acceleration or otherwise) and performance of the Secured Obligations, the Company hereby pledges and grants to the Collateral Agent for the benefit of the Holders and the Collateral Agent a security interest in all of the Company's right, title and interest in the following property, whether now owned by the Company or hereafter acquired and whether now existing or hereafter coming into existence (all being collectively referred to herein as "COLLATERAL"):

- (a) the shares of capital stock and other Equity Interests of the Issuers represented by the certificates identified in Annex 1 hereto and all Equity Interests of any other Person, now or hereafter owned by the Company, in each case together with the certificates evidencing the same (collectively, the "PLEDGED STOCK");
- (b) all shares, securities, moneys or property representing a dividend on any of the Pledged Stock, or representing a distribution or return of capital upon or in respect of the Pledged Stock, or resulting from a split-up, revision, reclassification or other like change of the Pledged Stock or otherwise received in exchange therefor, and any subscription warrants, rights or options issued to the holders of, or otherwise in respect of, the Pledged Stock;
- (c) without affecting the obligations of the Company under any provision prohibiting such action hereunder or under the Note Purchase Agreement, in the event of any consolidation or merger in which an Issuer or other Subsidiary of the Company is not the surviving corporation, all shares of each class of the capital stock and all other Equity Interests of the successor corporation (unless such successor corporation is the Company itself) formed by or resulting from such consolidation or merger (the Pledged Stock, together with all other certificates, shares, securities, properties or moneys as may from time to time be pledged hereunder pursuant to clause (a) or (b) above and this clause (c) being herein collectively called the "STOCK COLLATERAL");
- (d) all accounts and general intangibles (each as defined in the Uniform Commercial Code) of the Company constituting any right to the payment of money, including (but not limited to) all moneys due and to become due to the Company in respect of any loans or advances or for Inventory or Equipment or other goods sold or leased or for services rendered, all moneys due and to become due to the Company under any guarantee (including a letter of credit) of the purchase price of

Inventory or Equipment sold by the Company and all tax refunds (such accounts, general intangibles and moneys due and to become due being herein called collectively "ACCOUNTS");

- (e) all instruments, chattel paper or letters of credit (each as defined in the Uniform Commercial Code) of the Company evidencing, representing, arising from or existing in respect of, relating to, securing or otherwise supporting the payment of, any of the Accounts, including (but not limited to) promissory notes, drafts, bills of exchange and trade acceptances (herein collectively called "INSTRUMENTS");
 - (f) all Deposit Accounts of the Company;
- (g) all inventory (as defined in the Uniform Commercial Code) of the Company, all goods obtained by the Company in exchange for such inventory, and any products made or processed from such inventory including all substances, if any, commingled therewith or added thereto (herein collectively called "INVENTORY");
- (h) all Intellectual Property and all other accounts or general intangibles not constituting Intellectual Property or Accounts;
- (i) all equipment (as defined in the Uniform Commercial Code) of the Company, including all Motor Vehicles (herein collectively called "EQUIPMENT");
- (j) each contract and other agreement of the Company relating to the sale or other disposition of Inventory or Equipment;
- (k) all documents of title (as defined in the Uniform Commercial Code) or other receipts of the Company covering, evidencing or representing Inventory or Equipment (herein collectively called "DOCUMENTS"):
- (1) all rights, claims and benefits of the Company against any Person arising out of, relating to or in connection with Inventory or Equipment purchased by the Company, including, without limitation, any such rights, claims or benefits against any Person storing or transporting such Inventory or Equipment;
 - (m) all Commercial Tort Claims of the Company;
- (n) all securities accounts (as defined in Article 8 of the Uniform Commercial Code) of the Company and the financial assets (as defined in Article 8 of the Uniform Commercial Code) credited from time to time thereto; and
- (o) all other tangible and intangible personal property and fixtures of the Company, including, without limitation, all proceeds, products, offspring, accessions, rents, profits, income, benefits, substitutions and replacements of and to any of the property of the

Company described in the preceding clauses of this Section 3 (including, without limitation, any proceeds of insurance thereon and all causes of action, claims and warranties now or hereafter held by the Company in respect of any of the items listed above) and, to the extent related to any property described in said clauses or such proceeds, products and accessions, all books, correspondence, credit files, records, invoices and other papers, including without limitation all tapes, cards, computer runs and other papers and documents in the possession or under the control of the Company or any computer bureau or service company from time to time acting for the Company.

Section 4. COMPANY ACCOUNTS.

- 4.01. COMPANY ACCOUNTS. The Company has established the Company Accounts into which there shall be deposited from time to time the Cash and cash equivalents required to be deposited therein pursuant to Sections 8.21 and 13.8 of the Note Purchase Agreement, the proceeds of any of the Collateral required to be delivered to the Collateral Agent pursuant hereto and all Cash and cash equivalents at any time held by the Company. The Securities Account is under the control of the Collateral Agent. The balance from time to time in the Company Accounts shall constitute part of the Collateral hereunder and shall not constitute payment of the Secured Obligations until applied as hereinafter provided. However, at any time following the occurrence and during the continuance of Account Notice Event, the Collateral Agent at the written direction of the Majority Holders apply or cause to be applied the financial assets held in or credited to the Securities Account to the payment of the Secured Obligations in the manner specified in Section 5.13 hereof. Withdrawals from the Securities Account may be made at the written instruction of the Majority Holders pursuant to Section 13.11 of the Note Purchase Agreement
- 4.02. RELEASE OF COLLATERAL. The security interest in any Collateral shall automatically and without any action by the Collateral Agent or the Holders be released upon any sale of such Collateral by the Company pursuant to the terms of Section 13.8 of the Note Purchase Agreement and this Agreement.
- 4.03. INVESTMENT OF BALANCE IN SECURITIES ACCOUNT. Amounts on deposit in the Securities Account shall be invested from time to time in such Permitted Securities as the Company (or, after the occurrence and during the continuance of an Account Notice Event, the Majority Holders) shall determine, PROVIDED that at any time after the occurrence and during the continuance of an Account Notice Event, the Collateral Agent may in its discretion at any time and from time to time elect to liquidate any such Permitted Securities and to apply or cause to be applied the proceeds thereof to the payment of the Secured Obligations in the manner specified in Section 5.13 hereof.

Section 5. FURTHER ASSURANCES; REMEDIES. In furtherance of the grant of the pledge and security interest pursuant to Section 3 hereof, the Company hereby agrees with the Collateral Agent as follows:

5.01. DELIVERY AND OTHER PERFECTION. The Company shall:

- (a) if any of the shares, securities, moneys or property required to be pledged by the Company under clauses (a), (b) and (c) of Section 3 hereof are received by the Company, hold such Property in trust and forthwith either (x) transfer and deliver to the Collateral Agent such shares or securities so received by the Company (together with the certificates for any such shares and securities duly endorsed in blank or accompanied by undated stock powers duly executed in blank, all in form and substance reasonably satisfactory to the Collateral Agent), all of which thereafter shall be held by the Collateral Agent (except in the case of cash dividends which shall be held in a Company Account in accordance with the Note Purchase Agreement), pursuant to the terms of this Agreement, as part of the Collateral or (y) take such other action as may be necessary or appropriate or as the Collateral Agent may reasonably request to duly record the Lien created hereunder in such shares, securities, moneys or property in said clauses (a), (b) and (c);
- (b) deliver and pledge to the Collateral Agent any and all Instruments pledged hereunder, endorsed and/or accompanied by such instruments of assignment and transfer in such form and substance as may be required to perfect the pledge thereof hereunder or as the Collateral Agent may reasonably request; PROVIDED, that so long as no Default shall have occurred and be continuing, the Company may retain for collection in the ordinary course any Instruments received by the Company in the ordinary course of business and the Collateral Agent shall, promptly upon request of the Company, make appropriate arrangements for making any Instrument pledged by the Company hereunder available to the Company for purposes of presentation, collection or renewal (any such arrangement to be effected, to the extent deemed appropriate by the Collateral Agent, against trust receipt or like document);
- (c) give, execute, deliver, file and/or record any financing statement, continuation statement, notice, instrument, document, agreement or other papers that may be necessary or desirable or that the Collateral Agent may reasonably request to create, preserve, perfect or validate the security interest granted pursuant hereto or to enable the Collateral Agent to exercise and enforce its rights hereunder with respect to such pledge and security interest, including, without limitation, causing any or all of the Stock Collateral to be transferred of record into the name of the Collateral Agent or its nominee (and the Collateral Agent agrees that if any Stock Collateral is transferred into its name or the name of its nominee, the Collateral Agent will thereafter promptly give to the Company copies of any

notices and communications received by it with respect to the Stock Collateral), PROVIDED that notices to account debtors in respect of any Accounts or Instruments shall be subject to the provisions of clause (q) below;

- (d) upon the acquisition after the date hereof by the Company of any Equipment covered by a certificate of title or ownership, cause the Collateral Agent to be listed as the lienholder on such certificate of title and within 120 days of the acquisition thereof deliver evidence of the same to the Collateral Agent;
- (e) keep full and accurate books and records relating to the Collateral, and stamp or otherwise mark such books and records in such manner as may be necessary to reflect the security interests granted by this Agreement;
- (f) permit representatives of the Collateral Agent, upon reasonable notice, at any time during normal business hours to inspect and make abstracts from its books and records pertaining to the Collateral, and permit representatives of the Collateral Agent to be present at the Company's place of business to receive copies of all communications and remittances relating to the Collateral, and forward copies of any notices or communications received by the Company with respect to the Collateral, all to the extent and in such manner as the Collateral Agent may reasonably request; and
- (g) upon the occurrence and during the continuance of any Default, upon request of the Collateral Agent, promptly notify (and the Company hereby authorizes the Collateral Agent so to notify) each account debtor in respect of any Accounts or Instruments that such Collateral has been assigned to the Collateral Agent hereunder, and that any payments due or to become due in respect of such Collateral are to be made directly to the Securities Account.
- 5.02. OTHER FINANCING STATEMENTS AND LIENS. Without the prior written consent of the Collateral Agent, the Company shall not file or suffer to be on file, or authorize or permit to be filed or to be on file, in any jurisdiction, any financing statement or like instrument with respect to the Collateral in which the Collateral Agent is not named as the sole secured party.
- 5.03. PRESERVATION OF RIGHTS. The Collateral Agent shall not be required to take steps necessary to preserve any rights against third parties to any of the Collateral.
 - 5.04. SPECIAL PROVISIONS RELATING TO STOCK COLLATERAL.
 - (a) So long as no Event of Default shall have occurred and be continuing, the Company shall have the right to exercise all voting, consensual and other powers of ownership pertaining to the Stock Collateral for all purposes not inconsistent with the terms of this Agreement, the Note Purchase Agreement, the Notes or any other related instrument or agreement referred to herein or therein, PROVIDED that the Company agrees that it will not vote the Stock Collateral in any

manner that is inconsistent with the terms of any Note Document or any other instrument or agreement; and the Collateral Agent shall execute and deliver to the Company or cause to be executed and delivered to the Company all such proxies, powers of attorney, dividend and other orders, and all such instruments, without recourse, as the Company may reasonably request for the purpose of enabling the Company to exercise the rights and powers that it is entitled to exercise pursuant to this Section 5.04(a).

(b) If any Account Notice Event shall have occurred, then so long as such Account Notice Event shall continue, and whether or not the Collateral Agent exercises any available right to declare any Secured Obligation due and payable or seeks or pursues any other relief or remedy available to it under applicable law or under this Agreement, the Note Purchase Agreement, the Notes or any other agreement relating to such Secured Obligation, to the extent not already delivered pursuant to Section 5.01, all dividends and other distributions on the Stock Collateral shall be paid directly to the Collateral Agent and deposited by it into the Securities Account, subject to the terms of this Agreement, and, if the Collateral Agent shall so request in writing, the Company agrees to execute and deliver to the Collateral Agent appropriate additional dividend, distribution and other orders and documents to that end.

5.05. EVENTS OF DEFAULT, ETC. Subject to the last sentence of this Section 5.05, during the period during which (i) with respect to any cash contained in the BGLS Checking Account and all cash dividends and cash proceeds of any of the Stock Collateral, an Account Notice Event shall have occurred and be continuing and (ii) with respect to all other Collateral, an Event of Default shall have occurred and be continuing:

- (a) the Company shall, at the request of the Collateral Agent, assemble the Collateral owned by it at such place or places, reasonably convenient to both the Collateral Agent and the Company, designated in its request;
- (b) the Collateral Agent may make any reasonable compromise or settlement deemed desirable with respect to any of the Collateral and may extend the time of payment, arrange for payment in installments, or otherwise modify the terms of, any of the Collateral;
- (c) the Collateral Agent shall have all of the rights and remedies with respect to the Collateral of a secured party under the Uniform Commercial Code (whether or not the Uniform Commercial Code is in effect in the jurisdiction where the rights and remedies are asserted) and such additional rights and remedies to which a secured party is entitled under the laws in effect in any jurisdiction where any rights and remedies hereunder may be asserted, including, without

limitation, the right, to the maximum extent permitted by law, to exercise all voting, consensual and other powers of ownership pertaining to the Collateral as if the Collateral Agent were the sole and absolute owner thereof (and the Company agrees to take all such action as may be appropriate to give effect to such right);

- (d) the Collateral Agent in its discretion may, in its name or in the name of the Company or otherwise, demand, sue for, collect or receive any money or property at any time payable or receivable on account of or in exchange for any of the Collateral, but shall be under no obligation to do so; and
- (e) the Collateral Agent may, upon ten business days' prior written notice to the Company of the time and place, with respect to the Collateral or any part thereof that shall then be or shall thereafter come into the possession, custody or control of the Collateral Agent or any of its agents, sell, lease, assign or otherwise dispose of all or any part of such Collateral, at such place or places as the Collateral Agent deems best, and for cash or for credit or for future delivery (without thereby assuming any credit risk), at public or private sale, without demand of performance or notice of intention to effect any such disposition or of the time or place thereof (except such notice as is required above or by applicable statute and cannot be waived), and the Collateral Agent or anyone else may be the purchaser, lessee, assignee or recipient of any or all of the Collateral so disposed of at any public sale (or, to the extent permitted by law, at any private sale) and thereafter hold the same absolutely, free from any claim or right of whatsoever kind, including any right or equity of redemption (statutory or otherwise), of the Company, any such demand, notice and right or equity being hereby expressly waived and released. The Collateral Agent may, without notice or publication, adjourn any public or private sale or cause the same to be adjourned from time to time by announcement at the time and place fixed for the sale, and such sale may be made at any time or place to which the sale may be so adjourned.

The proceeds of each collection, sale or other disposition under this Section 5.05 shall be applied in accordance with Section 5.13 hereof.

5.06. REGISTRATION RIGHTS.

(a) If the Collateral Agent shall determine to exercise its right to sell any or all of the Stock Collateral, and if in the opinion of the Collateral Agent it is necessary or advisable to have the Stock Collateral, or that portion thereof to be sold, registered under the provisions of the Securities Act, the Company will cause the issuer thereof to (i) execute and deliver, and cause the directors and officers of such issuer to execute and deliver, all such instruments and documents, and do or cause to be done all such other acts as may be, in the opinion of the Collateral Agent, necessary or advisable to register the Stock Collateral, or that portion thereof to be sold,

under the provisions of the Securities Act, (ii) use its best efforts to cause the registration statement relating thereto to become effective and to remain effective for a period of one year from the date of the first public offering of the Stock Collateral, or that portion thereof to be sold and (iii) make all amendments thereto and/or to the related prospectus which, in the opinion of the Collateral Agent, are necessary or advisable, all in conformity with the requirements of the Securities Act and the rules and regulations of the Securities and Exchange Commission applicable thereto. The Company agrees to cause such issuer to comply with the provisions of the securities or "Blue Sky" laws of any and all jurisdictions which the Collateral Agent shall designate and to make available to its security holders, as soon as practicable, an earnings statement (which need not be audited) which will satisfy the provisions of Section 11(a) of the Securities Act.

- (b) The Company recognizes that the Collateral Agent may be unable to effect a public sale of any or all the Stock Collateral, by reason of certain prohibitions contained in the Securities Act and applicable state securities laws or otherwise, and may be compelled to resort to one or more private sales thereof to a restricted group of purchasers which will be obliged to agree, among other things, to acquire such securities for their own account for investment and not with a view to the distribution or resale thereof. The Company acknowledges and agrees that any such private sale may result in prices and other terms less favorable than if such sale were a public sale and, notwithstanding such circumstances, agrees that any such private sale shall be deemed to have been made in a commercially reasonable manner. The Collateral Agent shall be under no obligation to delay a sale of any of the Stock Collateral for the period of time necessary to permit the issuer thereof to register such securities for public sale under the Securities Act, or under applicable state securities laws, even if such issuer would agree to do so.
- (c) The Company agrees to use its best efforts to do or cause to be done all such other acts as may be necessary to make such sale or sales of all or any portion of the Stock Collateral pursuant to this Section 5.06 valid and binding and in compliance with any and all other applicable law. The Company further agrees that a breach of any of the covenants contained in this Section 5.06 will cause irreparable injury to the Collateral Agent and the Holders, that the Collateral Agent and the Holders have no adequate remedy at law in respect of such breach and, as a consequence, that each and every covenant contained in this Section 5.06 shall be specifically enforceable against the Company, and the Company hereby waives and agrees not to assert any defenses against an action for specific performance of such covenants except for a defense that no Event of Default has occurred under the Note Purchase Agreement.

5.07. DEFICIENCY. If the proceeds of sale, collection or other realization of or upon the Collateral pursuant to Sections 5.05 or 5.06 hereof are insufficient to cover the costs and expenses of such realization and the payment in full of the Secured Obligations, the Company shall remain liable for any deficiency.

5.08. REMOVALS, ETC. Without at least 30 days' prior written notice to the Collateral Agent, the Company shall not (i) maintain any of its books and records with respect to the Collateral at any office or maintain its principal place of business at any place, or permit any Inventory or Equipment to be located anywhere, other than at the address indicated beneath the signature of the Company to the Note Purchase Agreement or at one of the locations identified in Annex 3 hereto or in transit from one of such locations to another, (ii) change its name, or the name under which it does business, identity or corporate structure to such an extent that any financing statement filed in connection with this Agreement would become misleading and (iii) change its jurisdiction of organization or the location of its chief executive office or sole place of business from that identified in Annex 3.

5.09. MAINTENANCE OF INSURANCE.

- (a) At any time the Company holds Material amounts of Inventory and Equipment, the Company shall maintain, with financially sound and reputable companies, insurance policies (i) insuring the Inventory and Equipment (including Motor Vehicles) against loss by fire, explosion, theft and such other casualties as may be reasonably satisfactory to the Majority Holders and (ii) insuring the Company, the Collateral Agent and the Holders against liability for personal injury and property damage relating to such Inventory and Equipment (including Motor Vehicles), such policies to be in such form and amounts and having such coverage as may be reasonably satisfactory to the Collateral Agent and the Majority Holders.
- (b) All such insurance shall (i) provide that no cancellation, material reduction in amount or material change in coverage thereof shall be effective until at least 30 days after receipt by the Collateral Agent of written notice thereof, (ii) name the Collateral Agent as insured party or loss payee, (iii) if reasonably requested by the Collateral Agent, include a breach of warranty clause and (iv) be reasonably satisfactory in all other respects to the Collateral Agent.
- (c) The Company shall deliver to the Collateral Agent and the Holders a report of a reputable insurance broker with respect to such insurance substantially concurrently with each delivery of the Company's audited annual financial statements and such supplemental reports with respect thereto as the Collateral Agent may from time to time reasonably request.
- 5.10. PAYMENT OF OBLIGATIONS. The Company shall pay and discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all taxes, assessment and governmental charges or levies imposed upon the Collateral or in respect of income or profits

therefrom, as well as all claims of any kind (including, without limitation, claims for labor, materials and supplies) against or with respect to the Collateral, except that no such charge need be paid if the amount or validity thereof is currently being contested in good faith by appropriate proceedings, reserves in conformity with GAAP with respect thereto have been provided on the books of the Company and such proceedings could not reasonably be expected to result in the sale, forfeiture or loss of any material portion of the Stock Collateral or any interest therein. In its reasonable discretion, the Collateral Agent may discharge taxes and other encumbrances at any time levied or placed on any of the Collateral, make repairs thereto and pay any necessary filing fees. The Company agrees to reimburse the Collateral Agent on demand for any and all expenditures so made. The Collateral Agent shall have no obligation to the Company to make any such expenditures, nor shall the making thereof relieve the Company of any default.

Notwithstanding anything to the contrary contained herein, the Company shall remain liable under each contract or agreement comprised in the Collateral to be observed or performed by the Company thereunder. The Collateral Agent shall not have any obligation or liability under any such contract or agreement by reason of or arising out of this Agreement or the receipt by the Collateral Agent of any payment relating to any of the Collateral, nor shall the Collateral Agent be obligated in any manner to perform any of the obligations of the Company under or pursuant to any such contract or agreement, to make inquiry as to the nature or sufficiency of any payment received by the Collateral Agent in respect of the Collateral or as to the sufficiency of any performance by any party under any such contract or agreement, to present or file any claim, to take any action to enforce any performance or to collect the payment of any amounts which may have been assigned to the Collateral Agent or to which the Collateral Agent may be entitled at any time or times. The Collateral Agent's sole duty with respect to the custody, safe keeping, and physical preservation of the Collateral in its possession, under the Uniform Commercial Code of the applicable jurisdiction or otherwise, shall be to deal with such Collateral in the same manner as the Collateral Agent deals with similar property for its own account.

- 5.11. NOTICES. The Company shall advise the Collateral Agent and the Holders promptly, in reasonable detail, of:
 - (a) any Lien (other than security interests created hereby or Liens permitted under the Note Purchase Agreement) on any of the Collateral which would adversely affect the ability of the Collateral Agent to exercise any of its remedies hereunder; and
 - (b) of the occurrence of any other event which could reasonably be expected to have a material adverse effect on the aggregate value of the Collateral or on the security interests created hereby.
- 5.12. PRIVATE SALE. The Collateral Agent shall incur no liability as a result of the sale of the Collateral, or any part thereof, at any private sale pursuant to Section 5.05 or 5.06 hereof conducted in a commercially reasonable manner. The Company hereby waives any claims against the Collateral

Agent arising by reason of the fact that the price at which the Collateral may have been sold at such a private sale was less than the price that might have been obtained at a public sale or was less than the aggregate amount of the Secured Obligations, even if the Collateral Agent accepts the first offer received and does not offer the Collateral to more than one offeree.

5.13. APPLICATION OF PROCEEDS. Except as otherwise herein expressly provided, the proceeds of any collection, sale or other realization of all or any part of the Collateral pursuant hereto, and any other cash at the time held by the Collateral Agent shall be applied by the Collateral Agent:

FIRST, to the payment of the costs and expenses of such collection, sale or other realization, including reasonable out-of-pocket costs and expenses of the Collateral Agent and the fees and expenses of its agents and counsel and other professional advisors, and all expenses incurred and advances made by the Collateral Agent in connection therewith;

NEXT, to the payment in full of the Secured Obligations; and

FINALLY, to the payment to the Company, or its successors or assigns, or as a court of competent jurisdiction may direct, of any surplus then remaining.

As used in this Section 5, "PROCEEDS" of Collateral shall mean cash, securities and other property realized in respect of, and distributions in kind of, Collateral, including any thereof received under any reorganization, liquidation or adjustment of debt of the Company or any issuer of or obligor on any of the Collateral.

- 5.14. ATTORNEY-IN-FACT. Without limiting any rights or powers granted by this Agreement to the Collateral Agent while no Event of Default has occurred and is continuing, upon the occurrence and during the continuance of any Event of Default the Collateral Agent is hereby appointed the attorney-in-fact of the Company for the purpose of carrying out the provisions of this Section 5 and taking any action and executing any instruments that the Collateral Agent may deem necessary or advisable to accomplish the purposes hereof, which appointment as attorney-in-fact is irrevocable and coupled with an interest. Without limiting the generality of the foregoing, so long as the Collateral Agent shall be entitled under this Section 5 to make collections in respect of the Collateral, the Collateral Agent shall have the right and power to receive, endorse and collect all checks made payable to the order of the Company representing any dividend, payment or other distribution in respect of the Collateral or any part thereof and to give full discharge for the same.
- 5.15. TERMINATION. When all Secured Obligations shall have been paid in full, this Agreement shall terminate, and the Collateral Agent shall forthwith cause to be assigned, transferred and delivered, against receipt but without any recourse, warranty or representation whatsoever, any remaining Collateral and money received in respect thereof, to or on the order of the Company. The Collateral Agent shall also execute and deliver to the Company upon

such termination such Uniform Commercial Code termination statements, certificates for terminating the Liens on the Motor Vehicles and such other documentation as shall be reasonably requested by the Company to effect the termination and release of the Liens on the Collateral.

- 5.16. FURTHER ASSURANCES. The Company agrees that, from time to time it will execute and deliver such further documents and do such other acts and things as may be necessary or appropriate or as the Collateral Agent may reasonably request in order fully to effect the purposes of this Agreement.
- 5.17. RELEASE OF MOTOR VEHICLES. So long as no Default shall have occurred and be continuing, upon the request of the Company, the Collateral Agent shall execute and deliver to the Company such instruments as the Company shall reasonably request to remove the notation of the Collateral Agent as lienholder on any certificate of title for any Motor Vehicle; PROVIDED that any such instruments shall be delivered, and the release effective only upon receipt by the Collateral Agent of a certificate from the Company stating that the Motor Vehicle the lien on which is to be released is to be sold or has suffered a casualty loss (with title thereto passing to the casualty insurance company therefor in settlement of the claim for such loss).

Section 6. MISCELLANEOUS.

- 6.01. NO WAIVER. No failure on the part of the Collateral Agent to exercise, and no course of dealing with respect to, and no delay in exercising, any right, power or remedy hereunder shall operate as a waiver thereof; nor shall any single or partial exercise by the Collateral Agent of any right, power or remedy hereunder preclude any other or further exercise thereof or the exercise of any other right, power or remedy. The remedies herein are cumulative and are not exclusive of any remedies provided by law.
- 6.02. NOTICES. All notices, requests, consents and demands hereunder shall be in writing and telecopied or delivered to the intended recipient at its "Address for Notices" specified pursuant to Section 19 of the Note Purchase Agreement and shall be deemed to have been given at the times specified in said Section 19.
- 6.03. AMENDMENTS, ETC. The terms of this Agreement may be waived, altered or amended only by an instrument in writing duly executed by the Company and the Collateral Agent in accordance with Section 18 of the Note Purchase Agreement. Any such amendment or waiver shall be binding upon the Collateral Agent, each holder of any of the Secured Obligations and the Company.
- 6.04. SUCCESSORS AND ASSIGNS. This Agreement shall be binding upon and inure to the benefit of the respective successors and assigns of the Company, the Collateral Agent and each holder of any of the Secured Obligations (PROVIDED, however, that the Company shall not assign or transfer its rights hereunder without the prior written consent of the Collateral Agent).
- 6.05. CAPTIONS. The captions and section headings appearing herein are included solely for convenience of reference and are not intended to affect the interpretation of any provision of this Agreement.

- 6.06. COUNTERPARTS. This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument and either of the parties hereto may execute this Agreement by signing any such counterpart. Delivery of an executed counterpart of a signature page of this Agreement by facsimile transmission shall be as effective as delivery of a manually executed counterpart of this Agreement.
- 6.07. GOVERNING LAW. This Agreement shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the law of the State of New York excluding choice-of-law principles of the law of such State that would require the application of the laws of a jurisdiction other than such State.
- 6.08. AGENTS AND ATTORNEYS-IN-FACT. The Collateral Agent may employ agents and attorneys-in-fact in connection herewith and shall not be responsible for the negligence or misconduct of any such agents or attorneys-in-fact selected by it in good faith.
- 6.09. SEVERABILITY. If any provision hereof is invalid and unenforceable in any jurisdiction, then, to the fullest extent permitted by law, (i) the other provisions hereof shall remain in full force and effect in such jurisdiction and shall be liberally construed in favor of the Collateral Agent in order to carry out the intentions of the parties hereto as nearly as may be possible and (ii) the invalidity or unenforceability of any provision hereof in any jurisdiction shall not affect the validity or enforceability of such provision in any other jurisdiction.
- 6.10. ENVIRONMENTAL INDEMNIFICATION OF COLLATERAL AGENT AND HOLDERS. The Company hereby agrees to indemnify the Collateral Agent and the Holders from, and hold the Collateral Agent and the Holders harmless against, any losses, liabilities, claims, damages or expenses arising under any Environmental Law as a result of the past, present or future operations of the Company or any of its Subsidiaries following the exercise by the Collateral Agent of any of its rights and remedies under this Agreement.
- 6.11. INDEMNIFICATION OF NV HOLDINGS. The Company hereby agrees to indemnify NV Holdings for, and hold it harmless against, any claim, demand, expense (including but not limited to attorneys' fees), loss or liability incurred by it arising solely out of or in connection with its being an affiliate of the Company. The Company hereby acknowledges that the indemnity contained in this Section 6.11 is for the benefit of NV Holdings and that NV Holdings is relying on said indemnity as a basis for entering into the NV Holdings Pledge Agreement.
- 6.12. INTEGRATION. This Agreement and the other Note Documents represent the agreement of the Company, the Collateral Agent and the Holders with respect to the subject matter hereof and thereof, and there are no promises, undertakings, representations or warranties by the Collateral Agent or any Holder relative to subject matter hereof and thereof not expressly set forth or referred to herein or in the other Note Documents.

- 6.13. ACKNOWLEDGEMENTS. The Company hereby acknowledges that:
- (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Note Documents to which it is a party;
- (b) neither the Collateral Agent nor any Holder has any fiduciary relationship with or duty to the Company arising out of or in connection with this Agreement or any of the other Note Documents, and the relationship between the Company, on the one hand, and the Collateral Agent and Holders, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and
- (c) no joint venture is created hereby or by the other Note Documents or otherwise exists by virtue of the transactions contemplated hereby among the Holders or among the Company and the Holders.
- 6.14. ADDITIONAL SUBSIDIARIES. The Company shall cause each new Subsidiary of the Company to execute and deliver an acknowledgment and undertaking substantially in the form of the acknowledgments and undertakings attached hereto and to do such other acts and things as the Collateral Agent may reasonably request in order fully to effect the purposes of this Agreement.

IN WITNESS WHEREOF, the parties hereto have caused this Pledge and Security Agreement to be duly executed and delivered as of the day and year first above written.

BGLS INC.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen Title: Executive Vice President

UNITED STATES TRUST COMPANY OF NEW YORK, as Collateral Agent

By: /s/ PATRICIA GALLAGHER

Name: Patricia Gallagher

Title: Assistant Vice President

FOR GOOD AND VALUABLE CONSIDERATION, the receipt and sufficiency of which are hereby acknowledged, New Valley Holdings, Inc., a Delaware corporation ("NV HOLDINGS") hereby acknowledges receipt of a copy of the foregoing Pledge and Security Agreement dated as of May 14, 2001 and agrees as follows (with capitalized terms used and not otherwise defined below that are defined in such Pledge and Security Agreement having the meanings ascribed to such terms therein):

- NV Holdings will promptly note on its books the security interests granted under such Pledge and Security Agreement in the applicable Pledged Stock.
- 2. NV Holdings hereby waives any rights or requirement at any time hereafter to receive a copy of such Pledge and Security Agreement in connection with the registration of any Stock Collateral in the name of the Collateral Agent or its nominee or the exercise of voting rights by the Collateral Agent.
- 3. NV Holdings agrees (a) to take all actions contemplated to be taken by it under the Pledge and Security Agreement, (b) to comply with all provisions thereof and (c) not to participate in any transaction that is inconsistent with its obligations hereunder or thereunder.

The undersigned hereby represents and warrants that (a) the execution, delivery and performance of this Acknowledgment and Undertaking has been duly authorized by all necessary action on the part of the undersigned, (b) the undersigned has duly executed and delivered this Acknowledgment and Undertaking and (c) this Acknowledgment and Undertaking is its legal, valid and binding obligation, enforceable against the undersigned in accordance with its terms. This Acknowledgment and Undertaking shall be governed by, and construed in accordance with, the law of the State of New York.

This Acknowledgment and Undertaking may not be terminated, revoked, amended or otherwise modified without the prior written consent of the Collateral Agent, which is an express third-party beneficiary hereof.

Dated: May 14, 2001

NEW VALLEY HOLDINGS, INC.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen

FOR GOOD AND VALUABLE CONSIDERATION, the receipt and sufficiency of which are hereby acknowledged, Vector Tobacco (USA) Ltd., a Delaware corporation ("VTUSA") hereby acknowledges receipt of a copy of the foregoing Pledge and Security Agreement dated as of May 14, 2001 and agrees as follows (with capitalized terms used and not otherwise defined below that are defined in such Pledge and Security Agreement having the meanings ascribed to such terms therein):

- 1. VTUSA will promptly note on its books the security interests granted under such Pledge and Security Agreement in the applicable Pledged Stock.
- 2. VTUSA hereby waives any rights or requirement at any time hereafter to receive a copy of such Pledge and Security Agreement in connection with the registration of any Stock Collateral in the name of the Collateral Agent or its nominee or the exercise of voting rights by the Collateral Agent.
- 3. VTUSA agrees (a) to take all actions contemplated to be taken by it under the Pledge and Security Agreement, (b) to comply with all provisions thereof and (c) not to participate in any transaction that is inconsistent with its obligations hereunder or thereunder.

The undersigned hereby represents and warrants that (a) the execution, delivery and performance of this Acknowledgment and Undertaking has been duly authorized by all necessary action on the part of the undersigned, (b) the undersigned has duly executed and delivered this Acknowledgment and Undertaking and (c) this Acknowledgment and Undertaking is its legal, valid and binding obligation, enforceable against the undersigned in accordance with its terms. This Acknowledgment and Undertaking shall be governed by, and construed in accordance with, the law of the State of New York.

This Acknowledgment and Undertaking may not be terminated, revoked, amended or otherwise modified without the prior written consent of the Collateral Agent, which is an express third-party beneficiary hereof.

Dated: May 14, 2001

VECTOR TOBACCO (USA) LTD.

By: /s/ SAMUEL M. VEASEY

Name: Samuel M. Veasey

Title: Vice President - Finance

FOR GOOD AND VALUABLE CONSIDERATION, the receipt and sufficiency of which are hereby acknowledged, Vector Research Ltd., a Delaware corporation ("RESEARCH") hereby acknowledges receipt of a copy of the foregoing Pledge and Security Agreement dated as of May 14, 2001 and agrees as follows (with capitalized terms used and not otherwise defined below that are defined in such Pledge and Security Agreement having the meanings ascribed to such terms therein):

- 1. Research will promptly note on its books the security interests granted under such Pledge and Security Agreement in the applicable Pledged Stock.
- 2. Research hereby waives any rights or requirement at any time hereafter to receive a copy of such Pledge and Security Agreement in connection with the registration of any Stock Collateral in the name of the Collateral Agent or its nominee or the exercise of voting rights by the Collateral Agent.
- 3. Research agrees (a) to take all actions contemplated to be taken by it under the Pledge and Security Agreement, (b) to comply with all provisions thereof and (c) not to participate in any transaction that is inconsistent with its obligations hereunder or thereunder.

The undersigned hereby represents and warrants that (a) the execution, delivery and performance of this Acknowledgment and Undertaking has been duly authorized by all necessary action on the part of the undersigned, (b) the undersigned has duly executed and delivered this Acknowledgment and Undertaking and (c) this Acknowledgment and Undertaking is its legal, valid and binding obligation, enforceable against the undersigned in accordance with its terms. This Acknowledgment and Undertaking shall be governed by, and construed in accordance with, the law of the State of New York.

This Acknowledgment and Undertaking may not be terminated, revoked, amended or otherwise modified without the prior written consent of the Collateral Agent, which is an express third-party beneficiary hereof.

Dated: May 14, 2001

VECTOR RESEARCH LTD.

By: /s/ JOSELYNN D. VAN SICLEN

Name: Joselynn D. Van Siclen Title: Vice President

FOR GOOD AND VALUABLE CONSIDERATION, the receipt and sufficiency of which are hereby acknowledged, Brooke (Overseas) Ltd., a Delaware corporation ("BROOKE OVERSEAS") hereby acknowledges receipt of a copy of the foregoing Pledge and Security Agreement dated as of May 14, 2001 and agrees as follows (with capitalized terms used and not otherwise defined below that are defined in such Pledge and Security Agreement having the meanings ascribed to such terms therein):

- 1. Brooke Overseas will promptly note on its books the security interests granted under such Pledge and Security Agreement in the applicable Pledged Stock.
- 2. Brooke Overseas hereby waives any rights or requirement at any time hereafter to receive a copy of such Pledge and Security Agreement in connection with the registration of any Stock Collateral in the name of the Collateral Agent or its nominee or the exercise of voting rights by the Collateral Agent.
- 3. Brooke Overseas agrees (a) to take all actions contemplated to be taken by it under the Pledge and Security Agreement, (b) to comply with all provisions thereof and (c) not to participate in any transaction that is inconsistent with its obligations hereunder or thereunder.

The undersigned hereby represents and warrants that (a) the execution, delivery and performance of this Acknowledgment and Undertaking has been duly authorized by all necessary action on the part of the undersigned, (b) the undersigned has duly executed and delivered this Acknowledgment and Undertaking and (c) this Acknowledgment and Undertaking is its legal, valid and binding obligation, enforceable against the undersigned in accordance with its terms. This Acknowledgment and Undertaking shall be governed by, and construed in accordance with, the law of the State of New York.

This Acknowledgment and Undertaking may not be terminated, revoked, amended or otherwise modified without the prior written consent of the Collateral Agent, which is an express third-party beneficiary hereof.

Dated: May 14, 2001

BROOKE (OVERSEAS) LTD.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen

FOR GOOD AND VALUABLE CONSIDERATION, the receipt and sufficiency of which are hereby acknowledged, New Valley Corporation, a Delaware corporation ("NEW VALLEY") hereby acknowledges receipt of a copy of the foregoing Pledge and Security Agreement dated as of May 14, 2001 and agrees as follows (with capitalized terms used and not otherwise defined below that are defined in such Pledge and Security Agreement having the meanings ascribed to such terms therein):

- 1. New Valley will promptly note on its books the security interests granted under such Pledge and Security Agreement in the applicable Stock
- 2. New Valley hereby waives any rights or requirement at any time hereafter to receive a copy of such Pledge and Security Agreement in connection with the registration of any Stock Collateral in the name of the Collateral Agent or its nominee or the exercise of voting rights by the Collateral Agent.
- 3. New Valley agrees (a) to take all actions contemplated to be taken by it under the Pledge and Security Agreement, (b) to comply with all provisions thereof and (c) not to participate in any transaction that is inconsistent with its obligations hereunder or thereunder.

The undersigned hereby represents and warrants that (a) the execution, delivery and performance of this Acknowledgment and Undertaking has been duly authorized by all necessary action on the part of the undersigned, (b) the undersigned has duly executed and delivered this Acknowledgment and Undertaking and (c) this Acknowledgment and Undertaking is its legal, valid and binding obligation, enforceable against the undersigned in accordance with its terms. This Acknowledgment and Undertaking shall be governed by, and construed in accordance with, the law of the State of New York.

This Acknowledgment and Undertaking may not be terminated, revoked, amended or otherwise modified without the prior written consent of the Collateral Agent, which is an express third-party beneficiary hereof.

Dated: May 14, 2001

NEW VALLEY CORPORATION

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen

FOR GOOD AND VALUABLE CONSIDERATION, the receipt and sufficiency of which are hereby acknowledged, Brooke Group Holding Inc., a Delaware corporation ("BROOKE HOLDING") hereby acknowledges receipt of a copy of the foregoing Pledge and Security Agreement dated as of May 14, 2001 and agrees as follows (with capitalized terms used and not otherwise defined below that are defined in such Pledge and Security Agreement having the meanings ascribed to such terms therein):

- 2. Brooke Holding hereby waives any rights or requirement at any time hereafter to receive a copy of such Pledge and Security Agreement in connection with the registration of any Stock Collateral in the name of the Collateral Agent or its nominee or the exercise of voting rights by the Collateral Agent.
- 3. Brooke Holding agrees (a) to take all actions contemplated to be taken by it under the Pledge and Security Agreement, (b) to comply with all provisions thereof and (c) not to participate in any transaction that is inconsistent with its obligations hereunder or thereunder.

The undersigned hereby represents and warrants that (a) the execution, delivery and performance of this Acknowledgment and Undertaking has been duly authorized by all necessary action on the part of the undersigned, (b) the undersigned has duly executed and delivered this Acknowledgment and Undertaking and (c) this Acknowledgment and Undertaking is its legal, valid and binding obligation, enforceable against the undersigned in accordance with its terms. This Acknowledgment and Undertaking shall be governed by, and construed in accordance with, the law of the State of New York.

This Acknowledgment and Undertaking may not be terminated, revoked, amended or otherwise modified without the prior written consent of the Collateral Agent, which is an express third-party beneficiary hereof.

Dated: May 14, 2001

BROOKE GROUP HOLDING INC.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen

PLEDGED STOCK

ISSUER	CERTIFICATE NOS.	REGISTERED OWNER	NUMBER OF SHARES
New Valley Corporation	NV 1712	BGLS Inc.	83,628 shares of common stock, par value \$.01 per share.
New Valley Corporation	NV 1710	BGLS Inc.	1,974 shares of common stock, par value \$.01 per share.
New Valley Corporation	W 2096	BGLS Inc.	5,924 warrants to purchase shares of Common Stock, par value \$.01 per share.
New Valley Corporation	W 2098	BGLS Inc.	1,254,425 warrants to purchase shares of Common Stock, par value \$.01 per share.
Vector Tobacco (USA), LTD.	1	BGLS Inc.	100 shares of common stock, par value \$.01 per share.
Brooke (Overseas) Ltd.	2	BGLS Inc.	10 shares of common stock, par value \$.01 per share.
Vector Research Ltd.	2	BGLS Inc.	100 shares, \$.01 par value per share.
Brooke Group Holding Inc.	2	BGLS Inc.	1,000 shares, \$.10 par value per share.
New Valley Holdings, Inc.	1	BGLS Inc.	100 shares, \$.01 par value per share.

FILINGS AND OTHER ACTIONS REQUIRED TO PERFECT SECURITY INTERESTS

UNIFORM COMMERCIAL CODE FILINGS

Delaware Secretary of State Florida Secretary of State

PATENT AND TRADEMARK FILINGS

None

ACTIONS WITH RESPECT TO STOCK COLLATERAL

Delivery to Collateral Agents of certificates representing Pledged Stock in the State of New York along with a stock power endorsed in blank.

OTHER ACTIONS

- Execution and delivery of the Account Control Agreement by the Company, the Collateral Agent and the securities intermediary party thereto.
- 2. Delivery of pledged notes to the Collateral Agent in New York with effective endorsements in blank.

CHIEF EXECUTIVE OFFICE

LIST OF LOCATIONS

100 S.E. Second Street Miami, Florida 33131 34

ANNEX 4

INTELLECTUAL PROPERTY

None.

INSTRUMENTS

\$25,000,000 Secured Revolving Demand Promissory Note, dated as of March 6, 2001, made by Vector Tobacco (USA) Ltd. and Vector Tobacco Ltd. in favor of BGLS Inc. and Vector Group Ltd.

DEPOSIT ACCOUNTS

Account #01596108029 at Bank of America, N.A.

PLEDGE AND SECURITY AGREEMENT

PLEDGE AND SECURITY AGREEMENT, dated as of May 14, 2001 (as amended, modified or supplemented from time to time, this "AGREEMENT") between New Valley Holdings, Inc., a corporation duly organized and validly existing under the laws of the State of Delaware (the "NV HOLDINGS") and United States Trust Company of New York, a New York banking corporation, as collateral agent (together with its successors and assigns, the "COLLATERAL AGENT") on behalf of the holders (the "HOLDERS") of the 10% Senior Secured Notes due March 31, 2006 (the "NOTES") issued pursuant to that certain Note Purchase Agreement, dated as of May 14, 2001 (as amended, supplemented or modified from time to time, the "NOTE PURCHASE AGREEMENT") between BGLS Inc., (the "COMPANY") and the Holders.

Section 1. DEFINITIONS. Terms defined in the Note Purchase Agreement are used herein as defined therein. In addition, as used herein:

"ACCOUNT CONTROL AGREEMENT" means the Account Control Agreement, dated as of the date hereof, substantially in the form of Exhibit B-2 to the Note Purchase Agreement among the Company, the Collateral Agent and Bank of America, N.A., as amended, modified and supplemented from time to time.

"ACCOUNTS" has the meaning ascribed thereto in Section 3(d) hereof.

"BROOKE HOLDING" means Brooke Group Holding, Inc. a Delaware corporation, and any successor thereto.

"COLLATERAL" has the meaning ascribed thereto in Section 3 hereof.

"COMMERCIAL TORT CLAIM" means (i) prior to July 1, 2001, a claim of NV Holdings arising in tort and (ii) on or after July 1, 2001, has the meaning ascribed thereto in the Uniform Commercial Code.

"COPYRIGHT COLLATERAL" means all Copyrights, whether now owned or hereafter acquired by NV Holdings.

"COPYRIGHTS" means all copyrights, copyright registrations and applications for copyright registrations, including, without limitation, all renewals, extensions, income, royalties, damages and payments now or hereafter due and/or payable under and with respect thereto, including, without limitation, damages and payments for past or future infringements thereof, the right to sue for past, present and future infringements thereof, and all rights corresponding thereto throughout the world.

"DEPOSIT ACCOUNT" means any deposit account (as defined in the Uniform Commercial Code).

"DOCUMENTS" has the meaning ascribed thereto in Section 3(k) hereof.

"ENVIRONMENTAL LAWS" means any and all present and future Federal, state, local and foreign laws, rules or regulations, and any orders or decrees, in each case as now or hereafter in effect, relating to the regulation or protection of human health, safety or the environment or to emissions, discharges, releases or threatened releases of pollutants, contaminants, chemicals or toxic or hazardous substances or wastes into the indoor or outdoor environment, including, without limitation, ambient air, soil, surface water, ground water, wetlands, land or subsurface strata, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of pollutants, contaminants, chemicals or toxic or hazardous substances or wastes.

"EQUIPMENT" has the meaning ascribed thereto in Section 3(i) hereof.

"EQUITY INTERESTS" means, with respect to any Person, any capital stock of such Person and shares, interests, participations or other ownership interests (however designated) of any Person and any rights (other than debt securities convertible into corporate stock), warrants and options to purchase any of the foregoing, including (without limitation) each class of common stock and preferred stock of such Person if such Person is a corporation and each general and limited partnership interest of such Person if such Person is a partnership.

"FARM PRODUCTS" has the meaning ascribed thereto in the Uniform Commercial Code.

"INITIAL PLEDGED NEW VALLEY SHARES" means all Equity Interests of New Valley represented by the certificates identified in Annex 1 hereto.

"INSTRUMENTS" has the meaning ascribed thereto in Section 3(e) hereof.

"INTELLECTUAL PROPERTY" means, collectively, all Copyright Collateral, all Patent Collateral and all Trademark Collateral.

"INVENTORY" has the meaning ascribed thereto in Section 3(g) hereof.

"ISSUERS" means, collectively, the respective corporations identified in Annex 1 hereto under the caption "Issuer".

"MOTOR VEHICLES" means motor vehicles, tractors, trailers and other like property whether or not the title thereto is governed by a certificate of title or ownership.

"NEW VALLEY" means New Valley Corporation, a Delaware corporation, and any successor thereto. $\,$

"NEW VALLEY COMMON STOCK" means common shares of New Valley, par value \$.01 per share.

"NEW VALLEY CONVERTIBLE SECURITIES" means any securities or rights that are convertible into or exchangeable for Equity Interests of New Valley.

"NOTES" has the meaning ascribed thereto in the first paragraph hereof.

"NV HOLDINGS" means New Valley Holdings, Inc., a Delaware corporation, and any successor thereto.

"PATENT COLLATERAL" means all Patents, whether now owned or hereafter acquired by NV Holdings.

"PATENTS" means all patents and patent applications, including, without limitation, the inventions and improvements described and claimed therein together with the reissues, divisions, continuations, renewals, extensions and continuations-in-part thereof, all income, royalties, damages and payments now or hereafter due and/or payable under and with respect thereto, including, without limitation, damages and payments for past or future infringements thereof, the right to sue for past, present and future infringements thereof, and all rights corresponding thereto throughout the world.

"PERSON" means an individual, partnership, corporation, limited liability company, association, trust, unincorporated organization, or a government or agency or political subdivision thereof.

"PLEDGED STOCK" has the meaning ascribed thereto in Section 3(a) hereof. $\,$

"RESEARCH" means Vector Research Ltd., a Delaware corporation, and any successor thereto.

"SECURED OBLIGATIONS" means, collectively, all obligations and liabilities of any kind or nature, present or future, absolute or contingent, of (i) the Company arising under the Note Documents, (ii) NV Holdings arising under this Agreement or any other undertaking or agreement delivered by NV Holdings in connection with any other Note Document, (iii) VTUSA arising under any undertaking or agreement delivered by VTUSA in connection with any Note Document, (iv) Brooke Overseas arising under any undertaking or agreement delivered by Brooke Overseas in connection with any Note Document, (v) New Valley arising under any undertaking or agreement delivered by New Valley in connection with any Note Document, (vi) Brooke Holding arising under the Brooke Holding Pledge Agreement or any other undertaking or agreement delivered by Research in connection with any Note Document, (viii) Vector arising under the Vector Pledge Agreement or any other undertaking or agreement delivered by Vector in connection with any other Note Document or (ix) Liggett arising under any undertaking or agreement delivered by Liggett in connection with any Note Document.

"STOCK COLLATERAL" means, collectively, the Collateral described in clauses (a) through (c) of Section 3 hereof and the proceeds of and to any such property and, to the extent related to any such property or such proceeds, all books, correspondence, credit files, records, invoices and other papers.

"TRADEMARK COLLATERAL" means all Trademarks, whether now owned or hereafter acquired by NV Holdings. Notwithstanding the foregoing, the Trademark Collateral does not and shall not include any Trademark that would be rendered invalid, abandoned, void or unenforceable by reason of its being included as part of the Trademark Collateral.

"TRADEMARKS" means all trade names, trademarks and service marks, logos, trademark and service mark registrations, and applications for trademark and service mark registrations, including, without limitation, all renewals of trademark and service mark registrations, all rights corresponding thereto throughout the world, the right to recover for all past, present and future infringements thereof, all other rights of any kind whatsoever accruing thereunder or pertaining thereto, together, in each case, with the product lines and goodwill of the business connected with the use of, and symbolized by, each such trade name, trademark and service mark.

"UNIFORM COMMERCIAL CODE" means the Uniform Commercial Code as in effect from time to time in the State of New York.

"VECTOR" means Vector Group Ltd., a Delaware corporation, and any successor thereto.

"VTUSA" means Vector Tobacco (USA) Ltd., a Delaware corporation, and any successor thereto.

Section 2. REPRESENTATIONS AND WARRANTIES. NV Holdings represents and warrants and covenants to the Collateral Agent for the benefit of the Holders and the Collateral Agent that:

- (b) NV Holdings has the corporate power and authority and the legal right to execute and deliver, to perform its obligations under, and to grant the security interest in the Collateral pursuant to, this Agreement and has taken all necessary corporate action to authorize its execution, delivery and performance of, and grant of the security interest in the Collateral pursuant to, this Agreement;
- (c) this Agreement constitutes a legal, valid and binding obligation of NV Holdings, enforceable in accordance with its terms, except as enforceability may be affected by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing;
- (d) the execution, delivery and performance by NV Holdings of this Agreement will not (i) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of NV Holdings under, any indenture, mortgage, deed of trust, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other agreement or instrument to which NV

Holdings is bound or by which NV Holdings or any of its properties is bound or affected, (ii) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental or Regulatory Authority applicable to NV Holdings or (iii) violate any provision of any statute or other rule or regulation of any Governmental or Regulatory Authority applicable to NV Holdings;

- (e) no consent, approval or authorization of, or registration, filing (other than the filing of any financing statements contemplated in any Note Document) or declaration with, any Governmental or Regulatory Authority is required in connection with the execution, delivery or performance by NV Holdings of this Agreement;
- (f) no litigation, investigation or proceeding of or before any arbitrator or Governmental or Regulatory Authority is pending or, to the knowledge of NV Holdings, threatened by or against NV Holdings against any of its properties or revenues with respect to this Agreement or any of the transactions contemplated hereby;
- (g) except for the Security Interest, NV Holdings is the sole legal and equitable owner of the Collateral, holds the same free and clear of all Liens, charges, encumbrances and security interests of any kind and nature and will make no other assignment, pledge, mortgage, hypothecation, transfer or other disposition of the Collateral, other than the sale or application of the Collateral in compliance with the provisions of Sections 8.24 and 13.8 of the Note Purchase Agreement;
- (h) NV Holdings has good, right and legal title to the Collateral and will defend its title thereto against the claims of all Persons whomsoever and will maintain and preserve the Security Interest as long as this Agreement shall remain in full force and effect;
- (i) no financing statement or other public notice with respect to all or any part of the Collateral is on file or of record in any public office, except as have been filed by the Collateral Agent pursuant to this Agreement or as are permitted by the Note Purchase Agreement;
- (j) except for the filing of UCC-1 financing statements, no consent or approval of, or other action by, and no notice to or filing with, any Governmental or Regulatory Authority or securities exchange, was or is necessary as a condition (i) to the validity of the pledge provided for herein or for the execution, delivery or performance of this Agreement by NV Holdings or (ii) other than filings with (A) the United States Patent and Trademark Office in connection with the assignment of the Patents, Trademarks and Copyrights or (B) the

Securities Act, and regulations thereunder and any applicable blue sky laws, for the exercise by the Collateral Agent of the voting or other rights provided for in this Agreement or the remedies in respect of the Collateral pursuant to this Agreement;

- (k) as of the date hereof: (i) there are 22,813,063 shares of New Valley Common Stock issued and outstanding and such shares are the only issued and outstanding Equity Interests of New Valley and (ii) there are no issued and outstanding New Valley Convertible Securities, and New Valley is not subject to any obligation, contingent or otherwise, to issue in the future any additional Equity Interests or any such New Valley Convertible Securities except:
 - (1) warrants exercisable for 17,867,566 shares of New Valley Common Stock; and
 - (2) options to purchase 1,271,088 shares of New Valley Common Stock and options to purchase warrants exercisable for 584,000 shares of New Valley Common Stock.
- (1) the Initial Pledged New Valley Shares are duly authorized, validly issued, fully paid and nonassessable;
- (m) except for the Note Documents, NV Holdings is not now and will not become a party to any voting trust or other agreement or undertaking with respect to the exercise of the voting or consent rights associated with any of the Stock Collateral (other than an agreement or undertaking in respect of a sale of the Stock Collateral, which sale is or will be in compliance with the provisions of Section 13.8 of the Note Purchase Agreement), and there are no other restrictions on the exercise of such rights;
- (n) except for the Note Documents, NV Holdings is not now and will not become a party to any agreement or undertaking pursuant to which any Issuer or any other Person could have the right to purchase any of the Stock Collateral (other than an agreement or undertaking in respect of a sale of the Stock Collateral, which sale is or will be in compliance with the provisions of Section 13.8 of the Note Purchase Agreement), and at the time of delivery of any Stock Collateral to the Collateral Agent there will not be (and NV Holdings will not thereafter permit to exist) any other restrictions on the transfer of the Stock Collateral;
- (o) except for the Note Documents, NV Holdings has the right to vote, pledge and grant a security interest in, or otherwise transfer, the Collateral free of any liens (except as set forth in (g) above);
- (p) the security interests granted pursuant to this Agreement(i) upon completion of the filings and other actions specified on Annex 2

(which, in the case of all filings and other documents referred to on said Annex, have been delivered to the Collateral Agent in duly executed form) will constitute valid perfected security interests in all of the Collateral in favor of the Collateral Agent enforceable in accordance with the terms hereof against all creditors of NV Holdings and all Persons purporting to purchase any Collateral from NV Holdings and (ii) are prior to all other Liens on the Collateral in existence on the date hereof except for Liens permitted by the Note Purchase Agreement which have priority over the Liens on the Collateral held by the Collateral Agent by operation of law;

- (q) the grant and perfection of the Security Interest in the Stock Collateral for the benefit of the Collateral Agent and the Holders, in accordance with the terms hereof, will not violate the registration requirements of the Securities Act, any provisions of any other applicable federal, state or foreign securities laws, or any applicable general corporation law or other applicable law;
- (r) each Issuer is a corporation duly organized, validly existing and in good standing under the laws of the state of its incorporation and has full power and authority to own or lease its properties and assets and to carry on its business;
- (s) the stock powers delivered to the Collateral Agent with respect to the Pledged Stock have been, and at the time of the delivery of any other stock powers to the Collateral Agent in connection with the pledge of any additional Stock Collateral such stock powers will be, duly executed;
- (t) the pledge of the Initial Pledged Shares pursuant to this Agreement, together with the transfer of such Initial Pledged Shares to the Collateral Agent, creates a valid and perfected first priority security interest in the Stock Collateral, in favor of the Collateral Agent for the benefit of the Collateral Agent and the Holders, securing the due and punctual payment and performance in full of the Secured Obligations in accordance with the respective terms thereof;
- (u) NV Holdings is incorporated in the State of Delaware and the location of its chief executive office or sole place of business on the date hereof is specified on Annex 3;
- (v) none of the Collateral consists of, or is the proceeds (as defined in the Uniform Commercial Code) of, Farm Products;
- (w) on the date hereof, the Inventory and the Equipment (other than Motor Vehicles) are kept at the locations listed on Annex 3;

6;

(x)

- (i) Annex 4 lists all Intellectual Property owned by NV Holdings in its own name on the date hereof;
- (ii) On the date hereof, all material Intellectual Property is valid, subsisting, unexpired and enforceable, has not been abandoned and does not infringe the intellectual property rights of any other Person:
- (iii) Except as set forth in Annex 4, on the date hereof, none of the Intellectual Property is the subject of any licensing or franchise agreement pursuant to which NV Holdings is the licensor or franchisor;
- (iv) No holding, decision or judgment has been rendered by any Governmental or Regulatory Authority which would limit, cancel or question the validity of, or NV Holdings' rights in, any Intellectual Property in any respect; and
- (v) No action or proceeding is pending or, to the knowledge of NV Holdings threatened, or the date hereof seeking to limit, cancel or question the validity of any Intellectual Property or NV Holdings' ownership interest therein;
- (y) subject to Section 4.01(b), all Instruments owned by NV Holdings are listed on Annex 5 and have been delivered to the Collateral Agent;
- (z) each Instrument which is a promissory note issued to and held by NV Holdings (including any promissory note evidencing loans made by NV Holdings to any of its Affiliates or Associates, any Vector Expanded Affiliate or any Group Executive) constitutes the legal, valid and binding obligation of the obligor with respect thereto, enforceable in accordance with its terms, subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing;
- (aa) any instruments of assignment and transfer delivered pursuant to Section 4.01(b) have been, or will be, duly executed;
 - (bb) all Deposit Accounts of NV Holdings are listed on Annex
 - (cc) NV Holdings has no Commercial Tort Claims; and

(dd) NV Holdings shall not sell, assign, transfer or otherwise dispose of, or pledge, hypothecate or otherwise encumber, any of its Equity Interests in New Valley which are pledged pursuant to the this Agreement, unless any such sale, assignment, transfer or disposition is in compliance with the provisions of Section 10.1 and Section 13.8 of the Note Purchase Agreement.

Section 3. COLLATERAL. As collateral security for the prompt payment in full when due (whether at stated maturity, by acceleration or otherwise) and performance of the Secured Obligations, NV Holdings hereby pledges and grants to the Collateral Agent for the benefit of the Holders and the Collateral Agent a security interest in all of NV Holdings' right, title and interest in the following property, whether now owned by NV Holdings or hereafter acquired and whether now existing or hereafter coming into existence (all being collectively referred to herein as "COLLATERAL"):

- (a) the shares of capital stock and other Equity Interests of the Issuers represented by the certificates identified in Annex 1 hereto and all Equity Interests of any other Person, now or hereafter owned by NV Holdings, in each case together with the certificates evidencing the same (collectively, the "PLEDGED STOCK");
- (b) all shares, securities, moneys or property representing a dividend on any of the Pledged Stock, or representing a distribution or return of capital upon or in respect of the Pledged Stock, or resulting from a split-up, revision, reclassification or other like change of the Pledged Stock or otherwise received in exchange therefor, and any subscription warrants, rights or options issued to the holders of, or otherwise in respect of, the Pledged Stock;
- (c) without affecting the obligations of NV Holdings under any provision prohibiting such action hereunder or under the Note Purchase Agreement, in the event of any consolidation or merger in which an Issuer or other Subsidiary of NV Holdings is not the surviving corporation, all shares of each class of the capital stock and all other Equity Interests of the successor corporation (unless such successor corporation is NV Holdings itself) formed by or resulting from such consolidation or merger (the Pledged Stock, together with all other certificates, shares, securities, properties or moneys as may from time to time be pledged hereunder pursuant to clause (a) or (b) above and this clause (c) being herein collectively called the "STOCK COLLATERAL");
- (d) all accounts and general intangibles (each as defined in the Uniform Commercial Code) of NV Holdings constituting any right to the payment of money, including (but not limited to) all moneys due and to become due to NV Holdings in respect of any loans or advances or for Inventory or Equipment or other goods sold or leased or for services

rendered, all moneys due and to become due to NV Holdings under any guarantee (including a letter of credit) of the purchase price of Inventory or Equipment sold by NV Holdings and all tax refunds (such accounts, general intangibles and moneys due and to become due being herein called collectively "ACCOUNTS");

- (e) all instruments, chattel paper or letters of credit (each as defined in the Uniform Commercial Code) of NV Holdings evidencing, representing, arising from or existing in respect of, relating to, securing or otherwise supporting the payment of, any of the Accounts, including (but not limited to) promissory notes, drafts, bills of exchange and trade acceptances (herein collectively called "INSTRUMENTS");
 - (f) all Deposit Accounts of NV Holdings;
- (g) all inventory (as defined in the Uniform Commercial Code) of NV Holdings, all goods obtained by NV Holdings in exchange for such inventory, and any products made or processed from such inventory including all substances, if any, commingled therewith or added thereto (herein collectively called "INVENTORY");
- (h) all Intellectual Property and all other accounts or general intangibles not constituting Intellectual Property or Accounts;
- (i) all equipment (as defined in the Uniform Commercial Code) of NV Holdings, including all Motor Vehicles (herein collectively called "EQUIPMENT");
- (j) each contract and other agreement of NV Holdings relating to the sale or other disposition of Inventory or Equipment;
- (k) all documents of title (as defined in the Uniform Commercial Code) or other receipts of NV Holdings covering, evidencing or representing Inventory or Equipment (herein collectively called "DOCUMENTS");
- (1) all rights, claims and benefits of NV Holdings against any Person arising out of, relating to or in connection with Inventory or Equipment purchased by NV Holdings, including, without limitation, any such rights, claims or benefits against any Person storing or transporting such Inventory or Equipment;
 - (m) all Commercial Tort Claims of NV Holdings;
- (n) all securities accounts (as defined in Article 8 of the Uniform Commercial Code) and the financial assets (as defined in Article 8 of the Uniform Commercial Code) credited from time to time thereto; and

(o) all other tangible and intangible personal property and fixtures of NV Holdings, including, without limitation, all proceeds, products, offspring, accessions, rents, profits, income, benefits, substitutions and replacements of and to any of the property of NV Holdings described in the preceding clauses of this Section 3 (including, without limitation, any proceeds of insurance thereon and all causes of action, claims and warranties now or hereafter held by NV Holdings in respect of any of the items listed above) and, to the extent related to any property described in said clauses or such proceeds, products and accessions, all books, correspondence, credit files, records, invoices and other papers, including without limitation all tapes, cards, computer runs and other papers and documents in the possession or under the control of NV Holdings or any computer bureau or service company from time to time acting for NV Holdings.

Section 4. FURTHER ASSURANCES; REMEDIES. In furtherance of the grant of the pledge and security interest pursuant to Section 3 hereof, NV Holdings hereby agrees with the Collateral Agent as follows:

4.01. DELIVERY AND OTHER PERFECTION. NV Holdings shall:

- (a) if any of the shares, securities, moneys or property required to be pledged by NV Holdings under clauses (a), (b) and (c) of Section 3 hereof are received by NV Holdings, hold such Property in trust and forthwith either (x) transfer and deliver to the Collateral Agent such shares or securities so received by NV Holdings (together with the certificates for any such shares and securities duly endorsed in blank or accompanied by undated stock powers duly executed in blank, all in form and substance reasonably satisfactory to the Collateral Agent), all of which thereafter shall be held by the Collateral Agent (except in the case of cash dividends which shall be transferred to the Company and held in a Company Account in accordance with the Note Purchase Agreement), pursuant to the terms of this Agreement, as part of the Collateral or (y) take such other action as may be necessary or appropriate or as the Collateral Agent may reasonably request to duly record the Lien created hereunder in such shares, securities, moneys or property in said clauses (a), (b) and (c);
- (b) deliver and pledge to the Collateral Agent any and all Instruments pledged hereunder, endorsed and/or accompanied by such instruments of assignment and transfer in such form and substance as may be required to perfect the pledge thereof hereunder or as the Collateral Agent may reasonably request; PROVIDED, that so long as no Default shall have occurred and be continuing, NV Holdings may retain for collection in the ordinary course any Instruments received by NV Holdings in the ordinary course of business and the Collateral Agent

shall, promptly upon request of NV Holdings, make appropriate arrangements for making any Instrument pledged by NV Holdings hereunder available to NV Holdings for purposes of presentation, collection or renewal (any such arrangement to be effected, to the extent deemed appropriate by the Collateral Agent, against trust receipt or like document);

- (c) give, execute, deliver, file and/or record any financing statement, continuation statement, notice, instrument, document, agreement or other papers that may be necessary or desirable or that the Collateral Agent may reasonably request to create, preserve, perfect or validate the security interest granted pursuant hereto or to enable the Collateral Agent to exercise and enforce its rights hereunder with respect to such pledge and security interest, including, without limitation, causing any or all of the Stock Collateral to be transferred of record into the name of the Collateral Agent or its nominee (and the Collateral Agent agrees that if any Stock Collateral is transferred into its name or the name of its nominee, the Collateral Agent will thereafter promptly give to NV Holdings copies of any notices and communications received by it with respect to the Stock Collateral), PROVIDED that notices to account debtors in respect of any Accounts or Instruments shall be subject to the provisions of clause (g) below;
- (d) upon the acquisition after the date hereof by NV Holdings of any Equipment covered by a certificate of title or ownership, cause the Collateral Agent to be listed as the lienholder on such certificate of title and within 120 days of the acquisition thereof deliver evidence of the same to the Collateral Agent;
- (e) keep full and accurate books and records relating to the Collateral, and stamp or otherwise mark such books and records in such manner as may be necessary to reflect the security interests granted by this Agreement;
- (f) permit representatives of the Collateral Agent, upon reasonable notice, at any time during normal business hours to inspect and make abstracts from its books and records pertaining to the Collateral, and permit representatives of the Collateral Agent to be present at NV Holdings' place of business to receive copies of all communications and remittances relating to the Collateral, and forward copies of any notices or communications received by NV Holdings with respect to the Collateral, all to the extent and in such manner as the Collateral Agent may reasonably request; and
- (g) upon the occurrence and during the continuance of any Default, upon request of the Collateral Agent, promptly notify (and NV Holdings hereby authorizes the Collateral Agent so to notify) each account debtor in respect of any Accounts or Instruments that such Collateral has been assigned to the Collateral Agent hereunder, and

that any payments due or to become due in respect of such Collateral are to be made directly to the Securities Account.

- 4.02. OTHER FINANCING STATEMENTS AND LIENS. Without the prior written consent of the Collateral Agent, NV Holdings shall not file or suffer to be on file, or authorize or permit to be filed or to be on file, in any jurisdiction, any financing statement or like instrument with respect to the Collateral in which the Collateral Agent is not named as the sole secured party.
- 4.03. PRESERVATION OF RIGHTS. The Collateral Agent shall not be required to take steps necessary to preserve any rights against third parties to any of the Collateral.

4.04. SPECIAL PROVISIONS RELATING TO STOCK COLLATERAL.

- (a) So long as no Event of Default shall have occurred and be continuing, NV Holdings shall have the right to exercise all voting, consensual and other powers of ownership pertaining to the Stock Collateral for all purposes not inconsistent with the terms of this Agreement, the Note Purchase Agreement, the Notes or any other instrument or agreement referred to herein or therein, PROVIDED that NV Holdings agrees that it will not vote the Stock Collateral in any manner that is inconsistent with the terms of any Note Document or any other related instrument or agreement; and the Collateral Agent shall execute and deliver to NV Holdings or cause to be executed and delivered to NV Holdings all such proxies, powers of attorney, dividend and other orders, and all such instruments, without recourse, as NV Holdings may reasonably request for the purpose of enabling NV Holdings to exercise the rights and powers that it is entitled to exercise pursuant to this Section 4.04(a).
- (b) If any Account Notice Event shall have occurred, then so long as such Account Notice Event shall continue, and whether or not the Collateral Agent exercises any available right to declare any Secured Obligation due and payable or seeks or pursues any other relief or remedy available to it under applicable law or under this Agreement, the Note Purchase Agreement, the Notes or any other agreement relating to such Secured Obligation, to the extent not already delivered pursuant to Section 4.01, all dividends and other distributions on the Stock Collateral shall be paid directly to the Collateral Agent and deposited by it into the Securities Account subject to the terms of this Agreement, and, if the Collateral Agent shall so request in writing, NV Holdings agrees to execute and deliver to the Collateral Agent appropriate additional dividend, distribution and other orders and documents to that end.
- 4.05. EVENTS OF DEFAULT, ETC. Subject to the last sentence of this Section 4.05, during the period during which (i) with respect to moneys contained in the NV Holdings Checking Account and all cash dividends and cash proceeds of any of the Stock Collateral, an Account Notice Event shall have

occurred and be continuing and (ii) with respect to all other Collateral, an Event of Default shall have occurred and be continuing:

- (a) NV Holdings shall, at the request of the Collateral Agent, assemble the Collateral owned by it at such place or places, reasonably convenient to both the Collateral Agent and NV Holdings, designated in its request;
- (b) the Collateral Agent may make any reasonable compromise or settlement deemed desirable with respect to any of the Collateral and may extend the time of payment, arrange for payment in installments, or otherwise modify the terms of, any of the Collateral;
- (c) the Collateral Agent shall have all of the rights and remedies with respect to the Collateral of a secured party under the Uniform Commercial Code (whether or not the Uniform Commercial Code is in effect in the jurisdiction where the rights and remedies are asserted) and such additional rights and remedies to which a secured party is entitled under the laws in effect in any jurisdiction where any rights and remedies hereunder may be asserted, including, without limitation, the right, to the maximum extent permitted by law, to exercise all voting, consensual and other powers of ownership pertaining to the Collateral as if the Collateral Agent were the sole and absolute owner thereof (and NV Holdings agrees to take all such action as may be appropriate to give effect to such right);
- (d) the Collateral Agent in its discretion may, in its name or in the name of NV Holdings or otherwise, demand, sue for, collect or receive any money or property at any time payable or receivable on account of or in exchange for any of the Collateral, but shall be under no obligation to do so; and
- (e) the Collateral Agent may, upon ten business days' prior written notice to NV Holdings of the time and place, with respect to the Collateral or any part thereof that shall then be or shall thereafter come into the possession, custody or control of the Collateral Agent or any of its agents, sell, lease, assign or otherwise dispose of all or any part of such Collateral, at such place or places as the Collateral Agent deems best, and for cash or for credit or for future delivery (without thereby assuming any credit risk), at public or private sale, without demand of performance or notice of intention to effect any such disposition or of the time or place thereof (except such notice as is required above or by applicable statute and cannot be waived), and the Collateral Agent or anyone else may be the purchaser, lessee, assignee or recipient of any or all of the Collateral so disposed of at any public sale (or, to the extent permitted by law, at any private sale) and thereafter hold the same absolutely, free from any claim or right of whatsoever kind, including any right or equity of redemption (statutory or otherwise), of NV Holdings, any such demand,

notice and right or equity being hereby expressly waived and released. The Collateral Agent may, without notice or publication, adjourn any public or private sale or cause the same to be adjourned from time to time by announcement at the time and place fixed for the sale, and such sale may be made at any time or place to which the sale may be so adjourned.

The proceeds of each collection, sale or other disposition under this Section 4.05 shall be applied in accordance with Section 4.13 hereof.

4.06. REGISTRATION RIGHTS.

- (a) If the Collateral Agent shall determine to exercise its right to sell any or all of the Stock Collateral, and if in the opinion of the Collateral Agent it is necessary or advisable to have the Stock Collateral, or that portion thereof to be sold, registered under the provisions of the Securities Act, NV Holdings will cause the issuer thereof to (i) execute and deliver, and cause the directors and officers of such issuer to execute and deliver, all such instruments and documents, and do or cause to be done all such other acts as may be, in the opinion of the Collateral Agent, necessary or advisable to register the Stock Collateral, or that portion thereof to be sold, under the provisions of the Securities Act, (ii) use its best efforts to cause the registration statement relating thereto to become effective and to remain effective for a period of one year from the date of the first public offering of the Stock Collateral, or that portion thereof to be sold and (iii) make all amendments thereto and/or to the related prospectus which, in the opinion of the Collateral Agent, are necessary or advisable, all in conformity with the requirements of the Securities Act and the rules and regulations of the Securities and Exchange Commission applicable thereto. NV Holdings agrees to cause such issuer to comply with the provisions of the securities or "Blue Sky" laws of any and all jurisdictions which the Collateral Agent shall designate and to make available to its security holders, as soon as practicable, an earnings statement (which need not be audited) which will satisfy the provisions of Section 11(a) of the Securities Act.
- (b) NV Holdings recognizes that the Collateral Agent may be unable to effect a public sale of any or all the Stock Collateral, by reason of certain prohibitions contained in the Securities Act and applicable state securities laws or otherwise, and may be compelled to resort to one or more private sales thereof to a restricted group of purchasers which will be obliged to agree, among other things, to acquire such securities for their own account for investment and not with a view to the distribution or resale thereof. NV Holdings acknowledges and agrees that any such private sale may result in prices and other terms less favorable than if such sale were a public sale and, notwithstanding such circumstances, agrees that any such private sale shall be deemed to have been made in a commercially reasonable manner. The Collateral Agent shall be under no obligation to delay a

sale of any of the Stock Collateral for the period of time necessary to permit the issuer thereof to register such securities for public sale under the Securities Act, or under applicable state securities laws, even if such issuer would agree to do so.

- (c) NV Holdings agrees to use its best efforts to do or cause to be done all such other acts as may be necessary to make such sale or sales of all or any portion of the Stock Collateral pursuant to this Section 4.06 valid and binding and in compliance with any and all other applicable law. NV Holdings further agrees that a breach of any of the covenants contained in this Section 4.06 will cause irreparable injury to the Collateral Agent and the Holders, that the Collateral Agent and the Holders have no adequate remedy at law in respect of such breach and, as a consequence, that each and every covenant contained in this Section 4.06 shall be specifically enforceable against NV Holdings, and NV Holdings hereby waives and agrees not to assert any defenses against an action for specific performance of such covenants except for a defense that no Event of Default has occurred under the Note Purchase Agreement.
- 4.07. DEFICIENCY. If the proceeds of sale, collection or other realization of or upon the Collateral pursuant to Sections 4.05 or 4.06 hereof are insufficient to cover the costs and expenses of such realization and the payment in full of the Secured Obligations, NV Holdings shall remain liable for any deficiency; PROVIDED, HOWEVER, that the maximum liability of NV Holdings hereunder shall in no event exceed the amount of Secured Obligations which may be guaranteed by NV Holdings under applicable laws relating to the insolvency of debtors.
- 4.08. REMOVALS, ETC. Without at least 30 days' prior written notice to the Collateral Agent, NV Holdings shall not (i) maintain any of its books and records with respect to the Collateral at any office or maintain its principal place of business at any place, or permit any Inventory or Equipment to be located anywhere, other than at the address indicated beneath the signature of NV Holdings to the Note Purchase Agreement or at one of the locations identified in Annex 3 hereto or in transit from one of such locations to another, (ii) change its name, or the name under which it does business, identity or corporate structure to such an extent that any financing statement filed in connection with this Agreement would become misleading and (iii) change its jurisdiction of organization or the location of its chief executive office or sole place of business from that identified in Annex 3.

4.09. MAINTENANCE OF INSURANCE.

- (a) At any time NV Holdings holds Material amounts of Inventory and Equipment, NV Holdings shall maintain, with financially sound and reputable companies, insurance policies (i) insuring the Inventory and Equipment (including Motor Vehicles) against loss by fire, explosion, theft and such other casualties as may be reasonably satisfactory to the Required Holders and (ii) insuring NV Holdings, the Collateral Agent and the Holders against liability for personal injury and property damage relating to such Inventory and Equipment (including Motor Vehicles), such policies to be in such form and amounts and having such coverage as may be reasonably satisfactory to the Collateral Agent and the Required Holders.
- (b) All such insurance shall (i) provide that no cancellation, material reduction in amount or material change in coverage thereof shall be effective until at least 30 days after receipt by the Collateral Agent of written notice thereof, (ii) name the Collateral Agent as insured party or loss payee, (iii) if reasonably requested by the Collateral Agent, include a breach of warranty clause and (iv) be reasonably satisfactory in all other respects to the Collateral Agent.
- (c) NV Holdings shall deliver to the Collateral Agent and the Holders a report of a reputable insurance broker with respect to such insurance substantially concurrently with each delivery of NV Holdings' audited annual financial statements and such supplemental reports with respect thereto as the Collateral Agent may from time to time reasonably request.
- 4.10. PAYMENT OF OBLIGATIONS. NV Holdings shall pay and discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all taxes, assessments and governmental charges or levies imposed upon the Collateral or in respect of income or profits therefrom, as well as all claims of any kind (including, without limitation, claims for labor, materials and supplies) against or with respect to the Collateral, except that no such charge need be paid if the amount or validity thereof is currently being contested in good faith by appropriate proceedings, reserves in conformity with GAAP with respect thereto have been provided on the books of NV Holdings and such proceedings could not reasonably be expected to result in the sale, forfeiture or loss of any material portion of the Stock Collateral or any interest therein. In its reasonable discretion, the Collateral Agent may discharge taxes and other encumbrances at any time levied or placed on any of the Collateral, make repairs thereto and pay any necessary filing fees. NV Holdings agrees to reimburse the Collateral Agent on demand for any and all expenditures so made. The Collateral Agent shall have no obligation to NV Holdings to make any such expenditures, nor shall the making thereof relieve NV Holdings of any default.

Notwithstanding anything to the contrary contained herein, NV Holdings shall remain liable under each contract or agreement comprised in the Collateral to be observed or performed by NV Holdings thereunder. The Collateral Agent

shall not have any obligation or liability under any such contract or agreement by reason of or arising out of this Agreement or the receipt by the Collateral Agent of any payment relating to any of the Collateral, nor shall the Collateral Agent be obligated in any manner to perform any of the obligations of NV Holdings under or pursuant to any such contract or agreement, to make inquiry as to the nature or sufficiency of any payment received by the Collateral Agent in respect of the Collateral or as to the sufficiency of any performance by any party under any such contract or agreement, to present or file any claim, to take any action to enforce any performance or to collect the payment of any amounts which may have been assigned to the Collateral Agent or to which the Collateral Agent may be entitled at any time or times. The Collateral Agent's sole duty with respect to the custody, safe keeping, and physical preservation of the Collateral in its possession, under the Uniform Commercial Code of the applicable jurisdiction or otherwise, shall be to deal with such Collateral in the same manner as the Collateral Agent deals with similar property for its own account.

- 4.11. NOTICES. NV Holdings shall advise the Collateral Agent and the Holders promptly, in reasonable detail, of:
 - (a) any Lien (other than security interests created hereby or Liens permitted under the Note Purchase Agreement) on any of the Collateral which would adversely affect the ability of the Collateral Agent to exercise any of its remedies hereunder; and
 - (b) of the occurrence of any other event which could reasonably be expected to have a material adverse effect on the aggregate value of the Collateral or on the security interests created hereby.
- 4.12. PRIVATE SALE. The Collateral Agent shall incur no liability as a result of the sale of the Collateral, or any part thereof, at any private sale pursuant to Section 4.05 or 4.06 hereof conducted in a commercially reasonable manner. NV Holdings hereby waives any claims against the Collateral Agent arising by reason of the fact that the price at which the Collateral may have been sold at such a private sale was less than the price that might have been obtained at a public sale or was less than the aggregate amount of the Secured Obligations, even if the Collateral Agent accepts the first offer received and does not offer the Collateral to more than one offeree.
- 4.13. APPLICATION OF PROCEEDS. Except as otherwise herein expressly provided, the proceeds of any collection, sale or other realization of all or any part of the Collateral pursuant hereto, and any other cash at the time held by the Collateral Agent shall be applied by the Collateral Agent:

FIRST, to the payment of the costs and expenses of such collection, sale or other realization, including reasonable out-of-pocket costs and expenses of the Collateral Agent and the fees and expenses of its agents and counsel and other professional advisors, and all expenses incurred and advances made by the Collateral Agent in connection therewith;

NEXT, to the payment in full of the Secured Obligations; and

FINALLY, to the payment to NV Holdings, or its successors or assigns, or as a court of competent jurisdiction may direct, of any surplus then remaining.

As used in this Section 4, "PROCEEDS" of Collateral shall mean cash, securities and other property realized in respect of, and distributions in kind of, Collateral, including any thereof received under any reorganization, liquidation or adjustment of debt of NV Holdings or any issuer of or obligor on any of the Collateral.

- 4.14. ATTORNEY-IN-FACT. Without limiting any rights or powers granted by this Agreement to the Collateral Agent while no Event of Default has occurred and is continuing, upon the occurrence and during the continuance of any Event of Default the Collateral Agent is hereby appointed the attorney-in-fact of NV Holdings for the purpose of carrying out the provisions of this Section 4 and taking any action and executing any instruments that the Collateral Agent may deem necessary or advisable to accomplish the purposes hereof, which appointment as attorney-in-fact is irrevocable and coupled with an interest. Without limiting the generality of the foregoing, so long as the Collateral Agent shall be entitled under this Section 4 to make collections in respect of the Collateral, the Collateral Agent shall have the right and power to receive, endorse and collect all checks made payable to the order of NV Holdings representing any dividend, payment or other distribution in respect of the Collateral or any part thereof and to give full discharge for the same.
- 4.15. TERMINATION. When all Secured Obligations shall have been paid in full, this Agreement shall terminate, and the Collateral Agent shall forthwith cause to be assigned, transferred and delivered, against receipt but without any recourse, warranty or representation whatsoever, any remaining Collateral and money received in respect thereof, to or on the order of NV Holdings. The Collateral Agent shall also execute and deliver to NV Holdings upon such termination such Uniform Commercial Code termination statements, certificates for terminating the Liens on the Motor Vehicles and such other documentation as shall be reasonably requested by NV Holdings to effect the termination and release of the Liens on the Collateral.
- 4.16. FURTHER ASSURANCES. NV Holdings agrees that, from time to time it will execute and deliver such further documents and do such other acts and things as may be necessary or appropriate or as the Collateral Agent may reasonably request in order fully to effect the purposes of this Agreement.
- 4.17. RELEASE OF MOTOR VEHICLES. So long as no Default shall have occurred and be continuing, upon the request of NV Holdings, the Collateral Agent shall execute and deliver to NV Holdings such instruments as NV Holdings shall reasonably request to remove the notation of the Collateral Agent as lienholder on any certificate of title for any Motor Vehicle; PROVIDED that any such instruments shall be delivered, and the release effective only upon receipt by the Collateral Agent of a certificate from NV Holdings stating that the Motor

Vehicle the lien on which is to be released is to be sold or has suffered a casualty loss (with title thereto passing to the casualty insurance company therefor in settlement of the claim for such loss).

- 4.18. RECEIPT OF CASH. Subject to Section 8.21(b) of the Note Purchase Agreement, NV Holdings shall immediately dividend or otherwise transfer all Cash or cash equivalents it receives to the Company.
- 4.19. SALE OF NEW VALLEY EQUITY INTERESTS. Other than in accordance with the Note Purchase Agreement, NV Holdings shall not sell any Equity Interests in New Valley.
 - 4.20. CERTAIN COVENANTS.
 - (a) NV Holdings shall not incur any Indebtedness.
 - (b) The sole function of NV Holdings shall be to hold Equity Interests in New Valley, and NV Holdings shall not sell any Equity Interests in New Valley other than in accordance with the Note Purchase Agreement and this Agreement. NV Holdings shall not conduct any business activities other than to hold such Equity Interests, to make such payments and to conduct those incidental and customary activities required in order to oversee and manage its interests in New Valley.
 - (c) Except as contemplated in this Agreement, NV Holdings shall not, directly or indirectly, create, incur, assume or permit to exist any Lien on or with respect to any of its assets, including without limitation, the New Valley Assets, or any income or profits therefrom including any dividends and distributions received thereon or distributable with respect thereto, or interest therein.
 - (d) NV Holdings shall not make, directly or indirectly, any Investments in cash or by delivery of Property (x) in any Person, whether by the acquisition of Indebtedness or other obligation, security, or Equity Interest, or by loan, advance, extension of credit, or capital contribution, or otherwise, or (y) in any Property except (i) purchases of Equity Interests in New Valley so long as such Equity Interests shall be pledged hereunder, and (ii) Investments in Publicly-held Companies to the extent received as a result of any dividend in kind, distribution, spin off, split up, corporate reorganization, or similar transaction with respect to New Valley in which such Investment is received without the payment of further consideration by NV Holdings so long as such Investment is pledged hereunder.
 - (e) NV Holdings shall not (i) consolidate with or merge with or into any other Person, (ii) transfer (by lease, assignment, sale or otherwise) in a single transaction or through a series of transactions, all or substantially all of its properties and assets, as an entirety or substantially as an entirety, to another Person or group of Persons or (iii) adopt a Plan of Liquidation.

- 4.21. WAIVER OF SURETYSHIP DEFENSES. To the extent this Agreement is deemed a guarantee, the liability of NV Holdings hereunder shall be absolute, unconditional and irrevocable irrespective of, and without being lessened or limited by:
 - (a) the occurrence of any Event of Default under, or any lack of validity, legality or enforceability of any provision of any Note Document, or any other agreement or document;
 - (b) the failure of any Holder:
 - (i) to assert any claim or demand or to enforce any right or remedy against the Company or any other Person (including any other guarantor of the Secured Obligations) under the provisions of any Note Document, or otherwise, or
 - (ii) to exercise any right or remedy against any other guarantor of or any other Person pledging collateral securing any of the Secured Obligations;
 - (c) any change in the time, manner or place of payment of, or in any term of, all or any of the Secured Obligations, or any other extension, compromise, indulgence or renewal of any Secured Obligation;
 - (d) any reduction, limitation, variation, impairment, discontinuance or termination of the Secured Obligations for any reason (other than by reason of any payment which is not required to be rescinded), including any claim of waiver, release, discharge, surrender, alteration or compromise, and shall not be subject to (and NV Holdings hereby waives any right to or claim of) any defense or setoff, counterclaim, recoupment or termination whatsoever by reason of the invalidity, illegality, nongenuineness, irregularity, compromise, unenforceability of, or any other event or occurrence affecting, the Secured Obligations or otherwise (other than by reason of any payment which is not required to be rescinded);
 - (e) any amendment to, rescission, waiver or other modification of, or any consent to any departure from, any of the terms of the Secured Obligations or any guarantees or security;
 - (f) any addition, exchange, release, discharge, realization or non-perfection of any collateral security in respect of the Secured Obligations;
 - (g) any amendment to, rescission, waiver or other modification of, or release or addition of, or consent to any departure from, any other guarantee held by the Holders as security for any of the Secured Obligations;

- (h) the loss of or in respect of or the unenforceability of any other guarantee or other security which the Holders may now or hereafter hold in respect of the Secured Obligations, whether occasioned by the fault of the Holders or otherwise;
- (i) any change in the name of the Company or in the constitutive documents, capital structure, capacity or constitution of the Company, the bankruptcy or insolvency of the Company, the sale of any or all of the Company's business or assets or the Company being consolidated, merged or amalgamated with any other Person;
- (j) any failure on the part of the Company or any other Person to perform or comply with any term of the Note Purchase Agreement, the Notes, any of the Secured Obligations or any other agreement or document:
- (k) any suit or other action brought by any beneficiaries or creditors of, or by, the Company or any other Person for any reason whatsoever, including without limitation any suit or action in any way attacking or involving any issue, matter or thing in respect of any Note Document, any of the Secured Obligations or any other agreement or document;
- (1) any lack or limitation of status or of power, incapacity or disability of the Company or any trustee or agent thereof; or
- (m) any other circumstance (other than final and indefeasible payment in full) which might otherwise constitute a defense available to, or a legal or equitable discharge of, the Company, Brooke Holding, Vector, any other surety or guarantor.
- 4.22. RELEASE OF COLLATERAL. The security interest in any Collateral shall automatically and without any action by the Collateral Agent or the Holders be released upon any sale of such Collateral by NV Holdings pursuant to the terms of the Note Purchase Agreement and this Agreement.

Section 5. MISCELLANEOUS.

- 5.01. NO WAIVER. No failure on the part of the Collateral Agent to exercise, and no course of dealing with respect to, and no delay in exercising, any right, power or remedy hereunder shall operate as a waiver thereof; nor shall any single or partial exercise by the Collateral Agent of any right, power or remedy hereunder preclude any other or further exercise thereof or the exercise of any other right, power or remedy. The remedies herein are cumulative and are not exclusive of any remedies provided by law.
- 5.02. NOTICES. All notices, requests, consents and demands hereunder shall be in writing and telecopied or delivered to the intended recipient at its "Address for Notices" specified pursuant to Section 19 of the Note Purchase Agreement (or, in the case of NV Holdings, at the address

specified therein for the Company) and shall be deemed to have been given at the times specified in said Section 19.

- 5.03. AMENDMENTS, ETC. The terms of this Agreement may be waived, altered or amended only by an instrument in writing duly executed by NV Holdings and the Collateral Agent in accordance with Section 18 of the Note Purchase Agreement. Any such amendment or waiver shall be binding upon the Collateral Agent, each holder of any of the Secured Obligations and NV Holdings.
- 5.04. SUCCESSORS AND ASSIGNS. This Agreement shall be binding upon and inure to the benefit of the respective successors and assigns of NV Holdings, the Collateral Agent and each holder of any of the Secured Obligations (PROVIDED, however, that NV Holdings shall not assign or transfer its rights hereunder without the prior written consent of the Collateral Agent).
- 5.05. CAPTIONS. The captions and section headings appearing herein are included solely for convenience of reference and are not intended to affect the interpretation of any provision of this Agreement.
- 5.06. COUNTERPARTS. This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument and either of the parties hereto may execute this Agreement by signing any such counterpart. Delivery of an executed counterpart of a signature page of this Agreement by facsimile transmission shall be as effective as delivery of a manually executed counterpart of this Agreement.
- 5.07. GOVERNING LAW. This Agreement shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the law of the State of New York excluding choice-of-law principles of the law of such State that would require the application of the laws of a jurisdiction other than such State.
- 5.08. AGENTS AND ATTORNEYS-IN-FACT. The Collateral Agent may employ agents and attorneys-in-fact in connection herewith and shall not be responsible for the negligence or misconduct of any such agents or attorneys-in-fact selected by it in good faith.
- 5.09. SEVERABILITY. If any provision hereof is invalid and unenforceable in any jurisdiction, then, to the fullest extent permitted by law, (i) the other provisions hereof shall remain in full force and effect in such jurisdiction and shall be liberally construed in favor of the Collateral Agent in order to carry out the intentions of the parties hereto as nearly as may be possible and (ii) the invalidity or unenforceability of any provision hereof in any jurisdiction shall not affect the validity or enforceability of such provision in any other jurisdiction.
- 5.10. ENVIRONMENTAL INDEMNIFICATION OF COLLATERAL AGENT AND HOLDERS. NV Holdings hereby agrees to indemnify the Collateral Agent and the Holders from, and hold the Collateral Agent and the Holders harmless against, any losses, liabilities, claims, damages or expenses arising under any

Environmental Law as a result of the past, present or future operations of NV Holdings or any of its Subsidiaries following the exercise by the Collateral Agent of any of its rights and remedies under this Agreement.

- 5.11. INTEGRATION. This Agreement and the other Note Documents represent the agreement of NV Holdings, the Collateral Agent and the Holders with respect to the subject matter hereof and thereof, and there are no promises, undertakings, representations or warranties by the Collateral Agent or any Holder relative to subject matter hereof and thereof not expressly set forth or referred to herein or in the other Note Documents.
 - 5.12. ACKNOWLEDGEMENTS. NV Holdings hereby acknowledges that:
 - (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Note Documents to which it is a party;
 - (b) neither the Collateral Agent nor any Holder has any fiduciary relationship with or duty to NV Holdings arising out of or in connection with this Agreement or any of the other Note Documents, and the relationship between NV Holdings, on the one hand, and the Collateral Agent and Holders, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and
 - (c) no joint venture is created hereby or by the other Note Documents or otherwise exists by virtue of the transactions contemplated hereby among the Holders or among NV Holdings and the Holders.
- 5.13. ADDITIONAL SUBSIDIARIES. NV Holdings shall cause each new Subsidiary of NV Holdings to execute and deliver an acknowledgment and undertaking substantially in the form of the acknowledgments and undertakings attached hereto and to do such other acts and things as the Collateral Agent may reasonably request in order fully to effect the purposes of this Agreement.

IN WITNESS WHEREOF, the parties hereto have caused this Pledge and Security Agreement to be duly executed and delivered as of the day and year $\frac{1}{2} \left(\frac{1}{2} \right)$ first above written.

NEW VALLEY HOLDINGS, INC.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen Title: Executive Vice President

UNITED STATES TRUST COMPANY OF NEW YORK, as Collateral Agent

/s/ PATRICIA GALLAGHER

Name: Patricia Gallagher Title: Assistant Vice President

ACKNOWLEDGMENT AND UNDERTAKING

FOR GOOD AND VALUABLE CONSIDERATION, the receipt and sufficiency of which are hereby acknowledged, New Valley Corporation, a Delaware corporation ("NEW VALLEY") hereby acknowledges receipt of a copy of the foregoing Pledge and Security Agreement dated as of May 14, 2001 and agrees as follows (with capitalized terms used and not otherwise defined below that are defined in such Pledge and Security Agreement having the meanings ascribed to such terms therein):

- 1. New Valley will promptly note on its books the security interests granted under such Pledge and Security Agreement in the applicable Pledged Stock.
- 2. New Valley hereby waives any rights or requirement at any time hereafter to receive a copy of such Pledge and Security Agreement in connection with the registration of any Stock Collateral in the name of the Collateral Agent or its nominee or the exercise of voting rights by the Collateral Agent.
- 3. New Valley agrees (a) to take all actions contemplated to be taken by it under the Pledge and Security Agreement, (b) to comply with all provisions thereof and (c) not to participate in any transaction that is inconsistent with its obligations hereunder or thereunder.

The undersigned hereby represents and warrants that (a) the execution, delivery and performance of this Acknowledgment and Undertaking has been duly authorized by all necessary action on the part of the undersigned, (b) the undersigned has duly executed and delivered this Acknowledgment and Undertaking and (c) this Acknowledgment and Undertaking is its legal, valid and binding obligation, enforceable against the undersigned in accordance with its terms. This Acknowledgment and Undertaking shall be governed by, and construed in accordance with, the law of the State of New York.

This Acknowledgment and Undertaking may not be terminated, revoked, amended or otherwise modified without the prior written consent of the Collateral Agent, which is an express third-party beneficiary hereof.

Dated: May 14, 2001

NEW VALLEY CORPORATION

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen

Title: Executive Vice President

PLEDGED STOCK

ISSUER	CERTIFICATE NOS.	REGISTERED OWNER	NUMBER OF SHARES
New Valley Corporation	NV 1709	New Valley Holdings, Inc.	396,996 shares of common stock, par value \$.01 per share.
New Valley Corporation	NV 1711	New Valley Holdings, Inc.	12,366,520 shares of common stock, par value \$.01 per share.
New Valley Corporation	W 2095	New Valley Holdings, Inc.	1,190,988 warrants to purchase shares of Common Stock, par value \$.01 per share.
New Valley Corporation	W 2097	New Valley Holdings, Inc.	618,326 warrants to purchase shares of Common Stock, par value \$.01 per share.

FILINGS AND OTHER ACTIONS REQUIRED TO PERFECT SECURITY INTERESTS

UNIFORM COMMERCIAL CODE FILINGS

Delaware Secretary of State Florida Secretary of State

PATENT AND TRADEMARK FILINGS

None.

ACTIONS WITH RESPECT TO STOCK COLLATERAL

Delivery to Collateral Agents of certificates representing Pledged Stock in the State of New York along with stock powers endorsed in blank.

OTHER ACTIONS

CHIEF EXECUTIVE OFFICE

LIST OF LOCATIONS

209 B Baynard Building 3411 Silverside Road Wilmington, Delaware 19810 30

ANNEX 4

INTELLECTUAL PROPERTY

None.

31

ANNEX 5

INSTRUMENTS

None.

DEPOSIT ACCOUNTS

Account #01596321075 at Bank of America, N.A.

PLEDGE AND SECURITY AGREEMENT

PLEDGE AND SECURITY AGREEMENT, dated as of May 14, 2001 (as amended, modified or supplemented from time to time, this "AGREEMENT") between Brooke Group Holding Inc., a corporation duly organized and validly existing under the laws of the State of Delaware (the "BROOKE HOLDING") and United States Trust Company of New York, a New York banking corporation, as collateral agent (together with its successors and assigns, the "COLLATERAL AGENT") on behalf of the holders (the "HOLDERS") of the 10% Senior Secured Notes due March 31, 2006 (the "NOTES") issued pursuant to that certain Note Purchase Agreement, dated as of May 14, 2001 (as amended, supplemented or modified from time to time, the "NOTE PURCHASE AGREEMENT") between the BGLS, Inc (the "COMPANY") and the Holders.

Section 1. DEFINITIONS. Terms defined in the Note Purchase Agreement are used herein as defined therein. In addition, as used herein:

"ACCOUNT CONTROL AGREEMENT" means the Account Control Agreement, dated as of the date hereof, substantially in the form of Exhibit B-2 to the Note Purchase Agreement among the Company, the Collateral Agent and Bank of America, N.A., as amended, modified and supplemented from time to time.

"ACCOUNTS" has the meaning ascribed thereto in Section 3(d) hereof.

"BROOKE HOLDING" means Brooke Group Holding, Inc. a Delaware corporation, and any successor thereto.

"COLLATERAL" has the meaning ascribed thereto in Section 3 hereof.

"COMMERCIAL TORT CLAIM" means (i) prior to July 1, 2001, a claim of Brooke Holding arising in tort and (ii) on or after July 1, 2001, has the meaning ascribed thereto in the Uniform Commercial Code.

"COPYRIGHT COLLATERAL" means all Copyrights, whether now owned or hereafter acquired by Brooke Holding.

"COPYRIGHTS" means all copyrights, copyright registrations and applications for copyright registrations, including, without limitation, all renewals, extensions, income, royalties, damages and payments now or hereafter due and/or payable under and with respect thereto, including, without limitation, damages and payments for past or future infringements thereof, the right to sue for past, present and future infringements thereof, and all rights corresponding thereto throughout the world.

"DEPOSIT ACCOUNT" means any deposit account (as defined in the Uniform Commercial Code).

"DOCUMENTS" has the meaning ascribed thereto in Section 3(k) hereof.

"ENVIRONMENTAL LAWS" means any and all present and future Federal, state, local and foreign laws, rules or regulations, and any orders or decrees, in each case as now or hereafter in effect, relating to the regulation or protection of human health, safety or the environment or to emissions, discharges, releases or threatened releases of pollutants, contaminants, chemicals or toxic or hazardous substances or wastes into the indoor or outdoor environment, including, without limitation, ambient air, soil, surface water, ground water, wetlands, land or subsurface strata, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of pollutants, contaminants, chemicals or toxic or hazardous substances or wastes.

"EQUIPMENT" has the meaning ascribed thereto in Section 3(i) hereof.

"EQUITY INTERESTS" means, with respect to any Person, any capital stock of such Person and shares, interests, participations or other ownership interests (however designated) of any Person and any rights (other than debt securities convertible into corporate stock), warrants and options to purchase any of the foregoing, including (without limitation) each class of common stock and preferred stock of such Person if such Person is a corporation and each general and limited partnership interest of such Person if such Person is a partnership.

"FARM PRODUCTS" has the meaning ascribed thereto in the Uniform Commercial Code.

"INITIAL PLEDGED LIGGETT SHARES" means all Equity Interests of Liggett represented by the certificates of Liggett identified in Annex 1 hereto.

"INSTRUMENTS" has the meaning ascribed thereto in Section 3(e) hereof.

"INTELLECTUAL PROPERTY" means, collectively, all Copyright Collateral, all Patent Collateral and all Trademark Collateral.

"INVENTORY" has the meaning ascribed thereto in Section 3(g) hereof.

"ISSUERS" means, collectively, the respective corporations identified in Annex 1 hereto under the caption "Issuer".

"LIGGETT" means Liggett Group, Inc. a Delaware corporation, and any successor thereto.

"LIGGETT CONVERTIBLE SECURITIES" means any securities or rights that are convertible into or exchangeable for Equity Interests of Brooke Holding.

"MOTOR VEHICLES" means motor vehicles, tractors, trailers and other like property whether or not the title thereto is governed by a certificate of title or ownership.

"NEW VALLEY" means New Valley Corporation, a Delaware corporation, and any successor thereto. $\,$

"NOTES" has the meaning ascribed thereto in the first paragraph hereof.

"NV HOLDINGS" means New Valley Holdings, Inc., a Delaware corporation, and any successor thereto.

"PATENT COLLATERAL" means all Patents, whether now owned or hereafter acquired by Brooke Holding.

"PATENTS" means all patents and patent applications, including, without limitation, the inventions and improvements described and claimed therein together with the reissues, divisions, continuations, renewals, extensions and continuations-in-part thereof, all income, royalties, damages and payments now or hereafter due and/or payable under and with respect thereto, including, without limitation, damages and payments for past or future infringements thereof, the right to sue for past, present and future infringements thereof, and all rights corresponding thereto throughout the world.

"PERSON" means an individual, partnership, corporation, limited liability company, association, trust, unincorporated organization, or a government or agency or political subdivision thereof.

"PLEDGED STOCK" has the meaning ascribed thereto in Section 3(a) hereof. $\,$

"RESEARCH" means Vector Research Ltd., a Delaware corporation, and any successor thereto.

"SECURED OBLIGATIONS" means, collectively, all obligations and liabilities of any kind or nature, present or future, absolute or contingent, of (i) the Company arising under the Note Documents, (ii) NV Holdings arising under the NV Holdings Pledge Agreement or any other undertaking or agreement delivered by NV Holdings in connection with any other Note Document, (iii) VTUSA arising under any undertaking or agreement delivered by VTUSA in connection with any Note Document, (iv) Brooke Overseas arising under any undertaking or agreement delivered by Brooke Overseas in connection with any Note Document, (v) New Valley arising under any undertaking or agreement delivered by New Valley in connection with any Note Document, (vi) Brooke Holding arising under this Agreement or any other undertaking or agreement delivered by Brooke Holding, (vii) Research arising under any undertaking, or agreement delivered by Research in connection with any Note Document, (viii) Vector arising under the Vector Pledge Agreement or any other undertaking or agreement delivered by Vector in connection with any other Note Document or (ix) Liggett arising under any undertaking or agreement delivered by Liggett in connection with any Note Document.

"STOCK COLLATERAL" means, collectively, the Collateral described in clauses (a) through (c) of Section 3 hereof and the proceeds of and to any such property and, to the extent related to any such property or such proceeds, all books, correspondence, credit files, records, invoices and other papers.

"TRADEMARK COLLATERAL" means all Trademarks, whether now owned or hereafter acquired by Brooke Holding. Notwithstanding the foregoing, the

Trademark Collateral does not and shall not include any Trademark that would be rendered invalid, abandoned, void or unenforceable by reason of its being included as part of the Trademark Collateral.

"TRADEMARKS" means all trade names, trademarks and service marks, logos, trademark and service mark registrations, and applications for trademark and service mark registrations, including, without limitation, all renewals of trademark and service mark registrations, all rights corresponding thereto throughout the world, the right to recover for all past, present and future infringements thereof, all other rights of any kind whatsoever accruing thereunder or pertaining thereto, together, in each case, with the product lines and goodwill of the business connected with the use of, and symbolized by, each such trade name, trademark and service mark.

"UNIFORM COMMERCIAL CODE" means the Uniform Commercial Code as in effect from time to time in the State of New York.

"U.S. LEGAL TENDER" means such coin or currency of the United States of America as at the time of payment is legal tender for the payment of public and private debts.

"VECTOR" means Vector Group Ltd., a Delaware corporation and any successor thereto.

"VTUSA" means Vector Tobacco (USA) Ltd., a Delaware corporation, and any successor thereto.

Section 2. REPRESENTATIONS AND WARRANTIES. Brooke Holding represents and warrants and covenants to the Collateral Agent for the benefit of the Holders and the Collateral Agent that:

- (a) Brooke Holding is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation;
- (b) Brooke Holding has the corporate power and authority and the legal right to execute and deliver, to perform its obligations under, and to grant the security interest in the Collateral pursuant to, this Agreement and has taken all necessary corporate action to authorize its execution, delivery and performance of, and grant of the security interest in the Collateral pursuant to, this Agreement;
- (c) this Agreement constitutes a legal, valid and binding obligation of Brooke Holding, enforceable in accordance with its terms, except as enforceability may be affected by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing;

- (d) the execution, delivery and performance by Brooke Holding of this Agreement will not (i) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of Brooke Holding under, any indenture, mortgage, deed of trust, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other agreement or instrument to which Brooke Holding is bound or by which Brooke Holding or any of its properties is bound or affected, (ii) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental or Regulatory Authority applicable to Brooke Holding or (iii) violate any provision of any statute or other rule or regulation of any Governmental or Regulatory Authority applicable to Brooke Holding;
- (e) no consent, approval or authorization of, or registration, filing (other than the filing of any financing statements contemplated in any Note Document) or declaration with, any Governmental or Regulatory Authority is required in connection with the execution, delivery or performance by Brooke Holding of this Agreement;
- (f) no litigation, investigation or proceeding of or before any arbitrator or Governmental or Regulatory Authority is pending or, to the knowledge of Brooke Holding, threatened by or against Brooke Holding against any of its properties or revenues with respect to this Agreement or any of the transactions contemplated hereby;
- (g) except for the Security Interest, Brooke Holding is the sole legal and equitable owner of the Collateral, holds the same free and clear of all Liens, charges, encumbrances and security interests of any kind and nature and will make no other assignment, pledge, mortgage, hypothecation, transfer or other disposition of the Collateral, other than the sale or application of the Collateral in compliance with the provisions of Sections 8.24 and 13.8 of the Note Purchase Agreement;
- (h) Brooke Holding has good, right and legal title to the Collateral and will defend its title thereto against the claims of all Persons whomsoever and will maintain and preserve the Security Interest as long as this Agreement shall remain in full force and effect;
- (i) no financing statement or other public notice with respect to all or any part of the Collateral is on file or of record in any public office, except as have been filed by the Collateral Agent pursuant to this Agreement or as are permitted by the Note Purchase Agreement;
- (j) except for the filing of UCC-1 financing statements, no consent or approval of, or other action by, and no notice to or filing with, any Governmental or Regulatory Authority or securities exchange, was or is necessary as a condition (i) to the validity of the pledge

provided for herein or for the execution, delivery or performance of this Agreement by Brooke Holding or (ii) other than filings with (A) the United States Patent and Trademark Office in connection with the assignment of the Patents, Trademarks and Copyrights or (B) the Securities Act, and regulations thereunder and any applicable blue sky laws, for the exercise by the Collateral Agent of the voting or other rights provided for in this Agreement or the remedies in respect of the Collateral pursuant to this Agreement;

- (k) as of the date hereof: (i) the Initial Pledged Liggett Shares are the only issued and outstanding shares of capital stock of Liggett and (ii) there are no issued and outstanding Liggett Convertible Securities, and Liggett is not subject to any obligation, contingent or otherwise, to issue in the future any additional Equity Interests or any such Liggett Convertible Securities;
- (1) the Initial Pledged Liggett Shares are duly authorized, validly issued, fully paid and nonassessable;
- (m) except for the Note Documents, Brooke Holding is not now and will not become a party to any voting trust or other agreement or undertaking with respect to the exercise of the voting or consent rights associated with any of the Stock Collateral (other than an agreement or undertaking in respect of a sale of the Stock Collateral, which sale is or will be in compliance with the provisions of Section 13.8 of the Note Purchase Agreement), and there are no other restrictions on the exercise of such rights;
- (n) except for the Note Documents, Brooke Holding is not now and will not become a party to any agreement or undertaking pursuant to which any Issuer or any other Person could have the right to purchase any of the Stock Collateral (other than an agreement or undertaking in respect of a sale of the Stock Collateral, which sale is or will be in compliance with the provisions of Section 13.8 of the Note Purchase Agreement), and at the time of delivery of any Stock Collateral to the Collateral Agent there will not be (and Brooke Holding will not thereafter permit to exist) any other restrictions on the transfer of the Stock Collateral;
- (o) except for the Note Documents, Brooke Holding has the right to vote, pledge and grant a security interest in, or otherwise transfer, the Collateral free of any liens (except as set forth in (g) above):
- (p) the security interests granted pursuant to this Agreement (i) upon completion of the filings and other actions specified on Annex 2 (which, in the case of all filings and other documents referred to on said Annex, have been delivered to the Collateral Agent in duly executed form) will constitute valid perfected security interests in all of the Collateral in favor of the Collateral Agent enforceable in

accordance with the terms hereof against all creditors of Brooke Holding and all Persons purporting to purchase any Collateral from Brooke Holding and (ii) are prior to all other Liens on the Collateral in existence on the date hereof except for Liens permitted by the Note Purchase Agreement which have priority over the Liens on the Collateral held by the Collateral Agent by operation of law;

- (q) the grant and perfection of the Security Interest in the Stock Collateral for the benefit of the Collateral Agent and the Holders, in accordance with the terms hereof, will not violate the registration requirements of the Securities Act, any provisions of any other applicable federal, state or foreign securities laws, or any applicable general corporation law or other applicable law;
- (r) each Issuer is a corporation duly organized, validly existing and in good standing under the laws of the state of its incorporation and has full power and authority to own or lease its properties and assets and to carry on its business;
- (s) the stock powers delivered to the Collateral Agent with respect to the Pledged Stock have been, and at the time of the delivery of any other stock powers to the Collateral Agent in connection with the pledge of any additional Stock Collateral such stock powers will be, duly executed;
- (t) the pledge of the Initial Pledged Shares pursuant to this Agreement, together with the transfer of such Initial Pledged Shares to the Collateral Agent, creates a valid and perfected first priority security interest in the Stock Collateral, in favor of the Collateral Agent for the benefit of the Collateral Agent and the Holders, securing the due and punctual payment and performance in full of the Secured Obligations in accordance with the respective terms thereof;
- (u) Brooke Holding is incorporated in the State of Delaware and the location of its chief executive office or sole place of business on the date hereof is specified on Annex 3;
- (v) none of the Collateral consists of, or is the proceeds (as defined in the Uniform Commercial Code) of, Farm Products;
- (w) on the date hereof, the Inventory and the Equipment (other than Motor Vehicles) are kept at the locations listed on Annex 3;

(x)

(i) Annex 4 lists all Intellectual Property owned by Brooke Holding in its own name on the date hereof;

- (ii) On the date hereof, all material Intellectual Property is valid, subsisting, unexpired and enforceable, has not been abandoned and does not infringe the intellectual property rights of any other Person;
- (iii) Except as set forth in Annex 4, on the date hereof, none of the Intellectual Property is the subject of any licensing or franchise agreement pursuant to which Brooke Holding is the licensor or franchisor;
- (iv) No holding, decision or judgment has been rendered by any Governmental or Regulatory Authority which would limit, cancel or question the validity of, or Brooke Holding's rights in, any Intellectual Property in any respect; and
- (v) No action or proceeding is pending or, to the knowledge of Brooke Holding threatened, or the date hereof seeking to limit, cancel or question the validity of any Intellectual Property or Brooke Holding's ownership interest therein;
- (y) subject to Section 4.01(b), all Instruments owned by Brooke Holding are listed on Annex 5 and have been delivered to the Collateral Agent;
- (z) each Instrument which is a promissory note issued to and held by Brooke Holding (including any promissory note evidencing loans made by Brooke Holding to any of its Affiliates or Associates, any Vector Expanded Affiliate or any Group Executive) constitutes the legal, valid and binding obligation of the obligor with respect thereto, enforceable in accordance with its terms, subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing;
- (aa) any instruments of assignment and transfer delivered pursuant to Section 4.01(b) have been, or will be, duly executed;
- (bb) all Deposit Accounts of Brooke Holding are listed on Annex 6; $\,$
 - (cc) Brooke Holding has no Commercial Tort Claims; and
- (dd) Brooke Holding shall not sell, assign, transfer or otherwise dispose of, pledge, hypothecate or otherwise encumber, any of its Equity Interests in Liggett which are pledged pursuant to this Agreement, unless any such sale, assignment, transfer or disposition is

in compliance with provisions of Section 10.1 and Section 13.8 of the Note Purchase Agreement.

Section 3. COLLATERAL. As collateral security for the prompt payment in full when due (whether at stated maturity, by acceleration or otherwise) and performance of the Secured Obligations, Brooke Holding hereby pledges and grants to the Collateral Agent for the benefit of the Holders and the Collateral Agent a security interest in all of Brooke Holding's right, title and interest in the following property, whether now owned by Brooke Holding or hereafter acquired and whether now existing or hereafter coming into existence (all being collectively referred to herein as "COLLATERAL"):

- (a) the shares of capital stock and other Equity Interests of the Issuers represented by the certificates identified in Annex 1 hereto and all Equity Interests of any other Person, now or hereafter owned by Brooke Holding, in each case together with the certificates evidencing the same (collectively, the "PLEDGED STOCK");
- (b) all shares, securities, moneys or property representing a dividend on any of the Pledged Stock, or representing a distribution or return of capital upon or in respect of the Pledged Stock, or resulting from a split-up, revision, reclassification or other like change of the Pledged Stock or otherwise received in exchange therefor, and any subscription warrants, rights or options issued to the holders of, or otherwise in respect of, the Pledged Stock;
- (c) without affecting the obligations of Brooke Holding under any provision prohibiting such action hereunder or under the Note Purchase Agreement, in the event of any consolidation or merger in which an Issuer or other Subsidiary of Brooke Holding is not the surviving corporation, all shares of each class of the capital stock and all other Equity Interests of the successor corporation (unless such successor corporation is Brooke Holding itself) formed by or resulting from such consolidation or merger (the Pledged Stock, together with all other certificates, shares, securities, properties or moneys as may from time to time be pledged hereunder pursuant to clause (a) or (b) above and this clause (c) being herein collectively called the "STOCK COLLATERAL");
- (d) all accounts and general intangibles (each as defined in the Uniform Commercial Code) of Brooke Holding constituting any right to the payment of money, including (but not limited to) all moneys due and to become due to Brooke Holding in respect of any loans or advances or for Inventory or Equipment or other goods sold or leased or for services rendered, all moneys due and to become due to Brooke Holding

under any guarantee (including a letter of credit) of the purchase price of Inventory or Equipment sold by Brooke Holding and all tax refunds (such accounts, general intangibles and moneys due and to become due being herein called collectively "ACCOUNTS");

- (e) all instruments, chattel paper or letters of credit (each as defined in the Uniform Commercial Code) of Brooke Holding evidencing, representing, arising from or existing in respect of, relating to, securing or otherwise supporting the payment of, any of the Accounts, including (but not limited to) promissory notes, drafts, bills of exchange and trade acceptances (herein collectively called "INSTRUMENTS");
 - (f) all Deposit Accounts of Brooke Holding;
- (g) all inventory (as defined in the Uniform Commercial Code) of Brooke Holding, all goods obtained by Brooke Holding in exchange for such inventory, and any products made or processed from such inventory including all substances, if any, commingled therewith or added thereto (herein collectively called "INVENTORY");
- (h) all Intellectual Property and all other accounts or general intangibles not constituting Intellectual Property or Accounts;
- (i) all equipment (as defined in the Uniform Commercial Code) of Brooke Holding, including all Motor Vehicles (herein collectively called "EQUIPMENT");
- (j) each contract and other agreement of Brooke Holding relating to the sale or other disposition of Inventory or Equipment;
- (k) all documents of title (as defined in the Uniform Commercial Code) or other receipts of Brooke Holding covering, evidencing or representing Inventory or Equipment (herein collectively called "DOCUMENTS");
- (1) all rights, claims and benefits of Brooke Holding against any Person arising out of, relating to or in connection with Inventory or Equipment purchased by Brooke Holding, including, without limitation, any such rights, claims or benefits against any Person storing or transporting such Inventory or Equipment;
 - (m) all Commercial Tort Claims of Brooke Holding;
- (n) all securities accounts (as defined in Article 8 of the Uniform Commercial Code) and the financial assets (as defined in Article 8 of the Uniform Commercial Code) credited from time to time thereto; and
- (o) all other tangible and intangible personal property and fixtures of Brooke Holding, including, without limitation, all proceeds, products, offspring, accessions, rents, profits, income, benefits, substitutions and replacements of and to any of the property

of Brooke Holding described in the preceding clauses of this Section 3 (including, without limitation, any proceeds of insurance thereon and all causes of action, claims and warranties now or hereafter held by Brooke Holding in respect of any of the items listed above) and, to the extent related to any property described in said clauses or such proceeds, products and accessions, all books, correspondence, credit files, records, invoices and other papers, including without limitation all tapes, cards, computer runs and other papers and documents in the possession or under the control of Brooke Holding or any computer bureau or service company from time to time acting for Brooke Holding.

Section 4. FURTHER ASSURANCES; REMEDIES. In furtherance of the grant of the pledge and security interest pursuant to Section 3 hereof, Brooke Holding hereby agrees with the Collateral Agent as follows:

4.01. DELIVERY AND OTHER PERFECTION. The Company shall:

- (a) if any of the shares, securities, moneys or property required to be pledged by Brooke Holding under clauses (a), (b) and (c) of Section 3 hereof are received by Brooke Holding, hold such Property in trust and forthwith either (x) transfer and deliver to the Collateral Agent such shares or securities so received by Brooke Holding (together with the certificates for any such shares and securities duly endorsed in blank or accompanied by undated stock powers duly executed in blank, all in form and substance reasonably satisfactory to the Collateral Agent), all of which thereafter shall be held by the Collateral Agent (except in the case of cash dividends which shall be transferred to the Company and held in a Company Account in accordance with the Note Purchase Agreement), pursuant to the terms of this Agreement, as part of the Collateral or (y) take such other action as may be necessary or appropriate or as the Collateral Agent may reasonably request to duly record the Lien created hereunder in such shares, securities, moneys or property in said clauses (a), (b) and (c);
- (b) deliver and pledge to the Collateral Agent any and all Instruments pledged hereunder, endorsed and/or accompanied by such instruments of assignment and transfer in such form and substance as may be required to perfect the pledge thereof hereunder or as the Collateral Agent may reasonably request; PROVIDED, that so long as no Default shall have occurred and be continuing, Brooke Holding may retain for collection in the ordinary course any Instruments received by Brooke Holding in the ordinary course of business and the Collateral Agent shall, promptly upon request of Brooke Holding, make appropriate arrangements for making any Instrument pledged by Brooke Holding hereunder available to Brooke Holding for purposes of presentation, collection or renewal (any such arrangement to be effected, to the extent deemed appropriate by the Collateral Agent, against trust receipt or like document);

- (c) give, execute, deliver, file and/or record any financing statement, continuation statement, notice, instrument, document, agreement or other papers that may be necessary or desirable or that the Collateral Agent may reasonably request to create, preserve, perfect or validate the security interest granted pursuant hereto or to enable the Collateral Agent to exercise and enforce its rights hereunder with respect to such pledge and security interest, including, without limitation, causing any or all of the Stock Collateral to be transferred of record into the name of the Collateral Agent or its nominee (and the Collateral Agent agrees that if any Stock Collateral is transferred into its name or the name of its nominee, the Collateral Agent will thereafter promptly give to Brooke Holding copies of any notices and communications received by it with respect to the Stock Collateral), PROVIDED that notices to account debtors in respect of any Accounts or Instruments shall be subject to the provisions of clause (g) below;
- (d) upon the acquisition after the date hereof by Brooke Holding of any Equipment covered by a certificate of title or ownership, cause the Collateral Agent to be listed as the lienholder on such certificate of title and within 120 days of the acquisition thereof deliver evidence of the same to the Collateral Agent;
- (e) keep full and accurate books and records relating to the Collateral, and stamp or otherwise mark such books and records in such manner as may be necessary to reflect the security interests granted by this Agreement;
- (f) permit representatives of the Collateral Agent, upon reasonable notice, at any time during normal business hours to inspect and make abstracts from its books and records pertaining to the Collateral, and permit representatives of the Collateral Agent to be present at Brooke Holding's place of business to receive copies of all communications and remittances relating to the Collateral, and forward copies of any notices or communications received by Brooke Holding with respect to the Collateral, all to the extent and in such manner as the Collateral Agent may reasonably request; and
- (g) upon the occurrence and during the continuance of any Default, upon request of the Collateral Agent, promptly notify (and Brooke Holding hereby authorizes the Collateral Agent so to notify) each account debtor in respect of any Accounts or Instruments that such Collateral has been assigned to the Collateral Agent hereunder, and that any payments due or to become due in respect of such Collateral are to be made directly to the Securities Account.
- 4.02. OTHER FINANCING STATEMENTS AND LIENS. Without the prior written consent of the Collateral Agent, Brooke Holding shall not file or suffer to be on file, or authorize or permit to be filed or to be on file, in any

jurisdiction, any financing statement or like instrument with respect to the Collateral in which the Collateral Agent is not named as the sole secured party.

4.03. PRESERVATION OF RIGHTS. The Collateral Agent shall not be required to take steps necessary to preserve any rights against third parties to any of the Collateral.

4.04. SPECIAL PROVISIONS RELATING TO STOCK COLLATERAL.

- (a) So long as no Event of Default shall have occurred and be continuing, Brooke Holding shall have the right to exercise all voting, consensual and other powers of ownership pertaining to the Stock Collateral for all purposes not inconsistent with the terms of this Agreement, the Note Purchase Agreement, the Notes or any other instrument or agreement referred to herein or therein, PROVIDED that Brooke Holding agrees that it will not vote the Stock Collateral in any manner that is inconsistent with the terms of any Note Document or any such other related instrument or agreement; and the Collateral Agent shall execute and deliver to Brooke Holding or cause to be executed and delivered to Brooke Holding all such proxies, powers of attorney, dividend and other orders, and all such instruments, without recourse, as Brooke Holding may reasonably request for the purpose of enabling Brooke Holding to exercise the rights and powers that it is entitled to exercise pursuant to this Section 4.04(a).
- (b) If any Account Notice Event shall have occurred, then so long as such Account Notice Event shall continue, and whether or not the Collateral Agent exercises any available right to declare any Secured Obligation due and payable or seeks or pursues any other relief or remedy available to it under applicable law or under this Agreement, the Note Purchase Agreement, the Notes or any other agreement relating to such Secured Obligation, to the extent not already delivered pursuant to Section 4.01, all dividends and other distributions on the Stock Collateral shall be paid directly to the Collateral Agent and deposited by it into the Securities Account, subject to the terms of this Agreement, and, if the Collateral Agent shall so request in writing, Brooke Holding agrees to execute and deliver to the Collateral Agent appropriate additional dividend, distribution and other orders and documents to that end.
- 4.05. EVENTS OF DEFAULT, ETC. Subject to the last sentence of this Section 4.05, during the period during which (i) with respect to any cash contained in the Brooke Holding Checking Account and all cash dividends and cash proceeds of any of the Stock Collateral, an Account Notice Event shall have occurred and be continuing and (ii) with respect to all other Collateral, an Event of Default shall have occurred and be continuing:
 - (a) Brooke Holding shall, at the request of the Collateral Agent, assemble the Collateral owned by it at such place or places, reasonably convenient to both the Collateral Agent and Brooke Holding, designated in its request;

- (b) the Collateral Agent may make any reasonable compromise or settlement deemed desirable with respect to any of the Collateral and may extend the time of payment, arrange for payment in installments, or otherwise modify the terms of, any of the Collateral;
- (c) the Collateral Agent shall have all of the rights and remedies with respect to the Collateral of a secured party under the Uniform Commercial Code (whether or not the Uniform Commercial Code is in effect in the jurisdiction where the rights and remedies are asserted) and such additional rights and remedies to which a secured party is entitled under the laws in effect in any jurisdiction where any rights and remedies hereunder may be asserted, including, without limitation, the right, to the maximum extent permitted by law, to exercise all voting, consensual and other powers of ownership pertaining to the Collateral as if the Collateral Agent were the sole and absolute owner thereof (and Brooke Holding agrees to take all such action as may be appropriate to give effect to such right);
- (d) the Collateral Agent in its discretion may, in its name or in the name of Brooke Holding or otherwise, demand, sue for, collect or receive any money or property at any time payable or receivable on account of or in exchange for any of the Collateral, but shall be under no obligation to do so; and
- (e) the Collateral Agent may, upon ten business days' prior written notice to Brooke Holding of the time and place, with respect to the Collateral or any part thereof that shall then be or shall thereafter come into the possession, custody or control of the Collateral Agent or any of its agents, sell, lease, assign or otherwise dispose of all or any part of such Collateral, at such place or places as the Collateral Agent deems best, and for cash or for credit or for future delivery (without thereby assuming any credit risk), at public or private sale, without demand of performance or notice of intention to effect any such disposition or of the time or place thereof (except such notice as is required above or by applicable statute and cannot be waived), and the Collateral Agent or anyone else may be the purchaser, lessee, assignee or recipient of any or all of the Collateral so disposed of at any public sale (or, to the extent permitted by law, at any private sale) and thereafter hold the same absolutely, free from any claim or right of whatsoever kind, including any right or equity of redemption (statutory or otherwise), of Brooke Holding, any such demand, notice and right or equity being hereby expressly waived and released. The Collateral Agent may, without notice or publication, adjourn any public or private sale or cause the same to be adjourned from time to time by announcement at the time and place fixed for the

sale, and such sale may be made at any time or place to which the sale may be so adjourned.

The proceeds of each collection, sale or other disposition under this Section 4.05 shall be applied in accordance with Section 4.13 hereof.

4.06. REGISTRATION RIGHTS.

- (a) If the Collateral Agent shall determine to exercise its right to sell any or all of the Stock Collateral, and if in the opinion of the Collateral Agent it is necessary or advisable to have the Stock Collateral, or that portion thereof to be sold, registered under the provisions of the Securities Act, Brooke Holding will cause the issuer thereof to (i) execute and deliver, and cause the directors and officers of such issuer to execute and deliver, all such instruments and documents, and do or cause to be done all such other acts as may be, in the opinion of the Collateral Agent, necessary or advisable to register the Stock Collateral, or that portion thereof to be sold, under the provisions of the Securities Act, (ii) use its best efforts to cause the registration statement relating thereto to become effective and to remain effective for a period of one year from the date of the first public offering of the Stock Collateral, or that portion thereof to be sold and (iii) make all amendments thereto and/or to the related prospectus which, in the opinion of the Collateral Agent, are necessary or advisable, all in conformity with the requirements of the Securities Act and the rules and regulations of the Securities and Exchange Commission applicable thereto. The Company agrees to cause such issuer to comply with the provisions of the securities or "Blue Sky" laws of any and all jurisdictions which the Collateral Agent shall designate and to make available to its security holders, as soon as practicable, an earnings statement (which need not be audited) which will satisfy the provisions of Section 11(a) of the Securities Act.
- (b) The Company recognizes that the Collateral Agent may be unable to effect a public sale of any or all the Stock Collateral, by reason of certain prohibitions contained in the Securities Act and applicable state securities laws or otherwise, and may be compelled to resort to one or more private sales thereof to a restricted group of purchasers which will be obliged to agree, among other things, to acquire such securities for their own account for investment and not with a view to the distribution or resale thereof. The Company acknowledges and agrees that any such private sale may result in prices and other terms less favorable than if such sale were a public sale and, notwithstanding such circumstances, agrees that any such private sale shall be deemed to have been made in a commercially reasonable manner. The Collateral Agent shall be under no obligation to delay a sale of any of the Stock Collateral for the period of time necessary to permit the issuer thereof to register such securities for public sale under the Securities Act, or under applicable state securities laws, even if such issuer would agree to do so.

- (c) The Company agrees to use its best efforts to do or cause to be done all such other acts as may be necessary to make such sale or sales of all or any portion of the Stock Collateral pursuant to this Section 4.06 valid and binding and in compliance with any and all other applicable law. The Company further agrees that a breach of any of the covenants contained in this Section 4.06 will cause irreparable injury to the Collateral Agent and the Holders, that the Collateral Agent and the Holders have no adequate remedy at law in respect of such breach and, as a consequence, that each and every covenant contained in this Section 4.06 shall be specifically enforceable against Brooke Holding, and Brooke Holding hereby waives and agrees not to assert any defenses against an action for specific performance of such covenants except for a defense that no Event of Default has occurred under the Note Purchase Agreement.
- 4.07. DEFICIENCY. If the proceeds of sale, collection or other realization of or upon the Collateral pursuant to Sections 4.05 or 4.06 hereof are insufficient to cover the costs and expenses of such realization and the payment in full of the Secured Obligations, Brooke Holding shall remain liable for any deficiency; PROVIDED, HOWEVER, that the maximum liability of Brooke Holding hereunder shall in no event exceed the amount of Secured Obligations which may be guaranteed by Brooke Holding under applicable laws relating to the insolvency of debtors.
- 4.08. REMOVALS, ETC. Without at least 30 days' prior written notice to the Collateral Agent, Brooke Holding shall not (i) maintain any of its books and records with respect to the Collateral at any office or maintain its principal place of business at any place, or permit any Inventory or Equipment to be located anywhere, other than at the address indicated beneath the signature of the Company to the Note Purchase Agreement or at one of the locations identified in Annex 3 hereto or in transit from one of such locations to another, (ii) change its name, or the name under which it does business, identity or corporate structure to such an extent that any financing statement filed in connection with this Agreement would become misleading and (iii) change its jurisdiction of organization or the location of its chief executive office or sole place of business from that identified in Annex 3.

4.09. MAINTENANCE OF INSURANCE.

(a) At any time Brooke Holding holds Material amounts of Inventory and Equipment, Brooke Holding shall maintain, with financially sound and reputable companies, insurance policies (i) insuring the Inventory and Equipment (including Motor Vehicles) against loss by fire, explosion, theft and such other casualties as may be reasonably satisfactory to the Required Holders and (ii) insuring Brooke Holding, the Collateral Agent and the Holders against liability for personal injury and property damage relating to such Inventory and Equipment (including Motor Vehicles), such policies to be in such form and amounts and having such coverage as may be reasonably satisfactory to the Collateral Agent and the Required Holders.

- (b) All such insurance shall (i) provide that no cancellation, material reduction in amount or material change in coverage thereof shall be effective until at least 30 days after receipt by the Collateral Agent of written notice thereof, (ii) name the Collateral Agent as insured party or loss payee, (iii) if reasonably requested by the Collateral Agent, include a breach of warranty clause and (iv) be reasonably satisfactory in all other respects to the Collateral Agent.
- (c) Brooke Holding shall deliver to the Collateral Agent and the Holders a report of a reputable insurance broker with respect to such insurance substantially concurrently with each delivery of Brooke Holding's audited annual financial statements and such supplemental reports with respect thereto as the Collateral Agent may from time to time reasonably request.

4.10. PAYMENT OF OBLIGATIONS. Brooke Holding shall pay and discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all taxes, assessments and governmental charges or levies imposed upon the Collateral or in respect of income or profits therefrom, as well as all claims of any kind (including, without limitation, claims for labor, materials and supplies) against or with respect to the Collateral, except that no such charge need be paid if the amount or validity thereof is currently being contested in good faith by appropriate proceedings, reserves in conformity with GAAP with respect thereto have been provided on the books of Brooke Holding and such proceedings could not reasonably be expected to result in the sale, forfeiture or loss of any material portion of the Stock Collateral or any interest therein. In its reasonable discretion, the Collateral Agent may discharge taxes and other encumbrances at any time levied or placed on any of the Collateral, make repairs thereto and pay any necessary filing fees. Brooke Holding agrees to reimburse the Collateral Agent on demand for any and all expenditures so made. The Collateral Agent shall have no obligation to Brooke Holding to make any such expenditures, nor shall the making thereof relieve Brooke Holding of any default.

Notwithstanding anything to the contrary contained herein, Brooke Holding shall remain liable under each contract or agreement comprised in the Collateral to be observed or performed by Brooke Holding thereunder. The Collateral Agent shall not have any obligation or liability under any such contract or agreement by reason of or arising out of this Agreement or the receipt by the Collateral Agent of any payment relating to any of the Collateral, nor shall the Collateral Agent be obligated in any manner to perform any of the obligations of Brooke Holding under or pursuant to any such contract or agreement, to make inquiry as to the nature or sufficiency of any payment received by the Collateral Agent in respect of the Collateral or as to the sufficiency of any performance by any party under any such contract or agreement, to present or file any claim, to take any action to enforce any performance or to collect the payment of any amounts which may have been assigned to the Collateral Agent or to which the Collateral Agent may be entitled at any time or times. The Collateral Agent's sole duty with respect to the custody, safe keeping, and physical preservation of the Collateral in its possession, under the Uniform Commercial Code of the applicable jurisdiction or otherwise, shall be to deal with such Collateral in the same manner as the

Collateral Agent deals with similar property for its own account.

- 4.11. NOTICES. Brooke Holding shall advise the Collateral Agent and the Holders promptly, in reasonable detail, of:
 - (a) any Lien (other than security interests created hereby or Liens permitted under the Note Purchase Agreement) on any of the Collateral which would adversely affect the ability of the Collateral Agent to exercise any of its remedies hereunder; and
 - (b) of the occurrence of any other event which could reasonably be expected to have a material adverse effect on the aggregate value of the Collateral or on the security interests created hereby.
- 4.12. PRIVATE SALE. The Collateral Agent shall incur no liability as a result of the sale of the Collateral, or any part thereof, at any private sale pursuant to Section 4.05 or 4.06 hereof conducted in a commercially reasonable manner. Brooke Holding hereby waives any claims against the Collateral Agent arising by reason of the fact that the price at which the Collateral may have been sold at such a private sale was less than the price that might have been obtained at a public sale or was less than the aggregate amount of the Secured Obligations, even if the Collateral Agent accepts the first offer received and does not offer the Collateral to more than one offeree.
- 4.13. APPLICATION OF PROCEEDS. Except as otherwise herein expressly provided, the proceeds of any collection, sale or other realization of all or any part of the Collateral pursuant hereto, and any other cash at the time held by the Collateral Agent shall be applied by the Collateral Agent:

FIRST, to the payment of the costs and expenses of such collection, sale or other realization, including reasonable out-of-pocket costs and expenses of the Collateral Agent and the fees and expenses of its agents and counsel and other professional advisors, and all expenses incurred and advances made by the Collateral Agent in connection therewith;

NEXT, to the payment in full of the Secured Obligations; and

FINALLY, to the payment to Brooke Holding, or its successors or assigns, or as a court of competent jurisdiction may direct, of any surplus then remaining.

As used in this Section 4, "PROCEEDS" of Collateral shall mean cash, securities and other property realized in respect of, and distributions in kind of, Collateral, including any thereof received under any reorganization, liquidation or adjustment of debt of Brooke Holding or any issuer of or obligor on any of the Collateral.

4.14. ATTORNEY-IN-FACT. Without limiting any rights or powers granted by this Agreement to the Collateral Agent while no Event of Default has occurred and is continuing, upon the occurrence and during the continuance of any Event of Default the Collateral Agent is hereby appointed the attorney-in-fact of Brooke Holding for the purpose of carrying out the

provisions of this Section 4 and taking any action and executing any instruments that the Collateral Agent may deem necessary or advisable to accomplish the purposes hereof, which appointment as attorney-in-fact is irrevocable and coupled with an interest. Without limiting the generality of the foregoing, so long as the Collateral Agent shall be entitled under this Section 4 to make collections in respect of the Collateral, the Collateral Agent shall have the right and power to receive, endorse and collect all checks made payable to the order of Brooke Holding representing any dividend, payment or other distribution in respect of the Collateral or any part thereof and to give full discharge for the same

- 4.15. TERMINATION. When all Secured Obligations shall have been paid in full, this Agreement shall terminate, and the Collateral Agent shall forthwith cause to be assigned, transferred and delivered, against receipt but without any recourse, warranty or representation whatsoever, any remaining Collateral and money received in respect thereof, to or on the order of Brooke Holding. The Collateral Agent shall also execute and deliver to Brooke Holding upon such termination such Uniform Commercial Code termination statements, certificates for terminating the Liens on the Motor Vehicles and such other documentation as shall be reasonably requested by Brooke Holding to effect the termination and release of the Liens on the Collateral.
- 4.16. FURTHER ASSURANCES. Brooke Holding agrees that, from time to time it will execute and deliver such further documents and do such other acts and things as may be necessary or appropriate or as the Collateral Agent may reasonably request in order fully to effect the purposes of this Agreement.
- 4.17. RELEASE OF MOTOR VEHICLES. So long as no Default shall have occurred and be continuing, upon the request of Brooke Holding, the Collateral Agent shall execute and deliver to Brooke Holding such instruments as Brooke Holding shall reasonably request to remove the notation of the Collateral Agent as lienholder on any certificate of title for any Motor Vehicle; PROVIDED that any such instruments shall be delivered, and the release effective only upon receipt by the Collateral Agent of a certificate from Brooke Holding stating that the Motor Vehicle the lien on which is to be released is to be sold or has suffered a casualty loss (with title thereto passing to the casualty insurance company therefor in settlement of the claim for such loss).
- ${\hbox{4.18. INDEBTEDNESS. Brooke Holding shall not incur any }} \\$ Indebtedness.
- 4.19. RECEIPT OF CASH. Subject to Section 8.21(b) of the Note Purchase Agreement, Brooke Holding shall immediately dividend or otherwise transfer all Cash or cash equivalents it receives to the Company.
- 4.20. SALE OF LIGGETT EQUITY INTERESTS. Other than in accordance with the Note Purchase Agreement, Brooke Holding shall not sell any Equity Interests in Liggett.

- 4.21. WAIVER OF SURETYSHIP DEFENSES. To the extent this Agreement is deemed a guarantee, the liability of Brooke Holding hereunder shall be absolute, unconditional and irrevocable irrespective of, and without being lessened or limited by:
 - (a) the occurrence of any Event of Default under, or any lack of validity, legality or enforceability of any provision of any Note Document, or any other agreement or document;
 - (b) the failure of any Holder:
 - (i) to assert any claim or demand or to enforce any right or remedy against the Company or any other Person (including any other guarantor of the Secured Obligations) under the provisions of any Note Document, or otherwise, or
 - (ii) to exercise any right or remedy against any other guarantor of or other Person pledging collateral securing any of the Secured Obligations;
 - (c) any change in the time, manner or place of payment of, or in any term of, all or any of the Secured Obligations, or any other extension, compromise, indulgence or renewal of any Secured Obligation;
 - (d) any reduction, limitation, variation, impairment, discontinuance or termination of the Secured Obligations for any reason (other than by reason of any payment which is not required to be rescinded), including any claim of waiver, release, discharge, surrender, alteration or compromise, and shall not be subject to (and Brooke Holding hereby waives any right to or claim of) any defense or setoff, counterclaim, recoupment or termination whatsoever by reason of the invalidity, illegality, nongenuineness, irregularity, compromise, unenforceability of, or any other event or occurrence affecting, the Secured Obligations or otherwise (other than by reason of any payment which is not required to be rescinded);
 - (e) any amendment to, rescission, waiver or other modification of, or any consent to any departure from, any of the terms of the Secured Obligations or any guarantees or security;
 - (f) any addition, exchange, release, discharge, realization or non-perfection of any collateral security in respect of the Secured Obligations;
 - (g) any amendment to, rescission, waiver or other modification of, or release or addition of, or consent to any departure from, any other guarantee held by the Holders as security for any of the Secured Obligations;

- (h) the loss of or in respect of or the unenforceability of any other guarantee or other security which the Holders may now or hereafter hold in respect of the Secured Obligations, whether occasioned by the fault of the Holders or otherwise;
- (i) any change in the name of Brooke Holding or in the constitutive documents, capital structure, capacity or constitution of the Company, the bankruptcy or insolvency of the Company, the sale of any or all of the Company's business or assets or Brooke Holding being consolidated, merged or amalgamated with any other Person;
- (j) any failure on the part of Brooke Holding or any other Person to perform or comply with any term of the Note Purchase Agreement, the Notes, any of the Secured Obligations or any other agreement or document;
- (k) any suit or other action brought by any beneficiaries or creditors of, or by, Brooke Holding or any other Person for any reason whatsoever, including without limitation any suit or action in any way attacking or involving any issue, matter or thing in respect of any Note Document, any of the Secured Obligations or any other agreement or document;
- (1) any lack or limitation of status or of power, incapacity or disability of the Company or any trustee or agent thereof; or
- (m) any other circumstance (other than final and indefeasible payment in full) which might otherwise constitute a defense available to, or a legal or equitable discharge of, the Company, NV Holdings, or Vector or any other surety or guarantor.
- 4.22. RELEASE OF COLLATERAL. The security interest in any Collateral shall automatically and without any action by the Collateral Agent or the Holders be released upon any sale of such Collateral by Brooke Holding pursuant to the terms of the Note Purchase Agreement and this Agreement.

Section 5. MISCELLANEOUS.

- 5.01. NO WAIVER. No failure on the part of the Collateral Agent to exercise, and no course of dealing with respect to, and no delay in exercising, any right, power or remedy hereunder shall operate as a waiver thereof; nor shall any single or partial exercise by the Collateral Agent of any right, power or remedy hereunder preclude any other or further exercise thereof or the exercise of any other right, power or remedy. The remedies herein are cumulative and are not exclusive of any remedies provided by law.
- 5.02. NOTICES. All notices, requests, consents and demands hereunder shall be in writing and telecopied or delivered to the intended recipient at its "Address for Notices" specified pursuant to Section 19 of the Note Purchase Agreement (or, in the case of Brooke Holding, at the address

specified therein for the Company) and shall be deemed to have been given at the times specified in said Section 19.

- 5.03. AMENDMENTS, ETC. The terms of this Agreement may be waived, altered or amended only by an instrument in writing duly executed by Brooke Holding and the Collateral Agent in accordance with Section 18 of the Note Purchase Agreement. Any such amendment or waiver shall be binding upon the Collateral Agent, each holder of any of the Secured Obligations and Brooke Holding.
- 5.04. SUCCESSORS AND ASSIGNS. This Agreement shall be binding upon and inure to the benefit of the respective successors and assigns of Brooke Holding, the Collateral Agent and each holder of any of the Secured Obligations (PROVIDED, however, that Brooke Holding shall not assign or transfer its rights hereunder without the prior written consent of the Collateral Agent).
- 5.05. CAPTIONS. The captions and section headings appearing herein are included solely for convenience of reference and are not intended to affect the interpretation of any provision of this Agreement.
- 5.06. COUNTERPARTS. This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument and either of the parties hereto may execute this Agreement by signing any such counterpart. Delivery of an executed counterpart of a signature page of this Agreement by facsimile transmission shall be as effective as delivery of a manually executed counterpart of this Agreement.
- 5.07. GOVERNING LAW. This Agreement shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the law of the State of New York excluding choice-of-law principles of the law of such State that would require the application of the laws of a jurisdiction other than such State.
- 5.08. AGENTS AND ATTORNEYS-IN-FACT. The Collateral Agent may employ agents and attorneys-in-fact in connection herewith and shall not be responsible for the negligence or misconduct of any such agents or attorneys-in-fact selected by it in good faith.
- 5.09. SEVERABILITY. If any provision hereof is invalid and unenforceable in any jurisdiction, then, to the fullest extent permitted by law, (i) the other provisions hereof shall remain in full force and effect in such jurisdiction and shall be liberally construed in favor of the Collateral Agent in order to carry out the intentions of the parties hereto as nearly as may be possible and (ii) the invalidity or unenforceability of any provision hereof in any jurisdiction shall not affect the validity or enforceability of such provision in any other jurisdiction.
- 5.10. ENVIRONMENTAL INDEMNIFICATION OF COLLATERAL AGENT AND HOLDERS. Brooke Holding hereby agrees to indemnify the Collateral Agent and the Holders from, and hold the Collateral Agent and the Holders harmless against, any losses, liabilities, claims, damages or expenses arising under any Environmental Law as a result of the past, present or future operations of

Brooke Holding or any of its Subsidiaries following the exercise by the Collateral Agent of any of its rights and remedies under this Agreement.

- 5.11. INTEGRATION. This Agreement and the other Note Documents represent the agreement of Brooke Holding, the Collateral Agent and the Holders with respect to the subject matter hereof and thereof, and there are no promises, undertakings, representations or warranties by the Collateral Agent or any Holder relative to subject matter hereof and thereof not expressly set forth or referred to herein or in the other Note Documents.
 - 5.12. ACKNOWLEDGEMENTS. Brooke Holding hereby acknowledges

that:

- (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Note Documents to which it is a party;
- (b) neither the Collateral Agent nor any Holder has any fiduciary relationship with or duty to Brooke Holding arising out of or in connection with this Agreement or any of the other Note Documents, and the relationship between Brooke Holding, on the one hand, and the Collateral Agent and Holders, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and
- (c) no joint venture is created hereby or by the other Note Documents or otherwise exists by virtue of the transactions contemplated hereby among the Holders or among Brooke Holding and the Holders.
- 5.13. ADDITIONAL SUBSIDIARIES. Brooke Holding shall cause each new Subsidiary of Brooke Holding to execute and deliver an acknowledgment and undertaking substantially in the form of the acknowledgments and undertakings attached hereto and to do such other acts and things as the Collateral Agent may reasonably request in order fully to effect the purposes of this Agreement.

IN WITNESS WHEREOF, the parties hereto have caused this Pledge and Security Agreement to be duly executed and delivered as of the day and year $\frac{1}{2} \left(\frac{1}{2} \right)$ first above written.

BROOKE GROUP HOLDING INC.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen Title: Executive Vice President

UNITED STATES TRUST COMPANY OF NEW YORK, as Collateral Agent

/s/ PATRICIA GALLAGHER

Name: Patricia Gallagher Title: Assistant Vice President

ACKNOWLEDGMENT AND UNDERTAKING

FOR GOOD AND VALUABLE CONSIDERATION, the receipt and sufficiency of which are hereby acknowledged, Liggett Group, Inc. a Delaware corporation ("LIGGETT") hereby acknowledges receipt of a copy of the foregoing Pledge and Security Agreement dated as of May 14, 2001 and agrees as follows (with capitalized terms used and not otherwise defined below that are defined in such Pledge and Security Agreement having the meanings ascribed to such terms therein):

- 1. Liggett will promptly note on its books the security interests granted under such Pledge and Security Agreement in the applicable Pledged Stock.
- 2. Liggett hereby waives any rights or requirement at any time hereafter to receive a copy of such Pledge and Security Agreement in connection with the registration of any Stock Collateral in the name of the Collateral Agent or its nominee or the exercise of voting rights by the Collateral Agent.
- 3. Liggett agrees (a) to take all actions contemplated to be taken by it under the Pledge and Security Agreement, (b) to comply with all provisions thereof and (c) not to participate in any transaction that is inconsistent with its obligations hereunder or thereunder.

The undersigned hereby represents and warrants that (a) the execution, delivery and performance of this Acknowledgment and Undertaking has been duly authorized by all necessary action on the part of the undersigned, (b) the undersigned has duly executed and delivered this Acknowledgment and Undertaking and (c) this Acknowledgment and Undertaking is its legal, valid and binding obligation, enforceable against the undersigned in accordance with its terms. This Acknowledgment and Undertaking shall be governed by, and construed in accordance with, the law of the State of New York.

This Acknowledgment and Undertaking may not be terminated, revoked, amended or otherwise modified without the prior written consent of the Collateral Agent, which is an express third-party beneficiary hereof.

Dated: May 14, 2001

LIGGETT GROUP INC.

By: /s/ CHARLES M. KINGAN

Name: Charles M. Kingan Title: Vice President & CFO

PLEDGED STOCK

ISSUER	CERTIFICATE NOS.	REGISTERED OWNER	NUMBER OF SHARES
Liggett Group, Inc.	2	Brooke Group Holding, Inc.	1,000 shares of common stock, par value \$.10 per share.

FILINGS AND OTHER ACTIONS REQUIRED TO PERFECT SECURITY INTERESTS

UNIFORM COMMERCIAL CODE FILINGS

Delaware Secretary of State Florida Secretary of State

PATENT AND TRADEMARK FILINGS

None.

ACTIONS WITH RESPECT TO STOCK COLLATERAL

Delivery to Collateral Agents of certificates representing Pledged Stock in the State of New York along with a stock power endorsed in blank.

OTHER ACTIONS

CHIEF EXECUTIVE OFFICE

LIST OF LOCATIONS

100 S.E. Second Street Miami, Florida 33131 29

ANNEX 4

INTELLECTUAL PROPERTY

None.

30

ANNEX 5

INSTRUMENTS

None.

DEPOSIT ACCOUNTS

Account #03751459059 at Bank of America, N.A.

ACKNOWLEDGMENT AND PLEDGE AGREEMENT

ACKNOWLEDGEMENT AND PLEDGE AGREEMENT, dated as of May 14, 2001 (this "AGREEMENT"), by and among (i) Vector Group Ltd., a Delaware corporation ("VECTOR"), (ii) United States Trust Company of New York, a New York banking corporation, as collateral agent (together with its successors and assigns, the "COLLATERAL AGENT") on behalf of the holders (the "HOLDERS") of the 10% Senior Secured Notes Due March 31, 2006 of BGLS Inc. issued pursuant to that certain Note Purchase Agreement, dated as of May 14, 2001 (as amended, supplemented or modified from time to time) between the Company and the Holders and (iii) the Purchasers (as defined in the Note Purchase Agreement). Capitalized terms, unless otherwise defined herein, are used herein with the meanings ascribed to them in the Note Purchase Agreement.

- 1. ACKNOWLEDGEMENTS. Vector hereby acknowledges to the Holders that it has read and understood the Note Purchase Agreement, and hereby agrees, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, that it shall comply with Sections 7.5, 8.5, 8.25, 9.1, 9.3(b), 9.4 and 9.6 of the Note Purchase Agreement. Vector shall be liable to the Holders for any breach of such Sections of the Note Purchase Agreement to the extent of any benefit accruing directly to it as a result of such breach.
- 2. PLEDGE OF COLLATERAL. As collateral security for the prompt payment in full when due (whether at stated maturity, by acceleration or otherwise) and performance of the Secured Obligations (as defined in the BGLS Pledge Agreement), Vector hereby pledges and grants to the Collateral Agent for the benefit of the Holders and the Collateral Agent a security interest in all of Vector's right, title and interest in that certain \$25,000,000 Secured Revolving Demand Promissory Note, dated as of March 6, 2001, made by Vector Tobacco (USA) Ltd. and Vector Tobacco Ltd. in favor of BGLS Inc. and Vector (the "SECURED NOTE" or the "COLLATERAL").
- 3. NOTE POWER. Concurrently with the delivery to the Security Agent of the Secured Note, Vector shall deliver an undated note power covering the Secured Note, duly executed in blank by Vector.
 - 4. REPRESENTATIONS AND WARRANTIES. Vector represents and warrants that:
- (a) Vector is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation;
- (b) Vector has the corporate power and authority and the legal right to execute and deliver, to perform its obligations under, and to grant the security interest in the Collateral pursuant to, this Agreement and has taken all necessary corporate action to authorize its execution, delivery and performance of, and grant of the security interest in the Collateral pursuant to, this Agreement;
- (c) this Agreement constitutes a legal, valid and binding obligation of Vector, enforceable in accordance with its terms, and upon delivery to the Collateral Agent of the Secured Note, the security interest created pursuant to this Agreement will constitute a valid, perfected first priority security interest in the Collateral, enforceable in accordance with its

terms against all creditors of Vector and any Persons purporting to purchase any Collateral from Vector, except in each case as enforceability may be affected by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing;

- (d) the execution, delivery and performance by Vector of this Agreement will not (i) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of Vector under, any indenture, mortgage, deed of trust, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other agreement or instrument to which Vector is bound or by which Vector or any of its properties is bound or affected, (ii) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental or Regulatory Authority applicable to Vector or (iii) violate any provision of any statute or other rule or regulation of any Governmental or Regulatory Authority applicable to Vector;
- (e) no consent, approval or authorization of, or registration, filing (other than the filing of any financing statements contemplated in any Note Document) or declaration with, any Governmental or Regulatory Authority is required in connection with the execution, delivery or performance by Vector of this Agreement; and
- (f) no litigation, investigation or proceeding of or before any arbitrator or Governmental or Regulatory Authority is pending or, to the knowledge of Vector, threatened by or against Vector against any of its properties or revenues with respect to this Agreement or any of the transactions contemplated hereby.
- 5. COVENANTS. Until the Secured Obligations have been repaid in full, Vector shall do the following:
- (a) give, execute, deliver, file and/or record any financing statement, continuation statement, notice, instrument, document, agreement or other papers that may be necessary or desirable or that the Collateral Agent may reasonably request to create, preserve, perfect or validate the security interest granted pursuant hereto or to enable the Collateral Agent to exercise and enforce its rights hereunder with respect to such pledge and security interest;
- (b) upon the occurrence and during the continuance of any Default, upon request of the Collateral Agent, promptly notify (and Vector hereby authorizes the Collateral Agent so to notify) each maker of the Secured Note that the Secured Note has been assigned to the Collateral Agent hereunder, and that any payments due or to become due in respect of such Collateral are to be made directly to the Securities Account;
- (c) without the prior written consent of the Collateral Agent, Vector shall not file or suffer to be on file, or authorize or permit to be filed or to be on file, in any jurisdiction, any financing statement or like

instrument with respect to the Collateral in which the Collateral Agent is not named as the sole secured party.

- 6. PRESERVATION OF RIGHTS. The Collateral Agent shall not be required to take steps necessary to preserve any rights against third parties to any of the Collateral.
- 7. EVENTS OF DEFAULT, ETC. During the period during which an Event of Default shall have occurred and be continuing:
- (a) the Collateral Agent may make any reasonable compromise or settlement deemed desirable with respect to any of the Collateral and may extend the time of payment, arrange for payment in installments, or otherwise modify the terms of, any of the Collateral;
- (b) the Collateral Agent shall have all of the rights and remedies with respect to the Collateral of a secured party under the Uniform Commercial Code (whether or not the Uniform Commercial Code is in effect in the jurisdiction where the rights and remedies are asserted) and such additional rights and remedies to which a secured party is entitled under the laws in effect in any jurisdiction where any rights and remedies hereunder may be asserted, including, without limitation, the right, to the maximum extent permitted by law, to exercise all voting, consensual and other powers of ownership pertaining to the Collateral as if the Collateral Agent were the sole and absolute owner thereof (and Vector agrees to take all such action as may be appropriate to give effect to such right);
- (c) the Collateral Agent in its discretion may, in its name or in the name of Vector or otherwise, demand, sue for, collect or receive any money or property at any time payable or receivable on account of or in exchange for any of the Collateral, but shall be under no obligation to do so; and
- (d) the Collateral Agent may, upon ten business days' prior written notice to Vector of the time and place, with respect to the Collateral or any part thereof that shall then be or shall thereafter come into the possession, custody or control of the Collateral Agent or any of its agents, sell, lease, assign or otherwise dispose of all or any part of such Collateral, at such place or places as the Collateral Agent deems best, and for cash or for credit or for future delivery (without thereby assuming any credit risk), at public or private sale, without demand of performance or notice of intention to effect any such disposition or of the time or place thereof (except such notice as is required above or by applicable statute and cannot be waived), and the Collateral Agent or anyone else may be the purchaser, lessee, assignee or recipient of any or all of the Collateral so disposed of at any public sale (or, to the extent permitted by law, at any private sale) and thereafter hold the same absolutely, free from any claim or right of whatsoever kind, including any right or equity of redemption (statutory or otherwise), of Vector, any such demand, notice and right or equity being hereby expressly waived and released. The Collateral Agent may, without notice or publication, adjourn any public or

private sale or cause the same to be adjourned from time to time by announcement at the time and place fixed for the sale, and such sale may be made at any time or place to which the sale may be so adjourned.

- 8. WAIVER OF SURETYSHIP DEFENSES. To the extent this Agreement is deemed a guarantee, Vector hereby waves all defenses available to guarantors and sureties, including, without limitation, the following:
- (a) the occurrence of any Event of Default under, or any lack of validity, legality or enforceability of any provision of any Note Document or any other agreement or document;
 - (b) the failure of any Holder:
 - (i) to assert any claim or demand or to enforce any right or remedy against any Document Party or any other Person under the provisions of any Note Document, or otherwise, or
 - (ii) to exercise any right or remedy against any other guarantor of or other Person pledging collateral securing any of the Secured Obligations;
- (c) any change in the time, manner or place of payment of, or in any term of, all or any of the Secured Obligations, or any other extension, compromise, indulgence or renewal of any Secured Obligation;
- (d) any reduction, limitation, variation, impairment, discontinuance or termination of the Secured Obligations for any reason (other than by reason of any payment which is not required to be rescinded), including any claim of waiver, release, discharge, surrender, alteration or compromise, and shall not be subject to any defense or setoff, counterclaim, recoupment or termination whatsoever by reason of the invalidity, illegality, nongenuineness, irregularity, compromise, unenforceability of, or any other event or occurrence affecting, the Secured Obligations or otherwise (other than by reason of any payment which is not required to be rescinded);
- (e) any amendment to, rescission, waiver or other modification of, or any consent to any departure from, any of the terms of the Secured Obligations or any guarantees or security;
- (f) any addition, exchange, release, discharge, realization or non-perfection of any collateral security in respect of the Secured Obligations;
- (g) any amendment to, rescission, waiver or other modification of, or release or addition of, or consent to any departure from, any other guarantee held by the Holders as security for any of the Secured Obligations;
- (h) the loss of or in respect of or the unenforceability of any guarantee or other security which the Holders may now or hereafter hold in respect of the Secured Obligations, whether occasioned by the fault of the Holders or otherwise;

- (i) any change in the name of the Company or in the constitutive documents, capital structure, capacity or constitution of the Company, the bankruptcy or insolvency of the Company, the sale of any or all of the Company's business or assets or the Company being consolidated, merged or amalgamated with any other Person;
- (j) any failure on the part of the Company or any other Person to perform or comply with any term of the Note Purchase Agreement, the Notes, any of the Secured Obligations or any other agreement or document;
- (k) any suit or other action brought by any beneficiaries or creditors of, or by, the Company or any other Person for any reason whatsoever, including without limitation any suit or action in any way attacking or involving any issue, matter or thing in respect of any Note Document, any of the Secured Obligations or any other agreement or document;
- (1) any lack or limitation of status or of power, incapacity or disability of the Company or any trustee or agent thereof; or
- (m) any other circumstance (other than final and indefeasible payment in full) which might otherwise constitute a defense available to, or a legal or equitable discharge of, the Company, Brooke Holding or NV Holdings or any surety or any other guarantor of the foregoing.
- 9. NO WAIVER OF RIGHTS. No failure on the part of any Holder or the Collateral Agent to exercise, and no course of dealing with respect to, and no delay in exercising, any right, power or remedy hereunder shall operate as a waiver thereof; nor shall any single or partial exercise by any Holder or the Collateral Agent of any right, power or remedy hereunder preclude any other or further exercise thereof or the exercise of any other right, power or remedy. The remedies herein are cumulative and are not exclusive of any remedies provided by law.
- 10. NOTICES. All notices and communications provided for hereunder shall be in writing and sent (a) by telecopy if the sender on the same day sends a confirming copy of such notice by a recognized overnight delivery service (charges prepaid), or (b) by registered or certified mail with return receipt requested (postage prepaid), or (c) by a recognized overnight delivery service (with charges prepaid). Any such notice must be sent:
- (a) if to any Holder, to such Holder or it at the address specified for such communications in SCHEDULE A to the Note Purchase Agreement, or at such other address as it shall have specified to Vector and the Collateral Agent in writing,
- (b) if to the Collateral Agent, to the Collateral Agent at such address as is set forth on its signature page thereto or at such other address or at such other address as the Collateral Agent shall have specified to each Holder and to Vector in writing, or

(c) if to Vector, to Vector at its address set forth on its signature page thereto to the attention of the General Counsel, or at such other address as Vector shall have specified to each Holder and to the Collateral Agent.

Notices under this SECTION 10 will be deemed given when actually received if sent by telecopy, upon the succeeding Business Day if sent by overnight courier and three days after deposit in the U.S. mail if sent by registered or certified mail.

- 11. AMENDMENTS, ETC. The terms of this Agreement may be waived, altered or amended only by an instrument in writing duly executed by Vector, the Majority Holders and the Collateral Agent. Any such amendment or waiver shall be binding upon the Collateral Agent, each holder of any of the Secured Obligations and Vector.
- 12. SUCCESSORS AND ASSIGNS. This Agreement shall be binding upon and inure to the benefit of the respective successors and assigns of Vector, the Holders, the Collateral Agent and each holder of any of the Secured Obligations (PROVIDED, however, that Vector shall not assign or transfer its rights hereunder without the prior written consent of the Collateral Agent).
- 13. CAPTIONS. The captions and section headings appearing herein are included solely for convenience of reference and are not intended to affect the interpretation of any provision of this Agreement.
- 14. COUNTERPARTS. This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument and any of the parties hereto may execute this Agreement by signing any such counterpart. Delivery of an executed counterpart of a signature page of this Agreement by facsimile transmission shall be as effective as delivery of a manually executed counterpart of this Agreement.
- 15. GOVERNING LAW. This Agreement shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the law of the State of New York excluding choice-of-law principles of the law of such State that would require the application of the laws of a jurisdiction other than such State.
- 16. AGENTS AND ATTORNEYS-IN-FACT. The Collateral Agent may employ agents and attorneys-in-fact in connection herewith and shall not be responsible for the negligence or misconduct of any such agents or attorneys-in-fact selected by it in good faith.
- 17. SEVERABILITY. If any provision hereof is invalid and unenforceable in any jurisdiction, then, to the fullest extent permitted by law, (i) the other provisions hereof shall remain in full force and effect in such jurisdiction and shall be liberally construed in favor of the Collateral Agent in order to carry out the intentions of the parties hereto as nearly as may be possible and (ii) the invalidity or unenforceability of any provision hereof in any jurisdiction shall not affect the validity or enforceability of such provision in any other jurisdiction.
- 18. ENVIRONMENTAL INDEMNIFICATION OF COLLATERAL AGENT AND HOLDERS. Vector hereby agrees to indemnify the Collateral Agent and the Holders from, and hold the Collateral Agent and the Holders harmless against, any losses, liabilities, claims, damages or expenses arising under any Environmental Law (as

defined in the BGLS Pledge Agreement) as a result of the past, present or future operations of the Company or any of its Subsidiaries following the exercise by the Collateral Agent of its rights and remedies under any Note Document.

19. INDEMNIFICATION OF NV HOLDINGS. Vector hereby agrees to indemnify NV Holdings for, and hold it harmless against, any claim, demand, expense (including but not limited to attorneys' fees), loss or liability incurred by it arising solely out of or in connection with its being an affiliate of Vector.

Vector hereby acknowledges that the indemnity contained in this Section 19 for the benefit of NV Holdings and that NV Holdings is relying on said indemnity as a basis for entering into the NV Holdings Pledge Agreement.

- 20. INTEGRATION. This Agreement and the other Note Documents represent the agreement of Vector, the Collateral Agent and the Holders with respect to the subject matter hereof and thereof, and there are no promises, undertakings, representations or warranties by the Collateral Agent or any Holder relative to subject matter hereof and thereof not expressly set forth or referred to herein or in the other Note Documents.
 - 21. ACKNOWLEDGEMENTS. Vector hereby acknowledges that:
- (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Note Documents to which it is a party;
- (b) neither the Collateral Agent nor any Holder has any fiduciary relationship with or duty to Vector arising out of or in connection with this Agreement or any of the other Note Documents, and the relationship between Vector, on the one hand, and the Collateral Agent and Holders, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and
- (c) no joint venture is created hereby or by the other Note Documents or otherwise exists by virtue of the transactions contemplated hereby among the Holders or among Vector and the Holders.

IN WITNESS WHEREOF, Vector Group Ltd. has executed this Acknowledgment and Undertaking, intending to be legally bound, as of May 14,

VECTOR GROUP LTD.

/s/ RICHARD J. LAMPEN By:

Name: Richard J. Lampen Title: Executive Vice President

100 S.E. Second Street Miami, Florida 33131 Telephone: (305) 579-8000 Facsimile: (305) 579-8009

Attention:

Richard J. Lampen Executive Vice President

UNITED STATES TRUST COMPANY OF NEW YORK

By: /s/ PATRICIA GALLAGHER

Name: Patricia Gallagher Title: Assistant Vice President

ADDRESS FOR NOTICES:

114 West 47th Street, 25th Floor New York, New York 10036-1532 Telephone: (212) 852-1664 Facsimile: (212) 852-1626

Attention: Patricia Gallagher

TCW LEVERAGED INCOME TRUST, L.P.

By: TCW Advisers (Bermuda), Ltd.

as its General Partner

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: TCW Investment Management Company

as Investment Adviser

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title: Group Managing Director

TCW LEVERAGED INCOME TRUST II, L.P.

By: TCW (LINC II), L.P.

as its General Partner

By: TCW Advisers (Bermuda), Ltd. its General Partner

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: TCW Investment Management Company as Investment Adviser

/s/ MARK ATTANASIO By:

Name: Mark Attanasio

Title: Group Managing Director

TCW LINC III CBO LTD.

By: TCW Investment Management Company

as Collateral Manager

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name:

Mark Attanasio Group Managing Director Title:

POWRs 1997-2 (Participating Obligations with Residuals 1997-2)

By: Citibank Global Asset Management

Its Investment Advisor

By: TCW Asset Management Company

Its Portfolio Manager

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio

Title: Group Managing Director

CAPTIVA II FINANCE LTD. By: TCW Advisors, Inc.

Its Financial Manager

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall
Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio Title: Group Managing Director

AIMCO CDO, SERIES 2000-A

By: Allstate Investment Management Company Its Collateral Manager

By: TCW Asset Management Company Its Investment Advisor

By: /s/ DARRYL L. SCHALL

Name: Darryl L. Schall Title: Managing Director

By: /s/ MARK ATTANASIO

Name: Mark Attanasio Title: Group Managing Director

ACCOUNT CONTROL AGREEMENT

ACCOUNT CONTROL AGREEMENT (this "AGREEMENT") dated as of May 14, 2001 is made by and among BGLS INC., a company duly organized under the laws of Delaware (the "COMPANY"); United States Trust Company of New York, a New York banking corporation, acting in its capacity as collateral agent for the purchasers of the Notes (in such capacity, together with its successors and permitted assigns, the "COLLATERAL AGENT"); and Bank of America, N.A. (together with its successors and permitted assigns, the "SECURITIES INTERMEDIARY"). In addition, all terms used herein and defined in the Uniform Commercial Code of the State of New York (as amended and in effect from time to time, the "UNIFORM COMMERCIAL CODE") shall have the respective meanings given to those terms in the Uniform Commercial Code, except where the context otherwise requires.

SECTION 1. ESTABLISHMENT OF THE ACCOUNT.

- (a) The parties hereto hereby confirm that the Company has caused the Securities Intermediary to establish, and the Securities Intermediary, acting as a "securities intermediary" (such term as used herein as defined in Section 8-102(a)(14)(ii) of the Uniform Commercial Code), has established account number 72-40-400-0372144 in the name of the Company (such account and any successor account, the "Account").
- (b) The Account shall be maintained on the books and records of the Securities Intermediary as a "securities account" (such term as used herein as defined in Section 8-501(a) of the Uniform Commercial Code), and the Securities Intermediary agrees to treat the Company as the "entitlement holder" (as defined in Section 8-102(a)(7) of the Uniform Commercial Code) with respect to the Account.
- (c) The Securities Intermediary agrees that all cash (which for purposes of this Agreement is hereby designated as a "financial asset" pursuant to Section 8-102(9)(iii) of the Uniform Commercial Code) and all securities described on EXHIBIT D to this Agreement ("PERMITTED SECURITIES") delivered to the Securities Intermediary by the Company or the Collateral Agent pursuant to the Pledge and Security Agreement, dated as of May 14, 2001, by and between the Company and the Collateral Agent (as the same may be amended, supplemented or otherwise modified from time to time, the "PLEDGE AGREEMENT") that are a type eligible to be held by a Federal Reserve Bank and/or the Depository Trust Company (each such Permitted Security, an "ELIGIBLE ASSET") will be promptly credited to the Account.
- (d) The Securities Intermediary agrees that each item of property (whether cash, a security, an instrument or obligation, share, participation, interest or other property whatsoever) credited to or held in the Account shall be held and treated as a "financial asset" (such term as used herein as defined in Section 8-102(a)(9) of the Uniform Commercial Code).
- (e) All securities and other property underlying any financial assets credited to the Account shall be registered in the name of the Securities Intermediary, indorsed to the Securities Intermediary or in blank or credited to another securities account maintained in the name

of the Securities Intermediary and in no case shall any financial asset credited to the Account be registered in the name of the Company, payable to the order of the Company or specially indorsed to the Company.

SECTION 2. ENTITLEMENT ORDERS. The Company has granted to the Collateral Agent, pursuant to the Pledge Agreement, a security interest in all securities entitlements related to the Account, the Account itself and all financial assets held therein or credited thereto. In furtherance thereof, the Company hereby agrees that the Securities Intermediary may, and the Securities Intermediary hereby agrees that it shall, comply with "entitlement orders" (such term as used herein as defined in Section 8-102(a)(8) of the Uniform Commercial Code) originated by the Company or the Collateral Agent and relating to the Account without further consent by the Company or any other Person (as defined below). In the event that the Securities Intermediary receives contradictory entitlement orders from the Company and the Collateral Agent and the Securities Intermediary has not (subject to SECTION 9 of this Agreement) already completely implemented the entitlement order of the Company, the Company hereby agrees that the Securities Intermediary may, and the Securities Intermediary hereby agrees that it shall, follow the entitlement order of the Collateral Agent. As used herein, "PERSON" means an individual, partnership corporation, limited liability company, association, trust, unincorporated association or a government agency or political subdivision thereof.

SECTION 3. SUBORDINATION OF LIEN; WAIVER OF SET-OFF. The Securities Intermediary agrees that, except for payment of its fees, commissions, settlement of open orders, and repayment of any amount credited or paid to the Company and/or the Collateral Agent with respect to interest, dividends, or other income or proceeds at maturity or otherwise prior to receipt by the Securities Intermediary of finally collected funds therefor, it shall not assert any lien, encumbrance, claim or right against the Account or any financial assets held therein or credited thereto. The Company and the Collateral Agent agree that any lien, encumbrance, claim or right of the Securities Intermediary against the Account and/or any financial assets held therein or credited thereto for the payment of such fees, commissions, and settlements and repayment of any such amount credited or paid to the Company and/or the Collateral Agent shall be first and prior to the claims of the Collateral Agent and the Company to the Account and any financial assets held therein or credited thereto. Nothing in this Section 3 shall create any liability on the part of the Collateral Agent in respect of any amounts paid or credited to the Collateral Agent or otherwise, all of which liabilities shall be solely the obligation of the Company.

SECTION 4. CHOICE OF LAW. This Agreement shall be governed by the law of the State of New York without regard to its conflicts of laws principles. Regardless of any provision in any other agreement, for purposes of this Agreement, the Securities Intermediary and the other parties agree that the "securities intermediary's jurisdiction" (as defined in Section 8-110(e) of the Uniform Commercial Code) with respect to the Account (as well as the securities entitlements related thereto) shall be the State of New York, and this Agreement shall be deemed "an agreement between the securities intermediary and its entitlement holder" as described in Section 8-110(e)(1) of the Uniform Commercial Code.

SECTION 5. CONFLICT WITH OTHER AGREEMENTS. Other than trading receipts issued with respect to the crediting of financial assets to the Account prior to the date hereof, there are no other agreements entered into between the

Company and the Securities Intermediary with respect to the Account. In the event of any conflict between this Agreement (or any portion thereof) and any other agreement between the Company and the Securities Intermediary with respect to the Account now existing or hereafter entered into, the terms of this Agreement shall prevail.

SECTION 6. AMENDMENTS. No amendment or modification of this Agreement or waiver of any right hereunder shall be binding on any party hereto unless it is in writing and is signed by each of the parties hereto.

SECTION 7. NOTICE OF ADVERSE CLAIMS. Except for the claims and interest of the Collateral Agent and of the Company in the Account, the Securities Intermediary does not know of any claim to, or interest in, the Account or in any financial assets credited thereto or held therein. If the Securities Intermediary receives written notice that any Person asserts any lien, encumbrance or adverse claim (including any writ, garnishment, judgment, warrant of attachment, execution or similar process) against the Account or in any financial assets credited thereto or held therein, the Securities Intermediary shall promptly notify the Collateral Agent and the Company thereof.

SECTION 8. REPRESENTATIONS, WARRANTIES AND COVENANTS OF THE SECURITIES INTERMEDIARY. The Securities Intermediary hereby makes the following representations, warranties and covenants:

- (a) The Securities Intermediary is a "securities intermediary" (as defined in Section 8-102(a)(14)(ii) of the Uniform Commercial Code) and is acting in such capacity with respect to the Account.
- (b) The Account has been established as set forth in SECTION 1 of this Agreement and shall be maintained in the manner set forth herein until termination of this Agreement. The Securities Intermediary shall not change the name or account number of the Account without the prior written consent of the Collateral Agent, except as a result of computer system or accounting changes affecting accounts of the Securities Intermediary generally (in which case the Securities Intermediary shall provide prompt notice of such change to the Collateral Agent and the Company).
- (c) This Agreement is the legal, valid and binding obligation of the Securities Intermediary enforceable against the Securities Intermediary in accordance with its terms subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing.
- (d) The Securities Intermediary has not entered into, and hereby agrees that until the termination of this Agreement, it shall not enter into, any agreement with any other Person (other than the Collateral Agent) relating to the Account (and/or any financial assets credited thereto) pursuant to which the Securities Intermediary has agreed or would agree, as the case may be, to comply with entitlement orders made by such Person. The Securities Intermediary has not entered

into any other agreement with the Company or any other Person purporting to limit or condition the obligation of or the Securities Intermediary to comply with entitlement orders as set forth in this Agreement.

SECTION 9. MAINTENANCE OF ACCOUNT. In addition to, and not in lieu of, the obligation of the Securities Intermediary to honor entitlement orders as agreed in SECTION 2 of this Agreement, the Securities Intermediary agrees to maintain the Account as follows:

- (a) NOTICE OF SOLE CONTROL. If at any time the Collateral Agent delivers to the Securities Intermediary a Notice of Sole Control (a "NOTICE OF SOLE CONTROL") in substantially the form set forth in EXHIBIT A to this Agreement, the Securities Intermediary agrees that, after receipt of such notice and until it receives a Notice of End of Sole Control (a "NOTICE OF END OF SOLE CONTROL") in substantially the form set forth in EXHIBIT B to this Agreement, it shall comply only with entitlement orders and other instructions originated by the Collateral Agent. To the extent that an entitlement order given by the Company has not been fully implemented at any time that the Securities Intermediary receives a Notice of Sole Control, the Securities Intermediary shall immediately cease implementing such entitlement order of the Company. Following the receipt by the Securities Intermediary of a Notice of End of Sole Control and unless and until a subsequent Notice of Sole Control is received by the Securities Intermediary, then the Securities Intermediary shall again, subject to SECTION 2 of this Agreement, follow the entitlement request originated by the Company.
- (b) STATEMENTS AND CONFIRMATIONS. The Securities Intermediary shall promptly send copies of all statements, confirmations and other correspondence concerning the Account and/or any financial assets credited thereto simultaneously to each of the Company and the Collateral Agent at the address set forth in SECTION 14 of this Agreement.
- (c) TAX REPORTING. All items of income, gain, expense and loss recognized in the Account shall be reported to the Internal Revenue Service by the Company and all state and local taxing authorities under the name and taxpayer identification number of the Company.

SECTION 10. RESPONSIBILITIES OF THE SECURITIES INTERMEDIARY. The Securities Intermediary undertakes to perform only such duties as are expressly set forth herein and the Securities Intermediary shall not be bound by any agreement between the Collateral Agent and the Company or any other parties to which the Securities Intermediary is not a party, whether or not the Securities Intermediary has knowledge thereof. Notwithstanding any other provision of this Agreement, it is agreed by the parties hereto that the Securities Intermediary shall not be liable for any action taken by it or any of its directors, officers, agents or employees in accordance with this Agreement, including without limitation, any action so taken at the request of the Collateral Agent, except for the Securities Intermediary's or such Person's own gross negligence or willful misconduct. Accordingly, the Securities Intermediary shall not incur any such liability with respect to (i) any action taken or omitted to be taken in good faith upon the advice of its counsel or counsel for the Collateral Agent given with respect to the duties or responsibilities of the Collateral Agent hereunder, or (ii) any action taken or omitted to be taken in

reliance on any document, including any entitlement order, Notice of Sole Control or Notice of End of Sole Control provided for in this Agreement, not only as to its due execution and to the validity and effectiveness of its provisions, but also the truth and accuracy of any information contained therein, which the Securities Intermediary shall in good faith believe to be genuine, to have been signed or presented by the proper Person or Persons. The Securities Intermediary shall have no responsibility or liability for complying with a Notice of Sole Control or a Notice of End of Sole Control or for complying with entitlement orders concerning the financial assets originated by the Collateral Agent.

The Securities Intermediary shall be deemed to have exercised reasonable care in the custody and preservation of the Account and financial assets in its possession if the Account and financial assets are accorded treatment substantially equal to that which the Securities Intermediary accords to similar property held by it as Securities Intermediary or in a comparable capacity, it being understood that the Securities Intermediary shall not have any responsibility for (a) ascertaining or taking any action with respect to calls, conversions, exchanges, maturities or similar matters relative to any financial assets held in the Account, whether or not the Securities Intermediary has or is deemed to have knowledge or notice of such matters or (b) taking any steps to maintain the value of any property held in the Account or to preserve rights against any parties with respect thereto. The Securities Intermediary shall have no duty to see to the payment or discharge of any tax assessment or other governmental charge, or any other lien or encumbrance of any kind, owing, asserted or levied against, the Account or any financial assets held therein. The Securities Intermediary makes no representation as to the legality or validity of any document or agreement to which it is not a party or this Agreement (other than as to itself), the validity sufficiency of any financial assets held in the Account or the value of the Account or any financial asset held therein or credited thereto.

SECTION 11. INDEMNIFICATION.

- (a) The Company agrees to pay, indemnify and hold the Securities Intermediary and each director, officer, employee, agent, bailee or other person acting on behalf of the Securities Intermediary, and each stockholder of any thereof, harmless from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses (including, without limitation, the reasonable fees and disbursements of counsel and other advisers) or disbursements of any kind or nature whatsoever with respect to the execution, delivery, enforcement, performance and administration of this Agreement, including, without limitation, any amendment hereto; or in connection with the transactions contemplated by (i) this Agreement; or (ii) any other agreements to which the Company is a party and which are related to the financing of the Company that is closing in connection with the execution and delivery of this Agreement (including arising from the ordinary negligence of the person seeking indemnification), except to the extent caused by the gross negligence or willful misconduct of the person seeking indemnification.
- (b) The obligations of the Company under this SECTION 11 shall survive the termination or modification of the other provisions of this Agreement and shall survive the commencement of a case under any applicable bankruptcy law on behalf of or against the Company or any other proceeding for the reorganization, management, adjustment of

debt, dissolution or liquidation on behalf of or against any such Person and shall survive any dissolution of any such Person and shall survive the resignation or removal of the Securities Intermediary.

(c) The Collateral Agent shall have no liability (other than for its own gross negligence or willful misconduct) or duty to the Securities Intermediary, including without limitation under clauses (a) and (b) of this SECTION 11.

SECTION 12. SUCCESSORS AND ASSIGNS. The terms of this Agreement shall be binding upon, and shall inure to the benefit of, the parties hereto and their respective successors and permitted assigns.

SECTION 13. TERMINATION. The rights and powers granted herein to the Collateral Agent have been granted in order to perfect its security interests in the Account, are powers coupled with an interest and, except as otherwise provided by mandatory provisions of applicable law, shall neither be affected by the bankruptcy of the Company nor the lapse of time. The obligations of the Securities Intermediary hereunder shall continue in effect until the security interests of the Collateral Agent in the Account have been terminated and the Collateral Agent has notified the Securities Intermediary of such termination in writing, substantially in the form of EXHIBIT C to this Agreement. Upon receipt of such notice, the Collateral Agent shall have no further right to originate entitlement orders concerning the Account.

Notwithstanding the foregoing, the Securities Intermediary may cease to act as Securities Intermediary hereunder and terminate this Agreement by giving the Company and the Collateral Agent not less than thirty (30) days notice thereof.

SECTION 14. NOTICES. All notices, requests and other communications provided for herein shall be given or made in writing (including, without limitation, by telex or telecopy) delivered to the intended recipient at the "Address for Notices" as specified under its name on the signature page hereto or, as to any party, at such other address as shall be designated by such party in a notice to each other party. Except as otherwise provided in this Agreement, all such communications shall be deemed to have been duly given when transmitted by telex or telecopier (with a confirming copy sent the same day by recognized overnight delivery service, charges prepaid) or personally delivered or, in the case of a mailed notice, upon receipt, in each case given or addressed as aforesaid.

SECTION 15. COUNTERPARTS. This Agreement may be executed in any number of counterparts, all of which shall constitute one and the same instrument, and any party hereto may execute this Agreement by signing and delivering one or more counterparts. Delivery of an executed counterpart of a signature page of this Agreement by facsimile transmission shall be effective as delivery of a manually executed counterpart of this Agreement.

SECTION 16. TITLES AND HEADINGS. Titles and headings to Sections and subsections herein are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.

SECTION 17. ENTIRE AGREEMENT. This Agreement is the entire agreement among the parties hereto regarding the subject matter hereof and supersedes any prior agreements and contemporaneous oral agreements regarding its subject matter.

 $\hbox{IN WITNESS WHEREOF, the Company, the Securities Intermediary and the Collateral Agent have caused this Agreement to be duly executed by their}\\$ duly authorized officers all as of the date first above written.

BGLS INC.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen
Title: Executive Vice President

ADDRESS FOR NOTICES:

100 S.E. Second Street Miami, Florida 33131

Telephone: (305) 579-8000 Facsimile: (305) 579-8009

Attention: Richard J. Lampen

Executive Vice President

UNITED STATES TRUST COMPANY OF NEW YORK, as Collateral Agent

By: /s/ PATRICIA GALLAGHER

Name: Patricia Gallagher Title: Assistant Vice President

Address for Notices:

114 West 47th Street, 25th Floor New York, New York 10036-1532 Telephone: (212) 852-1164 Facsimile: (212) 852-1626

Attention: Patricia Gallagher

BANK OF AMERICA, N.A.

as Securities Intermediary

By: /s/ DONALD CANFIELD

Name: Donald Canfield

Title: SVP

Address for Notices

Bank of America, N.A. 231 LaSalle Street Mail Code: IL-231-04-02 Chicago, Illinois 60697 Attention: Donald Canfield

Sheila Sanders Frank Hemeter

Facsimile: (312) 987-0500 Telephone: (312) 828-7574 (Canfield) Telephone: (312) 828-1423 (Sanders) Telephone: (312) 828-7548 (Hemeter)

EXHIBIT A

[Letterhead of United States Trust Company of New York]

[Date]

Bank of America, N.A. 231 LaSalle Street Mail Code: IL-231-04-02 Chicago, Illinois 60697 Facsimile: (704) 987-0500 Attention: Donald Canfield

Sheila Sanders Frank Hemeter

Re: NOTICE OF SOLE CONTROL

Ladies and Gentlemen:

As referenced in the Account Control Agreement, dated May 14, 2001, among BGLS Inc., us and you (a copy of which is attached) we hereby give you notice of our sole control over account number ______ (the "ACCOUNT") and all financial assets credited thereto. You are hereby instructed not to accept, and to cease implementing immediately, any entitlement orders or other instructions with respect to the Account or the financial assets credited thereto from any person other than the undersigned, unless otherwise ordered by a court of competent jurisdiction.

Very truly yours,

United States Trust Company of New York

By:

Name:

Title:

cc: BGLS Inc.

EXHIBIT B

[Letterhead of United States Trust Company of New York]

[Date]

Bank of America, N.A.
231 LaSalle Street
Mail Code: IL-231-04-02
Chicago, Illinois 60697
Facsimile: (704) 987-0500
Attention: Donald Canfield
Sheila Sanders

Frank Hemeter

Re: NOTICE OF END OF SOLE CONTROL

Ladies and Gentlemen:

As referenced in the Account Control Agreement, dated May 14, 2001 (the "AGREEMENT"), among BGLS Inc., us and you (a copy of which is attached) we hereby give you notice of the end of our sole control over account number ______ (the "ACCOUNT") and all financial assets credited thereto. Accordingly, subject to the delivery of a Notice of Sole Control in the future, the limitations of SECTION 8(A) of this Agreement as to your following entitlement orders of the Company shall no longer apply.

Very truly yours,

United States Trust Company of New York

ву:

Name: Title:

cc: BGLS Inc.

EXHIBIT C

[Letterhead of United States Trust Company of New York]

[Date]

Bank of America, N.A.
231 LaSalle Street
Mail Code: IL-231-04-02
Chicago, Illinois 60697
Facsimile: (704) 987-0500
Attention: Donald Canfield
Sheila Sanders
Frank Hemeter

Re: NOTICE OF TERMINATION

Ladies and Gentlemen:

As referenced in the Account Control Agreement, dated May 14, 2001 among BGLS Inc., us and you (a copy of which is attached) we hereby give you notice that the security interests of the Collateral Agent in account number ______ (the "ACCOUNT") have been terminated and the Collateral Agent has no further right to originate entitlement orders concerning the Account.

Very truly yours,

United States Trust Company of New York

By:
Name:

Title:

cc: BGLS Inc.

Exhibit D

PERMITTED SECURITIES

- 1. Direct noncallable obligations of, or noncallable obligations guaranteed by, the United States of America or agencies thereof for timely payment of which obligation or guarantee the full faith and credit of the United States of America or such agency is pledged.
- 2. Investments in commercial paper or master notes maturing within 270 days from the date of acquisition thereof and having (or the issuer thereof having), at such date of acquisition, the highest credit rating obtainable from Standard & Poor's Rating Group or from Moody's Investor Services, Inc.
- 3. Investments in certificates of deposit, banker's acceptances, time deposits or "late cash" deposits/funding agreements maturing or puttable to the issuer thereof within 180 days from the date of acquisition thereof issued or guaranteed by or placed with, and money market deposit accounts issued or offered by, any domestic office of any commercial bank organized under the laws of, or licensed to conduct a banking or trust business in, the United States of America or any State thereof which has a combined capital and surplus and undivided profits of not less than \$500,000,000 whose long-term debt is rated at least "A" or the equivalent thereof by both Standard & Poor's Rating Group and Moody's Investors Service, Inc.
- 4. Fully collateralized repurchase agreements with a term of not more than 30 days for securities described in clause (1) above and entered into with a financial institution satisfying the criteria described in clause (3) above.
- 5. Investments in the Nations Funds Cash Reserves Money Market Mutual Fund.

1

STOCK PURCHASE AGREEMENT dated as of May 16, 2001 by and between VECTOR GROUP LTD.

and

HIGH RIVER LIMITED PARTNERSHIP with respect to 1,639,344 shares of common stock of VECTOR GROUP LTD.

TABLE OF CONTENTS

 $\,$ This Table of Contents is not part of the Agreement to which it is attached but is inserted for convenience only.

		PAGE NO.
ARTICLE I SALES	OF SHARES AND CLOSING	1
1.01	PURCHASE AND SALE	1
1.02	PURCHASE PRICE	
1.03	CLOSING	1
ARTICLE II REPRI	ESENTATIONS AND WARRANTIES OF THE COMPANY	2
2.01	CORPORATE EXISTENCE OF THE COMPANY	2
2.02	AUTHORITY	2
2.03	CAPITAL STOCK	2
2.04	NO CONFLICTS	2
2.05	GOVERNMENTAL APPROVALS AND FILINGS	3
2.06	SEC REPORTS AND FINANCIAL STATEMENTS	3
2.07	BROKERS	3
ARTICLE III REPI	RESENTATIONS AND WARRANTIES OF PURCHASER	3
3.01	ORGANIZATION OF PURCHASER	3
3.02	AUTHORITY	4
3.03	NO CONFLICTS	4
3.04	GOVERNMENTAL APPROVALS AND FILINGS	4
3.05	PURCHASE FOR INVESTMENT	5
3.06	BROKERS	5
ARTICLE IV COND	ITIONS	5
4.01	CONDITIONS TO OBLIGATION OF EACH PARTY TO EFFECT THE CLOSING	5
ARTICLE V COVEN	ANTS	6
5.01	SHELF REGISTRATION STATEMENT	6
5.02	SPECIFIC PERFORMANCE; LIQUIDATED DAMAGES	6
5.03	REGISTRATION EXPENSES	7
5.04	INDEMNIFICATION	7
5.05	HSR	9

			PAGI NO
	5.06	LOCKUP	10
	5.07	STOCK EXCHANGE LISTING	10
ARTTCI F	VI DEEINI	TIONS	10
	6.01	DEFINITIONS	10
ARTICLE	VII MISCE		13
	7.01	ENTIRE AGREEMENT	
	7.02	EXPENSES	13
	7.03	WAIVER	
	7.04	AMENDMENT	13
	7.05	NO THIRD PARTY BENEFICIARY	
	7.06	NO ASSIGNMENT; BINDING EFFECT	13
	7.07	HEADINGS	13
	7.08	INVALID PROVISIONS	13
	7.09	GOVERNING LAW	14
	7.10	COUNTERPARTS	
	7.11	TERMINATION	14

This STOCK PURCHASE AGREEMENT dated as of May 16, 2001 is made and entered into by and between High River Limited Partnership, a Delaware limited partnership ("PURCHASER"), and Vector Group Ltd., a Delaware corporation (the "COMPANY"). Capitalized terms not otherwise defined herein have the meanings set forth in SECTION 6.01.

WHEREAS, Purchaser desires to purchase from the Company and the Company desires to sell to Purchaser 1,639,344 shares (the "SHARES") of common stock, par value \$.10 per share ("COMMON STOCK"), of the Company, on the terms and subject to the conditions set forth in this Agreement;

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth in this Agreement, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

ARTICLE I SALES OF SHARES AND CLOSING

- 1.01 PURCHASE AND SALE. The Company agrees to sell to Purchaser, and Purchaser agrees to purchase from the Company, the Shares at the Closing on the terms and subject to the conditions set forth in this Agreement.
- 1.02 PURCHASE PRICE. The aggregate purchase price for the Shares is \$50,000,000 (the "PURCHASE PRICE"), payable in immediately available United States funds at the Closing in the manner provided in SECTION 1.03.
- 1.03 CLOSING. The Closing will take place at the offices of Milbank, Tweed, Hadley & McCloy LLP, One Chase Manhattan Plaza, New York, NY 10005, on the Closing Date. At the Closing: (i) Purchaser will pay the Purchase Price by wire transfer of immediately available funds to such account as the Company has reasonably directed, and (ii) the Company will sell to Purchaser the Shares by delivering to Purchaser a certificate or certificates representing the Shares. The Shares will bear the following legend:

THE SECURITIES REPRESENTED BY THIS CERTIFICATE WERE ISSUED IN A PRIVATE PLACEMENT, WITHOUT REGISTRATION UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND IN RELIANCE UPON THE HOLDER'S REPRESENTATION THAT SUCH SECURITIES WERE BEING ACQUIRED FOR INVESTMENT AND NOT FOR RESALE. NO TRANSFER OF SUCH SECURITIES MAY BE MADE ON THE BOOKS OF THE COMPANY UNLESS ACCOMPANIED BY AN OPINION OF COUNSEL, SATISFACTORY TO THE COMPANY, THAT SUCH TRANSFER MAY BE EFFECTED WITHOUT REGISTRATION UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR THAT SUCH SECURITIES HAVE BEEN SO REGISTERED UNDER A REGISTRATION STATEMENT WHICH IS IN EFFECT AT THE DATE OF SUCH TRANSFER.

ARTICLE II

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company hereby represents and warrants to Purchaser as follows:

- 2.01 CORPORATE EXISTENCE OF THE COMPANY. The Company is a corporation duly incorporated, validly existing and in good standing under the Laws of the State of Delaware. The Company has full corporate power and authority to execute and deliver this Agreement and to perform the Company's obligations hereunder and to consummate the transactions contemplated hereby, including without limitation to sell and transfer (pursuant to this Agreement) the Shares.
- 2.02 AUTHORITY. The execution and delivery by the Company of this Agreement, and the performance by such party of its obligations hereunder, have been duly and validly authorized by the Board of Directors of the Company, no other corporate action on the part of the Company or its stockholders being necessary. This Agreement has been duly and validly executed and delivered by the Company and constitutes a legal, valid and binding obligation of the Company enforceable against the Company in accordance with its terms.
- 2.03 CAPITAL STOCK. The Shares are duly authorized, validly issued, fully paid and nonassessable. The delivery of a certificate or certificates at the Closing representing the Shares in the manner provided in SECTION 1.03 will transfer to Purchaser good and valid title to the Shares, free and clear of all Liens other than Liens created or suffered to exist by Purchaser.
- 2.04 NO CONFLICTS. The execution and delivery by the Company of this Agreement do not, and the performance by the Company of its obligations under this Agreement and the consummation of the transactions contemplated hereby will not:
- (a) conflict with or result in a violation or breach of any of the terms, conditions or provisions of the certificate of incorporation or by-laws (or other comparable corporate charter documents) of the Company;
- (b) conflict with or result in a violation or breach of any term or provision of any Law or Order applicable to the Company or any of its Assets and Properties (other than such conflicts, violations or breaches (i) which will not in the aggregate adversely affect the validity or enforceability of this Agreement or have a material adverse effect on the Business or Condition of the Company or (ii) as would occur solely as a result of the identity or the legal or regulatory status of Purchaser or any of its Affiliates); or
- (c) except as will not, individually or in the aggregate, be materially adverse to the Business or Condition of the Company or adversely affect the ability of the Company to consummate the transactions contemplated hereby or to perform its obligations hereunder, (i) conflict with or result in a violation or breach of, (ii) constitute (with or without notice or lapse of time or both) a default under, (iii) require the Company to obtain any consent, approval or action of, make any filing with or give any notice to any Person as a result or

under the terms of, (iv) result in or give to any Person any right of termination, cancellation, acceleration or modification in or with respect to, or (v) result in the creation or imposition of any Lien upon the Company or any of its Assets and Properties under, any Contract or License to which the Company is a party or by which any of its Assets and Properties is bound.

- 2.05 GOVERNMENTAL APPROVALS AND FILINGS. Except for the filings required by the Company under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations thereunder (the "HSR ACT") and the listing by the Company of the Shares on notice of issuance on The New York Stock Exchange and as required under SECTION 5.01, no other consent, approval or action of, filing with or notice to any Governmental or Regulatory Authority by the Company is required as a precondition to the execution, delivery and performance of this Agreement or the consummation of the transactions contemplated hereby except (i) where the failure to obtain any such consent, approval or action, to make any such filing or to give any such notice will not adversely affect the ability of the Company to consummate the transactions contemplated by this Agreement or to perform its obligations hereunder or have a material adverse effect on the Business or Condition of the Company, and (ii) those as would be required solely as a result of the identity or the legal or regulatory status of the Purchaser or any of its Affiliates.
- 2.06 SEC REPORTS AND FINANCIAL STATEMENTS. As of their respective dates, the Company SEC Reports did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. Since the respective dates of such filings, and except as disclosed therein, there has not been any change, event or development having, or that is reasonably expected to have, individually or in the aggregate, a material adverse effect on the Company.
- 2.07 BROKERS. All negotiations relative to this Agreement and the transactions contemplated hereby have been carried out by the Company directly with Purchaser without the intervention of any person on behalf of the Company in such manner as to give rise to any valid claim by any person against Purchaser or any Subsidiary for a finder's fee, brokerage commission or similar payment.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser hereby represents and warrants to the Company as follows:

3.01 ORGANIZATION OF PURCHASER. Purchaser is a limited partnership duly organized, validly existing and in good standing under the Laws of the State of Delaware. Purchaser is duly authorized to execute and deliver this Agreement and to perform Purchaser's obligations hereunder and to consummate the transactions contemplated hereby, including, without limitation, to buy (pursuant to this Agreement) the Shares.

- 3.02 AUTHORITY. The execution and delivery by Purchaser of this Agreement, and the performance by Purchaser of its obligations hereunder, have been duly and validly authorized, no other action on the part of Purchaser being necessary. This Agreement has been duly and validly executed and delivered by Purchaser and constitutes a legal, valid and binding obligation of Purchaser enforceable against Purchaser in accordance with its terms.
- 3.03 NO CONFLICTS. The execution and delivery by Purchaser of this Agreement do not, and the performance by Purchaser of its obligations under this Agreement and the consummation of the transactions contemplated hereby will not:
- (a) conflict with or result in a violation or breach of any of the terms, conditions or provisions of its partnership agreement (or other comparable organizational documents) of Purchaser;
- (b) conflict with or result in a violation or breach of any term or provision of any Law or Order applicable to Purchaser or any of its Assets and Properties (other than such conflicts, violations or breaches (i) which will not in the aggregate adversely affect the validity or enforceability of this Agreement or have a material adverse effect on the Business or Condition of Purchaser or (ii) as would occur solely as a result of the identity or the legal or regulatory status of the Company or any of its Affiliates); or
- (c) except as will not, individually or in the aggregate, be materially adverse to the Business or Condition of Purchaser or adversely affect the ability of Purchaser to consummate the transactions contemplated hereby or to perform its obligations hereunder, (i) conflict with or result in a violation or breach of, (ii) constitute (with or without notice or lapse of time or both) a default under, (iii) require Purchaser to obtain any consent, approval or action of, make any filing with or give any notice to any Person as a result or under the terms of, (iv) result in or give to any Person any right of termination, cancellation, acceleration or modification in or with respect to, or (v) result in the creation or imposition of any Lien upon Purchaser or any of its Assets and Properties under, any Contract or License to which Purchaser is a party or by which any of its Assets and Properties is bound.
- 3.04 GOVERNMENTAL APPROVALS AND FILINGS. Except for the filings required by Purchaser under the HSR Act, no consent, approval or action of, filing with or notice to any Governmental or Regulatory Authority by Purchaser is required as a precondition to the execution, delivery and performance of this Agreement or the consummation of the transactions contemplated hereby except (i) where the failure to obtain any such consent, approval or action, to make any such filing or to give any such notice will not adversely affect the ability of Purchaser to consummate the transactions contemplated by this Agreement or to perform its obligations hereunder or have a material adverse effect on the Business or Condition of Purchaser, and (ii) those as would be required solely as a result of the identity or the legal or regulatory status of the Company or any of its Affiliates.

3.05 PURCHASE FOR INVESTMENT.

- (a) The Shares will be acquired by Purchaser for its own account for the purpose of investment, it being understood that the right to dispose of such Shares shall be entirely within the discretion of Purchaser. Purchaser will refrain from transferring or otherwise disposing of any of the Shares, or any interest therein, in such manner as to cause the Company to be in violation of the registration requirements of the Securities Act or any applicable state securities or blue sky laws.
- (b) Purchaser acknowledges that it is an "accredited investor" as defined in Rule 501 of Regulation D under the Securities ${\sf Act.}$
- 3.06 BROKERS. All negotiations relative to this Agreement and the transactions contemplated hereby have been carried out by Purchaser directly with the Company without the intervention of any Person on behalf of the Purchaser in such manner as to give rise to any valid claim by any Person against the Company or any Subsidiary for a finder's fee, brokerage commission or similar payment.

ARTICLE IV

CONDITIONS

- 4.01 CONDITIONS TO OBLIGATION OF EACH PARTY TO EFFECT THE CLOSING. The respective obligations of each party hereunder to effect the Closing are subject to the fulfillment or waiver, at the Closing, of the following conditions:
- (a) REPRESENTATIONS AND WARRANTIES. Each of the representations and warranties made by the other party in this Agreement shall be true and correct in all material respects on and as of the Closing Date.
- (b) ORDERS AND LAWS. There shall not be pending or in effect on the Closing Date any Order or Law restraining, enjoining or otherwise prohibiting or making illegal the consummation of any of the transactions contemplated by this Agreement or which could reasonably be expected to otherwise result in a material diminution of the benefits of the transactions contemplated by this Agreement to such party, and there shall not be pending on the Closing Date any Action or Proceeding in, before or by any Governmental or Regulatory Authority which could reasonably be expected to result in the issuance of any such Order or the enactment, promulgation or deemed applicability to either party or the transactions contemplated by this Agreement of any such Law.
- (c) REGULATORY CONSENTS AND APPROVALS. All consents, approvals and actions of, filings with and notices to any Governmental or Regulatory Authority necessary to permit each party to perform its respective obligations under this Agreement and to consummate the transactions contemplated hereby (a) shall have been duly obtained, made or given, (b) shall be in form and substance reasonably satisfactory to each party, (c) shall not be subject to the satisfaction of any condition that has not been satisfied or waived and (d) shall be in full force

and effect, and all terminations or expirations of waiting periods under the HSR Act or imposed by any other Governmental or Regulatory Authority necessary for the consummation of the transactions contemplated by this Agreement shall have occurred, and the Shares shall have been accepted for listing on notice of issuance by The New York Stock Exchange.

ARTICLE V

COVENANTS

The Company covenants and agrees with Purchaser that the Company will comply with the covenants and provisions of this ARTICLE V, except to the extent Purchaser may otherwise consent in writing:

5.01 SHELF REGISTRATION STATEMENT.

- (a) The Company shall use best efforts to file with the Securities and Exchange Commission (the "Commission") by the Filing Date a Shelf Registration Statement pursuant to Rule 415 under the Securities Act (the "Shelf Registration Statement") on Form S-3 (or any successor form thereto) to register resales by Purchaser of the Shares. The Company shall use best efforts to cause such Shelf Registration Statement to be declared effective by the Commission as soon as possible after the Filing Date but in no event later than the date of termination of the Lock-Up Period (the "TERMINATION DATE"). The Company shall use best efforts to keep such Shelf Registration Statement continuously effective and usable until the date on which all of the Shares are sold or such earlier date as the Shares may be resold by Purchaser without registration under Rule 144(k) under the Securities Act (the "Final Date"). The Company shall deliver copies of the Prospectus to The New York Stock Exchange pursuant to Rule 153 under the Securities Act and to Purchaser on reasonable request.
- (b) Upon the occurrence of any event that would cause the Shelf Registration Statement (i) to contain a material misstatement or to omit a material fact required to be stated therein or necessary to make the statements made not misleading or (ii) not to be effective and usable for resale of the Shares until the Final Date, the Company shall notify Purchaser as soon as reasonably practicable thereafter and, within two Business Days of the occurrence of such event, file a supplement to the Prospectus included in (if a supplement is appropriate for such purpose) or, within four Business Days of the occurrence of such event, file an amendment to the Shelf Registration Statement, in the case of clause (i) immediately above correcting any such misstatement or omission, and in the case of either clause (i) or (ii) immediately above use best efforts to cause such amendment to be declared effective and such Shelf Registration Statement to become usable as soon as reasonably practicable thereafter.

5.02 SPECIFIC PERFORMANCE; LIQUIDATED DAMAGES.

(a) The Company and Purchaser agree that Purchaser shall be entitled to enforce specifically the obligations under SECTION 5.01(A) in any court of competent jurisdiction (this being in addition to any other remedy to which it

is entitled at law or equity). The Company and Purchaser further agree that Purchaser will suffer damages if the Shelf Registration Statement has not been declared effective on or prior to the Termination Date. Accordingly, if the Shelf Registration Statement has not been declared effective on or prior to the Termination Date, Liquidated Damages on the Shares shall accrue at a rate per share per day equal to the quotient obtained by dividing (x) \$30.50 by (y) the Post Lock-Up Period Days, until the Shelf Registration Statement has been declared effective or the Final Date has occurred, and shall be payable monthly; PROVIDED, HOWEVER, that the aggregate Liquidated Damages payable on the Shares may not exceed the Purchase Price; and PROVIDED, FURTHER, that if the Termination Date occurs subsequent to the six month anniversary of the Closing Date (the "SIX-MONTH DATE"), then any Liquidated Damages payable for the period between the Six-Month Date and the Termination Date shall be due and payable on the Termination Date.

- (b) Notwithstanding the foregoing, the Company shall not be required to pay such Liquidated Damages with respect to the Shares if the applicable default arises from the failure of the Company to file or cause to become effective within the time period specified above primarily by reason of the failure of the Purchaser to provide such information concerning Purchaser as (i) is required pursuant to the Securities Act and the regulations promulgated thereunder for inclusion in the Shelf Registration Statement or any Prospectus included therein or (ii) the SEC may request in connection with such Shelf Registration.
- 5.03 REGISTRATION EXPENSES. All fees and expenses incidental to the performance of or compliance with this Article V by the Company shall be borne by the Company whether or not the Shelf Registration is filed or becomes effective, other than underwriting discounts and commissions and transfer taxes, if any, in respect of the Shares, which shall be payable by Purchaser.

5.04 INDEMNIFICATION.

(a) INDEMNIFICATION BY THE COMPANY. The Company shall, to the full extent permitted by law, indemnify and hold harmless Purchaser, its directors and each Person, if any, who controls Purchaser within the meaning of the Securities Act against any losses, claims, damages, expenses or liabilities, joint or several (together, "Losses"), to which it or any such controlling Person may become subject under the Securities Act or otherwise, insofar as such Losses (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon any untrue statement or alleged untrue statement of any material fact contained in the Shelf Registration Statement or the Prospectus, or any omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein (in the case of the Prospectus, in the light of the circumstances under which they were made) not misleading, and the Company will reimburse Purchaser and each such controlling Person for any legal or any other expenses reasonably incurred by them in connection with investigating or defending any such Loss (or action or proceeding in respect thereof); PROVIDED that the Company shall not be liable in any such case to the extent that any such Loss (or action or

proceeding in respect thereof) arises out of or is based upon (x) an untrue statement or alleged untrue statement or omission or alleged omission made in any such registration statement, preliminary prospectus, final prospectus, summary prospectus, amendment or supplement in reliance upon and in conformity with written information furnished to the Company through an instrument duly executed by Purchaser specifically stating that it is for use in the preparation thereof or (y) Purchaser's failure to send or give a copy of the final Prospectus to the Persons asserting an untrue statement or alleged untrue statement or omission or alleged omission at or prior to the written confirmation of the sale of Shares to such Person if such statement or omission was corrected in such final Prospectus. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of Purchaser or any such controlling Person, and shall survive the transfer of Shares by Purchaser.

- (b) INDEMNIFICATION BY PURCHASER. Purchaser shall, to the full extent permitted by law, indemnify and hold harmless the Company, its directors and officers, and each other Person, if any, who controls the Company within the meaning of the Securities Act, against any Losses to which the Company or any such director or officer or controlling Person may become subject under the Securities Act or otherwise, insofar as such Losses (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon any untrue statement or alleged untrue statement of any material fact contained in the Shelf Registration Statement or the Prospectus, or any omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein (in the case of a Prospectus, in the light of the circumstances under which they were made) not misleading, if such untrue statement or alleged untrue statement or omission or alleged omission was made in reliance upon and in conformity with written information furnished to the Company through an instrument duly executed by Purchaser specifically stating that it is for use in the preparation of the Shelf Registration Statement or the Prospectus. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of the Company or any such director, officer or controlling Person and shall survive the transfer of the Shares by Purchaser.
- (c) NOTICES OF CLAIMS, ETC. Promptly after receipt by an Indemnified Party of notice of the commencement of any action or proceeding involving a claim referred to in the preceding PARAGRAPH (A) OR (B) of this SECTION 5.04, such Indemnified Party will, if a claim in respect thereof is to be made against an Indemnifying Party pursuant to such paragraphs, give written notice to the latter of the commencement of such action, PROVIDED that the failure of any Indemnified Party to give notice as provided herein shall not relieve the Indemnifying Party of its obligations under the preceding paragraphs of this SECTION 5.04, except to the extent that the Indemnifying Party is actually prejudiced by such failure to give notice. In case any such action is brought against an Indemnified Party, the Indemnifying Party shall be entitled to participate in and to assume the defense thereof, with counsel reasonably satisfactory to such Indemnified Party, and after notice from the Indemnifying Party to such Indemnified Party of its election so to assume the defense thereof, the Indemnifying Party shall not be liable to such Indemnified Party for any legal or other expenses subsequently incurred by the latter in connection with the defense thereof other than reasonable costs of investigation; PROVIDED that the Indemnified Party or Indemnified Parties shall have the right to employ one counsel to represent it or them if, in the reasonable judgment of the Indemnified Party or Indemnified Parties, it is

advisable for it or them to be represented by separate counsel by reason of having legal defenses which are different from or in addition to those available to the Indemnifying Party, and in that event the reasonable fees and expenses of such one counsel shall be paid by the Indemnifying Party. If the Indemnifying Party is not entitled to, or elects not to, assume the defense of a claim, it will not be obligated to pay the fees and expenses of more than one counsel for the Indemnified Parties with respect to such claim, unless in the reasonable judgment of the Indemnified Party a conflict of interest may exist between such Indemnified Party and any other Indemnified Parties with respect to such claim, in which event the Indemnifying Party shall be obligated to pay the fees and expenses of such additional counsel for the Indemnified Parties or counsels. No Indemnifying Party shall consent to entry of any judgment or enter into any settlement without the consent of the Indemnified Party which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such Indemnified Party of a release from all liability in respect to such claim or litigation. No Indemnifying Party shall be subject to any liability for any settlement made without its consent, which consent shall not be unreasonably withheld.

(d) CONTRIBUTION. If the indemnity and reimbursement obligation provided for in any paragraph of this SECTION 5.04 is unavailable or insufficient to hold harmless an Indemnified Party in respect of any Losses (or actions or proceedings in respect thereof) referred to therein, then the Indemnifying Party shall contribute to the amount paid or payable by the Indemnified Party as a result of such Losses (or actions or proceedings in respect thereof) in such proportion as is appropriate to reflect the relative fault of the Indemnifying Party on the one hand and the Indemnified Party on the other hand in connection with statements or omissions which resulted in such Losses. The relative fault shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the Indemnifying Party or the Indemnified Party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such untrue statement or omission. The parties hereto agree that it would not be just and equitable if contributions pursuant to this paragraph were to be determined by PRO RATA allocation or by any other method of allocation which does not take account of the equitable considerations referred to in the first sentence of this paragraph. The amount paid by an Indemnified Party as a result of the Losses referred to in the first sentence of this paragraph shall be deemed to include any legal and other expenses reasonably incurred by such Indemnified Party in connection with investigating or defending any Loss which is the subject of this paragraph. No Indemnified Party guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from the Indemnifying Party if the Indemnifying Party was not guilty of such fraudulent misrepresentation.

5.05 HSR. Each party will (i) take promptly all actions necessary to make the filings required of it or its Affiliates under the HSR Act, (ii) comply at the earliest practicable date with any request for additional information received by such party or its Affiliates from the Federal Trade Commission (the "FTC") or the Antitrust Division of the Department of Justice (the "ANTITRUST DIVISION") pursuant to the HSR Act, and (iii) cooperate with the other party in connection with such party's filings under the HSR Act and in connection with

resolving any investigation or other inquiry concerning the matters contemplated by this Agreement commenced by either the FTC or the Antitrust Division or state attorneys general.

5.06 LOCKUP.

- (a) Purchaser shall not and shall not permit any Affiliate or donee of Purchaser to offer or sell in the public market any Shares for a period of 12 months from the Closing Date (the "LOCKUP PERIOD").
- (b) Notwithstanding the foregoing, in the event that either Mr. Bennett S. LeBow or Mr. Howard Lorber, or any Affiliate or donee of either Mr. LeBow or Mr. Lorber, offers or sells in the public market during the Lock-Up Period more than 300,000 of the shares of Common Stock owned by such party as of the Closing Date, the Company shall so notify Purchaser and the Lock-Up Period will terminate.
- 5.07 STOCK EXCHANGE LISTING. The Company shall use its best efforts to cause the Shares to be approved for listing on notice of issuance on The New York Stock Exchange.

ARTICLE VI

DEFINITIONS

6.01 DEFINITIONS.

(a) DEFINED TERMS. As used in this Agreement, the following defined terms have the meanings indicated below:

"ACTIONS OR PROCEEDINGS" means any action, suit, proceeding, arbitration or Governmental or Regulatory Authority investigation.

"AFFILIATE" means any Person that directly, or indirectly through one of more intermediaries, controls or is controlled by or is under common control with the Person specified. For purposes of this definition, control of a Person means the power, direct or indirect, to direct or cause the direction of the management and policies of such Person whether by Contract or otherwise.

"AGREEMENT" means this Stock Purchase Agreement, as the same shall be amended from time to time.

"ANTITRUST DIVISION" has the meaning ascribed to it in SECTION 5.05.

"ASSETS AND PROPERTIES" of any Person means all assets and properties of every kind, nature, character and description (whether real, personal or mixed, whether tangible or intangible, and wherever situated), including the goodwill related thereto, operated, owned or leased by such Person.

"BUSINESS DAY" means any day other than a Saturday, a Sunday or any other day on which banking institutions are not authorized or required to close in New York City or Miami, Florida.

"BUSINESS OR CONDITION OF THE COMPANY" means the business, financial condition or results of operations of the Company and the Subsidiaries taken as a whole.

 $\mbox{"CLOSING"}$ means the closing of the transactions contemplated by SECTION 1.03.

"CLOSING DATE" means the third Business Day following the later of (i) the termination or expiration of the waiting period under the HSR Act or (ii) the Shares having been accepted for listing on notice of issuance by The New York Stock Exchange.

"COMMISSION" has the meaning ascribed to it in SECTION 5.01.

"COMMON STOCK" has the meaning ascribed to it in the forepart of this Agreement.

 $\mbox{\tt "COMPANY"}$ has the meaning ascribed to it in the forepart of this Agreement.

"COMPANY SEC REPORTS" means each form, report, schedule, registration statement, definitive proxy statement and other document (together with all amendments thereof and supplements thereto) filed by the Company with the SEC since December 31, 2000.

"CONTRACT" means any agreement, lease, license, evidence of indebtedness, mortgage, indenture, security agreement or other contract.

"FILING DATE" means May 31, 2001.

"FINAL DATE" has the meaning ascribed to it in SECTION

5.01(A).

"GOVERNMENTAL OR REGULATORY AUTHORITY" means any court, tribunal, arbitrator, authority, agency, commission, official or other instrumentality of the United States or any state, county, city or other political subdivision.

"HSR ACT" has the meaning ascribed to it in SECTION 2.05.

"INDEMNIFIED PARTY" means a party entitled to indemnity in accordance with SECTION $5.04\,.$

"INDEMNIFYING PARTY" means a party obligated to provide indemnity in accordance with SECTION 5.04.

"LAWS" means all laws, statutes, rules, regulations, ordinances and other pronouncements having the effect of law of the United States or any state, county, city or other political subdivision or of any Governmental or Regulatory Authority.

1.02.

"LICENSE" means all licenses, permits, certificates of authority, authorizations, approvals, registrations, franchises and similar consents granted or issued by any Governmental or Regulatory Authority.

"LIENS" means any mortgage, pledge, assessment, security interest, lease, lien, adverse claim, levy, charge or other encumbrance of any kind, or any conditional sale Contract, title retention Contract or other Contract to give any of the foregoing.

"ORDER" means any writ, judgment, decree, injunction or similar order of any Governmental or Regulatory Authority (in each such case whether preliminary or final).

"PERSON" means any natural person, corporation, limited liability company, general partnership, limited partnership, proprietorship, other business organization, trust, union, association or Governmental or Regulatory Authority.

"POST LOCK-UP PERIOD DAYS" means the number of days between (x) the earlier of (i) the Termination Date or (ii) the Six-Month Date and (y) the Final Date.

"PROSPECTUS" shall mean the prospectus included in the Shelf Registration Statement, as amended or supplemented by any prospectus supplement, with respect to the terms of the offering of any portion of the securities covered by such Shelf Registration Statement, and all other amendments and supplements to the Prospectus, including post-effective amendments, and all material incorporated by reference or deemed to be incorporated by reference in such Prospectus.

"PURCHASE PRICE" has the meaning ascribed to it in SECTION

 $\mbox{\sc "PURCHASER"}$ has the meaning ascribed to it in the forepart of this Agreement.

"SECURITIES ACT" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"SEC" means Securities and Exchange Commission.

 $\,$ "SHARES" has the meaning ascribed to it in the forepart of this Agreement.

"SHELF REGISTRATION STATEMENT" has the meaning ascribed to it in SECTION 5.01.

"SIX-MONTH DATE" has the meaning ascribed to it in SECTION

5.02. "SUBSIDIARY" means any Person in which the Company, directly

or indirectly through Subsidiaries or otherwise, beneficially owns more than 50% of either the equity interests in, or the voting control of, such Person.

"TERMINATION DATE" has the meaning ascribed to it in SECTION 5.01.

ARTICLE VII

MISCELLANEOUS

- 7.01 ENTIRE AGREEMENT. This Agreement supersedes all prior discussions and agreements between the parties with respect to the subject matter hereof and contains the sole and entire agreement between the parties hereto with respect to the subject matter hereof.
- 7.02 EXPENSES. Except as otherwise expressly provided in this Agreement, whether or not the transactions contemplated hereby are consummated, each party will pay its own costs and expenses incurred in connection with the negotiation, execution and closing of this Agreement and the transactions contemplated hereby.
- 7.03 WAIVER. Any term or condition of this Agreement may be waived at any time by the party that is entitled to the benefit thereof, but no such waiver shall be effective unless set forth in a written instrument duly executed by or on behalf of the party waiving such term or condition. No waiver by any party of any term or condition of this Agreement, in any one or more instances, shall be deemed to be or construed as a waiver of the same or any other term or condition of this Agreement on any future occasion. All remedies, either under this Agreement or by Law or otherwise afforded, will be cumulative and not alternative.
- 7.04 AMENDMENT. This Agreement may be amended, supplemented or modified only by a written instrument duly executed by or on behalf of each party hereto.
- 7.05 NO THIRD PARTY BENEFICIARY. The terms and provisions of this Agreement are intended solely for the benefit of each party hereto and their respective successors or permitted assigns, and it is not the intention of the parties to confer third-party beneficiary rights upon any other Person.
- 7.06 NO ASSIGNMENT; BINDING EFFECT. Neither this Agreement nor any right, interest or obligation hereunder may be assigned by any party hereto without the prior written consent of the other party hereto and any attempt to do so will be void, except for assignments and transfers by operation of Law. Subject to the preceding sentence, this Agreement is binding upon, inures to the benefit of and is enforceable by the parties hereto and their respective successors and assigns.
- $7.07\ \text{HEADINGS}.$ The headings used in this Agreement have been inserted for convenience of reference only and do not define or limit the provisions hereof.
- 7.08 INVALID PROVISIONS. If any provision of this Agreement is held to be illegal, invalid or unenforceable under any present or future Law, and if the rights or obligations of any party hereto under this Agreement will not be materially and adversely affected thereby, (a) such provision will be fully severable, (b) this Agreement will be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof, and (c) the remaining provisions of this Agreement will remain in full force and effect and will not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom.

- 7.09 GOVERNING LAW. This Agreement shall be governed by and construed in accordance with the Laws of the State of New York applicable to a Contract executed and performed in such State, without giving effect to the conflicts of laws principles thereof.
- 7.10 COUNTERPARTS. This Agreement may be executed in any number of counterparts, each of which will be deemed an original, but all of which together will constitute one and the same instrument.
- 7.11 TERMINATION. This Agreement may be terminated, and the transactions contemplated hereby may be abandoned, at any time after 60 days after execution of this Agreement by either party upon notification of the non-terminating party by the terminating party if the Closing shall not have occurred on or before such date and such failure to consummate is not caused by a breach of this Agreement by the terminating party.

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the duly authorized officer of each party hereto as of the date first above written.

HIGH RIVER LIMITED PARTNERSHIP
By: Barberry Corp., General Partner

By: /s/ EDWARD E. MATTNER

Name: Edward E. Mattner Title: Authorized Signatory

VECTOR GROUP LTD.

By: /s/ RICHARD J. LAMPEN

Name: Richard J. Lampen

Title: Executive Vice President

[Stock Purchase Agreement with Vector Group Ltd. dated May 16, 2001]

[CITIGATE SARD VERBINNE LETTERHEAD]

NEWS

FOR IMMEDIATE RELEASE

Contact: George Sard/Anna Cordasco/Paul Caminiti

Citigate Sard Verbinnen

212/687-8080

VECTOR GROUP RAISES \$100 MILLION IN PRIVATE EQUITY AND DEBT PLACEMENTS

PROCEEDS TO FUND ADVERTISING AND PROMOTION OF NEW OMNI(TM) CIGARETTE PRODUCTS

MIAMI, FL, MAY 16, 2001 - Vector Group Ltd. (NYSE: VGR) today announced that it has raised approximately \$100 million through private debt and equity offerings.

The Company entered into an agreement today with High River Limited Partnership, an investment entity owned by Carl Icahn, under which High River will purchase 1,639,344 shares of Vector Group common stock at \$30.50 per share, the market price when negotiations between the parties concluded subject to agreement on documentation. Pursuant to the agreement, High River has agreed not to sell or transfer the shares in the public markets for a one-year period. Following the purchase of the new shares, Carl Icahn will own approximately 13% of Vector Group's outstanding shares.

In addition, on May 8, 2001, a subsidiary of the Company agreed to the material terms for the issuance at a discount of 10% senior secured notes due March 31, 2006 to certain institutional investors. The sale of the notes in a private placement through Jefferies & Company, Inc. was completed on May 14, 2001. The proceeds from the offering were approximately \$50 million before fees and expenses.

Vector Group intends to use the proceeds of these offerings to fund the advertising and promotion of Vector Tobacco's new Omni(TM) and Omni Free(TM) cigarette products and for general corporate purposes. Vector Tobacco has the

rights to a process that enables the production of tobacco that is virtually free of nicotine and virtually free of tobacco specific nitrosamines (TSNAs), a potent carcinogen found in tobacco. In addition, Vector Tobacco has developed a proprietary technology that significantly reduces one of the most serious cancer-causing agents in cigarettes, carcinogenic polycyclic aromatic hydrocarbon (PAH) compounds. Vector Group expects to introduce these new products later this year and in early 2002 under the names Omni(TM) and Omni Free(TM). Both Omni(TM) and Omni Free(TM) are lighted, smoked and taste the same as conventional cigarettes.

Vector Group is a holding company that indirectly owns Liggett Group Inc., Vector Tobacco and a controlling interest in New Valley Corporation.

THIS PRESS RELEASE CONTAINS CERTAIN FORWARD-LOOKING STATEMENTS WITHIN THE MEANING OF THE PRIVATE SECURITIES LITIGATION REFORM ACT OF 1995. THE COMPANY HAS TRIED, WHENEVER POSSIBLE, TO IDENTIFY THESE FORWARD-LOOKING STATEMENTS USING WORDS SUCH AS "ANTICIPATES", "BELIEVES", "ESTIMATES", "EXPECTS", "PLANS", "INTENDS" AND SIMILAR EXPRESSIONS. THESE STATEMENTS REFLECT THE COMPANY'S CURRENT BELIEFS AND ARE BASED UPON INFORMATION CURRENTLY AVAILABLE TO IT. ACCORDINGLY, SUCH FORWARD-LOOKING STATEMENTS INVOLVE KNOWN AND UNKNOWN RISKS UNCERTAINTIES AND OTHER FACTORS WHICH COULD CAUSE THE COMPANY'S ACTUAL RESULTS, PERFORMANCE OR ACHIEVEMENTS TO DIFFER MATERIALLY FROM THOSE EXPRESSED IN, OR IMPLIED BY, SUCH STATEMENTS. THESE RISKS, UNCERTAINTIES AND CONTINGENCIES INCLUDE, WITHOUT LIMITATION, THE CHALLENGES INHERENT IN NEW PRODUCT DEVELOPMENT INITIATIVES, THE COMPANY'S ABILITY TO RAISE THE CAPITAL NECESSARY TO GROW ITS BUSINESS, POTENTIAL DISPUTES CONCERNING THE COMPANY'S INTELLECTUAL PROPERTY, POTENTIAL DELAYS IN OBTAINING ANY NECESSARY GOVERNMENT APPROVALS OF THE COMPANY'S PROPOSED VIRTUALLY NICOTINE-FREE TOBACCO PRODUCTS, POTENTIAL DELAYS IN OBTAINING THE TOBACCO NEEDED TO PRODUCE THE COMPANY'S PROPOSED NEW PRODUCTS, MARKET ACCEPTANCE OF THE COMPANY'S PROPOSED NEW PRODUCTS, COMPETITION FROM COMPANIES WITH GREATER RESOURCES THAN THE COMPANY AND THE COMPANY'S DEPENDENCE ON KEY EMPLOYEES. SEE ADDITIONAL DISCUSSION UNDER "RISK FACTORS" IN ITEM 1 OF THE COMPANY'S ANNUAL REPORT ON FORM 10-K FOR THE YEAR ENDED DECEMBER 31, 2000, AND OTHER FACTORS DETAILED FROM TIME TO TIME IN THE COMPANY'S OTHER FILINGS WITH THE SECURITIES AND EXCHANGE COMMISSION. THE COMPANY UNDERTAKES NO OBLIGATION TO UPDATE OR ADVISE UPON ANY SUCH FORWARD-LOOKING STATEMENTS TO REFLECT EVENTS OR CIRCUMSTANCES AFTER THE DATE OF THIS PRESS RELEASE OR TO REFLECT THE OCCURRENCE OF UNANTICIPATED EVENTS.

#